A Grammar of Nanga

Dogon language family Mali

Jeffrey Heath University of Michigan

draft dated August 2010 (converted to Unicode) not finished or definitive, use caution in citing I will later add index, consecutive numbering, page breaks, etc. comments and questions welcomed

author's email schweinehaxen@hotmail.com

color codes (excluding headings)

black: new material for this grammar blue: transcription of Nanga forms

green: transcription of other Malian languages and of reconstructions

red: comments to myself (e.g. data to collect or reanalyse)

pink: raw data not yet incorporated into text prose

yellow highlight: needs checking or commentary

Contents

1	Introd	luction	1
	1.1 Do	gon languages	1
	1.2 Na:	nga language	1
	1.3 En	vironment	2
	1.4 Pre	evious and contemporary study of Nanga	3
	1.4.1	Fieldwork	3
	1.4.2	Acknowledgements	3
2	Skotol	1	1
_		osody	
		lectable verbs	
		un phrase (NP)	
		stposition phrase (PP)	
		in clauses and constituent order	
		minalized clauses and constituent order	
		lative clauses.	
		erclausal syntax	
		•	
3		ology	
		neral	
		ernal phonological structure of stems and words	
	3.2.1	Syllables	
	3.2.2	1110411041 041404410	
		nsonants	
	3.3.1	Alveopalatals (c, j, n)	
	3.3.2	Voiced velar stop g and g-Spirantization (g→f)	
	3.3.3	Back nasals (ŋ ɲ)	
	3.3.4	Voiceless labials (p, f)	
	3.3.5	Laryngeals (h, ?)	
	3.3.6	Sibilants (s, š)	
	3.3.7	Nasalized sonorants (r ⁿ , w ⁿ , y ⁿ)	
	3.3.8	Consonant clusters1	
	3.3.8	1	
	3.3.8		
	3.3.8	3.3 Medial geminated CC clusters	3

3.3.8.4	Medial non-geminate CC clusters	14
3.3.8.5	Medial triple CCC clusters	14
3.3.8.6	Final CC clusters	15
3.4 Vowels	3	15
3.4.1 She	ort and (oral) long vowels	15
3.4.2 Na	salized vowels	16
3.4.3 Ini	tial vowels	17
3.4.4 Ste	m-final vowels (/u/ is rare)	17
3.4.5 Vo	calic harmony	17
3.5 Segmen	ntal phonological rules	18
3.5.1 Tra	ns-syllabic consonantal processes	18
3.5.1.1	Nasalization-Spreading	18
3.5.1.2	Consonantal metathesis in suffixal derivatives of verbs	20
3.5.2 Vo	calism of suffixally derived verbs	20
3.5.2.1	Suffixal Vowel-Spreading	20
3.5.2.2	Presuffixal V ₂ -Raising	20
3.5.3 Vo	calic rules sensitive to syllabic or metrical structure	20
3.5.3.1	Vowel-lengthening	20
3.5.3.2	Epenthesis	20
3.5.3.3	Post-Sonorant Syncope (verbs)	20
3.5.4 De	letion of final u (u-Apocope)	21
3.5.5 Lo	cal consonant cluster rules	21
3.5.5.1	Derhoticization (/r ⁿ / to n)	21
3.5.5.2	Rhotic Assimilation	21
3.5.5.3	Rhotic-Cluster Lateralization (/rr/→ll)	21
3.5.6 Vo	wel-vowel and vowel-semivowel sequences	22
3.5.6.1	Hiatus between adjacent vowels in reduplications	22
3.5.6.2	VV-Contraction	22
3.5.7 Lo	cal vowel-consonant interactions	22
3.5.7.1	/i/ > u before labial	22
3.5.7.2	Monophthongization (/iy/ to i:, /uw/ to u:)	22
3.6 Cliticiz	ation	23
3.6.1 Pho	onology of =m clitic ('be')	23
3.7 Tones.		24
3.7.1 Lex	xical tone patterns	24
3.7.1.1	At least one H-tone in each stem	24
3.7.1.2	Lexical tones of verbs	
3.7.1.3	Lexical tone patterns for unsegmentable noun stems	25
3.7.1.4	Lexical tone patterns for adjectives and numerals	30
3.7.1.5	Default H-tone, or autosegmental mapping?	31
3.7.1.6	Tone-break location for bitonal non-verb stems	31
3.7.1.7	Tone-break location for tritonal non-verb stems	32

	3.7.2	Grammatical tone patterns	33
	3.7.2	1 Grammatical tones for verb stems	33
	3.7.2	2 Grammatical tones for noun stems	33
	3.7.2		
	3.7.3	Tonal morphophonology	35
	3.7.3		35
	3.7.3		
	3.7.3	3 Tone-Dissimilation (decimal numerals)	37
	3.7.3	.4 Atonal-Morpheme Tone-Spreading	37
	3.7.4	Low-level tone rules	38
	3.7.4	.1 Rising-Tone Mora-Addition	38
	3.7.4	\mathcal{E}	
	3.7.4	,	
	3.7.4		
	3.7.4	ϵ	
	3.7.4		
		nation contours	
	3.8.1	Phrase and clausefinal nonterminal contours (\uparrow , \Rightarrow , $\Rightarrow \uparrow$, =	⇒¥) 41
	3.8.2	Adverbs and particles with lexically specified prolongation (
	3.8.3	Dying-quail intonational effect : (1Pl, 2Pl)	41
4		al, pronominal, and adjectival morphology	
4	4.1 Nou	ins	45
4	4.1 Nou 4.1.1	Simple nouns	45
4	4.1 Not 4.1.1 4.1.2	Simple nouns ('woman', 'child', 'person', 'thing')	45 45
4	4.1 Not 4.1.1 4.1.2 4.1.3	Simple nouns	45 45 46
4	4.1 Not 4.1.1 4.1.2 4.1.3 4.1.4	Simple nouns Irregular nouns ('woman', 'child', 'person', 'thing') 'So-and-so' (àmâ:n)	45 46 47
4	4.1 Not 4.1.1 4.1.2 4.1.3 4.1.4 4.1.5	Simple nouns Irregular nouns ('woman', 'child', 'person', 'thing')	45 45 46 47 48
4	4.1 Not 4.1.1 4.1.2 4.1.3 4.1.4 4.1.5 4.1.6	Simple nouns Irregular nouns ('woman', 'child', 'person', 'thing') 'So-and-so' (àmâ:n) Initial Cv- reduplication in nouns Final -Cv: and -Cvy reduplications in nouns Nouns with full-stem iteration	45 46 47 47 48
4	4.1 Not 4.1.1 4.1.2 4.1.3 4.1.4 4.1.5 4.1.6 4.1.7	Simple nouns Irregular nouns ('woman', 'child', 'person', 'thing') 'So-and-so' (àmâ:n) Initial Cv- reduplication in nouns Final -Cv: and -Cvy reduplications in nouns Nouns with full-stem iteration Frozen initial a- or an- in nouns	45 46 47 47 48 49
4	4.1 Not 4.1.1 4.1.2 4.1.3 4.1.4 4.1.5 4.1.6 4.1.7 4.2 Der	Simple nouns Irregular nouns ('woman', 'child', 'person', 'thing')	45 46 47 47 48 50 51
4	4.1 Not 4.1.1 4.1.2 4.1.3 4.1.4 4.1.5 4.1.6 4.1.7 4.2 Der 4.2.1	Simple nouns Irregular nouns ('woman', 'child', 'person', 'thing')	45 46 47 47 48 50 51
4	4.1 Not 4.1.1 4.1.2 4.1.3 4.1.4 4.1.5 4.1.6 4.1.7 4.2 Der 4.2.1 4.2.2	Simple nouns Irregular nouns ('woman', 'child', 'person', 'thing')	45 46 47 48 49 50 51
4	4.1 Not 4.1.1 4.1.2 4.1.3 4.1.4 4.1.5 4.1.6 4.1.7 4.2 Der 4.2.1 4.2.2 4.2.2	Simple nouns Irregular nouns ('woman', 'child', 'person', 'thing')	45 45 47 47 48 50 51 51
4	4.1 Not 4.1.1 4.1.2 4.1.3 4.1.4 4.1.5 4.1.6 4.1.7 4.2 Der 4.2.1 4.2.2 4.2.2 4.2.3	Simple nouns Irregular nouns ('woman', 'child', 'person', 'thing')	45 46 47 48 50 51 51 51
4	4.1 Not 4.1.1 4.1.2 4.1.3 4.1.4 4.1.5 4.1.6 4.1.7 4.2 Der 4.2.1 4.2.2 4.2.2 4.2.3 4.2.4	Simple nouns Irregular nouns ('woman', 'child', 'person', 'thing')	45 46 47 48 49 51 51 51 51
4	4.1 Not 4.1.1 4.1.2 4.1.3 4.1.4 4.1.5 4.1.6 4.1.7 4.2 Der 4.2.1 4.2.2 4.2.2 4.2.3 4.2.4 4.2.5	Simple nouns Irregular nouns ('woman', 'child', 'person', 'thing')	45454748495151515151
4	4.1 Not 4.1.1 4.1.2 4.1.3 4.1.4 4.1.5 4.1.6 4.1.7 4.2 Der 4.2.1 4.2.2 4.2.2 4.2.3 4.2.4 4.2.5 4.3 Pro	Simple nouns Irregular nouns ('woman', 'child', 'person', 'thing')	454547474850515151515252 lefined.
4	4.1 Not 4.1.1 4.1.2 4.1.3 4.1.4 4.1.5 4.1.6 4.1.7 4.2 Der 4.2.1 4.2.2 4.2.2 4.2.3 4.2.4 4.2.5 4.3 Pro- 4.3.1	Simple nouns Irregular nouns ('woman', 'child', 'person', 'thing')	454547474851515151525252
4	4.1 Not 4.1.1 4.1.2 4.1.3 4.1.4 4.1.5 4.1.6 4.1.7 4.2 Der 4.2.1 4.2.2 4.2.2 4.2.3 4.2.4 4.2.5 4.3 Pro- 4.3.1 4.4 Der	Simple nouns Irregular nouns ('woman', 'child', 'person', 'thing')	454547474851515151525252
4	4.1 Not 4.1.1 4.1.2 4.1.3 4.1.4 4.1.5 4.1.6 4.1.7 4.2 Der 4.2.1 4.2.2 4.2.2 4.2.3 4.2.4 4.2.5 4.3 Pro- 4.3.1	Simple nouns Irregular nouns ('woman', 'child', 'person', 'thing')	45454647485151515152 lefined5255

4.4.1.3 Anaphoric/logophoric demonstrative pronouns	4.4.1.2 Prenominal Anaphoric kú 'that (same)'	56
4.4.2.1 Locative adverbs	4.4.1.3 Anaphoric/logophoric demonstrative pronouns	58
4.4.2.2 Emphatic and Approximinative modifiers of adverbs	4.4.2 Demonstrative adverbs	58
4.4.3 Presentatives ('here's!')	4.4.2.1 Locative adverbs	58
4.5.1 Types of adjectives	4.4.2.2 Emphatic and Approximinative modifiers of adverbs	59
4.5.1 Types of adjectives	4.4.3 Presentatives ('here's!')	59
4.5.2 CvCv=> adverbials ('flat')	4.5 Adjectives	60
4.5.3 Iterated (fully reduplicated) adverbials 65 4.6 Participles 66 4.7 Numerals 66 4.7.1 Cardinal numerals 66 4.7.1.1 'One', 'same (one)', and 'other' 66 4.7.1.2 '2' to '10' 66 4.7.1.3 Decimal multiples ('10', '20',) and their combinations ('11', '59',) 66 4.7.1.4 Large numerals ('100', '1000',) and their composites 67 4.7.1.5 Currency 68 4.7.1.6 Distributive numerals 68 4.7.2 Ordinal adjectives 68 4.7.2.1 'First' and 'last' 69 4.7.2.2 Other ordinals (suffix -nɔ's) 69 4.7.3 Fractions and portions 70 5 Nominal and adjectival compounds 71 5.1 Nominal compounds 71 5.1.1 Compounds of type (n̄ n̄) 71 5.1.2 Compounds with final Verbal Noun, type (n̄ n̄) 71 5.1.3 Agentive compounds (n̄ n̄) 72 5.1.4 Possessive-type compounds (n̄ n̄) 72 <t< td=""><td>4.5.1 Types of adjectives</td><td>60</td></t<>	4.5.1 Types of adjectives	60
4.6 Participles 66 4.7 Numerals 66 4.7.1 Cardinal numerals 66 4.7.1.1 'One', 'same (one)', and 'other' 66 4.7.1.2 '2' to '10' 66 4.7.1.3 Decimal multiples ('10', '20',) and their combinations ('11', '59',) 66 64 4.7.1.4 Large numerals ('100', '1000',) and their composites 67 4.7.1.5 Currency 68 4.7.1.6 Distributive numerals 68 4.7.2 Ordinal adjectives 68 4.7.2.1 'First' and 'last' 69 4.7.2.2 Other ordinals (suffix -n5:) 69 4.7.3 Fractions and portions 70 5 Nominal and adjectival compounds 71 5.1 Nominal compounds 71 5.1.1 Compounds of type (ñ ñ) 71 5.1.2 Compounds with final Verbal Noun, type (ñ ñ) 71 5.1.4 Possessive-type compounds (ñ ñ) 72 5.1.5 Agentive compounds of type (x̄ v̄) 73 5.1.6 Compounds with yî: 'child' and ñjā or nàr"á 'fruit' 74 </td <td>4.5.2 CýCŷ=> adverbials ('flat')</td> <td>65</td>	4.5.2 CýCŷ=> adverbials ('flat')	65
4.7 Numerals	4.5.3 Iterated (fully reduplicated) adverbials	65
4.7.1 Cardinal numerals 66 4.7.1.1 'One', 'same (one)', and 'other' 66 4.7.1.2 '2' to '10' 66 4.7.1.3 Decimal multiples ('10', '20',) and their combinations ('11', '59',) 66 66 4.7.1.4 Large numerals ('100', '1000',) and their composites 67 4.7.1.5 Currency 68 4.7.1.6 Distributive numerals 68 4.7.2 Ordinal adjectives 68 4.7.2.1 'First' and 'last' 69 4.7.2.2 Other ordinals (suffix -nδ:) 69 4.7.3 Fractions and portions 70 5 Nominal and adjectival compounds 71 5.1.1 Compounds of type (n̄ n̄) 71 5.1.2 Compounds of type (n̄ n̄) 71 5.1.3 Compounds with final Verbal Noun, type (n̄ n̄) 71 5.1.4 Possessive-type compounds (n̄ n̄) 72 5.1.5 Agentive compounds of type (n̄ n̄) 72 5.1.5 Agentive compounds of type (n̄ n̄) 73 5.1.6 Compounds with yî: 'child' and n̄jā or nar"ā 'fruit' 74 5.1.7 'Man' (-ar"ā, ar"ā-), 'woman' (-ya, ya-, ye-) 75 5.1.8 Compounds with baŋā 'owner' 76 5.1.10 Natural-species compounds with medial linking element 77 <t< td=""><td>4.6 Participles</td><td>66</td></t<>	4.6 Participles	66
4.7.1.1 'One', 'same (one)', and 'other' 66 4.7.1.2 '2' to '10' 66 4.7.1.3 Decimal multiples ('10', '20',) and their combinations ('11', '59',) 66 67 4.7.1.4 Large numerals ('100', '1000',) and their composites 67 4.7.1.5 Currency 68 4.7.1.6 Distributive numerals 68 4.7.2 Ordinal adjectives 68 4.7.2.1 'First' and 'last' 69 4.7.2.2 Other ordinals (suffix -nŏ:) 69 4.7.3 Fractions and portions 70 5 Nominal and adjectival compounds 71 5.1 Nominal compounds of type (n̄ n̄) 71 5.1.2 Compounds of type (n̄ n̄) 71 5.1.3 Compounds with final Verbal Noun, type (n̄ n̄) 71 5.1.4 Possessive-type compounds (n̄ n̄) 72 5.1.5 Agentive compounds of type (n̄ n̄) 72 5.1.5 Agentive compounds with final Verbal Noun, type (n̄ n̄) 73 5.1.6 Compounds with yî: 'child' and n̄	4.7 Numerals	66
4.7.1.2 '2' to '10'		
4.7.1.3 Decimal multiples ('10', '20',) and their combinations ('11', '59',) 66 4.7.1.4 Large numerals ('100', '1000',) and their composites	4.7.1.1 'One', 'same (one)', and 'other'	66
'59',) 66 4.7.1.4 Large numerals ('100', '1000',) and their composites		
4.7.1.4 Large numerals ('100', '1000',) and their composites	4.7.1.3 Decimal multiples ('10', '20',) and their combinations	s ('11',
4.7.1.5 Currency 68 4.7.1.6 Distributive numerals 68 4.7.2 Ordinal adjectives 68 4.7.2.1 'First' and 'last' 69 4.7.2.2 Other ordinals (suffix -nŏ:) 69 4.7.3 Fractions and portions 70 5 Nominal and adjectival compounds 71 5.1 Nominal compounds 71 5.1.1 Compounds of type (n̄ n̄) 71 5.1.2 Compounds of type (n̄ n̄) 71 5.1.3 Compounds with final Verbal Noun, type (n̄ n̄) 71 5.1.4 Possessive-type compounds (n̄ n̄) 72 5.1.5 Agentive compounds of type (n̄ n̄) 73 5.1.6 Compounds with yî: 'child' and nja or nar na 'fruit' 74 5.1.7 'Man' (-ar na, ar na, ar na-), 'woman' (-ya:, ya:-, ya:-) 75 5.1.8 Compounds with baŋa 'owner' 76 5.1.10 Natural-species compounds with medial linking element 77 5.1.11 Instrumental relative compounds ('oil for rubbing') 77 5.1.12 Other phrasal compounds 70		
4.7.1.6 Distributive numerals 68 4.7.2 Ordinal adjectives 68 4.7.2.1 'First' and 'last' 69 4.7.2.2 Other ordinals (suffix -nɔ̃:) 69 4.7.3 Fractions and portions 70 5 Nominal and adjectival compounds 71 5.1 Nominal compounds 71 5.1.1 Compounds of type (n̄ n̄) 71 5.1.2 Compounds of type (n̄ n̄) 71 5.1.3 Compounds with final Verbal Noun, type (n̄ n̄) 71 5.1.4 Possessive-type compounds (n̄ n̄) 72 5.1.5 Agentive compounds of type (n̄ n̄) 72 5.1.5 Agentive compounds of type (n̄ n̄) 73 5.1.6 Compounds with yî: 'child' and nja or nar nar nar nar nar nar nar nar nar na	4.7.1.4 Large numerals ('100', '1000',) and their composites	67
4.7.2 Ordinal adjectives 68 4.7.2.1 'First' and 'last' 69 4.7.2.2 Other ordinals (suffix -nŏ:) 69 4.7.3 Fractions and portions 70 5 Nominal and adjectival compounds 71 5.1 Nominal compounds 71 5.1.1 Compounds of type (n̄ n̄) 71 5.1.2 Compounds of type (n̄ n̄) 71 5.1.3 Compounds with final Verbal Noun, type (n̄ n̄) 71 5.1.4 Possessive-type compounds (n̄ n̄) 72 5.1.5 Agentive compounds of type (x̄ v̄) 73 5.1.6 Compounds with yî: 'child' and nja or nàr na 'fruit' 74 5.1.7 'Man' (-àr na, àr na), 'woman' (-ya:, ya:-, ye) 75 5.1.8 Compounds with bàna 'owner' 76 5.1.9 Loose and tight compounds with de: ('authentic', 'entire') 76 5.1.10 Natural-species compounds with medial linking element 77 5.1.12 Other phrasal compounds 77		
4.7.2.1 'First' and 'last' 69 4.7.2.2 Other ordinals (suffix -nɔ̄:) 69 4.7.3 Fractions and portions 70 5 Nominal and adjectival compounds 71 5.1 Nominal compounds 71 5.1.1 Compounds of type (n̄ n̄) 71 5.1.2 Compounds of type (n̄ n̄) 71 5.1.3 Compounds with final Verbal Noun, type (n̄ n̄) 71 5.1.4 Possessive-type compounds (n̄ n̂) 72 5.1.5 Agentive compounds of type (x̄ v̄) 73 5.1.6 Compounds with yı̂: 'child' and njâ or narna 'fruit' 74 5.1.7 'Man' (-arna, arna-), 'woman' (-yā:, yā:-, yē-) 75 5.1.8 Compounds with baŋa 'owner' 76 5.1.9 Loose and tight compounds with de: ('authentic', 'entire') 76 5.1.10 Natural-species compounds with medial linking element 77 5.1.12 Other phrasal compounds 77		
4.7.2.2 Other ordinals (suffix -nɔ̄:)		
4.7.3Fractions and portions705Nominal and adjectival compounds715.1Nominal compounds715.1.1Compounds of type (\bar{n}\bar{n})715.1.2Compounds of type (\bar{n}\bar{n})715.1.3Compounds with final Verbal Noun, type (\bar{n}\bar{n})715.1.4Possessive-type compounds (\bar{n}\bar{n})725.1.5Agentive compounds of type (\bar{x}\bar{v})735.1.6Compounds with yî: 'child' and hjâ or nàr"\bar{n}\bar{a}' \text{fruit'}745.1.7'Man' (-\bar{a}r^n\bar{a}, \bar{a}r^n\bar{a}-), 'woman' (-y\bar{a}:, y\bar{a}:-, y\bar{e}-)755.1.8Compounds with b\bar{n}\bar{a}' \text{owner'}765.1.9Loose and tight compounds with d\bar{e}: ('authentic', 'entire')765.1.10Natural-species compounds with medial linking element775.1.11Instrumental relative compounds ('oil for rubbing')775.1.12Other phrasal compounds77		
5Nominal and adjectival compounds715.1Nominal compounds715.1.1Compounds of type (n̄ n̄)715.1.2Compounds of type (n̄ n̄)715.1.3Compounds with final Verbal Noun, type (n̄ n̄)715.1.4Possessive-type compounds (n̄ n̂)725.1.5Agentive compounds of type (x̄ v̄)735.1.6Compounds with yî: 'child' and nja or nar a' fruit'745.1.7'Man' (-ar a, ar a-), 'woman' (-ya:, ya:-, ye-)755.1.8Compounds with bana 'owner'765.1.9Loose and tight compounds with de: ('authentic', 'entire')765.1.10Natural-species compounds with medial linking element775.1.11Instrumental relative compounds ('oil for rubbing')775.1.12Other phrasal compounds77		
5.1 Nominal compounds	4.7.3 Fractions and portions	70
5.1 Nominal compounds	5 Naminal and adjectivel compounds	71
5.1.1 Compounds of type (\$\bar{n}\$ \bar{n}\$)		
5.1.2 Compounds of type (n n)	*	
5.1.3 Compounds with final Verbal Noun, type (\hat{n} \hat{n})	* * * * * * * * * * * * * * * * * * * *	
5.1.4 Possessive-type compounds (\bar{n} \hat{n})	1 31 ()	
5.1.5 Agentive compounds of type (x̄ v̄)		
5.1.6 Compounds with yî: 'child' and hjâ or nàr"á 'fruit'		
5.1.7 'Man' (-àr ⁿ á, àr ⁿ á-), 'woman' (-yá:, yà:-, yè-)		
5.1.8 Compounds with baná 'owner'		
5.1.9 Loose and tight compounds with dé: ('authentic', 'entire')		
5.1.10 Natural-species compounds with medial linking element		
5.1.11 Instrumental relative compounds ('oil for rubbing')		
5.1.12 Other phrasal compounds		
	- · · · · · · · · · · · · · · · · · · ·	
5.1.13 Unclassified nominal compounds. Error! Bookmark not defined.		
5.2 Adjectival compounds		
5.2.1 Bahuvrihi ("Blackbeard") compounds (\bar{n} \hat{a})		

5.2.1.1 With adjectival compound final	78
5.2.1.2 With numeral compound final	
5.2.1.3 Alternative bahuvrihi with -ni suffix	
5.2.2 Compounds of 5:- 'very' plus adjective	
6 Noun Phrase structure	81
6.1 Organization of NP constituents	81
6.1.1 Linear order	81
6.1.2 Headless NPs (absolute function of demonstratives, etc.)	84
6.1.3 Detachability (in relatives)	84
6.1.4 Internal bracketing and tone-dropping	84
6.2 Possessives	89
6.2.1 Alienable possession (P Q.L)	89
6.2.1.1 Nonpronominal NP as prenominal possessor	
6.2.1.2 Tonal treatment of modifiers following the possessed n	
6.2.1.3 Pronoun plus classifier as postnominal possessor	
6.2.2 Inalienable possession	
6.2.2.1 Kin terms and similar relationship terms	
6.2.2.2 No tone-dropping of numeral after pronominally posses	
term 99	
6.3 Noun plus adjective	100
6.3.1 Noun plus regular adjective	
6.3.2 Adjective gàmbî 'certain (ones)'	
6.3.3 Expansions of adjective	
6.3.3.1 Adjective sequences	
6.3.3.2 'Good to eat'	101
6.4 Noun plus cardinal numeral	
6.5 Noun plus determiner	
6.5.1 Prenominal kú 'the (afore-mentioned)'	
6.5.2 Postnominal demonstrative pronouns	
6.5.3 Definite morpheme plus noun	
6.6 Universal and distributive quantifiers	
6.6.1 'All' (kéréw, sóy)	
6.6.2 'Each' (kámâ)	
6.6.3 Universal and distributive quantifiers with negation	
6.6.4 Universal quantifier combined with a numeral	
6.7 Accusative -ŋ	
o., recusarive ij	10/
7 Coordination	111
7.1 Conjunction	
7.1.1 NP conjunction ('X and Y') with yo	
J	

7.1.1.1 Ordering of coordinands	111
7.1.1.2 Conjunction with final quantifier	111
7.1.2 "Conjunction" of verbs or VP's	
7.2 Disjunction	
7.2.1 'Or' (mà)	
7.2.2 Clause-level disjunction	112
8 Postpositions and adverbials	115
8.1 Dative and instrumental	115
8.1.1 Dative bay ('for', 'at the place of')	115
8.1.2 Instrumental yàŋà (ỳŋà)	116
8.2 Locational postpositions	117
8.2.1 Locative, allative, and ablative functions	117
8.2.2 Simple and complex PPs	
8.2.3 'In, on' (ga, ŋa, gɔ, ŋɔ, go, ŋo)	117
8.2.4 Locative clitic =yè (or final tone change) with place names	123
8.2.5 'On (the head of) X' ([X kû:] gò)	124
8.2.6 'Next to, beside X' ([X kéri] gà)	125
8.2.7 'In front of' ([X girè] gà)	125
8.2.8 'Behind/after X' ([X túndù] gò), 'about'	126
8.2.9 'Over X' ([X témbè] gà), 'under X' ([X dósù] gò)	126
8.2.10 'Between' ([X bérè-kèndè] gá)	127
8.2.11 'From X to Y'	128
8.3 Purposive-Causal dèr ⁿ i	128
8.4 Other adverbials (or equivalents)	129
8.4.1 Similarity (may ⁿ 'like')	
8.4.2 Extent ('a lot', 'a little')	130
8.4.3 Specificity	132
8.4.3.1 'Approximately'	132
8.4.3.2 'Exactly' (lék, cók, já:dì, té=>)	132
8.4.3.3 'Specifically' (té=>)	
8.4.4 Evaluation	
8.4.4.1 'Well' and 'badly'	
8.4.4.2 'Proper, right'	
8.4.5 Manner	
8.4.6 Spatiotemporal adverbials	134
8.4.6.1 Temporal adverbs	134
8.4.6.2 'First' (kǐyá)	135
8.4.6.3 Spatial adverbs	135
8.4.7 Expressive adverbials	
8.4.7.1 Adjectival intensifiers	
8.4.7.2 'Straight' ($d\acute{e}m = >$).	140

8.4.7.3	'Apart, separate' (déy ⁿ = >)	141
8.4.7.4	'Always' (àsú⇒), 'never' (à:bádá)	
8.4.7.5	'Exclusively, together' (só:"-só:")	
8.4.7.6	'All, entirely' (kéréw, sóy)	
8.4.8 Der	ived iterated adverbials	
8.4.8.1	Distributive adverbial iteration	143
8.4.8.2	'Scattered, here and there' (kân-kân)	143
9 Verbal dei	rivation	1/15
	ve verbs (-rî-)	
	al causative verbs	
	ductive causative with suffix -mi-	
	for causative suffixes $(-g\hat{i} \sim -\eta\hat{i}, -nd\hat{e} \sim -nd\hat{i})$	
	and Transitive	
9.3.1 Med	diopassive -yı̂- and Transitive -rı̂	152
	sive suffix -yé	
	or passive suffix -mî-	
	alent verbs without suffixal derivation	
	tival inchoative and factitive verbs	
	nal verbs	
	e verb-verb relationships	
	•	
10 Verbal in	ıflection	163
10.1 Inflect	ion of regular indicative verbs	163
10.1.1 Su	ffixes versus chained verbs	163
10.1.2 Ov	verview of categories	164
10.1.3 Ve	erb stem shapes	164
10.1.3.1	Cv:- verb stems	
10.1.3.2	yĭ:- 'see'	
10.1.3.3	Cuy(i)- and Ciy(i)- verbs	
10.1.3.4	nCv- and mCv- verbs	
10.1.3.5	Regular bisyllabic stems	
10.1.3.6	Trisyllabic stems	
	re indicative AN categories	
	rfective positive system (including perfect)	
10.2.1.1	Unsuffixed Perfective with all-low toned stem	
10.2.1.2	Perfective-1a -èrè-, Perfective-1b -ti	
10.2.1.3	Perfective-2 (-só-)	
10.2.1.4	Experiential Perfect 'have ever' (tá:-só-)	
10.2.1.5	Recent Perfect (jè-)	
10.2.1.6	Reduplicated Perfective (Cý-)	186

10.2.2	Imperfective positive system	187
10.2.2	Imperfective (positive) $(-\dot{m} - \sim -\dot{\eta} \sim -\dot{\eta} \sim -\dot{\eta}$	187
10.2.2		
10.2.2	2.3 Reduplicated Imperfective (Cŷ-)	191
10.2.2	2.4 Progressive (-sò-)	192
10.2.3	Negation of indicative verbs	195
10.2.3	Perfective Negative -rî-	195
10.2.3	3.2 Experiential Negative -tà:-rî-	203
10.2.3	•	
10.2.3	3.4 Imperfective Negative -ŋɔ̀:	204
10.2.3	\mathcal{E}	
10.3 Pro	nominal paradigms for non-imperative verbs	209
10.3.1	Subject pronominal suffixes	
10.3.2	Nonhuman versus 3Sg subject	210
10.3.3	Vowel-semivowel interactions of AN and pronominal suff	
10.3.4	Tones of subject pronominal suffixes	
10.4 Sta	tive form of verbs (reduplicated and unreduplicated)	
10.4.1	Stative positive	
10.4.2	\mathcal{E}	
	st-verbal temporal particles	
	Past clitic (≡bε-)	
10.5.1	1 4 5 7	
10.5.1	1 ,	
10.5.1	<i>d c j</i>	
10.5.1	U ,	
	'Still', 'up to now', (not) yet'	
•	peratives and Hortatives	
10.6.1	Imperatives and Prohibitives	
10.6.1	1 1	
10.6.1		
10.6.2	Positive hortatives (-mà, plural -mày ⁿ)	
10.6.3	Hortative Negative (-rá and variants, plural -rá-mày ⁿ)	
10.6.4	Third-person Hortative	
10.6.5	Third-Person Hortative with implied first person singula	r subject
	237	
11 X/D	1 1 4 4 4	220
	d predicate structure	
-	gular verbs and VP structure	
11.1.1	Verb types (valency)	
11.1.2	Valency of causatives	
11.1.3	Verb Phrase	
11.1.4	Fixed subject-verb combinations	241

11.1.5 Idiomatic and cognate objects	243
11.1.5.1 Formal relationships between cognate nominal and verb	243
11.1.5.2 Grammatical status of cognate nominal	247
11.2 'Be', 'become', 'have', and other statives	248
11.2.1 'It is' clitics	248
11.2.1.1 Positive 'it is' ($\equiv m$ -, $\equiv \eta$ -, $\equiv y \hat{\epsilon}$ -, $\equiv w$ -)	248
11.2.1.2 'It is not' (≡ndŏ:-)	252
11.2.1.3 Extensions of yέ-m- to 1Pl and 2Pl subject	253
11.2.2 Existential and locative quasi-verbs and particles	254
11.2.2.1 Existential (yá)	254
11.2.2.2 Locational quasi-verb (bù-, negative ngó-)	254
11.2.3 'Be put in/on' (kùr ⁿ ò-, gàr ⁿ à-, nànà-)	256
11.2.4 Morphologically regular verbs	257
11.2.4.1 'Remain, happen' (bě:-)	257
11.2.4.2 'Become, happen' (táŋî-)	257
11.2.4.3 'Want, like' (jòró-, nàmà-, m ^b ùrá-)	258
11.2.4.4 'Fear' (ú:-yî, ùwà-)	259
11.3 Quotative verb	260
11.3.1 'Say' (kiyé-)	260
11.4 Adjectival predicates	261
11.4.1 Positive adjectival predicates with 'it is' clitic	261
11.4.2 Negative adjectival and stative predicates (=ndó-)	262
11.5 Possessive predicates	263
11.5.1 'Have' (sò-)	263
11.5.2 'Belong to' predicates	264
11.6 Verb iteration	265
11.6.1 Uninflected iteration of type $[\hat{v}_1-\hat{v}_1(-\hat{v}_1 \dots)]$ in narratives	265
12 Comparatives	267
12.1 Asymmetrical comparatives	
12.1.1 Simple adjective with dé:rè=w 'than' and comparandum	
12.1.2 Verbal predicate plus dé:rè≡w 'than'	
12.1.3 'Surpass' (láwá-)	
12.1.4 'Be better, more' (dé:rè-)	
12.1.5 'Best' (kay)	
12.2 Symmetrical comparatives	
12.2.1 'Equal; be as good as' (bă:-)	
12.2.2 'Same (equal)' (kêw)	
12.2.3 'Attain, equal' (dŏ:-)	
12.3 'A fortiori' (sákô, yê:)	

13 Focali	zation and interrogation	275
	calization	
13.1.1	Subject focalization	275
13.1.2	Object focalization	276
13.1.3	Focalization of PP or other adverbial	
13.1.4	Focalization of postpositional complement	277
13.2 Int	errogatives	
13.2.1	Polar (yes/no) interrogatives (ma)	277
13.2.2	'Who?' (ǎŋ)	279
13.2.3	'What?' (kò-né), 'with what?', 'why?'	281
13.2.4	'Where?' (arnana)	
13.2.5	'When?' ([àŋgú tû:] gò, wàgàti àrnáŋá)	283
13.2.6	'How?' (àmây ⁿ)	284
13.2.7	'How much/many?' (à:ŋgǎy)	284
13.2.8	'Which?' (àngú)	285
13.2.9	'So-and-so' (àmá:nà)	286
13.2.10	Embedded interrogatives	286
14 Relati	vization	287
14.1 Ba	sics of relative clauses	287
14.1.1	Coordinated relatives with a shared head	287
14.1.2	Tone-dropping on final word(s) of NP in relative clause	288
14.1.3	Restrictions on the head noun in a relative clause	
14.1.4	Relative clause with conjoined NP as head	291
14.1.5	Headless relative clause	292
14.1.6	Preverbal subject pronominal in relative clause	292
14.1.7	Verbal participle	293
14.1.	7.1 Participles of positive perfective-system verbs	294
14.1.	7.2 Participles of positive imperfective-system and stat 296	ive verbs
14.1.	7.3 Participles of negative perfective-system verbs	299
14.1.′	7.4 Participles of negative imperfective-system and stat 301	ive verbs
14.1.		303
14.1.8	Relative clause involving verb- or VP-chain	304
14.1.9	Determiners following the participle	
14.1.10	Non-numeral quantifiers following the participle	
14.1.11	Indefinite relatives	306
14.2 Su	bject relative clause	306
14.2.1	Ordinary subject relative clause	306
14.3 Oh	iect relative clause	308

14.3	.1 Ordinary object relative clause	.308
14.4	Possessor relative clause	
14.5	Relativization on the complement of a postposition	.309
15 Ve	rb (VP) chaining and adverbial clauses	.311
15.1	Direct chains (without chaining morpheme)	.311
15.1	.1 Verbal Noun of directly chained verbs	.311
15.1	.2 Presence of AN suffix in nonfinal verb in direct chains	.311
15.1	.3 Arguments of directly chained verbs	.312
15.1	.4 Negation of direct verb chains	.312
15.1	.5 Direct chains including dògó- 'leave'	.312
15.1	.6 Direct chains including a motion verb or 'pick up, take'	.313
15.1	.7 Durative verb-iterations chained to a motion verb	.313
15.1	.8 Chains including mɔ̃:ndi:- 'be/do together'	.314
15.1	.9 Chaining with jéjè⇒ 'go with'	.314
15.2	Adverbial clauses with overt chaining or subordinating morpheme	.315
15.2	.1 Backgrounded durative clauses (mò)	.315
15.2	.2 Imperfective - n as subordinator	.316
15	5.2.2.1 Imperfective - j on activity verb plus time-of-day verb	.316
15	5.2.2.2 Imperfective -\hat{\mathre{m}} (-\hat{m}) plus b\hat{\mathre{m}} \tag{be}'	.317
15.2		
15.2	.4 Simultaneous same-subject jè: before motion verb	.318
15.2	Durativeέ: on complement of /dě:-/ 'be tired/do long time'.	.319
15.2	c.6 Clauses with /nà/ 'and then' (different subject, anterior)	.321
15.2	7 Clauses with /ý/ (same-subject, anterior)	.321
15.2	.8 /gáy/ or /-wndé/ '(and) then' (same-subject, anterior, future t 322	ime)
15.3	Other temporal adverbial clauses	.323
15.3	.1 'Since' clauses (sê dày ⁿ)	.323
15.4	Noun-headed temporal clause ('the time when')	.323
15.4	.1 Reverse anteriority clause with final múgò 'before'	.324
15.4	.2 'No sooner, than' (bú⇒, pú⇒)	.324
15.5	Spatial and manner adverbials	.326
15.5	.1 Spatial adverbial clause ('where')	.326
15.5		
15.5	.3 Headless adverbial clause as spatiotemporal or manner clause.	.326
15.5	'From X, until (or: all the way to) Y'	.327
15.5	.5 'As though' clause	.327
16 Co	nditional constructions	.329
	Hypothetical conditional with -nde 'if'	329

16.1.1 Regular antecedent clause with pronominal subject suffix	329
16.1.2 'Unless' antecedent	331
16.2 Alternative 'if' particles	332
16.2.1 'Even if' (dèr"i, mò)	
16.2.2 'As soon as' (tán)	
16.3 Willy-nilly and disjunctive antecedents ('whether X or Y')	
16.4 Counterfactual conditional	
17 Complement and purposive clauses	.335
17.1 Quotative complement	335
17.1.1 Direct versus indirect in quotative complements	335
17.1.2 'Say that' with inflectable 'say' verb (kiyέ-)	335
17.1.3 Quotative clitic /wa/	337
17.1.4 Jussive complement (reported imperative or hortative)	342
17.1.4.1 Quoted imperative	342
17.1.4.2 Embedded hortative	344
17.2 Factive (indicative) complements	346
17.2.1 'Know that' complement clause	
17.2.2 'See (find, hear) that'	
17.2.2.1 Direct-perception type (relative-clause complement)	349
17.2.2.2 Recognition (inference, hearsay) construction	
17.2.3 Factive complement with tîlây≡∅ 'it is certain'	
17.3 Verbal Noun (and other nominal) complements	
17.3.1 Structure of Verbal Noun Phrase	
17.3.2 'Prevent' (késé-, gă:ndî-)	351
17.3.3 'Dare' (dă:rî-)	
17.3.4 'Consent' (àwá)	
17.3.5 'Cease' (dògó-)	353
17.3.6 'Want' (jòró)	
17.3.7 'Forget' (iré)	354
17.3.8 Obligational (wá:jíbǐ 'duty')	
17.3.9 'Be afraid to' (ú:-yε-)	356
17.3.10 'Begin' (tóró-)	
17.3.11 'Finish' (dimé-)	358
17.4 Locative verbal noun or other nominal complement	
17.4.1 'Help' (bă:rî-)	
17.5 Chained-verb complement clause	
17.5.1 'Be able to, can' (bèré-)	
17.6 Purposive, causal, and locative clauses	
17.6.1 Purposive clauses with postposition derni 'for'	
17.6.1.1 Positive purposive clause with Imperfective -m and der ⁿ 1	
17.6.1.2 Negative clause with Imperative Negative -rá and dèr ⁿ 1	

17.6.1		
17.6.2	Purposive clauses with Imperfective participle, without d	
17.6.3	Purposive clause with with gó 'in' after motion verb	
17.6.4	Causal ('because') clause (ságù)	364
17.6.5	'Because of' (derni)	364
17.7 Cla	use-final nà: 'though'	364
18 Anapl	10ra	365
18.1 Ref	flexive	365
18.1.1	Reflexive object (á-ý, Plural â:-ỳ)	365
18.1.2	Reflexive PP complement (á, â:)	366
18.1.3	Reflexive possessor (á, â:)	
18.1.4	Emphatic pronouns	
18.1.4	1.1 With màr ⁿ á 'self'	368
	4.2 With tùmâ (tùmáy ⁿ) 'one/alone'	
18.1.4	4.3 With kû: 'head'	369
	gophoric and indexing pronouns	
18.2.1	True third person logophoric function	
18.2.2	Non-logophoric topic-indexing function	
18.3 Red	ciprocal	
18.3.1		
18.3.2	'Together' (bèndèy)	
	strictions on reflexives	
18.4.1	No antecedent-reflexive relation between coordinands	376
19 Gram	matical pragmatics	377
	pic	
19.1.1	Topic (gây, gâ)	
19.1.2	'Now' (náy ⁿ)	
19.1.3	'Also, even' (yaŋa)	
19.2 Pre	sentential discourse markers	
19.2.1	'As much as' (hálì ~ hálè)	
19.2.2	'Well,' (háyà)	
19.2.3	'So,' (wálà:)	
19.2.4	Clause-initial emphatic particle (péy, fés)	
19.2.5	'But' (mě:)	
19.2.6	'Indeed'	
19.2.7	'Lo,' (jágà⇒)	
	gmatic adverbials or equivalents	
19.3.1	'(Not) again', 'on the other hand'	
19.4 'Or	nly' particles	

20 Te	ext	387
19.7	Greetings	384
	Backchannel and uptake checks	
19.:	5.3 Clause-final dè	383
	5.2 Clause-final kòy	
19.:	5.1 Phrase-final já:ti 'exactly'	382
19.5	Phrase-final emphatics	382
19.4	4.2 'Just (one)' (léŋ)	382
19.4	4.1 'Only' (sǎy)	381

1 Introduction

1.1 Dogon languages

Dogon is a family of somewhere around 20 languages belonging to the vast Niger-Congo phylum. The internal structure of the family as a whole is not yet clear. At least one language traditionally classified as Dogon, namely Bangeri Me in the northwest, appears to be genetically divergent from the others.

1.2 Nanga language

The Nanga language (with "ng" pronounced as a velar nasal) is spoken in a relatively small area about halfway between Douentza and Bandiagara. The speakers of the language refer to themselves, and to their language, as nán, The villages are Anda [á:ndé], Namakoro [námákòrò], Pergesa [pègèsá], Kono [kɔ́:r¹ɔ́], Wakara [wakára], Boromi [bòròmí], Irani [iraní], Ouse [ú:sé], and Soroni [soroni]. Administratively, the Nanga-speaking zone is carved up among three districts (cercles) in the provice (région) of Mopti, namely the districts of Douentza (villages of Soroni and Ouse), Bandiagara (Anda, Wakara, Boromi, Irani, Namakoro, and Pergesa), and Koro (Kono), all of which currently form part of the région (i.e. province) of Mopti. There are markets in Wakara and Kono that participate in a five-day market-day sequence also involving the Tommo-So speaking villages of Mori and Kasa, and the mostly Jamsay-speaking village of Perge (not the same as Pergesa).

The major neighboring Dogon languages are Jamsay (to the northeast, and broadly to the east) and Tommo-So (=Tombo-So) to the west but also in a few villages to the immediate east of Anda and Ouse. Adult speakers of Nanga can generally speak these two languages.

The Nanga zone, and specifically the village of Anda where I worked, is also near the villages of Beni and Gamni, where another Dogon language, Ben Tey, is spoken. Intercommunication between Nanga and Ben Tey speakers is more often carried out in Nanga than in Ben Tey (in other words, this bilingual situation is asymmetrical). Nevertheless, Ben Tey and Nanga languages seem rather close genetically and structurally, and some Nanga speakers understand Ben Tey even if they don't speak it. Ben Tey is also closely related to Bankan Tey spoken in the village of Walo north of Douentza.

Fulfulde (the language of the Fulbe people and the primary language of Douentza) is also fairly well-known among Nanga speakers, particularly those who have had dealings with Fulbe herders or who frequent the markets.

1.3 Environment

Most of the Nanga villages are arranged in a north-south line. The centrally located village of Wakara is around N 14° 51′ by W 3° 2′. Going south from Wakara one hits Namakoro, Pergesa, and Kono in that order. Going north from Wakara one encounters Soroni, Ouse, and Anda in that order.

The Nanga-speaking zone is punctuated by inselbergs, mostly rather small (in comparison to the imposing cliffs found around Douentza and Sangha), with intervening lowlands. Historically, all of the villages were located on the lower slopes of the hills that Fulbe cavalry could not reach on horseback. During the last thirty or so years of the Twentieth Century, as security improved, several of the villages (Anda, Namakoro, Pergesa, Boromi, Irani) were moved to lower ground so that fields in the plains could be cultivated. As of 2007, the villages of Soroni, Ouse, and Kono remain in their original elevated position. The village of Wakara has divided on sectarian lines, with the traditional animists remaining on the hill while converts to Islam now live on the plain below (where a French-Arabic school has been built).

There is a small river fueled by springs that begins in Anda and (in season) empties into a pond at Beni. The other Nanga-speaking villages have wells, seasonal ponds, or mountain springs as water sources. The main productive activity is farming, chiefly millet and sorghum as grain staples. Fonio (*Digitaria exilis*) was once widely grown but is no longer favored. Maize and rice are grown in selected locations. Cow-peas (*Vigna unguiculata*) and roselle (*Hibiscus sabdariffa*) are planted in the same fields as the millet and sorghum. Other crops grown separately during the main farming season are sesame, peanuts, groundnuts (*Vigna subterranea*), and okra.

Off-season (contre-saison) gardening, where water is available, includes maize, onions, tomatoes, lettuce, okra, cassava, chile peppers, potatoes, and sweet potatoes. Sugar cane and watermelon are also grown in both seasons in favorable locations.

Fruit trees in the area include mango, papaya, banana, orange, lemon, grapefruit, guava, tamarind, date, and native fruits such as zaban (Saba senegalensis), wild grape (Lannea microcarpa), and detarium (Detarium microcarpum).

1.4 Previous and contemporary study of Nanga

There is no previous published work on this language.

1.4.1 Fieldwork

Fieldwork on Nanga was undertaken in several trips during 2004-2008 as part of a broader project focusing on northern and northeastern Dogon languages (Jamsay, Ben Tey, Bankan Tey, Toro Tegu, Nanga, Najamba). My data are from the village of Anda, which was relatively accessible from my base in Douentza.

1.4.2 Acknowledgements

The fieldwork on Bankan Tey is being carried out under grant BCS 0853364 from the National Science Foundation (NSF), Documenting Endangered Languages (DEL) program, 2009-12.

The larger work on Dogon languages began with grant PA-50643-04 from the National Endowment for the Humanities (NEH) for solo fieldwork on Jamsay. This led to the idea of a comparative Dogon linguistic project. The first phase thereof was funded by NSF, grant BCS 0537435, for the period 2006-08. The current grant referenced above is for the second phase. Completion of the overall project, i.e. detailed documentation of some 20 Dogon languages, will require a third funding phase.

The University of Michigan also provided important supplemental support.

My collaborators in the collective project have been Abbie Hantgan, Laura McPherson, Kirill Prokhorov, Steve Moran, and the late Stefan Elders. Our primary Malian assistant (and my Jamsay informant) is Minkailou Djiguiba.

I am indebted to the people of Anda village, including Aperou Moro (chef de village), Dogoyeri Moro, Mani Moro, Mandio Moro, and Seydou Moro. Many other villagers helped out the lexicographic work by bringing specimens of flora and fauna.

2 Sketch

2.1 Prosody

Nanga has a typical Dogon stem-level tonal system, with at least one (lexical) high tone element per stem. At the level of syllables, tones are H[igh], L[ow], $F[alling] = \langle HL \rangle$, $R[ising] = \langle LH \rangle$, and bell-shaped $\langle LHL \rangle$. There are no $\langle HLH \rangle$ syllables. The notation $\langle ... \rangle$ is used for single-syllable contour tones like $\langle HL \rangle$. The notation $\{...\}$ is used for more abstract stem-wide contours like $\{H\}$ and $\{HL\}$ whose specific realizations depend on the number of syllables and moras.

Verb stems are lexically {LH} or {H}. When a derivational suffix is added, the lexical contour is generally respected. In the verbal inflectional system, tones play an important role along with inflectional suffixes. Several of the suffixes impose a specific tone contour on the preceding stem, overriding lexical tones. There are two distinct -so- suffixes, Resultative -só- and Progressive -sò-, distinguished only by tone contours (on the stem and suffix) and lengthening of the stem-final vowel: súyó-só- 'has hit' (Resultative) versus súyò:-sò- 'is hitting'.

Nouns, adjectives, and numerals have little suffixal morphology, and what they have is generally straightforward phonologically. As in other Dogon languages, though, these non-verb stems undergo stem-wide tone contour overlays in various morphosyntactic positions. The most common overlaid contour is {L}, which I refer to as **tone-dropping**. A noun (or adjective) is tone-dropped before a modifying adjective, a determiner (demonstrative pronoun or Definite morpheme), or kámâ 'each'. There is a possessed-noun contour {HL} on nouns preceded by a possessor ending in a high tone. The combination of a noun with a possessor is treated as a **tonosyntactic island**, so the noun is not subject to the usual tone-dropping effects induced by adjectives or determiners to its right. However, the tonological-island effect does not apply to the combination of possessor and noun when this sequence is followed by kámâ 'each'; instead, the tone-dropping effect of kámâ does apply here to the noun and to a pronominal possessor.

In relative clauses, the final word of the core NP (i.e. noun plus any adjectives), and (simultaneously) a numeral also tone-dropped if the NP in question functions as head of the relative (§14.1.2).

2.2 Inflectable verbs

The verb stem may be underived or suffixally derived (Reversive, Causative). In indicative categories, the stem is followed by an aspect-negation (AN) suffix, then a pronominal-subject suffix. There is no audible AN suffix in the unsuffixed Perfective (all-low tone), so here the stem is followed by the usual pronominal-subject suffixes.

Negation is expressed within the AN suffix system, not by external Negative particles. In the indicative, there are two basic negative AN forms, Perfective Negative and Imperfective Negative, compared to a somewhat richer set of distinctions in the positive AN suffixes. The positive AN suffixes are in most cases absent from the negative forms.

There is a full set of Imperative and Hortative verb forms, positive and negative.

In relative clauses (see below), the regular inflected verb is replaced by a participle that agrees with the head NP in nominal features (but not person).

2.3 Noun phrase (NP)

Nouns and adjectives have no suffixal morphology. Unlike the more northerly Dogon languages, Nanga has no Singular/Plural distinction, nor a Human/Nonhuman or Animate/Inanimate distinction, within nominal or (modifying) adjectival morphology.

The **core NP** consists of a noun plus any following modifying adjectives. This may be preceded by a nonpronominal NP possessor or by Definite demonstrative $k\acute{u}$ (originally a Nonhuman or Inanimate possessor). The core NP may be followed by a numeral, another quantifier ('all', 'each'), and/or a determiner (Definite morpheme or demonstrative pronoun). The determiners do distinguish Animate Singular, Animate Plural, Inanimate Singular, and Inanimate Plural, and are therefore the primary vehicle for expressing these categories.

For all nouns, a nonpronominal (i.e. noun-headed) possessor NP appears to the left of the possessed noun. For nouns other than kin terms, a pronominal possessor is expressed by postposing a pronominally possessed form of a semantically generic noun ('thing', 'living being') functioning here as a possessive classifier. Kin terms, unlike other nouns, allow pronominal possessors to precede the possessed noun directly, with no classifier.

2.4 Postposition phrase (PP)

Dative bay (realized as báy or bày), Instrumental yàŋà, Locative ga (with various realizations), and other postpositions directly follow a NP. The NP is phonologically independent of the postposition, but the form of certain postpositions is affected, tonally and/or segmentally, by (the end of) the NP.

2.5 Main clauses and constituent order

Main clauses (and subordinated clauses) are verb-final and typically SOV (when the subject is nonprominal).

2.6 Nominalized clauses and constituent order

2.7 Relative clauses

As in all Dogon languages, relative clauses are a distinctive and important syntactic construction. For example, some types of spatiotemporal and manner adverbial clauses are relative clauses in form.

The syntax is similar to that of some other northeastern Dogon languages (e.g. Jamsay, Ben Tey).

The core of the head NP (noun plus any adjectives) along with a possessor and/or a numeral remains within the relative clause proper. Determiners and non-numeral quantifiers are peeled off of the head NP and are positioned to the right of the verb (i.e. participle). Within the clause-internal NP, the head noun and any following adjective(s) or numeral that are not already tone-dropped drop their tones. Repetition of the head NP as a possessed noun directly following the relative clause proper (as in Jamsay) has not been observed. The verb becomes a participle that is marked for aspect and negation but has no agreement morphology (except for 3Pl subject in negative non-subject relatives).

2.8 Interclausal syntax

3 Phonology

3.1 General

Syllables and metrical structure are briefly covered in §3.2. The phonemes and some basic facts about their distribution and combinations are presented in §3.3 (consonants) and §3.4 (vowels). Nontonal phonological rules are described in §3.5. Cliticization is briefly discussed in §3.6. Tonal and intonation systems are the subject of §3.7.

3.2 Internal phonological structure of stems and words

3.2.1 Syllables

Word-initial syllables may omit the consonantal onset. Therefore in the formulae given below, if the syllable in question is word-initial the initial C is optional.

Using v for a short vowel and v: for a long oral vowel, the normal shape of a monosyllabic stem is Cv: with long (oral or nasal) vowel (xx1.a). CvC ending in a sonorant is also attested in pure form, in a couple of verbs (xx1.b) and in a few non-verb stems (nouns and numerals) (xx1.c). The shape CvC also occurs in various expressive and emphatic elements, and in the more interjection-like cases the final C may be an obstruent (xx1.d). Many phonetic occurrences of CvC and Cv:C actually reflect apocope of a final high vowel, which can be shown most clearly for inflectable verb stems (xx1.e).

(xxx)		form	gloss	comments
	a.	yî:	'child'	
		pέ:	'get old'	
		tă: ⁿ	'taboo'	
		ŏ:	'the bush, outback'	
		Ě:	'well (for water)'	
	b.	núy ⁿ	'enter'	
		túy	'arrange in bunches'	

```
'two'
c. wŏy
    n\check{\mathfrak{z}}y^n
                'four'
    dèw
                'trap (noun)'
                'all'
d. sóy
                'seething (mad)'
    dóŋ-dóŋ
    kέk
                'completely, every inch'
                'purchase (noun)'
                                            variant èwi, cf. éwé 'buy'
e. ěw
                'rope'
                                            variant sèni
    sěŋ
    děw
                'cover'
                                            /dèwi/
                'add'
                                            /bă:rî/
    bă:r
```

The final syllable of a noncomposite multisyllabic word is Cv with short vowel (this is the only pattern allowed for verb stems), or CvC with a final sonorant.

3.2.2 Metrical structure

Segmental phonological rules generally point to a trochaic bias, i.e. bisyllabic [sw] with a strong followed by a weak. In CvCvCv trisyllabic verbs (in the bare stem and related inflections), the medial vowel (if high and short) is vulnerable to syncope.

3.3 Consonants

The inventory of consonantal phonemes is (xx1). Single parentheses enclose marginal phonemes. Double parentheses enclose extremely marginal phonemes.

(xx1) Consonants

```
1
                      2
                                          5
                                                        7
                                                               8
                                                                      9
                            3
                                                  6
labial
                                                        (\mathbf{w}^n)
                      b
                            m
                                    (f)
               p
alveolar
                      d
               t
                            n
alveopalatal c
                      j
                                    ((\S))
                            ŋ
velar
               k
                      g
                            ŋ
laryngeal
                                                               (h)
                                                                      ((?))
```

c is IPA [t], j is $[d_3]$, š is [], y is [].

key to columns: 1. aspirated voiceless stops (c is affricated); 2. voiced stops; 3.nasals, 4. voiceless fricatives (including sibilants); 5. laterals; 6-7. respectively unnasalized and nasalized sonorants; 8-9. laryngeals

3.3.1 Alveopalatals (c, j, n)

As in the other languages of the region (Dogon, montane Songhay, etc.), there is occasional fluctuation before front vowels $\{i \in \epsilon\}$ between /k/ and /c/, and between /g/ and /j/. Thus $g\check{e}:r^n\check{i}$ or $j\check{e}:r^n\check{i}$ 'take away'. However, I generally hear unpalatalized (though perhaps slightly affricated) velars /k/ and /g/ before front vowels in this language.

3.3.2 Voiced velar stop g and g-Spirantization ($g \rightarrow f$)

Spirantization of g to f in the frames [a_a] and [o_o], i.e., between low back vowels, is not systematic in this language, and I transcribe g rather than f.

3.3.3 Back nasals (η, η)

Velar ŋ occurs intervocalically in such words as nàŋá 'cow', bàŋá 'owner', dòŋóndî 'console', gòŋîrⁿî 'circulate', nùŋó-mî 'turn on', súŋúrⁿi 'ear', and nùŋá nùŋî 'sing a song'.

ŋ is an allophone of an underspecified nasal in the clusters ηg and ηk. For ηk I can cite only róŋké 'fail' (<Fulfulde) and jàjàŋkó: 'double grain spike' (perhaps composite). ηg is common intervocalically, and there are some alternations (within Nanga or among nearby Dogon languages) between η and ηg. Examples are yóŋgǐ 'soul', bòŋgɔ́ 'navel', and dáŋgárá 'thighbone'.

Palatoalveolar n is permitted word-initially (where yn is absent), as in nă: 'meal', nàmá 'malfunction (verb)', né:rnê 'evil dwarf', nùnîrnî 'be stationery', nó:rnî 'summons', nă:rnî 'call, summon', and nùmó 'shake (e.g. branch)'. Some other word-initial cases are némîrnî 'reins', néné 'toilet' (<Bambara), nùnûrnî 'quiver'. n also occurs intervocalically, but the examples are either historically composite, as in ògò-nònó 'camel', or they belong to "cultural" vocabulary that may well be borrowed: 5:nî 'sag under a load', tó:nê 'teasing'. One caste of griots is called né-néyn, and a term meaning 'nonsense' is né-némé, both probably with intial Cv reduplication. Interrogative 'what?' is based on né, the actual form being either nné or kò né.

3.3.4 Voiceless labials (p, f)

f is not a full-fledged phoneme in Nanga. It does occur in some loanwords, but even here it is often replaced by p. A pass through my working lexicon (July 2007) reveals only fú > varying with pú > 'all' (<Fulfulde), ná:figi or ná:pigi 'trouble-maker' (<Arabic via Fulfulde), and sátsánsêf 'sargeant' (<French). 'Rifle' is pronounced màrbâ or màrpâ.

3.3.5 Laryngeals (h, ?)

h occurs stem-initially in numerous nouns and verbs, nearly all of them Fulfulde loanwords: héyyèndé 'index finger', há:dî: 'stop (at border)', hámbé 'chew (tobacco)', húkûm 'tent', hîbbé 'be complete', hógô 'herd'. The ubiquitous regional particle meaning 'until' or 'all the way to' appears as hálī.

h is attested medially in the loanword jàhánámà 'hell', and it does not occur word- or syllable-finally.

3.3.6 Sibilants (s, \check{s})

There is no phonemic distinction between s and š (= IPA ʃ). Words like gùsî 'skin' have (unpalatalized) alveolar s in spite of the high front vowel. Loanwords like sátsánsêf 'sargeant' (French sergent-chef) and ĭnšá:lâw 'maybe' (<Arabic 'if God wills') are poorly-assimilated and usually have pronunciation variants.

3.3.7 Nasalized sonorants (rⁿ, wⁿ, yⁿ)

 \mathbf{r}^n (nasalized tap) is a true phoneme, though it is limited to word-internal intervocalic position. It (therefore) does not occur initially, finally, or in consonant clusters. When a short vowel following \mathbf{r}^n is syncopated, the \mathbf{r}^n is converted to \mathbf{n} , thus $\mathbf{súnur^n}$ i 'ear', $\mathbf{súnur^n}$ i nò or syncopated \mathbf{sunun} nò 'his/her ear'.

rⁿ occurs independently of other nasals in unclustered intervocalic position, as in bárⁿi 'red'. If there is a preceding nasal in the stem or uncompounded word, rⁿ but not oral r may occur, so 'ear' is súŋúrⁿi and 'give birth' is nàrⁿá. Sequences like #...ŋúri and #nàrá with oral r are not allowed within uncompounded stems, and this extends to any inflectional suffixes, so Perfective Negative suffix -ri becomes -rⁿi in nasalized environments: nàrⁿà-rⁿi 'she did not give birth'.

No conspicuous phonetic nasalization occurs in initial semivowels {y w} anticipating a following nasal with a separating vowel, e.g. in wanıı 'change direction', yongı 'soul'.

 y^n occurs independently of other nasals in intervocalic or word-final position: $\delta y^n \delta$ 'spin (cotton)', $w \delta y^n \delta$ 'boil', $s \delta y^n$ 'nauseating', $m \delta y^n$ '(joint) dislocation', $d \delta y^n - d \delta y^n$ 'red (intensifier)'. It also occurs in nasalizing environments, where y and y^n fall together as y^n , as in $n \delta y^n \delta y^n$ 'this year'. Since p (§3.3.3) occurs chiefly in initial position (excluding composites and borrowings), there is a partial complementarity between p and p suggestive of an original phoneme split. However, I can cite no synchronic alternations between p and p.

Nanga has m (along with Bankan Tey and Najamba) corresponding to autonomous w^n in some neighboring languages (Jamsay, Ben Tey, Toro Tegu), e.g. $n\acute{\epsilon}m\acute{\epsilon}$ 'taste (verb)' (Jamsay and Ben Tey $n\acute{\epsilon}w^n\acute{\epsilon}$), $n\grave{a}m\^{a}$ 'meat' (Jamsay $n\grave{b}w^n\acute{\delta}$, Ben Tey $n\grave{a}w^n\^{a}$:, Toro Tegu $n\grave{a}w^n\acute{a}$, but Bankan Tey $n\grave{a}m\^{a}$: and Najamba $n\grave{a}m\acute{a}$). Toro Tegu and Ben Tey have some synchronic alternations of w^n (intervocalic) with m (other positions).

3.3.8 Consonant clusters

3.3.8.1 Problematic /nd/ and /nn/ clusters

Nanga and to a lesser extent Bankan Tey seem to have expanded an original intervocalic *l or *r under some conditions into a cluster beginning with /n/. At any rate, there are numerous correspondences of Nanga /nd/ and /nn/ to simple sonorants in some other languages. An example is (xx1).

```
    (xx1) gloss Nanga comparative data
    'iron' 1:ndá Jamsay 1:r<sup>n</sup>έ, Ben Tey 1:r<sup>n</sup>έy, Bankan Tey η1r<sup>n</sup>éy
```

Since clusters like #nl and #nr are not acceptable, adding the extra initial /n/ seems to have forced a switch to an acceptable cluster /nn/ or /nd/.

Nanga has a number of important stems that begin with a nasal-nasal or nasal-stop cluster where some neighboring languages have an initial vowel instead of the initial nasal. The data in (xx2) are organized around possible

phonological source as inferred from the comparative data, but the reconstructions are first stabs.

(xx2)	gloss	Nanga	comparative data
	a. *r or *r ⁿ		
	'go up'	'ndέ	r: Jamsay & Ben Tey ùró, Bankan Tey ùrá
	4	(10 (6 10)	n: Toro Tegu ùnú\\ùnó l: Pergue ùló, Najamba ilé, Yanda Dom ?álé, Tommo So úló
	'house'	ńdô (indô)	r: Ben Tey & Bankan Tey úrò, Jamsay úró, Pergue îré
			l: Toro Tegu îló, Yanda Dom ?áló, Najamba ólé
	3Sg	ńnέ	r ⁿ : Ben Tey έr ⁿ έ, x, x, x
	'field'	ńnâ	r ⁿ : Ben Tey îr ⁿ à, Bankan Tey ɲîr ⁿ â:, x,
	'tooth'	nné (iné)	x, x r ⁿ : Jamsay ir ⁿ é, Ben Tey ir ⁿ ú, x, x, x
	b. *1 (or *r ?)		
	'go'	ńné	Ben Tey ló, Bankan Tey ìdó
	c. *(n)s		
	'seed'	njâ	Ben Tey & Bankan Tey isâ:, Jamsay sέy ⁿ
	d. *n(d)		
	'give'	ńdî	Ben Tey & Bankan Tey nî, Najamba ndê
	'name'	nnèr ⁿ i	Ben Tey inir ⁿ î:, Bankan Tey ŋinnî:, Najamba înèn
	e. *nj		
	'what?-Pl'	'nμέ	Ben Tey & Bankan Tey njé, Jamsay iné 'what?'

An initial short vowel is occasionally heard (e.g. indô for ndô). In these items, when the intervocalic sononant grew an extra preceding /n/, it may be that this nasal crept into and eventually occupied the moraic position of the original initial short vowel. Reduplications treat the verbs as vowel-initial, with i- as the reduplicative segment for the /nn/ and /nd/ verbs, as in Imperfective i-indî-m- 'give'. This is heard with a faint glottal stop separating the identical vowels: [iʔindim].

Closer phonetic study is needed of what I write as nd and nn. It may be that "nd" is really $[n^d]$, i.e., a single complex segment consisting of an n with a brief oral release. The stems with initial /nn/ are often pronounced with [?n], i.e. a preglottalized n, after a vowel or semivowel (but not phrase-initially).

3.3.8.2 *Initial* /mb/

The verb mbó- occurs with the noun meaning 'nose' (or 'snot') in the phrase kirné mbó 'blow one's nose'. The only potential cognate I have come across is Najamba îmbî. The reduplicated forms treat the verb as beginning in /u/, hence reduplicated Imperfective ù-úmbó-m-.

mbùră- (or mbù-ră-), probably containing a (Stative) Negative suffix -rv-but now probably frozen), means 'not want, dislike'. Cf. Ben Tey mì-rá-, Bankan Tey mbì-rá-.

I have no other cases of initial /mb/ (or / ηg /, other than grammatical formatives such as Negative $\dot{\eta}g\dot{o}$ - and some demonstratives).

3.3.8.3 Medial geminated CC clusters

Geminated clusters are rare in Nanga. Based on a search of the lexicon (July 2008 version), only /ll/ is well-attested medially within a stem: jálli- 'grab suddenly', a few irregular reversives like illi-ri- 'remember' (iré- 'forget'), and a few Fulfulde loans.

In loanwords I can also cite one example each of /bb/ and /yy/: hîbbé- 'be complete', héyyèndé 'index finger'. I have no cases of {cc dd gg hh jj kk mm ŋŋ pp rr ss tt ww} within a stem.

/nn/ occurs medially in Fulfulde loans like hînnê 'quantity'. In native Dogon stems, it occurs initially (e.g. ńné- 'go', see §3.3.8.1, above), but not medially within a stem.

Some additional geminate clusters may arise at compound boundaries or due to Syncope. I can cite den-nî: 'semen' (den'î 'penis', nî: 'water').

Nanga tends to simplify even the few geminated clusters that do occur. I heard $h\acute{o}g\^{o}$ 'animal pen' (in other Dogon languages $h\acute{o}gg\`{o}$, from Fulfulde). For siddi 'sulphur' (*sitti), hijji 'pilgrimage to Mecca', and gállo 'house with walled courtyard', the medial geminate was often simplified (sidi, hiji, gálo), though the high tone did not spread into the second syllable as it typically does with true CvCv nouns (and in $h\acute{o}g\^{o}$). This is consistent with the parallel tendency to lenite intervocalic voiceless stops to voiced stops (*t > d, etc.).

3.3.8.4 Medial non-geminate CC clusters

Homorganic nasal plus voiced stop clusters {mb nd ng} are common within stems, e.g. dùmbó- '(blade) be blunt', dóndóró 'ball-shaped', mèngîré- 'shape into balls'. /nd/ also occurs suffix-initially in inchoative derivatives, like gàrá-ndíyé- 'become pungent', and /mb/ occurs in girè-mbî 'blindness'.

Excluding borrowings and compound boundaries, we have the following.

Homorganic nasal plus voiceless stop. /nt/: àntôngó 'residue after first winnowing of pounded grain' (cognate verb tóngó), àntô: 'stick with hook', and àntá:rí (variant àtá:rí) 'hunt (noun)' may contain an original formative *àn-. /ŋk/in jà-jàŋkó: 'double millet grain spike' may also involve an original boundary (cf. Najamba jànjàn-kàbá).

/r/ plus consonant is found in a few items: /rg/ in pergé 'sheep', (kɔŋɔrnoy-)dargá 'breakfast'; /rŋ/ in búrŋɔ´ 'group of travelers' (etymology unknown); /rs/ in kársì 'gizzard' (etymology unknown).

/wr/ is well-attested: kéwrî- 'cut wood' (cf. Jamsay céréwé-) and káwrî- 'split (peanut)' (Jamsay káráwá-), séwrú- 'trim surface (of wood) with knife', pówrî- 'rub or scrape off' (Jamsay pórówó-), kówró- '(handle) be broken', káwrà 'shard', jòwrĭ-nî: 'sauce'. /wy/ in éw-yé- 'sit' is perhaps still segmentable (an archaic causative éw-ré- is the only synchronic evidence).

/yr/ occurs in bóyrè 'porridge', from Fulfulde.

Other non-geminate CC clusters are found in borrowings (chiefly from Fulfulde). Examples of such clusters include nasal plus sibilant /ms/, /ns/; rhotic plus various consonants /rb/, /rd/, /rt/, /rk/ /rm/, /rn/; lateral plus various consonants /lb/, /lp/, /lg/, /lk/, /lm/, /ls/; semivowel plus various consonants /wd/, /wt/, /ws/, /yb/, /yg/, /yk/, /yn/.

3.3.8.5 *Medial triple CCC clusters*

Examples are few, and involve tap /r/ or semivowel /y/ followed by a homorganic nasal-stop sequence.

kàrmbî 'horse's mouth bit' and kòrmbî 'rope around donkey's tail' are heard with no clear vowel after the tap. However, representations of the type kàrimbî with a medial short high vowel would not be far from the phonetic output.

háyndî 'be amazing' and related words and póyngôl 'light on the horizon', both from Fulfulde, are my examples beginning with y.

3.3.8.6 Final CC clusters

None.

3.4 Vowels

Excluding tonal markings, the vowel segments are those in (xx1).

(xx1)	short oral	long oral	nasalized	
	u	u:	_	
	O	o:		
	Э	o :	o:n	
	a	a:	a:n	
	ε	ε:	ε :	
	e	e:		
	i	i:	i: ⁿ	

3.4.1 Short and (oral) long vowels

Except in loanwords like bòmbôn 'candy' (French bonbon) and some contractions involving affixes or clitics, all short vowels are oral. Long vowels in monosyllabic stems may be nasal (following section) or oral. There are also some long oral vowels in other syllables.

Vowel-final monosyllabic stems take the form (C)v: with long vowel: yĭ: 'see', yî: 'child', pĕ: 'old', pɛ: 'get old', yă: 'women', dŏ: 'haunch', tŭ: 'each other', dɔ: 'arrive', ɔ: 'place', dĕ: 'mother', pɔ: 'pick (fruit)'.

In non-monosyllabic stems, short vowels predominate. In the case of verb stems (excluding borrowings), a long vowel is allowed only in the initial syllable, except insofar as a final /...iyi/ contracts to ...î:, as in 5g-î: 'become hot' (compare imperative 5gî-yà). Initial long vowels are illustrated by pú:rî 'caress', kó:só 'brush away', bɔ:sî 'mix (crushed millet) with water'. The long vowel is usually followed by a single consonant, but nasal-stop clusters are also allowed: ké:ndé 'make (well)'. There are many trisyllabic verb stems with all-short vowels like jèŋîrⁿî 'look' and jîgîré 'spin'.

Examples of long vowels in noun and other non-verb stems: té:ndî 'wooden bed', bá:sî 'misfortune', kà:sá 'wool' Monomorphemic stems like dóró:sì 'strap for slapping horse', músó:rò 'head shawl' (French *mouchoir*), and gùrá:nâ 'Coran (tome)', with a long vowel in a non-initial, non-final syllable, are borrowings, usually from Fulfulde. Nanga has e.g. pétèy 'flat and wide' and

pótòy 'flat (nose)' corresponding to Jamsay pétè⇒ and pátà⇒ (cf. Ben Tey pátà⇒).

A final falling-tone does not require lengthening of a short vowel: kòrô 'ax', tà:rî 'egg', sàwâ 'grass'. In neighboring languages, either the vowel is long (typical of Ben Tey, e.g. tàrî: 'egg' and sàwâ: 'grass') or the tone is simple (typical of Jamsay, e.g. tárú 'egg'). (However, there are no final short risingtoned syllables.

3.4.2 Nasalized vowels

Vowel nasalization is limited in Nanga. Leaving aside loanwords, affixal contractions, and an occasional expressive reduplication like $s\hat{\mathfrak{d}}^n$ - $s\hat{\mathfrak{d}}^n$ 'newborn', nasalization is confined to long-vowel monosyllables, plus a few expressive adverbials (transcribed with \Rightarrow to indicate prolongation). All examples known to me are in (xx1.a-b). The correlation with open vowels $\{a \ni \epsilon\}$ is typologically normal, but there are also a couple of cases with i.

```
(xxx) a. tá:<sup>n</sup>
                               '(goat) stand on hind legs to browse'
               pá<sup>n</sup>⇒
                               'wide open'
               pá:n
                               'find a mate or double for' (and other meanings)
               jâ:<sup>n</sup>
                               'normal, right'
                               'shed'
               tă:n
               gâ:n
                               'onion'
          b. tè:n
                               'honeycomb'
                               'looking straight at'
               s\acute{\epsilon}^n \Rightarrow
               kê:<sup>n</sup>
                               'inheritance'
                               'fonio' (grain)
          c. pš:n
          d. gĭ:n
                               'odor'
               bĭ:n
                               'cover up'
```

The ubiquitous yes/no and similar "grunted" utterances in (xx2) also show nasalization.

```
(xx2) \delta^n h \delta^n 'yes' \delta^n ? \delta^n 'no' h \check{a} :^n 'huh?'
```

The verb 'shoot' is tá:, cf. Ben Tey and Bankan Tey tá, but Jamsay tá:ⁿ. A similar case is dǐ: 'be tired' (in certain locutions), cf. Ben Tey, Bankan Tey, and Toro Tegu dấ 'be tired', but Jamsay dǐ:ⁿ 'be tired'.

3.4.3 Initial vowels

The word-initial Cv (Cv:, CvC) syllable may have its initial C position vacant, so there are many words beginning with a vowel. In an alphabetical printout of the lexicon (July 2008) I count 7.5 pages of entries beginning with a/a:, 7 beginning with e/e: and ε/ε : combined, 2.5 with i/i:, 9 with o/o: and o/o: combined, and 2.5 with u/u:. The vowel qualities e and o are relatively uncommon. A few examples follow.

```
a/a: àmá 'half-ripe', àgîy 'hold on to', á:njî 'yawn (verb)', á:r¹¹ 'shine'.
e/e: èmbî 'trap', èré 'competition', éw-yé 'sit', émbéré-pè:sì 'hedgehog',
è:njî 'roselle'.

ɛ/ɛ: émbî 'pinch', éwé 'buy', è:mbé 'sorghum', é:nĵî 'tomorrow'.
i/i: ìré 'be better', îsì 'fish', î:-rî- 'cause to stop', î:ndá 'iron'.
o/o: ómbó 'take off (garment)', òmìr¹¹ 'parent-in-law' ò:kî (variant ò:gî)
'tree sp. (Diospyros)', ó:ró 'hyrax, dassie (mammal)'

ɔ/ɔ: ɔ́gî 'hot', ɔ́ndò 'ax', ɔ́:rî '(herder) leave in morning with herd', ɔ́:sô
'tree sp.' (Grewia)'.

u/u: ùsú 'day (unit)', úrɔ́ 'skin and butcher', ú:yî 'be afraid of'.
```

3.4.4 Stem-final vowels (/u/ is rare)

Any vowel quality may end a verb, noun, or other stem. However, stem-final /u/ is effectively absent except in monosyllabic non-verb stems like kû: 'head', and in predicative forms of nonmonosyllabic adjectives whose modifying form ends in /i/. Nouns whose cognates in Jamsay etc. end in /u/ have final /i/ in Nanga: búrî 'bread', ósî 'road', gùsî 'skin', etc. Likewise with adjectives in their basic modifying forms, e.g. dúsî 'heavy', dùgî 'big, fat'. Verbs that end in a high vowel have final /i/, though in some phonological environments this is optionally realized as /u/.

3.4.5 Vocalic harmony

The vowels $\{e \ o\}$ and $\{\epsilon \ o\}$ constitute two opposed harmonic sets. Vowels of either set may co-occur with a, and with high vowels $\{i \ u\}$, which are therefore nonharmonic.

This type of vocalic harmony has little practical significance in Nanga. Within verb stems, the most important "harmonic" pattern is that exactly the same vowel is repeated, as in kémé 'build' and bògóró 'bellow'. Verbal inflectional suffixes do not harmonize with their stems. For example, the Imperfective Negative suffix -ŋò- does not change to -ŋò- after an {e o} stem: (kòyò-kè:sí) kóyó-ŋò: 'he/she doesn't shout'. Causative suffix complex -ndíyé-mî likewise fails to harmonize with the preceding stem in e.g. kéwé-ndíyé-mî 'make equal'. Causative -mî has Imperative -mò that disregards any preceding harmonic vowels: gŏ:-mî 'cause to go in', Imperative gŏ:-mò.

However, there are some harmonic examples involving both vowels of a harmonic set. In the following cases, Nanga o...e corresponds to e...e in neighboring languages: yòrî-yé 'extend credit' (Ben Tey and Bankan Tey yèrì-yé, Jamsay yènè-ŋé), ónjî-yé 'be ashamed' (Ben Tey énjî-yé, Bankan Tey ézî-yé, Jamsay é:-ŋé).

Many loanwords, especially from Fulfulde, respect $\{e \ o\}$ versus $\{\epsilon \ o\}$ vowel harmony, as already in Fulfulde itself. For example, in osiporê 'forestry official' (originally from French *Eaux et Forêts*), we see two o's and an e (along with a nonharmonic high vowel), while in torote or torode 'pestering' two o's co-occur with an ϵ . However, in compounds the two constituents are harmonically independent.

3.5 Segmental phonological rules

3.5.1 Trans-syllabic consonantal processes

3.5.1.1 Nasalization-Spreading

With exceptions discussed below, a nasal $\{m \ n \ n \ n\}$ or nasalized sonorant $\{r^n \ y^n \ w^n\}$ at the onset of a syllable, or a nasalized vowel within that syllable, induces nasalization on an intervocalic oral sonorant $\{r \ y \ w\}$ at the beginning of the following syllable. Phonetically, of course, the nasalization extends to adjoining vowels, so that in $n a r^n a$ 'give birth' all four segments are phonetically nasalized.

Semivowels $\{y \ w\}$ but not r are also (phonetically) nasalized by the same kinds of preceding segments in syllable-final position. I will indicate this syllable-final nasalization in transcriptions, but in stems like $n\acute{u}y^n$ 'go in' one could argue that the nasalization of the semivowel is a phonetic rather than a phonological fact. Semivowels (but not r) are inconsistently nasalized by an immediately following nasal. My assistant nasalized the semivowels in $p\acute{o}y^n\eta g\^{o}l$ 'light on horizon', $h\acute{a}y^nnd\grave{e}$ 'amazement', where the triple cluster is arguably broken after the nasal consonant, and in $t\acute{e}w^n\eta i$ 'rejoin', but not in cases involving simple yn, viz. $l\acute{a}yni$ - 'chant invocations', $d\acute{u}mb\acute{o}$ - $k\grave{e}mb\acute{e}y$ - $n\grave{i}$

'skinny-buttocks', ày-nó 'fatigue', and séynî- 'give good news'. These data are based on the assistant's artificially pronouncing póyⁿ, etc., as separate syllables.

Inside an unsegmentable stem, Nasalization-Spreading is a passive constraint. A sequence like nVr^nV (V=any vowel) as in $n ar^n a$ 'give birth' is acceptable, but a sequence #nVrV with oral /r/ is not. Examples respecting the constraint are $n u y^n a y^n$ 'this year', $m u r^n a$ 'sick person', $n u y^n a y^n$ 'now', and $y u a y^n$ 'a cattle disease'.

The exceptions that I have observed involve the **sequence m...r** with unnasalized r. Thus alongside the "correct" sequence in e.g. dómór¹ó 'criticism' and mòr¹ô 'wild date', there are several cases with "incorrect" m...r. The cases known to me are mùrá- 'not want', dòmúró 'shave around the edges', jòmòró 'foreskin', òmúrô 'tamarind', támórò 'dates', pómúró 'eke out', kómórò '(finger-)nail', màrá 'be lost', mùmúró 'dip (food) deeply (in sauce)', mèrè-gìré 'abdomen', mèrégê 'evil dwarf'. The m in several of these words is optionally (or dialectally) pronounced with a brief oral release, e.g. kómbórò, and there is some comparative support for *mb (or *mb). pá:mîrâl 'understanding' is an unassimilated Fulfulde loanword and shows oral r, but note the parallel borrowing pá:mîr¹è 'understanding' (cf. verb pá:mî 'understand').

The rule does not apply to r at the beginning of a cluster, even when there is a preceding nasal. This is presumably because rⁿ does not occur as first member of a cluster. However, since my examples involve m. the exception may also be related to the comments about m...r above. The examples in my data are marba (variant marpa) 'rifle' and múrsı 'revolt'.

Verbal suffixes beginning with r are subject to Nasalization-Spreading. **Perfective Negative** -ri- is illustrated by dàmá 'speak', Perfective Negative dàmà-rni 'did not speak', and by tá:n- 'spread out fingers', Perfective Negative tà:>-rni 'did not spread out fingers'. In nùynò-rni, which can mean either 'did not enter' (stem núy) or 'did not hear' (stem nǔy), the yn is clearly nasalized. The contrast with núy-èrè 'entered' (discussed below) suggests that the Perfective Negative is particularly favorable to Nasalization-Spreading.

Reversive derivational suffix -rî- is illustrated by kámá 'crumple' and Reversive kámî-rⁿî 'uncrumple', whose Perfective Negative kàmì-rⁿà-rⁿî 'did not uncrumple' shows recursive application of the rule.

Perfective allomorph -èrè is not subject to Nasalization-Spreading even in núy-èrè 'entered' (stem núy, contrast Perfective Negative nùyⁿò-rⁿi mentioned above), much less in e.g. mă:-èrè 'became dry' where the source nasal is farther away. Perhaps -èrè is treated phonologically as though beginning with a (nonnasal) consonant that blocks Nasalization-Spreading.

- 3.5.1.2 Consonantal metathesis in suffixal derivatives of verbs
- 3.5.2 Vocalism of suffixally derived verbs
- 3.5.2.1 Suffixal Vowel-Spreading
- 3.5.2.2 Presuffixal V₂-Raising
- 3.5.3 Vocalic rules sensitive to syllabic or metrical structure

3.5.3.1 Vowel-lengthening

Nanga has small number of stems that appear to show historically secondary lengthening of a vowel before a nasal-stop cluster in the first syllable of a bisyllabic stem.

(xxx)	kó:ndó	'curved (stick)'					
	kè:ndê	'cheek'	Ben Tey cèlê:, Bankan Tey kèndêy			lêy	
	mè:njé	'thin'	Ben	Tey	mènjê-w,	Bankan	Tey
mèzêy	1			-	-		_
	ò:mbó	'chin'	Ben Tey òmbôy, Bankan Tey òmbôy			ıbây	
	sà:mbá	'earth'	Ben	Tey	sùmŏy ⁿ ,	Bankan	Tey
sùmbăy	У						
•	kă:ndâ	'melon'	Ben	Tey	kànár ⁿ ày,	Bankan	Tey
kànđirá	á			-	•		-

There is no synchronic rule of lengthening vowels before such clusters, and some native Dogon items retain short vowels in comparable syllabic positions: néndè 'tongue', yùr-pémbî 'woman's wrap'.

3.5.3.2 Epenthesis

No epenthesis processes have been observed.

3.5.3.3 Post-Sonorant Syncope (verbs)

Many verb stems of more than one syllable end in short i (sometimes varying with u). When followed by a C-initial inflectional suffix like Perfective-2 -só-or Perfective-1b -ti-, the short high vowel optionally deletes in allegro speech after an unclustered sonorant, e.g. dàgiri-só-y ~ dàgir-só-y 'I got ready'. This

optional syncope has no wider phonological consequences; in particular, rhotics do not assimilate to following coronals (§3.5.5.2).

3.5.4 Deletion of final u (u-Apocope)

There is no productive apocope of word-final short high vowels, even after unclustered sonorants. This is seen especially with adjectives like támî 'cold, slow' and its predicative form támû 'be cold, slow'. However, there are occasional forms suggesting (probably low-level) apocope: the first iteration in intensifier pár-párú 'shiny new'.

3.5.5 Local consonant cluster rules

3.5.5.1 Derhoticization ($/r^n/to n$)

In forms like $k\acute{a}r^n\iformalfine{1}{i}$ -'forms like $k\acute{a}r^n\iformalfine{1}{i}$ -'forms like $\'formalfine{1}{i}$ -'forms like $\'formalfine{1}{i}$ -'forms like $\'formalfine{1}{i}$ -'forms like $\'formalfine{1}{i}$ -'forms like 'forms like $\'formalfine{1}{i}$ -'forms like 'forms like

3.5.5.2 Rhotic Assimilation

In forms like $t\acute{ari}$ - $t\acute{i}$ - 'glued on' (Perfective-1b), Syncope of the medial vowel is not phonologically systematic (§3.5.3.3), and the tap r is released prior to the onset of the t in normal pronunciation. See, however, the discussion of Rhotic-Cluster Lateralization, just below.

3.5.5.3 Rhotic-Cluster Lateralization (/rr/→11)

Although there is no regular phonological rule to this effect, there are vestiges of a phonological development of the type ...rv-r... > ...r-r... (Syncope) > ...l-l..., and of the type ...rv-rⁿ... > ...rⁿ-rⁿ... (Syncope) > ...l-l..., if my interpretation of Reversive derivatives is correct. For discussion of Reversive măl-lî- 'unseal' from màrⁿi- 'seal up', and of Reversives like kólli-rî- 'unhook' from kóri- 'hook, hang up', (xx4) in §9.1, below. Since the regular Reversive suffix -rî- has been re-added to e.g. kólli-rî-, native speakers may well interpret the phonological process here as one where medial r is converted to ll before the

suffix (a kind of double dissimilation, in segmental quality and length, to the r of the suffix).

3.5.6 Vowel-vowel and vowel-semivowel sequences

3.5.6.1 Hiatus between adjacent vowels in reduplications

Hiatus is not typical of Nanga phonology, but when a vowel-initial verb stem is reduplicated, the two occurrences of the same vowel are phonetically separated by a glottal stop. Example: éw-yé- 'sit', reduplicated Imperfective è-éw-yé-ŋ 'he/she will sit', phonetic [èʔéwjéŋ]. I do not write the glottal in ordinary transcriptions.

In unreduplicated Perfectives (§10.1.2.2), the 3Sg form may end in ...o-e, ...a-ε, or ...o-ε. The vowel is composite but has no hiatus. Examples: nò-è 'he/she drank', sà-è 'he/she replied', wò-è 'he/she caught'. Similar combinations occurs with Perfective -èrè- (§10.1.2.3), as in gó-èrè- 'went out'.

3.5.6.2 VV-Contraction

3.5.7 Local vowel-consonant interactions

3.5.7.1 Alternations between i and u

Stem- or suffix-final short high vowels $\{i\ u\}$ in verbal morphology usually assimilate to a following suffixal non-homorganic semivowel. This happens, for example, when Perfective-1b -ti- is followed by 2Sg -w (or 2Pl -w.:.), and when bù- 'be' is followed by 1Sg -y (or 1Pl -y.:.). Thus -tù-w, bì-y, see §10.3.3.

These assimilations feed into Monophthongization, see just below.

3.5.7.2 Monophthongization (/iy/ to i:, /uw/ to u:)

In syllable-final position, /iy/ is heard as i:, and /uw/ is heard as u:. In my normal transcription I use the full spellings with vowel and semivowel.

For discussion of the phonological representation of verbs like tîy- (tî:-) 'send', see §10.1.1.

Subject-pronominal suffixes like 1Sg -y and 2Sg -w are often involved in Monophthongization. Examples are 1Sg Perfective Negative -rî-ý [rî:] and 2Sg 'be' bù-w [bù:] 'you are (somewhere)'. The number of forms subject to Monophthongization is swollen by the effects of vowel-semivowel

assimilations, see the immediately preceding section. Thus gò:-rú-w [gò:rú:] 'you-Sg did not go out' from /-rî-w/.

3.6 Cliticization

There is no sharp phonological difference between clitics, suffixes, or postposed particles such as postpositions. I use the term clitic (boundary symbol =) in cases where a morpheme (or morpheme cluster) that would ordinarily be thought of as postposed is attached to an independently occurring stem (noun, verb stem or inflected verb, etc.), with some evidence of phonological interaction.

I transcribe as clitics the following: the inflectable 'it is' morpheme and its negative counterpart (see just below, and §11.xxx); the locational form of place names (§8.1); and the inflectable Past morpheme =be- that is added to (partially) inflected verb forms. Based on phonological interactions, one could argue that some additional postpositions are really clitics.

3.6.1 Phonology of =m clitic ('be')

For the paradigm of this clitic, which allows pronominal inflections, see §11.2.1, below. The first and second person forms are based on =m-. The third person forms (after vowels) are $3Sg = \eta$ and phonetic variants, $3Pl = \emptyset$, and Inanimate =w.

Of phonological interest is the fact that, after a vowel, 1Sg = m-i frequently contracts to =m, while 2Sg = m-i frequently contracts to =m-w. This results in a unique (for Nanga) opposition of final plain versus labialized consonant. The labialization in the 2Sg form is difficult for the unpracticed ear to hear, but native speakers immediately correct poorly pronounced versions.

Some Imperfective inflected forms of verbs also end in an inflected form of =m, so the opposition of plain and labialized is quite important.

3.7 Tones

3.7.1 Lexical tone patterns

3.7.1.1 At least one H-tone in each stem

Regular stems (nouns, verbs, adjectives, numerals) have at least one high-tone component. For example, a monosyllable may be H-, F-, or R-toned, but not L-toned.

3.7.1.2 Lexical tones of verbs

In their lexically basic form (for criteria see §10.1.1), verbs may be {H} toned (xx1.a) or may have a rising {LH} contour. For Cv: monosyllabics, only {H} is observed. Bisyllabics and longers stems, including Cvy reduced from /Cvyi/, {LH} is realized as Cu&y, CvCv, CvCcv, Cvcv, and CvCvcv. While some other Dogon languages of the zone known to me delay the tone break until the final mora of the stem, in Nanga the break occurs at the second vocalic mora, i.e. midway through the long vowel of Cvcv, and in the second syllable of CvCv, CvCCv, and CvCvcv.

```
(xx1)
                    stem
                                            gloss
              a. all-H-toned
                                            'sow, plant (seeds)'
                    tá:
                    núy<sup>n</sup>
                                            'go in'
                    súyó
                                            'hit'
                    kîmé
                                            'tremble'
                    kέsέ
                                            'cut'
                    niy<sup>n</sup>έ
                                            'sleep'
                                            'burn' (also 'give meat to')
                    kúwó-mî
                    éwré-ndîyé
                                            'become small'
```

b. rising-toned gŏ: 'go out' nš: 'drink' 'catch' wŏ: jàrⁿá 'tap' 'fall' yègέ 'run' yàgá bă:rî 'add' (or 'help') 'scrub (one's body)' yă:sî dă:-ndî 'instruct' bă:rá-mî-'have (sb) help (sb)' nàmbi 'stomp on (hide)' 'remove foot from' nàmbi-ri winjé-mi 'swing (a whip)' 'poke with fingers' wègisi dùgú-ndîyé 'become big'

If the stem begins with a stop, we always have all-high tone if the stop is voiceless, and nearly always rising tone contour if the stop is voiced. Occasional exceptions to the latter generalization involve loanwords like já:jíyî- 'come back home from the pasture' and gápé- 'win (match or contest)'. In Nanga the correlation (in verb stems) of tones with initial stop voicing is quite strong.

For vocalic sequences in verb stems, see §10.xxx.

3.7.1.3 Lexical tone patterns for unsegmentable noun stems

"Nouns" is interpreted broadly here, including some noun-like adverbs.

Among monosyllabic nouns (which have at least two moras), R-tone is common but all other possibilities except all-low occur. In (xx1), representative examples are given for H, F, and R, along with all known examples of <LHL> (excluding deverbal derivatives).

(xx1) Monosyllabic nouns

stem	gioss
a. H (rare) gáy ⁿ	'courage
b. F	
pî:	'wealth'

```
jâ:n
                           'just deserts'
                           'large awl'
       gô:
c. R
       tă:n
                           'shed'
                           'horse'
       sŏŋ
                           'fence'
      jă:
                           'hip'
      dŏ:
                           'hand'
       nă:
d. <LHL> (all known exx.)
                           'fire'
       gõ:
       tε̃:n
                           'honeycomb'
       mε̃:
                           'soft mud'
                           'awareness'
       sõ:
       g \tilde{\mathfrak{I}} y^n
                           'last year'
       dèw
                           'trap'
```

See also the discussion of Cv-Cv: reduplicated nouns in §3.xxx.

For bisyllabic nouns, the monotonal contour is HH. Bitonal contours may be of the general type {HL} or {LH}. {HL} is realized as HL or HF. HL is typical of bisyllabic nouns with heavy first syllable and light second syllable, i.e. CvCCv or Cv:Cv. HF is usual for bisyllabic nouns with two light syllables, i.e. CvCv, or with a final heavy syllable, e.g. CvCvC or CvCv:. A final falling tone does not require lengthening of a final short vowel. {LH} contour is realized as LH or as LR. The only attested tritonal contour is {LHL}, which for bisyllabic nouns is realized as RF or LF. There may be one case of quadritonal {HLHL} in 'ostrich' (xxx.h) with H<LHL> tones, but this may well have originated as a compound.

(xx2) Bisyllabic nouns

```
stem gloss

a. {H} as HH
gîr"á 'harvest'
ségî 'dues'
má:ndî 'belief'
wé:rî 'tendon'
tó:sî 'testicles'

b. {HL} as HF
```

```
CvCv
                         'flour'
      pîr<sup>n</sup>â
      mîr<sup>n</sup>â
                         'voice'
                         'griot's calling out of lineage'
      tîgâ
                         'francolin'
      súgî
                         'squirrel'
      kîyâ
 final heavy syllable
      sápôl
                         'row'
                         'curiosity'
      kúmpâm
      sînwâ:
                         'Chinese (person)' (Fr chinois)
                         'modern rifle (type)'
      báykâl
c. {HL} as HL
  Cv:Cv
                         'money'
      bú:đi
  CvCCv
                         'customary rite'
      témbi
  CvCCCv
      háy<sup>n</sup>ndὲ
                         'amazement'
d. {LH} as LH
 final Cv
      kùw<sup>n</sup>á
                         'crowned crane' (bird)
                         'bird'
      li:gi
                         'mourning'
      pàndi
                         'rainbow'
      nà:mbá
      nèr<sup>m</sup>i
                         'dog'
e. {LH} as LR
 final CvC
      lèrěw
                         'entirely' (adverb)
f. {LHL} as LF
  CvCv
      jàmâ
                         'crowd'
      dùr<sup>n</sup>ô
                         'eagle-owl'
                         'act of forging'
      dùwâ
  CvCCv
                         'rifle'
      màrpâ
                         'leech'
      tòndô
                         'chicken'
      ènjê
  Cv:Cv
```

```
'egg'
      tà:rî
                       'kneading stick'
      kò:rô
 Cv:CCv
                       'worm, grub'
      ti:nji
 CvCv:
                       'frog'
      òηî:
      sàgô:
                       'starling sp.'
f. {LHL} as RF
 CvCCv
                       'God'
      děnjê ~ jěnjê
 Cv:Cv
      kě:rê
                       'money'
                       'steam-cooked millet meal'
      să:râ
                       'young (person)'
      sŏ:rô
h. {HLHL} as H<LHL>
 CvCv:
      sákő:
                       'ostrich' (variant ságo:)
```

The bisyllabic data, especially the HF pattern, suggest a tendency for tone contrasts to be expressed **near the right edge**.

Trisyllabic noun stems show more clearly the details of how the tone break in a bitonal contour pushes toward the right edge. HHH is well-attested but not very common (xx3.a); it is more typical of verbs. The basic falling contour, abstractly {HL}, is **HHL** (xx3.b) which corresponds structurally to bisyllabic HF. This suggests that the H of {HL} can push up to one full syllable to the right from its onset, but is then bounded. However, I did record **HHF** primarily in trisyllabic stems (mostly borrowed from Fulfulde) ending in a CvC syllable, even if the final consonant is not always pronounced in Nanga (xx3.c). The basic rising contour, abstractly {LH}, is **LLH** (xx3.d), not #LHH as for verbs.

Of the possible **tritonal contours** {LHL} and {HLH}, only {LHL} is well-attested. It is normally realized as LHF or as LLF (phonologically unpredictable, therefore lexical choice). 'Mango' is RHL (xx3.The other tritonal possibility, {HLH}, is attested in two nouns (xx3.h).

(xx3) Trisyllabic nouns

```
stem gloss

a. {H} as HHH
tá-tágá 'arrogance'
```

```
pógiri
                         'belly strap'
      \acute{a}d\acute{u}r^n\acute{o}
                         'life'
      dáŋgárá
                         'thigh'
b. {HL} as HHL
final Cv
      úsúr<sup>n</sup>ò
                         'wind'
                         'day laborer'
      sámár<sup>n</sup>i
                         'remorse'
      dúgúri
      dóró:sì
                         'strap'
c. {HL} as HHF
final C\hat{v}C
      tógîyêm
                         'pot used to heat metal underneath'
                         'first rains'
      kórósôl
      pá:mîrâl
                         'understanding'
final C\hat{v} from *C\hat{v}C or *C\hat{v}:
      súgúlâ
                         'remorse' (variant súgúlâm)
      bálámbâ
                         'champion' (variant bálámbân, cf. Ben Tey
                         bálámbâ:)
      káráwâ
                         'wooden milk bucket'
                                                        (variant ká:râ,
                         Fulfulde karawal)
d. {LH} as LLH
                         'domestic animal'
      gàsègé
      nàr"iy" é
                         'orphan'
      àr<sup>n</sup>àw<sup>n</sup>5
                         'marabou stork'
e. {LHL} as LHF
      dòndîyê
                         'cat'
      gòmbórô
                         'chest'
                         'Islamic law'
      sàriyê
      wòtúmbâ
                         'mound'
f. {LHL} as LLF
                         'rifle-cock tightener' (màrpâ 'rifle')
      màrpà-jigirî
                         'craziness'
      pì-pirî
      dògòrô
                         'cave cemetery'
                         'finger'
      làsirî
      sò-sòrî
                         'branch whip'
      bàgàrâ
                         'hubbub'
                         'pistol'
      pitòlô
```

```
nèmbirê
                      'act of pleading'
g. {LHL} as LHL
     làsá:sì
                      'modern rifle'
h. {LHLH} as RLH (only example)
                      'mango' (with variants)
     măngòró
i. {HLH} as HLH (only example)
     héyyèndé
                      'index finger'
```

Quadrisyllabics are prosodically composite, or borrowed (usually from Fulfulde). If composite (usually the break is in the middle), we can get tone contours like LL-HH that ostensibly violate the rule that tone contrasts are pushed to the right edge. This is of course because rules that apply to unsegmentable stems do not apply across compound boundaries. Quadrisyllabic borrowings usually have HHHL or HL-HL contours (the latter suggests that native speakers structure them prosodically like compounds).

3.7.1.4 Lexical tone patterns for adjectives and numerals

Adjectives and numerals have basically the same tone types as nouns, allowing for gaps due to a limited inventory, and due to a predominance of mono- and bisyllabic rather than longer stems.

Examples, beginning with monosyllables, are in (xx1).

(xx1) Adjectives and Numerals

stem	gloss
monosyllabic, R pě: yă: wŏy	'old' 'female' 'two'
bisyllabic, HH	
sóró	'straight'
úmá	'alive'
pîrî	'white'
pé:ré	'innocent'
nánáy ⁿ	'respectable'

```
bisyllabic, HF
      \acute{a}r^n \hat{a}
                          'male'
      órî
                          'smooth, sleek'
      kúrê
                          'six'
      pέ:rù
                          'ten'
bisyllabic, LH
                          'easy'
      nà:rná
      kùnjú
                          'middle-aged'
      dùgî
                          'big'
                          'thousand'
      mùsú
bisyllabic, LF
      tègê
                          'young'
      nòmî
                          'difficult'
      gòmî
                          'bad'
bisyllabic, LR
                          'five'
      nimi:
      tè:sĭ:
                          'nine'
```

3.7.1.5 Default H-tone, or autosegmental mapping?

Particularly for verbs, the effective restriction of lexical tones to {H} and {LH} gives us the option of taking the H-tone element as a default that need not be specified lexically. In this view, the {LH} verbs are those that have an initial L-tone element.

It would be much more difficult to make such an analysis work for non-verb stems, which have a greater range of lexical tone contours.

3.7.1.6 Tone-break location for bitonal non-verb stems

For monosyllabic stems, a contour tone {HL} or {LH} is of course realized on the single syllable: tă:" 'shed', pî: 'wealth'.

Bisyllabic and longer non-verb stems (nouns, adjectives, numerals) with {LH} contour have the tone break **near the right edge**: CvCv, CvCcv, Cv:Cv, Cv:Cv, Cv:Cv;, CvCvCv, Cv:Cv;, CvCvCv, Cv:Cv;, cvCvCv, Examples are kùwⁿá 'crowned crane', pàndí 'mourning', nò:mbó 'rainbow', gàsègé 'animal'.

Bisyllabic and longer non-verb stems with {HL} contour likewise have the break near the right edge. For prosodically light CvCv and nCv stems (with short vowels), the usual pronunciation is CvCv and nCv, with the H-tone component spreading into the onset of the second syllable. Bisyllabics with a final heavy syllable likewise have the tone break just before the final mora. Bisyllabics with a heavy first syllable and light second syllable have the break at the syllable boundary. Examples: súgî 'francolin', bú:dì 'money', sînwâ: 'Chinese person'. Trisyllabics are usually HHL-toned if they have a final light syllable, and HH<HL>-toned if they have a final heavy syllable, but some loanwords (mostly from Fulfulde) that have lost a final sonorant are still pronounced with a final falling tone; see (xx3.c) in §3.7.1.3.

The tendency of tone breaks in the basic lexical forms of noun, adjective, and numeral stems to occur as close as possible to the right edge is also observed with most {HL} and {LH} contours that are overlaid on such stems by morphosyntactic context (§xxx).

3.7.1.7 Tone-break location for tritonal non-verb stems

{HLH} is very rare. I can cite héyyèndé 'index finger', which occurs with similar tones in some other languages.

{LHL} is more variable in tone-break location than the other tone contours considered above. One consistent pattern is CvCv for bimoraic CvCv stems, where the final vowel has falling tone. Likewise CvCv: when the first syllable is light and the second is heavy. Examples: dùrnô 'eagle-owl', sàgô: 'starling sp.'. However, CvCCv can appear, depending on the lexical item, as CvCCv or CvCCv, compare tòndô 'leech' and màrpâ 'rifle' with děnjê (variant jěnjê) 'God'. Likewise with Cv:C(C)v stems: tà:rî 'egg' and kò:rô 'kneading stick', but sŏ:rò 'young (person)' and kě:rè 'money'. There is a similar lexical choice of tone-break location in {LHL} trisyllabics, namely between LH<HL> with the tone break near the left edge and LL<HL> with the tone break as close as possible to the right edge. Examples are dòndîyê 'cat' and nèmbìrê 'act of pleading'. The LHL contour in làsá:sì '(modern) rifle' is a variant on LH<HL> when the medial syllable has a long vowel.

3.7.2 Grammatical tone patterns

3.7.2.1 Grammatical tones for verb stems

Verbs are lexically {H} or {LH}. The tone break in {LH} stems, as seen in the bare stem (e.g. in verb chains) and in positive Perfective forms, is near the left edge: Cv;, CvCv, Cv:Cv, CvCv, CvCv, etc. For examples see §10.1.3.

3.7.2.2 Grammatical tones for noun stems

When preceded by a possessor NP, the possessed noun is subject to an **overlaid** $\{HL\}$ **tone contour**. However, if the possessor NP ends in a L-tone component (i.e. in a syllable with L, F, or <LHL tone), the possessed noun loses its initial H-tone component and is therefore all-L. Since pronominal possession ('my house', 'their dog') is expressed by a postposed element (e.g. 'house my-thing' = 'my house'), the possessed noun in this case has its regular lexical tones. These patterns are illustrated in (xx1), where the noun 'cat' has LHF tone in isolation and before the pronominal possessor, but appears with HHL tone after 'woman' and with all-L tone after 'man'.

```
(xx1) a. dòndiyê 'cat' 'my cat'

b. yă-ŋ dóndiyè 'a woman's cat' 'a man's cat'
```

A fuller set of nouns is given in (xx2) to show the phonological realization of the basic {HL} possessed-noun pattern. (xx2.a) shows that a monosyllabic lexically F-toned noun has no audible change when the possessed-noun {HL} contour is superimposed. In (xx2.b), the tonal overlay is audible but phonetically subtle, from unpossessed HF to possessed HL. In (xx2.c), the noun has unpossessed and possessed HHL, so as in (xx2.a) there is no audible difference. In (xx2.d), the tonal contrast between unpossessed and possessed forms is clear since the unpossessed noun has a contour other than {HL}.

(xx2) §		gloss	lexical form	rm possessed (after yǎ-ŋ 'woman'	
	a.	'wealth'	pî:	yă-ŋ pî:	
	b.	'house' 'money'	ńdô bú:đi	yă-ŋ ńdô yă-ŋ bú:đi	

c.	'donkey'	súmáŋà	yǎ-ŋ súmáŋà
d.	'dog'	nèr ^m i	yă-ŋ nér ⁿ î
	'courtyard'	dámbî	yǎ-ŋ dámbì
	'blood'	gòndùgó	yǎ-ŋ góndúgò
	'mother'	dě:	yǎ-ŋ dê:
	'hawk'	tè-têw	yǎ-ŋ té-têw

As indicated above, all of the possessed forms drop to all-low when the preceding possessor NP ends in a L-tone component: árnâ gòndùgò 'a man's blood', etc.

A noun stem drops its tones to all-L when followed by a modifying adjective, a Definite morpheme, or a demonstrative pronoun. Thus $n \tilde{\epsilon} r^{n} \tilde{i}$ 'dog' becomes low-toned in $n \tilde{\epsilon} r^{n} \tilde{i}$ 'a good dog', $n \tilde{\epsilon} r^{n} \tilde{i}$ n\(\tilde{\epsilon}\) 'the dog', and $n \tilde{\epsilon} r^{n} \tilde{i}$ w\(\tilde{\epsilon}\) 'this dog'. See \(\xi xxx\), below.

A noun does not interact tonally with a following cardinal numeral or other quantifier: nèrⁿì kúrê 'six dogs', nèrⁿì kéréw 'all the dogs'.

A noun that has escaped tone-dropping within the core NP or from a possessor is tone-dropped as head of a relative clause: nèr^{tr}i î: sùyò-sè né 'the dog that we hit-Past'. See §14.xxx.

3.7.2.3 Grammatical tones for adjectives and numerals

An adjective drops its tones when followed within the core NP by another modifying adjective. Thus ner'i bar'i 'a red (= brown) dog' has a high-toned adjective bar'i, but this drops to all-low toned when a second adjective is added: ner'i bar'i esi 'a good red (= brown) dog'.

A modifying adjective or a cardinal numeral following a noun is within the domain of the {HL} or all-low possessed-noun contour imposed by a preceding possessor NP. The noun appears with {HL} or all-low contour depending on the final tone of the possessor NP. Any adjective or numeral that follows the noun appears as all-low. Thus yǎ-ŋ [nér"i bàr"i] 'a woman's red (= brown) dog' (bár"i 'red'), yǎ-ŋ [nér"i kùrè] 'a woman's six dogs' (kúrê 'six'), yǎ-ŋ [nér"i bàr"i kùrè] 'a woman's six red (= brown) dogs'. See §xxx, below.

An adjective or cardinal numeral within an unpossessed NP functioning as head NP of a relative drops its tones. [nèr"i bàr"i] î: sùyò-sè né 'the red (= brown) dog that we hit-Past' (nèr"i bár"i 'red dog'), [nèr"i kùrè] î: sùyò-sè né 'the six dogs that we hit-Past' (nèr"i kúrê 'six dogs').

3.7.3 Tonal morphophonology

3.7.3.1 Autosegmental tone association (verbs)

In the lexically basic form, verbs have either all-high {H} or rising {LH} tone contour. The {H} contour spreads over the entire verb stem. The {LH} contour is realized as <LH> on monosyllabic stems, as LH on bimoraic bisyllabics (CvCv) and on CvCCv bisyllabics, as LH(H...) on trisyllabic or longer stems with initial short-voweled Cv or CvC syllable, and as <LH>H(H...) on bisyllabic or longer stems with initial-syllable long vowel. The generalization is that the break from low to high tone in the {LH} contour is before the second vocalic mora.

3.7.3.2 Break point in {HL} and {LH} overlaid tone contours

This section considers the tonal break point between H and L when a {HL} or {LH} contour is overlaid on a stem or word. The most useful stems to test are trisyllabics, e.g. CvCvCv, since these stems allow us to determine whether the break point remains near the left edge or near the right edge. The patterns are summarized in (xx1); see the indicated sections for the data. Except for one stylistically marked narrative verb-stem iteration, which limits the initial H of {HL} to one syllable, all overlaid {HL| and {LH} contours have break points after the second syllable in trisyllabic stems. In other words, the location of tone break is defined with respect to the right, not left, edge.

(xx1) {HL} contour tonal break point

grammatical context	CvCvCv tone
a. {HL} contour	
break point up to 2 syllables from left	
possessed noun after H-final possessor	[CýCý][Cỳ]
possessive-type $(\bar{n} \ \hat{n})$ compound (§5.1.4)	[CýCý][Cỳ]
bahuvrihi (n â) compounds (§5.2.1)	[CýCý][Cỳ]
Imperative of {H}-toned stems (§10.6.1.1)	[CýCý][Cỳ]
break point one syllable from left	30 3
first of 2+ iterated verb stems (§11.6.1)	[Cý][CỳCỳ]
b. {LL} contour	
break point up to 2 syllables from left	
agentive $(\bar{x} \hat{v}\text{-Ppl})$ compounds (§5.1.5)	[CvCv][Cv]

When the relevant stem has only one or two syllables, the break point is positioned in such a way that the final tone element is not obliterated. For the narrative verb-stem iterations, this is an issue only for monosyllabic verbs, which have a falling tone on the first iteration, e.g. $\hat{C}\hat{v}$:

For the {HL} patterns that have trisyllabic HHL realizations, we again get falling-tone on monosyllabics: $C\hat{v}$:, $C\hat{v}C$. Bisyllabics with heavy first syllable and monomoraic final syllable, i.e. Cv:Cv, CvCCv, and Cv:Cv, position the tone break at the internal syllabic boundary (Cv: $C\hat{v}$, etc.). Bisyllabics with final bimoraic syllable, e.g. CvCvC, CvCv:, CvCCvC, have the tone break inside the second syllable ($C\hat{v}C\hat{v}C$, etc.). Light bisyllabics with two monomoraic syllables, i.e. CvCv and vCv (with syllabic nasal), are usually pronounced (prepausally) with the H-tone spilling into the second syllable, but not obliterating the terminal L-tone, resulting in a final falling tone: $C\hat{v}C\hat{v}$, \hat{v} Cv. However, if such a light bisyllabic is a compound initial, or is followed by a modifying adjective, it is normally heard as $C\hat{v}C\hat{v}$, \hat{v} Cv.

For the {LH} pattern (deveral agentives), monosyllabics have rising tone (\check{Cv} :), and all bisyllabics have the break at the internal syllable boundary ($\check{Cv}\check{Cv}$). Since the inputs here are verb stems, they are either \check{Cv} : monosyllabics, or longer stems ending in a short \check{Cv} syllable, so they do not present the same full range of stem shapes as do nouns.

There are few non-verb stems that have four or more syllables and that are not treated phonologically as compounds. However, there are a few, and they too show the tone break near the right edge: yǎ-ŋ bîsîyémì 'a woman's acacia' (bîsîyémì), see §6.2.1.1. Because derivational suffixes can be stacked up in verbs, it is easier to find quadrisyllabic and even longer verbs, and their Imperative stems show the same right-edge tone break: éwré-ndîyé-mò 'make (it) small(er)!'.

Examples showing the primary tone-break patterns are in (xx2). $\{HL\}$ is illustrated with bahuvrihis. The $\{LH\}$ examples are agentives. The second stem in each compound is the relevant one.

```
a. monosyllabic
(xx2)
         {HL}
                            'two-headed' (wŏy 'two')
            kû:-wôy
         \{LH\}
            tònđi-tě:
                            'basket-weaver'
        b. bisyllabic
         {HL}
            kû:-dúgî
                            'big-headed' (dùgi 'big')
                            'five-armed' (nimi: 'five')
            nă:-nîmî:
            sègé-mé:njè
                            'thin-skeletoned' (mè:njé 'thin')
         {LH}
```

```
nàmà-tùrí 'meat-seller, butcher'

c. trisyllabic
{HL}
kû:-púrúgǐ 'with an off-white head' (púrúgî 'off-white')
{LH}
bǐdīgà-bǐdīgî 'magician'
```

3.7.3.3 Tone-Dissimilation (decimal numerals)

pé:rù 'ten' forms compounds with following single-digit numerals to produce '20', '30', ... '90' (§4.7.1.3). The 'ten' word appears segmentally in these compounds as peri- ('20', before a /y/), per- ('60' through '90', before velar or cornoal {k g s t}), and pe- ('30' through '50', before coronal {n t}). There is a close, but not perfect, inverse correlation of the final tone of 'ten' with the initial tone of the following numeral. The correlation is seen in pèri-yèy '20' (cf. wòy '2'), pé-tà:ndǐ: '30', pěn-nòyⁿ '40', pěn-nìmǐ:ⁿ '50', pèr-kúrê '60', pèr-súyê '70', and pèr-gá:rè '80', but the final decimal numeral in the series is an exception: pèr-tè:sǐ: '90'. Because numeral sequences like this are often recited in an incantational fashion, adjacent numerals often share phonological properties, and '90' may simply be following the shape pèr- of the three preceding members of the series.

3.7.3.4 Atonal-Morpheme Tone-Spreading

Atonal morphemes (suffixes, clitics, particles, postpositions) acquire their surface tone by spreading from the final tone of the morpheme on the left. The morphemes in question are mostly monosyllabic, but /yaŋa/ 'also, too' is bisyllabic, and both of its syllables get the tone that spreads from its left.

The process applies systematically to atonal suffixes and clitics. Relevant suffixes in verbal morphology are 1Sg -y, 2Sg -w, and some allomorphs of the 3Pl suffix. (3Sg suffix is zero, and 1Pl and 2Pl suffixes have their own unusual pitch contour.)

For nouns and pronouns, Accusative suffix -ŋ is atonal and therefore gets its tone by spreading.

The relevant clitics are forms of the 'it is' clitic ($\equiv m$ -, $\equiv w$ -, etc.). For example, with nă: 'hand' we get nă: $\equiv \dot{w}^n$ 'it is a hand', and with nàmâ 'meat' we get nàmâ $\equiv \dot{w}^n$ 'it is meat'.

Particles and postpositions that get their tones by spreading include Interrogative ma and Quotative wa. Interrogative ma in particular is highly

subject to intonational modification, so the (phonological) tone-spreading is not always clearly audible.

When tones spread from a stem or word that ends in a vowel with a contour tone, and when the targeted suffix or clitic consists of just a consonant, the resulting CvC or Cv:C syllable may require a phonetic adjustment regarding the location of the tone break; see Contour-Tone Stretching (§3.xxx, below).

Up to a point, the process is recursive. For example, in a word like yě:-só-w 'you-Sg came', the final -w morpheme gets its high tone from -só-, and when e.g. Quotative particle /wa/ is added, the high tone spreads to this particle: yě:-só-w wá '(it was said) you came'.

However, in the combination of Accusative -ŋ and /yaŋa/ 'also, even', the latter is low-toned even when the former has high tone due to spreading; nji-ŋ yaŋa 'me too' (§19.1.3). One way to interpret this is to take the Accusative as having underlying low tone.

3.7.4 Low-level tone rules

3.7.4.1 Rising-Tone Mora-Addition

A final **falling tone** may occur on a short Cv syllable; there is no lengthening (mora-addition) of this vowel. In addition to $C\acute{v}C\^{v}$ and $C\check{v}:C\^{v}$ stems (lexical nouns, some Imperative verb forms), where the final falling tone represents the spread of the preceding high tone into the onset of the final syllable, there are nouns like $\grave{u}s\grave{u}r\^{i}$ 'broom' that clearly have a falling tone expressed entirely on the final short vowel.

However, a **rising tone** cannot be expressed on a single mora, i.e. in a Cv syllable with short vowel. In cases like $n\check{u}$: 'person', preadjectival form $n\grave{u}$ with short vowel ($n\grave{u}$ èsî 'a good person'), it is reasonable to take $/n\check{u}/$ as the basic lexical representation, even though it never occurs on the surface as such. When tone-dropping applies, producing $n\grave{u}$ (e.g. before an adjective), there is no need to lengthen the vowel. When the lexical rising tone surfaces, the short vowel must be lengthened.

(xx1) Rising-Tone Mora Addition

In a final monomoraic Cv syllable, a rising tone forces addition of one mora (i.e. the vowel is lengthened).

The phonological behavior of nu: 'person' is atypical for a monosyllabic. Most monosyllabic nouns of the shape Cv: are not shortened when tone-dropped (or when an {HL} tone contour is overlaid on them). This suggests that

the vowel length of most Cv: stems is lexical rather than a product of Rising-Tone Mora Addition. Examples: ŏ: '(the) bush, outback' and compound ò:-dárⁿá 'full outback', ná: 'hand' and compound nà:-kéndè 'palm of hand'.

For non-verb stems of two or more syllables, final long vowels are uncommon, but there are cases of final \check{v} : that shorten when an overlaid contour converts it from <LH> to either L or <HL> tone. The two numerals in (xx2) have long, <LH>-toned final vowels in isolation or phrase-finally, but in bahuvrihi compounds, which overlay {HL} on the final stem, the final vowel appears as short and L-toned.

```
(xx2) stem gloss with {HL}-overlay

nimi: 'five' nă:-nîmî 'five-armed' (bahuvrihi)
tà:ndî: 'three' kû:-tá:ndî 'three-headed' (bahuvrihi)
```

I know of no stem with final long falling-toned vowel that shortens when it drops to L-tone. Nonmonosyllabic stems with a final long falling-toned vowel are rare and apparently confined to loanwords. An example is parpæ: 'perfume', which end in a full-fledged long vowel that does not shorten: yǎ-ŋ párpæ: 'a woman's perfume'. Others that behave similarly include the fauna terms ɔ̀ŋî: 'frog' and sàgô: 'blue-eared starling'.

Verb stems do not meet the phonological conditions for the shortening.

3.7.4.2 Contour-Tone Stretching

The final syllable of a word may contain a Cv or Cv: with a contour tone plus a suffixal or clitic sonorant consonant (atonal or tonal). A contour tone <HL> or <LH> in such a syllable is pronounced with the tone break near the end of the syllable.

For example, in combinations with the 'it is' clitic, the final semivowel bears the second tone component in combinations like $n\check{a}:=\check{w}^n$ 'this is a hand' and $n\grave{a}m\hat{a}=\check{w}^n$ 'it is meat'. I transcribe the noun stem in such cases with its regular (word-final) tones, but the actual pronunciations are closer to $[n\grave{a}:\check{w}^n]$ and $[n\grave{a}m\acute{a}\check{w}^n]$. So a minor rule stretching the tone break to just before the final mora is needed.

3.7.4.3 Final-Tone Resyllabification

There are no combinations of a CvC or CvC stem with a vowel-initial clitic or suffix that would trigger a resyllabification and consequent tonal jump.

3.7.4.4 Rightward H-Spreading

Whether special tone-spreading rules are needed for Nanga depends on a number of analytical decisions. If we take lexical tone contours to be mechanically separate from syllables and segments, we will need some rules or constraints to account for the surface tones. If the tones are already associated with segments or syllables in the lexicon, those rules are not needed. Similarly, how we analyse the imposition of possessed-noun tone contours, and of tone-dropping and other tonosyntactic processes, determines the need for spreading rules and their precise formulation.

3.7.4.5 Floating-Tone Linking

There are no systematical vowel-deletion rules (syncope, apocope), so there are no contexts where a tone stranded (de-linked) by such a deletion must relink to the left or right.

However, the 1Sg possessor morpheme is a floating L-tone preceding the possessive classifiers $g\hat{\sigma} \sim k\hat{\sigma}$ (inanimate) and $y\hat{\epsilon}$ (animate). The resulting forms are $k\hat{\sigma}$: and $y\hat{\epsilon}$:, respectively, i.e. with <LHL> tones (§6.2.1.3, §6.2.2.1).

3.7.4.6 Final-Cv R-to-H Reduction

I know of no clear cases where a final rising tone is reduced (e.g. in a monomoraic syllable) to a high tone. This occurs in some other Dogon languages in nominal morphology, where a noun stem (arguably /CvCv/lexically) is realized as CvCv- without suffix and as CvCv-x with a suffix

consisting of a (nasal) consonant (x). The only noun in Nanga that has such a suffix is /yă/ 'woman', with its archaic singular yă-ŋ and its unsuffixed plural yă:, but here (perhaps since the stem is monosyllabic) the unsuffixed form is lengthened, allowing full expression of the rising tone. See Rising-Tone Mora-Addition (§3.xxx, above).

3.8 Intonation contours

3.8.1 Phrase and clause--final nonterminal contours $(\uparrow, \Rightarrow, \Rightarrow \uparrow, \Rightarrow \searrow)$

In texts, and to a lesser extent in elicited sentential examples, arrows are used to indicate intonationally significant high (\uparrow) or low (\downarrow) terminal pitch, prolongation of the final syllable (\Rightarrow), or the combination of prolongation with either high ($\Rightarrow \uparrow$) or low ($\Rightarrow \downarrow$) terminal pitch. Prolongation and/or high terminal pitch are typical of nonfinal phrases or clauses in parallel with a final phrase or clause, which may have ordinary phrasal intonation or may have unusually low pitch. Intonational prolongation is also common with interrogative particle /ma/ and 'or' disjunctive particle /ma/, which are not clearly separable.

3.8.2 Adverbs and particles with lexically specified prolongation (⇒)

Prolongation of the coda of the final syllable is also baked into certain adverbials as part of their lexical form. Fewer forms are affected in Nanga than in, say, Jamsay. The Jamsay adverbials $d\acute{e}m \Rightarrow$ 'straight (direction)' and $d\acute{e}y^n \Rightarrow$ 'apart, separate' are pronounced with prolongation of the final sonorant, but in Nanga I hear $d\acute{e}m$ and $d\acute{e}y^n$ without notable prolongation.

I did observe prolongation in some other expressive adverbials and adjectival intensifiers, e.g. késékéréy⇒ 'very dry (intensifier)', pɔ́tɔ̂⇒ 'flat and small', and bende-bende⇒ 'brick-shaped'. See §xxx for more examples.

3.8.3 Dying-quail intonational effect ∴ (1Pl, 2Pl)

A dying-quail intonation pattern by which the final syllable of a word is prolonged and undergoes a slow pitch decline has been found in Jamsay (most systematically at the end of each coordinand in a pronominal or NP conjunction). It also occurs in Ben Tey, where it converts singular to plural pronominal categories for first, second, and Logophoric persons. The symbol for this is \therefore , placed at the end of the word.

In Jamsay, only the coda of the final syllable of the word is affected, so if the word ends in a CvC syllable with final sonorant, as in $n\check{\epsilon}$ -m 'women' as a coordinand, the prolongation and final pitch decline in the conjoined form $n\check{\epsilon}$ -m. apply to the m, not to the syllabic nucleus ϵ . In Jamsay, the initial pitch of the relevant syllable respects the phonological tone, which may be high or low. If high, the pitch decline is conspicuous; if low, the main audible effect is prolongation.

In Nanga there is an analogue to this, but it is somewhat different phonetically. It is limited to verbs and clitics conjugated for 1Pl and 2Pl subject, which (as in Ben Tey) are intonational modifications of the corresponding singulars. The independent pronouns, 1Pl î: (cf. 1Sg ĭ:n), 2Pl û: (cf. 2Sg ú), and Logophoric Pl â: (cf. singular á) are also (in part) tonal modifications of the singulars. However, in Nanga I did not observe the exaggerated prolongation and corresponding slow pitch decline in these pronouns, as in the cases I consider to have the dying-quail feature.

In verbal inflections, however, we do get terminal pitch patterns close to the dying-quail examples in the other languages. Consider the data in (xx1), where the square brackets simulate the phonetic pitch effects, and \therefore in the phonological transcription indicates dying-quail intonation.

```
1P1
(xx1)
                  1Sg
                        2Sg
                                             2P1
           sùyè- 'hit' (unsuffixed Perfective)
                                       sùyò-ỳ∴ [sùjòóòj]
                  sùyà-ỳ
                       sùyò-w
                                             sùyò-w∴ [sùjòóòw]
          súyó-jè- 'have already hit'
                  súyó-jè-y
                                       súyó-jè-ỳ∴ [sújódʒèéèj]
                       súyó-jè-w
                                             súyó-jè-w∴ [sújódʒèéèw]
       c. súyó-só- 'hit' (Perfective-2)
                  súyó-só-ý
                                       súyó-só-ý∴ [sújósòóòj]
                       súyó-só-w
                                             súyó-só-w∴
                                                          sújásòóòw
```

In these transcriptions, it should be understood that \therefore override the phonological tone shown in the final syllable (and earlier syllables under some conditions). Therefore we get phonetic [LHL] pitch not only in (xx1.a-b), where the singulars show L-toned final syllable, but also in (xx1.c), where the singulars have H-toned final syllable.

Consider now (xx2). In these paradigms, observe in particular that the 1Pl and 2Pl forms show a **high tone on the penultimate syllable**, including the nasal onset in (xx2.a), in contrast to a low tone in the 1Sg and 2Sg (and the third

person forms, not shown). In addition, the stem-final vowel is lengthened in the Perfective Negative.

```
(xx2)
           form 1Sg 2Sg
                                      1Pl 2Pl
       a. ngó- 'not be (somewhere)'
                 ngó-ý
                                      ŋgó-y∴ [ŋgòóòj]
                       ngó-w
                                           ŋgó-w∴ [ŋgòóòw]
       b. mbură- 'dislike' (§11.2.6.3)
                 mbùrà-ý
                                     mbùrà-ý∴ [mbúrààj]
                       mbùrà-w
                                           mbùrà-w∴ [mbúrààw]
       c. sùyò-rî- 'did not hit'
                 sùyò-rî-ý [sùjòrî:]
                                     sùyò-ri-y∴ [sùjó:ri:]
                       sùyò-rú-w [sùjòrú:] sùyò-rú-w∴ [sùjó:rù:]
```

These forms create a transcriptional conundrum. One choice is to transcribe more or less phonetically, e.g. 1Pl sùyó-rǐ-y for 'we did not hit'. The other is to transcribe the 1Pl and 2Pl forms as intonational modifications of the corresponding singulars, e.g. 1Pl sùyò-rî-y:, bringing out the morphophonological structure but disguising the phonetics. I choose the latter course.

In (xx2.a), the dying-quail intonation actually is quadripartite [H-LHL], with the initial high pitch realized on the penultimate syllable. In (xx2.b-c), dying quail is [H-L] if analysed as being realized on the final two syllables, or [L-H-L] if analysed as being realized on the last three syllables.

In (xx2.c), I can hear no lengthening of 1Pl -rì-ỳ [rì:] vis-à-vis 1Sg -rî-y [rî:], the latter already being pronounced like a long vowel (as the /iy/ sequence monophthongizes). Likewise for 2Pl versus 2Sg.

The various realizations of dying-quail intonation are summarized in (xx3). The 2Pl forms, not shown, are parallel prosodically to the 1Pl forms shown. Verb stems used to illustrate the regular inflectional suffixes are súy5- 'hit', supplement by túwé- 'die' (Perfective -1b), nǔy¹ 'hear' as an irregular Imperfective Negative, and bàmbî-yî 'carry on back' in the Stative.

(xx3) Nanga dying-quail intonation with 1Pl -y:

a. [LHL] pitch, erasing input tone, on final syllable only on otherwise L-toned final syllable
unsuffixed Perfective sùyò-ỳ∴ [sùjòóòj]
Perfective-la túwé-èrè-y∴ [túwéèrèéèj]
Perfective-lb súyó-ti-y∴ [sújótìñ]

```
Recent Perfect jè-
                               súyó-jè-ỳ∴
                                                     [sújódzèéèj]
    Redup Perfective
                               sú-sùyò-y∴
                                                     [súsùjòóòj]
    Imperfective
                               (sù-)súy\acute{o}-mi-y^n:.
                                                    [(sù)sújómìîi]
    Progressive
                                                    [(sù)sújò(ò)sòóòj]
                               (sù-)súyò(:)-sò-y∴
    Stative
                               bá-bàmbà-ỳ∴
                                                     [bábàmbàáàj]
    Impf Negative
                               súyó-ŋò:-y<sup>n</sup>∴
                                                     [sújóŋòóòj<sup>n</sup>]
                                                    [sòóòj]
    'have'
                               yá sò-y∴
    'be present'
                               yá bì-y∴
                                                     [yábìíìi]
    'we are Fulbe'
                               pîrâ≡mi-y<sup>n</sup>∴
                                                     [miii]
                [cf. pîrâ=m-ĭ 'I am a Fulbe']
 on otherwise H-toned final syllable
    Perfective-2 -só-
                               súyó-só-ý∴
                                                     [sújósòóòj]
    'we are women'
                               yă:≡mî-y<sup>n</sup>∴
                                                    [miii]
                [cf. yă-ŋ≡m-î 'I am a woman']
 on otherwise R-toned final syllable
    'it is not'
                               ≡ndò-y∴
                                                    [ndòóòj]
b. [LH-L] with [LH] realized on the penult, both vowels lengthened
 on otherwise L-toned penult and final
                               súyà:-sò≡ndò-ỳ∴
                                                    [súyð:sòóndòòj]
    Progressive Negative
 on otherwise L-toned penult and H-toned final
    Stative Negative
                               bàmbà≡ndó-ý∴
                                                     [bàmbàándòòj]
                               sò≡ndó-ý∴
    'not have'
                                                    [sòóndòòj]
c. [H-L] with [H] realized on the penult, only the ultimate lengthened
 on otherwise L-toned penult and R-toned final
    'dislike' (irreg)
                               mbùrà-ý∴
                                                    [mbúrààj]
 on otherwise L-toned penult and H-toned final
                               nù-ŋɔ́-y<sup>n</sup>∴
    Impf Negative ('hear')
                                                    [núŋɔ̀ɔ̀j<sup>n</sup>]
    'not be'
                               ὴgó-ý∴
                                                    [ńgòòj]
 on otherwise H-toned penult and final
    'love'
                               m̀bá≡m-iy∴
                                                     [mbámii]
    'be small'
                               èwrέ≡mî-y∴
                                                    [èwrémi:]
d. [H-L] with H on penult (for monosyllables only at the end of the
    syllable), penult and (perhaps) ultimate lengthened
 on otherwise H-toned final following \{L\}-toned stem
    Perfective Negative
                               sùyò-rî-y∴
                                                    [sùjź:ri:]
                               tà:-rî-y∴
                                                    [tǎ:rì:]
```

4 Nominal, pronominal, and adjectival morphology

4.1 Nouns

4.1.1 Simple nouns

With the exception of 'woman', discussed below (§4.1.2), nouns in Nanga have a single form that is used with both singular and plural reference. Number is therefore expressed not in the noun (or adjective), but in postnominal determiners (Definite or demonstrative), which distinguish Animate Singular, Animate Plural, Inanimate Singular, and Inanimate Plural.

```
(xx1) a. nèr"i '(a) dog; dogs'
nèr"i né 'the dog' (singular)
nèr"i bû: 'the dogs'

b. tùmá 'tree; trees'
tùmà gú 'the tree'
tùmà ý 'the trees'
```

In free form (without following modifiers), Cv: is an allowable nominal stem shape (xx2.a). Short-voweled Cv is found only in the variant k5 'thing' (more often k5ŋ, or an extended form k5-kámâ 'thing, anything') along with its plural yế (xx2.b). The noun for 'person' also reduces to a short-voweled form before a modifier (xx2.c). ná 'time(s)' occurs only before a quantifier (xx2.d). For 'thing' and 'person' see the more detailed discussion in §4.1.2, below.

(xx2)	stem (free)	Definite	gloss
a.	kû: jǎ: nǒ: dǒ: nǎ: jǎ: tǎ: tǎ: cô: ô:	kù: gú jà: gú nò: gú dò: gú nà: gú jà: gú tà: " gú dù: gú ò: gú	'head' 'fence' 'mouth' 'rag cushion' 'hand' 'meal' 'taboo' 'load' 'medication' 'place'

```
ὲ: gú
                                        'well'
    Ě:
    yă:
                      yà: nέ
                                        'woman'
                      bà: nέ
                                        'father'
    bă:
                      dè: nέ
                                        'mother'
    dě:
b. nŭ:
                      nù nέ
                                        'person' (see §4.1.2)
                                        'thing' (see §4.1.2)
c. k \acute{o} \eta \sim k \acute{o}
                      kò gú
                                        'things' (see §4.1.2)
    yέ
                      yè gú
d. nà
                                        'time(s)'
```

Certain nouns that now behave as true Cv: stems in most combinations (xx2.a) preserve vestiges of a short-voweled form in certain compounds. For short-voweled yà- 'woman' in compounds, see §5.1.6. Regarding nă: 'hand', the terms for two types of shoulderbag, nà-pé:ré (variant nà-pégîré) and nà-kòmbó (cf. kòmbó 'war'?) are suggestive, but I cannot find cognates elsewhere. bă: 'father' and dě: 'mother' occur with short vowel in parallel uncle/aunt terms (the finals are the adjectives 'big' and 'small'): bà dîyâ 'father's elder brother', bà tègê 'father's younger brother', dè dîyâ 'mother's elder sister', dè tègê 'mother's younger sister'.

4.1.2 Irregular nouns ('woman', 'child', 'person', 'thing')

For 'woman', the basic form is yă:, which is used in simple-noun form for the plural 'women', and in all combinations (even with singular reference) with a following word or particle within the NP: yà: tùmâ 'one woman', yà: èsî '(a) good woman', yà: né ' the woman', yà: wǒŋ 'this woman'. See also the compounds in §5.1.7, below. However, as a simple noun (usually indefinite), the singular is expressed as yǎ-ŋ, as in yǎ-ŋ gó-èrè-Ø 'a woman went out'. This is the only survivor of an original nominal and adjectival suffixal morphology, which in other Dogon languages distinguishes Animate (or Human) Singular, Animate (or Human) Plural, and Inanimate (or Nonhuman).

wăn 'the counterparty', plural wã: or wă:-yè

The antonym árⁿâ 'man' is a regular noun with no special singular form.

'Child' is yî: (for its compounds, see §5.1.6). Examples with following elements include yǐ: èsî '(a) good child', yǐ: nế 'the child', and yǐ: tùmâ 'one child', The plural 'children' is yǐ-tègê, a somewhat frozen and semantically

specialized combination with tègê 'small' (more common synonym: èwré), cf. also noun tègê 'childhood'.

nǔ: 'person' is used for singular or plural reference. It does, however, reduce its vowel to short before another element. If the syntactic context calls for tone-dropping, we get nù. Where no tone-dropping applies, we get hightoned nú (functioning as the compression of a rising tone into a single mora). Examples with following elements: nù né 'the person', nù mòsí '(a) bad person', nú wǒy 'two people', nú kúrê 'six people'.

For 'thing', the singular form without a modifier is kóŋ or variant kó, with optional plural yé. The singular is kó (ior tone-dropped kò) before a modifier: kò gú 'the thing', kò tùmâ 'one thing', kò èsî 'a good thing', kó tà:ndĩ: 'three things'. The suppletive plural yé optionally replaces kó before nonsingular quantifiers: yé tà:ndĩ:.

 $k \acute{o} \eta \sim k \acute{o}$ 'thing' occurs in the common phrase $k \acute{o}$ kámâ 'each/any thing', which can be glossed '(not) anything' or 'nothing' when the predicate is negative. In positive contexts ('each/any thing'), the plural $y \grave{e} k \acute{a} m \acute{a}$ 'any things' is also possible.

 $k \acute{s} \eta \sim k \acute{s}$ 'thing' is undoubtedly related historically to the Inanimate Singular pronoun $k \acute{u}$, and to Inanimate Singular Definite $g \acute{u}$ ($k \acute{u}$, \acute{w}). Its plural $y \acute{\epsilon}$ is similarly related to Inanimate Plural Definite \acute{y} .

4.1.3 'So-and-so' (àmâ:n)

The 'So-and-so' word, used in generic contexts as a variable representing any personal name, is àmâ:n.

4.1.4 Initial Cv- reduplication in nouns

Corresponding to Ci- reduplications in Jamsay and Ben Tey, Nanga (like some other languages of the zone) has Cv- with a copy of both the source consonant and vowel.

In (xx1), we see three different tonal patterns for a basic type with Cv: base and an initial Cv- reduplication. All known examples are given. (xx2.b) is onomatopoeic. In (xx2.c), the ethnic term Bobo is most likely an accidental member of this reduplicated class. However, the four fauna terms do seem to form a clearly-recognized reduplicative set; 'agama lizard' seems to have mutated (cognates are unreduplicated bisyllabics pointing to *kèngú or the like) on the analogy of the other fauna terms.

(xx2) Reduplications with <LHL>

```
form
                               gloss
a. C_1 \hat{v}_1 - C_1 \hat{v}_{:1}
                               'beard'
        bè-bề:
b. C_1 \acute{\mathbf{v}}_1 - C_1 \dddot{\mathbf{v}}_1
       gá-gã:n
                               'crow'
c. C_1\hat{v}_1 - C_1\hat{v}_{:1}
        bò-bô:
                               'Bobo'
                               'griot (with war tomtoms)'
        gò-gô:
                               'grasshopper' (all species)
        kà-kâ:
        kè-kê:
                               'agama lizard' (multiple species/varieties)
        kὲ-kê:
                               'beetle, bug' (all species)
        tà-tâ:
                               'hyena' (extends to leopard)
d. C_1\hat{v}_1-C_1\hat{v}_1C
        tè-têw
                               'large raptors (hawks etc.)'
```

Numerous trisyllabic and longer nouns have an initial Cv- reduplication, though in most cases an unreduplicated form is not attested: kò-kósô 'dried leaves on ground', kî-kînjî 'broken pieces of seed spike', jà-jàŋkó: 'double grain spike', mò-mòrnô 'scorpion', nà-nà:rná 'spider', kó-kòsô 'cough (noun)' (cf. verb kósó-) and many others.

CvC-CvC appears in gòn-gŏn 'giant millipede'.

4.1.5 Final -Cv; and -Cvy reduplications in nouns

There are two minor final-reduplication patterns in nouns. The known examples, all of them fauna terms, are given in (xx1). Alternatively, -kɔ̃: in the term for 'louse' (xx1.a) could be taken as derived from kɔ̃:- 'eat (meal)'. The term for 'stone partridge' is onomatopoeic, so any semantic connection to verbs begiri- 'shake (grain)' and bẽ:- 'cut off the end of' is accidental.

(xx1) Final Reduplication (nouns).

```
a. ...-C<sub>1</sub>v:<sub>1</sub>
bègîrî-bὲ: 'stone partridge'
kòròŋ-kὸ: 'louse'
```

```
b. ... C_1\hat{v}y_1

pété-pêy 'grasshopper sp. (Oedaleus)'

sénér<sup>n</sup>é-sêy<sup>n</sup> 'grasshopper sp. (Kraussella)'
```

4.1.6 Nouns with full-stem iteration

A number of patterns are illustrated in (xx1). Several are at least vaguely onomatopoeic. There is usually no unreduplicated counterpart with related meaning.

In (xx1), the vocalism is identical in the two iterations. The tone pattern LL-HL occurs in several body-part terms (xx1.a). The remaining examples are onomatopoeic or otherwise expressive and have various tone contours.

(xx1) Full-Stem Iteration (nouns)

```
a. LL-HL or LL-HF tones
      kòrò-kórô
                        'bell'
      nà:-[tòŋò-tóŋô]
                       'elbow' (nă: 'hand, arm')
      kùnjò-kúnjò
                       'knee'
      dùmò-dúmô
                       'heel'
b. all-H
      sógóy-sógóy
                       'rattling'
c. F-F
                       'bow-wow!'
      wôw-wôw
                       'cheep!' (birds chirping)
      \hat{siw}^n - \hat{siw}^n
      kòyô wô:y-wô:y 'disconsolately weeping' (kòyô 'weeping')
d. other
      jùngùdú-júngúdú 'call of wild pigeon'
      kòyô bà:gà-bà:gà
                            'sobbing' (kòyô 'weeping')
```

The examples in (xx2) involve vocalic changes. In the single iteration cases (xx2.a-b), i.e. of type X-X', one or both non-low vowels in the first occurrence are replaced by a-vowels in the second. pú:sù-pâ:s can be interpreted as apocopated pú:sù-pá:sù. When both vowels, not just the first, are replaced by a-vowels (xx2.b), there is also a tonal change. In the double iteration type (xx2.c), i.e. of type X-X'-X, the first and third occurrences are identical (and have at least one H-tone) and the medial occurrence has a-vowels (and is all-L-toned).

```
a. one vowel replaced by a
(xx2)
                   kèndè-[pú:sù-pâ:s]
                                             'lungs' (kéndè 'liver, heart')
                  sèri-[búndi-bándi]
                                             'cream of millet with tamarind'
            b. both vowels replaced by a
                  kóró-kàrá
                                             'in clusters'
                  sùŋùr<sup>n</sup>ù-[gèrè-gárá]
                                             'earring' (súŋúr''i 'ear')
            c. H-L-H and LH-LL-LH with medial a-vocalism
                  kòró-kàrà-kòró
                                             'hubbub'
                  cî:n-cà:n-cî:n
                                             'creaking sound'
                                             'loud chatter'
                  hó:-hà:-hó:
                   5:-à:-5:
                                             '(sound of bullfrog croaking)'
```

4.1.7 Frozen initial a- or an- in nouns

A considerable number of noun stems begin with à- or àn-, which might originally have been a segmentable affix. For example, the term for 'stick with curved end' is either bèrè kó:ndó (with béré 'stick' followed by an adjective) or àkó:ndó (à-kó:ndó). A fuller list of candidates for this segmentation is in (xx1). Though historically unrelated, the number of such cases is increased accidentally by loanwords of Arabic origin, a few of which are listed in (xx1.c).

(xx1)	form	gloss	comment
	a. with à-		
	àbélkò àdăŋ àbà:gĭr ⁿ é à-kó:ndó àmíy ⁿ é àpûm àsĭmá	'game played with knife' 'somewhat deaf person' 'grass sp.' (Rottboellia) 'stick with curved end' 'inflection' 'mumps' 'dandy'	k5:nd5 'curved'
	àtá:n d ì	'constellation with 3 stars'	tà:ndĭ: 'three'
	a. with an-		
	àndiri àndèngè kèsi ànjèri àntá:ri àntô:	'rival' 'baggy pants' 'wrestling' 'hunt (noun)' 'stick with hooked end'	variant àtà:rí

àntòngó 'grains left after first winnowing'

c. Arabic loans (vowel-initial or with Definite *al-), selected examples àmá:nà 'promise, vow' ànsá:rá 'white person'

4.2 Derived nominals

4.2.1 Characteristic derivative (-gi)

A noun-to-noun derivation, denoting a person characterized by whatever the core noun denotes, is expressed by Characteristic suffix -gi. The core noun **drops its tones** before the suffix.

(xxx)	noun	gloss	Characteristic	gloss
	mùr ⁿ á	'disease'	mùr ⁿ à-gi	'sick person'
	gúmjέ	'hump'	gùmjè-gî	'hunchback'

4.2.2 Verbal Nouns

4.2.2.1 Regular Verbal Noun (-ndé)

The basic Verbal Noun formation involves a suffix $-nd\hat{\epsilon}$. The suffixal vowel is invariant and does not harmonize with stem vowels (xx1). The tone of the stem before $-nd\hat{\epsilon}$ is all-high, even when the stem otherwise has a rising tone contour (xx1.b).

(xx1)		verb	gloss	Verbal Noun
	a.	nő:- wő:- gő:- késé- súyó-	'drink' 'catch' 'go out' 'cut' 'hit'	nó:-ndé wó:-ndé gó:-ndé késé-ndé súyó-ndé
	b.	jàr ⁿ á-	'tap'	jár ⁿ á-ndé

4.2.3 Instrument nominals

Many instrument nominals are compounds including an incorporated object or similar compound initial; for them, see $\S5.xxx$. Some uncompounded instrument nominals related to verbs are in (xx1). The basic form, also seen in the compounds, is with low-toned stem ending in \hat{i} (xx1.a). (xx1.b) may be a case of apocope from *bùw \hat{i} , with the tones shifted to the surviving syllable. An alternation of $\frac{1}{2}$ with $\frac{1}{2}$ with $\frac{1}{2}$ is seen in (xx1.c).

(xx1)		verb	gloss	nominal	gloss
	a.	ďi:sé- jŭw-ró-	'file' 'turn (sth)'	ďi:sî jù:rî	'file' (tool) 'blacksmith's bellow'
	b.	bùwó-	'scrub'	bũw	'scrubber'
	c.	kó:sî-	'scrape'	kò:sî	'scraper' (tool)

There are several other nominals of similar form denoting instruments that have no synchronic relationship to a verb, e.g. gusirî 'pointed instrument for undoing braids'.

There are also several other (non-instrument) nominals of similar shape, functioning as cognate nominals; see (xx1.e) in §11.1.6.1.

4.2.4 Uncompounded agentives

Agentive nominals are normally compounds, with incorporated object or other nominal as compound initial; see §xxx.

'Hunter' is àtá:rî (variant àntá:rî), the same form used as a cognate nominal '(the) hunt' in the phrase àtá:rî tá:rî- 'go on (= engage in) a hunt'.

4.3 Pronouns

4.3.1 Basic personal pronouns

The basic morphological series are those in (xx1).

- (xx1) a. independent (also used for preverbal subject [e.g. in relative clauses], and optionally for object)
 - b. accusative (optional for direct object)

- c. pronominal-subject suffix on verbs
- d. possessor form, also used for complements of postpositions

The **independent**, **subject**, and accusative (= direct object) forms are given in (xx2). 3AnSg and 3AnPl are animate categories (including humans and animals), while Inan[imate] applies to plants and non-living things. Logo[phoric] pronouns are coindexed to the attributed author of quoted speech or thought (§xxx).

The preverbal subject forms (which are used in relative clauses) are identical to the independent forms. In regular main clauses the pronominal subject category is expressed by a suffix on the verb. The Accusative suffix -ŋ is usually reduced to nasalization of the preceding vowel. It may be entirely inaudible, in which case there is no audible difference between accusative and independent (or preverbal subject) forms, except in the 1Sg.

(xx2) Personal Pronouns

	subject			
	indep.	accusative	[_Verb]	[Verb]
1Sg	ĭ: ⁿ	ǹjî-ŋ	ĭ: ⁿ	-ỳ
1Pl	î:	î:-ŋ	î:	-ỳ ∴
2Sg	ú	ú-ŋ	ú	-ẁ
2Pl	û:	û:-ŋ	û:	-ẁ ∴
3AnSg	ńnέ	ńné-ŋ	ńné	-∅
3AnPl	bû:	bû:-ŋ	bû:	[varies by AN category]
InanSg	kú	kú-ŋ	kú	-∅
InanPl	kû:	kû:-ŋ	kû:	=3AnPl
LogoSg	á	á-ŋ	á	-Ø
LogoPl	â:	â:-ŋ	â:	=3AnPl

For the subject-pronominal suffixes, including the various 3Pl allomorphs (- \dot{a} , - \dot{v} , - $\dot{\epsilon}$, etc.), see §10.xxx.

Forms used as **possessors** and as **datives**, are in (xx3). The possessor forms include a classifier whose basic form is $/k\hat{o} \sim g\hat{o}/$ for Inanimate Singular and $/y\hat{e}/$ for Animate Singular and all Plural categories; see §xxx. The rarely used **inanimate possessor forms are merged** with the much more common 3AnSg and 3AnPl forms, but the animate/inanimate distinction is maintained in datives.

There is a **special 3Sg possessor** form no that does not allow a possessive classifier. In (xx3), the possessor forms are shown with N representing the preceding possessed noun (which retains its regular tones).

(xx3) Possessor and Dative Pronouns

possessor						
		AnSg/Anl	Pl/InanPl	InanSg	dative	
a.	1Sg 1Pl	N ỹ: N î: yè		N kồ: N î: gò	bàr ⁿ í ĭ: bày	
b.	2Sg 2Pl	N ú yê N û: yè		N ú gô N û: gò	ú báy û: bày	
c.	3AnSg 3AnPl	N nò N bû: yè		N nò N bû: gò	ńné báy bû: bày	
d.	InanSg InanPl		[=3AnSg] [=3AnPl]		kú báy kû: bày	
e.	LogoSg LogoPl		[see below	-	á báy â: bày	

The Logophoric pronouns have a slightly different morphosyntax as possessors. With kin terms and a few other human relationship terms ('friend'), the Logophoric possessor precedes the possessed noun (N), which therefore takes overlaid possessed-noun tone contour, {HL} after á and all-low after â:. With other animate possessed nouns (e.g. 'slave', 'sheep'), and with inanimates, the morphosyntax is the same as for the other pronouns (other than 3AnSg).

(xx4)		kin	AnSg/AnPl/InanPl	InanSg
	LogoSg	á N	N á yê	N á gô
	LogoPl	â: N	N â: yè	N â: gò

Examples: á dérè 'his/her-Logo elder sibling(s)', â: dèrè 'their-Logo elder sibling(s), pèrgé á yê 'his/her-Logo sheep', pèrgé â: yè 'their-Logo sheep', ńdô á gô 'his/her-Logo house(s)', ńdô â: gô 'their-Logo house(s)'.

4.4 Demonstratives

4.4.1 Definite morphemes

The forms in (xxx) behave like demonstratives, but are best glossed with definite 'the' in English. They indicate that the referent is already established in the discourse or in the common knowledge of the speech-act participants. In the forms given they require a preceding noun (or NP); in absolute function, an independent personal pronoun must be used (§4.xxx).

```
(xx1) Animate Sg nế
Animate Pl bû: (for bù gồ in factive complements, see §17.2.1)

Inanimate Sg kú (gú, w)
Inanimate Pl ý
```

The modified noun drops its tones before a Definite morpheme (as it would before a true demonstrative). In recordings, this tone-dropping may be the most easily audible cue that $\dot{\mathbf{w}}$ or $\dot{\mathbf{y}}$ is present, especially after more-or-less homorganic stem-final vowels. Examples with $\dot{\mathbf{ar}}^{n}\mathbf{a}$ 'man' and $\dot{\mathbf{bere}}$ 'stick'.

```
(xxx) a. àr<sup>n</sup>à nế
b. àr<sup>n</sup>à bû:
c. bèrè kú (bèrè gú, bèrè w)
d. bèrè ý
the man'
the stick'
the stick'
```

These Definite morphemes are also used at the end of (restrictive) relative clauses, agreeing with the head NP (§14.xxx).

4.4.1.1 'This/that' (deictic demonstrative pronouns)

Deictic demonstratives, which may be accompanied by a pointing gesture and in any event normally index proximate location, are given in (xxx).

```
(xx1) Animate Sg wŏŋ
Animate Pl wê:

Inanimate Sg ŋgú
Inanimate Pl yĕy
```

(xx1) a. wŏŋ

A preceding modified noun drops its tones. Examples with ár¹¹â 'man' and béré 'stick'.

```
(xxx) a. àr<sup>n</sup>à wŏŋ 'this/that man'
b. àr<sup>n</sup>à wê: 'these/those men'

c. bèrè ŋgú 'this/that stick'
d. bèrè yĕy 'these/those sticks'
```

Unlike the Definite morphemes, these deictics may also be used absolutely (i.e. with no overt preceding noun), see §6.xxx..

jóró-ŋò-yⁿ

```
Dem.AnSg like-Impf.Neg-1SgS
'I don't like that one (animate).'

b. ngú nji-n gòè-

Dem.AnSg 1SgO jab.Perf.L-3SgS
'That (object) jabbed (=pricked) me.'
```

4.4.1.2 Prenominal Anaphoric kú 'that (same)'

Anaphoric demonstrative kú, a special use of the Inanimate Singular pronoun, can be used prenominally in discourse-anaphoric sense: 'that (same, aforementioned) ...'. I will gloss it as "Dem" in interlinears. It is treated as a (pronominal) possessor, specifically like a pronominal possessor preceding a kin term (inalienable possession). kú therefore induces {HL} tone contour on the immediately following noun. The noun may be animate (including human) or inanimate in reference. A Definite morpheme with the appropriate animacy

and number features follows the noun, but does not affect the tone of the noun (i.e. there is no tone-dropping).

- $\begin{array}{cccc} (xxx) & a. & k\acute{u} & t\^{a}.^n & g\acute{u} \\ & Dem & shed.HL & Def.InanSg \\ & `that (same) shed` (t\widecheck{a}.^n) \end{array}$
 - b. kú ńdò gú
 Dem house.HL Def.InanSg
 'that (same) house' (ńdó)
 - c. kú ńdò ý
 Dem house.HL Def.InanPl
 'those (same) houses' (ńdó)
 - d.. kú nû: nế
 Dem person.HL Def.AnSg
 'that (same) person' (nú)
 - e.. kú yâ: nế
 Dem woman.HL Def.AnSg
 'that (same) woman' (yă:)
 - f. kú yâ: bû:
 Dem woman.HL Def.AnPl
 'those (same) women' (yă:)

Any such expression may be simplified by omitting the actual noun stem, using $k\hat{s}$ (inanimate) or $y\hat{\epsilon}$ (animate) as a substitute; these **generic morphemes** are also used in pronominal-possessor constructions ($\S6.xxx$).

- (xxx) a. kú kô gú
 Dem Inan.HL Def.InanSg
 'that (same) one [inanimate]'
 - b. kú yê nế
 Dem An.HL Def.AnSg
 'that (same) one [animate]'

kú, in spite of its probable origin as a "possessor" of sorts ('its ...'), is compatible with a true possessor NP (nonpronominal or pronominal). The possessor has its normal form and position: immediately before the possessed

noun if it is a nonpronominal NP, or a pronominal possessor with kin terms, and with generic k5 or yé following the noun if a pronominal alienable possessor. The tone of the possessed noun is determined by the immediately preceding element, hence {HL} if immediately after kú as in (xxx.a), again {HL} after a pronominal possessor with inalienable possession as in (xxx.b), and {L} after a nonpronominal possessor NP as in (xxx.c).

- (xxx) a. kú ńdò [ú gô] gú
 Dem house.HL [2Sg Poss.InanSg] Def.InanSg
 'that (same) house of yours-Sg'
 - b. kú ú léšì nế
 Dem 2SgP uncle.HL Def.AnSg
 'that (same) uncle of yours'
 - c. kú á:mádù ndò gú
 Dem Amadou house.L Def.InanSg
 'that (same) house of Amadou's'

4.4.1.3 Anaphoric/logophoric demonstrative pronouns

For third person logophoric (or other anaphoric) pronouns á (Sg) and â: (Pl), see §18.xxx).

4.4.2 Demonstrative adverbs

4.4.2.1 Locative adverbs

The simple spatial deictic adverbs, with stationery locative function (unless given ablative or allative function by a motion verb) are in (xx1). gá in ŋgà gá was originally the Locative postposition, but it seems quite frozen here.

(xx1) a. ŋgá 'here'
b ŋgà-gá 'over there' (pointing)
c. ká 'there' (discourse-definite)
yá

Another set of forms is used in the sense '(over) this/that way' or 'around here/there', indicating direction of motion or approximate location. The suffix $-r^{n}$ is added to the forms in (xx1), above. This is followed by ηa , an allomorph of the Locative postposition. The prolongation in the => variants perhaps reflects compensatory lengthening after the loss of this morpheme.

```
(xx2) a. ŋár<sup>n</sup>i ŋà yár<sup>n</sup>i=>
b. ŋgà-gá-r<sup>n</sup>i ŋà ygà-gá-r<sup>n</sup>i=>
c. kár<sup>n</sup>i ŋà kár<sup>n</sup>i=> yár<sup>n</sup>i ŋà yár<sup>n</sup>i=>
(over) that way, around there' (pointing)
(around there, in that direction (definite)'
(b. ŋgà-gá-r<sup>n</sup>i=>
(around there, in that direction (definite)'
```

4.4.2.2 Emphatic and Approximinative modifiers of adverbs

An emphatic particle té with prolonged vowel is used to insist on a specific location. A particle yànà is attested in approximative sense.

```
(xxx) a. ÿgá té⇒
here Emph
'right here'
```

b. ŋgá yàŋà
here around
'around here'

4.4.3 Presentatives ('here's ...!')

There are two basic presentative forms (xxx).

(xxx)	Sg or Pl	wùnér ⁿ é wùné	'here it is, here they are'
	Pl only	wèré	'there they are'

wùnérné and variant wùné are used, for example, when handing something to the addressee, while wèré is most often used to indicate that something lost or absent has just come into view. However, the distinction does not seem to be sharp.

When they have scope over a NP ('here's X!'), they may precede or follow the NP. For wúnérⁿé only, pronominal conjugation may be added using bù-'be'. The presentative particles may also be used preclausally, with scope over the whole clause (xxx.d-e); here the English free translation is awkward, but cf. French (e.g. *les voici qui dansent!*).

- (xxx) a. á:mádù wùnérⁿé
 Amadou here.is!
 'Here's Amadou!'
 - b. yı̈-tege were b-e child-Pl here.is! be-3PlS
 'Here are the children!'
 - e. wùnérⁿé bù-ỳ∴ here.is be-1PlS 'Here we are!'
 - d. wèré giyé giyé-m-è here.are! dance(noun) dance-Impf-3PlS 'Here they are, dancing!'
 - e. wùnérⁿé gĩyé gîyê-ⁿ here.is! dance(noun) dance-Impf.3SgS 'Here he/she is, dancing!'

4.5 Adjectives

For adjectival intensifiers, see

4.5.1 Types of adjectives

I use "adjective" loosely for an element that occurs chiefly in postnominal modifying function (forcing tone-dropping on the immediately preceding noun or adjective) and in predicative function ('be/become ADJ'). However, this broad categorization covers two major subtypes.

The first subtype, which one is tempted to call "true adjectives" since they are maximally distinct from nouns, is presented in (xx1). In addition to a postnominal modifying form, these stems have a simple predicative form for 3Sg inanimate and 3Sg animate subjects ('X is wide', etc.). In practice, several of these adjectives are restricted to either inanimate or animate subjects ('spacious' to inanimates, 'plump' to animates), but those that can apply to both inanimate and animate entities have a single 3Sg predicative form.

If the modifying form of the adjective ends in a non-high vowel (xx1.a) or in a semivowel (xx1.b), the predicative form is identical to the modifying form. The same is true of the one Ci: adjective (xx1.c), which is arguably also semivowel-final ($\frac{1}{\sin y}$). If the adjective ends in a high vowel, we get final i in postnominal modifying function, and final u in predicative function (xx1.d-f).

Of interest is the absence of HH-toned bisyllabic adjectives (except for color adjectives, discussed below), and of H-toned monosyllabic adjectives, in (xx1). The observed tone contours are rising (LH), falling (HF, monosyllabic F), and the one tritonal case (LF). See, however, the discussion of HH-toned color adjectives in (xx3), below. The switch between i and u does not change the lexical tone in the majority bitonal (HF and LH) patterns. However, in the one case of tritonal contour (LF), the final F-toned \hat{i} in modifying function (xx1.e) corresponds to H-toned \hat{u} in the predicative form (xx1.f).

(xx1) Adjectives

	gloss	modifying	predicati Inanimate	ive 3Sg Animate
a.	'distant'	wàgá	wàgá	wàgá
	'tall'	gàwá	gàwá	gàwá
	'long, tall'	gùró	gùró	gùrá
	'big, grand; many'	òwó	òwó	òwó
	'small, young'	èwré	èwré	èwré
b.	'spacious'	gâw	gâw	
c.	'pointed'	sî:	sî:	
		[arguably sîy	, cf. verb siyé	-ndîyé-]
d.	'narrow'	èmbî	èmbú	
	'narrow'	péŋî	péŋû	
	'good'	èsi	èsú	èsú
	'short'	déŋi	déŋû	déŋû
	'lightweight''	ér ^m i	ér ⁿ û	ér ⁿ û

	'easy, cheap'	nà:r ⁿ á	nà:r ⁿ á	
	'plump'	àmî		àmú
	'cold; slow'	támî	támû	támû
	'sweet; sharp'	έrî	érû	έrû
	'bitter'	gárî	gárû	
	'slightly coarse'	yágî	yágû	
	'hard'	már ^m i	már ⁿ û	
	'slightly bitter'	ásî	ásû	
	'salty, sour'	pári	párû	
e.	'big; fat'	dùgi	dùgú	dùgú
	'heavy'	dúsi	dúsû	dúsû
	'soft (skin)'	búrî	búrû	
	'hot, fast'	ógî	ógû	źgû
	'nasty, ugly'	mòsi	mòsú	mòsú
	'smooth, sleek'	órî	órû	
	'deep'	sóŋî	sóŋû	
	'coarse'	kúnji	kúnjû	
	'thick (liquid)'	kùri	kùrú	
f.	'difficult, costly'	nòmî	nòmú	nòmú

With other than 3Sg subject, the positive predicative form ('I am __', 'they are __', etc.) consists of the form used in modifying function plus the relevant conjugated form of the 'it is' clitic =m- (the 3Pl is a suppletive form =y\varepsilon). All of the negative predicative forms ('I am not __', 'they are not __', 'it/he/she is not __') are likewise formed by adding a conjugated form of the 'it is not' clitic to the form of the stem used in modifying function. This suggests that the modifying form is lexically basic, and I treat it accordingly in the lexicon. One complication is that bisyllabic adjectives with a rounded vowel {u o o} in the first syllable, as in (xx1.e) above, regularly assimilate the second-syllable i to u when followed by a (nonzero) clitic.

(xx2) illustrates the treatment of final vowels for two adjectives ending in high vowels ('heavy', 'lightweight') and for one adjective ending in a stable non-high vowel ('long, tall'). 'Heavy' furthermore illustrates Rounding-Assimilation before nonzero clitic (xx2.c).

```
(xx2) construction 'heavy' 'lightweight' 'long, tall'

a. postnominal modifying
'a __ person' nù: dúsì nù: ɛrʰi nù: gàwá
```

```
b. special 3Sg-subject positive predicative form
    'he is
                       dúsû
                                                        gàwá
c. cliticized forms based on (a), but subject to assimilatory rounding
    'they are'
                       dúsû≡yὲ
                                        έr<sup>n</sup>i=yὲ
                                                        gàwá≡yέ
    'I am ___'
                       dúsû≡m-ì
                                        έr<sup>m</sup>i≡m-i
                                                        gàwá≡m-î
    'he is not '
                       dùsù≡ndó-Ø
                                        èr"i≡ndó-Ø
                                                        gàwà≡ndó-Ø
```

The basic **color terms** in (xx3) can pattern like the adjectives in (xx1), above. In particular, they do have a special suffixless 3Sg predicative form. However, this 3Sg form was recorded only for inanimate subject (xx3.a), and even for inanimates it less common (with color adjectives) than a morphological construction with Inanimate \equiv w, Animate Singular \equiv ŋ, and Animate Plural \equiv yɛ (xx3.b). Of interest is the fact that the modifying forms are HH-toned while the special 3Sg forms in (xx3.a) are HL-toned.

(xx3) Adjectives

	gloss	modifying	Inanimate	predicative AnimateSg	_
a.	'red' 'black' 'white'	bár ⁿ î jémî pîrî	bár ⁿ û jémû pîrû	_ _ _	_ _ _
b.	'red' 'black' 'white'	bár ⁿ î jémî pîrî	bár ⁿ ú≡ẃ jɛ́mú≡ẃ pírí≡ẃ	bár ⁿ î=ŋ́ jɛ́mî=ŋ́ pîrî≡ŋ́	bár ⁿ î=yé jémî=yé pîrî≡yé

The stems in (xx4), below are common in postnominal modifying function. However, in predicative function I was only able to elicit the 'it is' clitic even for 3Sg subject. These stems are therefore more "noun-y" than those in (xx1), above. There are also some subtle differences in stem shapes. The long-voweled monosyllabic type Cv: is clearly represented in (xx4.a), but only dubiously (with \hat{i} : interpretable as $/\hat{i}y/\hat{j}$ in (xx1). The adjective 'new' (xx4.b) is heard with HF tone rather than the HL typical of the adjectives in (xx1). The adjectives in (xx4.c) have a Cv:mbv shape (with repeated vowel) that is heavier than anything seen in (xx1). However, there are plenty of adjectives in (xx4.d) that have very normal adjectival shapes, and the LF tone of gomi 'no good' (xx4.e) is matched by that of nomi 'difficult, costly' in (xx1.e). There is one consonant-final bisyllabic, $nandy^n$ 'respectable' (also 'useful', etc.); it is phonologically

possible to add the 3Sg clitics to this stem, as shown (xx4.f), but these combinations sound somewhat awkward, and one can alternatively use bú- 'be'.

(xx4) Adjectives with no special 3Sg predicative form

	gloss	modifying	predicative 3Sg	
			Inanimate	Animate
a.	'full' 'dry' 'old'	bá: mă: pě:	bá:≡ẃ mă:≡ẃ ⁿ pĚ:≡ẃ	pě:≡ή
b.	'new'	kándà	kándà≡ẁ	kándà≡ŋ̀
c.	'lean, emaciated' 'slender'	kó:mbó ké:mbé	kέ:mbέ≡ẃ	kó:mbó≡ή kέ:mbέ≡ή
d.	'runty' 'unripe, raw' 'half-ripe' 'weak, diluted' 'coarse' 'deserted, empty' 'living, alive' 'ripe (fruit)' 'innocent'	kèté kè:sî àmá sèré kùnjî kóró úmá bòró pé:ré	kè:sú=w àmá=w ⁿ sèré=ẃ kùnjú=ẃ kóró=ẃ bòró=ẃ	kèté≡ŋ́ kóró≡ŋ́ úmá≡ŋ́ pé:ré≡ŋ́
e.	'no good'	gòmî	gòmû≡ẁ	gòmî≡ŋ̀
f.	'respectable'	nánáy ⁿ	nánáy ⁿ ≡ẃ	nánáy ⁿ ≡ή

In the case of <code>iré</code> 'ripe' (postnominal in e.g. yù: <code>iré</code> 'ripe millet'), in predicative function I could elicit only forms of the verb <code>îré-</code>, e.g. <code>îré-èrè-</code> 'it has ripened (=is ripe)'.

Stems like sŏ:rò 'young (person)' and sátárá 'able-bodied (adult)' look like modifying adjectives in e.g. àrnà-yn sŏ:rò 'young man, adolescent boy' and yà: sátárá 'able-bodied woman'. However, these combinations are better analysed as noun-noun compounds, to judge by the form taken by the simple stems as predicates: sŏ:rô=ŋ 'he/she is young (=a young person)', sátárá=ŋ1 'he/she is an able-bodied adult'.

4.5.2 CýCŷ=> adverbials ('flat')

The forms in (xx1) are adverbials syntactically (they take $b\acute{u}$ - in predicative function), but translate as adjectives. They occur in predicative function with a following $b\acute{u}$ - 'be'. When $b\acute{u}$ - takes participial (relative-clause) form, the adverbial may modify a noun: $k\grave{u}$: $p\acute{e}t\^{e}$ => $b\acute{u}$ - $m\grave{i}$ 'flat-topped head'.

```
(xx1) form gloss

pótô=> 'flat (and small, but longer than wide, e.g. nose)'
pétê=> 'flat and wide'
```

The vocalic alternation reflects a minor sound-symbolic system also seen here and there among verbs. The final syllable is prolonged.

4.5.3 Iterated (fully reduplicated) adverbials

A small set of reduplicated adverbials denoting unusual shapes is presented in (xx1). The tone contour is LL-HH. The final syllable is prolonged. A single vowel is repeated in all four syllables.

```
    form gloss
    pèmbè-pémbé = > 'shaped like a flattened calabash' 'brick-shaped'
    sòrò-sóró = > 'elongated, ovoid, barrel-shaped'
    kèbè-kébé = > 'bar-shaped with flat sides'
```

A different pattern, with tone contour LL-HL, is seen in (xx2). This contour also occurs in several reduplicated noun stems denoting body parts (§4.xxx).

```
(xx2) form gloss

kù:-[gòrò-górò] 'having an oversized head' (kû: 'head')
gòmbò-gómbò 'projecting (rock)'
```

A form like jinjà-jinji 'noisy' (noun or adjective) is to be taken not as a simple reduplication, rather as an agentive with incorporated cognate object, cf. jinjâ jinji- 'make noise' (cognate nominal followed by verb).

4.6 Participles

For participles, i.e. a nominal form of the verb used in relative clauses, see §xxx, below.

4.7 Numerals

4.7.1 Cardinal numerals

```
4.7.1.1 'One', 'same (one)', and 'other'
```

'1' is expressed as an adjective tùmâ. The preceding noun drops tones, as before normal modifying adjectives: pèrgè tùmâ 'one sheep' (pèrgé), yà-ŋ tùmâ 'one woman' (yǎ-ŋ).

The adjective for 'other' is either bindé or bèndî. In practice they are used in postnominal modifying function, but not as adjectival predicates. pèrgè bèndî = pèrgè bindé 'a/the other sheep.'

4.7.1.2 '2' to '10'

The numerals from '2' to '10' are shown in (xx1).

(xx1)	gloss	form
	' 2'	wŏy
	'3'	tà:nđi:
	'4' '5'	nŏy ⁿ nìmi:
	3 '6'	nımı. kúrê
	·7 [,]	súyê
	' 8'	gá:rè
	'9'	tè:sĭ:
	'10'	pé:rù

4.7.1.3 Decimal multiples ('10', '20', ...) and their combinations ('11', '59', ...)

The multiples of '10' are given in (xx1).

(xx1) gloss form

```
'10'
                pέ:rù
'20'
                 pèrî-yěy
                pέ-tà:nďi:
'30'
'40'
                pěn-n<u>š</u>y<sup>n</sup>
'50'
                pěn-nimi:n
'60'
                 pèr-kúrê
'70'
                pèr-súyê
'80'
                pèr-gá:rè
'90'
                pèr-tè:si:
```

For the tonal patterns, see Tone-Dissimilation §3.7.xxx. The forms with final H- or R-tone ('20-50', '90') are sometimes heard with final low pitch when phrase-final. However, the terminal high-tone segment is always audible in careful pronunciation, and e.g. when followed by say 'only' or by an 'it is' clitic.

Combinations with a single-digit ('1-9') numeral, like '11' and '59', are expressed as shown in (xx2). The morpheme ságâ, used only with numerals, follows the single-digit numeral; I gloss it as 'plus'.

```
 \begin{array}{cccc} (xx2) & a. & p\hat{\epsilon}:r & [t\hat{u}m\hat{a} & s\hat{a}g\hat{a}] \\ & ten & [one & \textbf{plus}] \\ & & \text{`eleven'} \end{array}
```

- b. pé-nimi:n [tè:si: ságâ] ten-five [nine plus] 'fifty-nine'
- c. ùsú [pé-tà:ndi: tà:ndi: ságâ]
 day [ten-three three plus]
 'thirty-three days'

4.7.1.4 Large numerals ('100', '1000', ...) and their composites

The stems in (xx1) are nouns.

```
(xx1) gloss form

a. 'hundred' té:mdérê (<Fulfulde)

b. 'thousand' mùsú
```

```
c. 'million' milyô:" (<French)
```

With following numeral: té:mdérè wŏy 'two hundred', mùsú tà:ndĭ: 'three thousand'.

Numerals involving more than one level ('1-99', hundreds, thousands) did not require repetition of the modified noun in elicited examples, which occurred just once, at the beginning. Instead, the nonfinal segments showed prosodic prolongation (⇒).

```
(xxx) [pèrgé mùsú wǒy ⇒] [té:mdérè nǐmǐ: ⇒]
[sheep thousand two] [hundred five]
[pèrî-yěy]
[ten-two]

'two thousand, five hundred, (and) twenty sheep'
```

For seni '80' or '100' in currency terms, see just below.

4.7.1.5 *Currency*

The currency unit used in markets is bú:dù, equivalent to five CFA francs.

sèŋî is an archaic term meaning '100' still used by older people in connection with currency, hence '500 CFA francs' (synonym: bú:dì té:mdérè).

dògò-sèŋî 'Dogon hundred' means '80 (riyals)', i.e. '400 CFA francs'.

4.7.1.6 Distributive numerals

'One at a time', 'one apiece', and similar distributive phrases are expressed as tùmá-tùmá. Other adverbial distributives are likewise formed by iterating the numeral.

4.7.2 Ordinal adjectives

In addition to the ordinals described just below, note interrogative à:ŋgà-yɛ́ 'how-many-eth?' (French *quantième*), used to inquire what position or rank an individual is in.

4.7.2.1 'First' and 'last'

The basic forms are in (xxx).

```
(xxx) kiyá:w 'first'
dùmá 'last'
```

Examples: ndò kiyá:w 'the first house', ndò dùmá 'the last house' (cf. ndô 'house').

4.7.2.2 Other ordinals (suffix -nš:)

Other ordinals are formed by adding -nɔ̃: to the numeral, whose tones are dropped. There are irregular forms for 'second', 'third', and 'fourth'.

```
(xx1)
             form
                                                   gloss
        a. single-digit numeral
             wònjò-nɔ̃: (or: wònjŏw)
                                                   'second'
                                                   'third'
             tàndà-nɔ:
             nòr<sup>n</sup>ò-nŏ:
                                                   'fourth'
                                                   'fifth'
             nimi:-n5:
                                                   'sixth'
             kùrè-nš:
             sùyè-nɔ̃:
                                                   'seventh'
                                                   'eighth'
             gà:rè-n5:
            tè:sì:-n5:
                                                   'ninth'
                                                   'tenth'
            pè:r-n5:
        b. decimal
                                                   'twentieth'
             pèri-yèy-n5:
        c. decimal plus single-digit numeral
                                                   'eleventh'
             pê:r tùmà sàgà-n5:
        d. hundred
                                                   'hundredth'
             tè:mdèrè-n5:
        e. hundred plus '1-99' numeral (two levels)
             té:mdérè pèri-yèy-nš:
                                                   'hundred and twentieth'
```

In (xx1.c,e), I did not observe tone-dropping on the first part of the complex numeral, indicating that they are outside of the tone-dropping scope of the Ordinal suffix. In (xx1.b), the two components of the numeral are tightly fused and are treated as a frozen unit, so the whole sequence drops tones.

4.7.3 Fractions and portions

péjérè commonly means '(one) half'. Other (equal) fractions may be described generically as gósô 'portion, division'.

5 Nominal and adjectival compounds

5.1 Nominal compounds

5.1.1 Compounds of type $(\bar{n} \ \bar{n})$

Compounds of two nouns (one or both of which may be internally composite) where both the initial and the final have their regular tones are rare. I can, however, cite káŋê mútúkâl, literally "gold[metal] bushel," which denotes an ornament worn on the top of the ear as a pendant to an earring.

5.1.2 Compounds of type $(\hat{\mathbf{n}} \ \bar{\mathbf{n}})$

One of the two most common noun-noun compound types is one with low-toned compound initial (perhaps itself compounded) and a final noun with its regular tones. This pattern competes most directly with the possessor-type compounds (§5.1.4, below). In general, the more established and lexicalized a noun-noun compound is, the more likely it is to take the tonal form $[\hat{\mathbf{n}} \ \bar{\mathbf{n}}]$ rather than the possessor-type compound pattern.

Examples with initial from enjê 'chicken' are in (xx1).

```
(xx1)
        ènjè-tóndî
                             'chicken basket' (i.e. wicker chicken coop)
        ènjè-ńdô
                             'chicken coop (constructed with bricks or stones)'
                             'chicken's roosting place'
        ènjè-biyê
        ènjè-tènê
                             'chicken's drinking pan'
        èniè-gùwá
                             'chicken's foot' (also 'cock of musket')
        ènjè-káwsá
                             'chicken lice' (name of a disease with skin sores)
        ènjè-kòmórô
                             'chicken finger' (name of a tree [Piliostigma])
        ènjè-kéwré
                             'chicken eggplant' (bush sp. [Solanum])
```

5.1.3 Compounds with final Verbal Noun, type (n n)

Verbal nouns take direct-object complements in the form of regular NPs (xx1), so there are fewer compounds of this type in Nanga than in many other Dogon languages, where such complements often appear in low-toned compound-initial form.

```
(xx1) [yû: gîr<sup>n</sup>é-ndέ] nòmú
[millet harvest-VblN] be.difficult
'Harvesting millet is difficult.'
```

An alternative construction is a compound type with **low-toned initial** representing an incorporated object, and {HL} toned verb in nominal function (xx2). The compound initial in this case (unlike the verbal-noun construction) cannot be expanded with a determiner or quantifier.

```
(xx2) a. yù:-bárâ
millet.L-gather.Nom.HL
'act of gathering millet after the harvest and taking to the houses'
(cf. verb bàrá- 'gather')
```

- b. yù:-[dú:-yè]
 millet.L-[carry.on.head-MP.Nom.HL]
 'act of carrying millet on the head (from field to houses)'
- c. njà-bégîrè grain.L-winnow.Nom.HL 'winnowing grain'.

Further examples are lè:tèrè-tóŋô 'letter-writing', pèrgè-sémê 'slaughtering sheep', and nàŋà-págâ 'tying up cow(s)'. Such forms can be used as complements of bă:rî- 'help' (§17.5.1).

5.1.4 Possessive-type compounds $(\bar{n} \hat{n})$

In this compound type, the initial is treated like a possessor and the final like a possessed noun. That is, the initial has its regular tones, while the final has either $\{HL\}$ or all-low tone contour, depending on whether the initial ends in a high or low tone element. When eliciting new compounds, I found that my assistant often fluctuated between this pattern and the $[\hat{n} \ \bar{n}]$ type (§5.1.2, above), but the more lexicalized compounds in common use are generally $[\hat{n} \ \bar{n}]$.

Because there is no sharp distinction between possessor-possessed sequences and possessor-type compounds, I do not hyphenate the latter. Compare the compound (xx1.a) to the clearly possessive construction (xx1.b). As it happens, the final in these examples, duga-nja:, is already a (somewhat frozen) compound of [n n] type, but this point is not directly relevant here. The overlaid {HL} contour produces duga-nja.

```
    (xx1) a. nă: dúgá-njà hand necklace.HL 'hand-necklace' (term for 'thin bead bracelets') (dùgà-njâ:)
    b. yă-η dúgá-njà woman-Sg necklace.HL 'a woman's necklace'
```

Some other examples are in (xx2). In (xx2.a), the overlay of {HL} contour on músó:rò produces no audible change.

```
(xx2)
         a.
              [yùgùsì
                              jém]
                                               músó:rò
              [velvet.L
                                               shawl.HL
                              black]
               '(woman's) head shawl of black velvet' (yúgúsî, músó:rò)
         b.
              gồ:
                             òmir<sup>n</sup>i
                             parent.in.law.L
              fire
               'fire's in-law' (a plant, Waltheria) (omir<sup>n</sup>i)
         c. bòndi
                                  lî:gi
                                  bird.HL
              rain
               'rain-bird' (i.e., 'cuckoo') (lî:gî)
```

5.1.5 Agentive compounds of type $(\hat{\mathbf{x}} \hat{\mathbf{v}})$

Most agentive nominals are compounds including an incorporated compound initial, usually a noun denoting a typical direct object of the action, or a cognate nominal of the verb. The initial is low-toned. The Agentive form of the verb has {LH} tone contour realized as <LH>, LH, or LLH depending on the number of syllables. The Agentive form also ends in /i/ (which is therefore always hightoned), except for monosyllabic stems, which have their regular lexical vowel. In (xx1), the form of the inflectable verb with its typical object is given in parentheses after the free gloss. The monosyllabic finals are (xx1.f-g).

```
(xx1) a. kù:-kùwî
sorcery.L-practice.sorcery.Agent
'sorceror' (kŭ: kúwó-)
b. bìdigà-bìdigî
magic.tricks.L-do.magic.Agent
'magician' (bìgìgâ bìdigî-)
c. dàwri-bìri
```

```
magical.solution.L-work.Agent
'magician (generally maleficent)' (dăwrî bìré-)

d. nàmà-tùrî
    meat.L-sell.Agent
    'butcher' (meat-seller) (nàmâ túró-)

e. nàdò-kèmî
    house.L-build.Agent
    'construction worker, mason' (ndô kémé-)

f. tòndì-tě:
    basket.L-weave.Agent
    'basket-weaver' (tóndî té:-)

g. kònjò-nò:
    millet.beer.L-drink.Agent
```

The initial may itself be compound, as long as it has no determiner.

```
a. [nà:-bìrà]-bìrî
[hand.L-work(noun).L]-work.Agent
'manual worker' (nà:-bîrá bìré-)
b. [sàgì-nì:]-nɔ:
[dried.wild.grapes.L-water.L]-drink.Agent
'drinker of wild-grape juice' (sàgì-nì: nɔ:-)
```

'drinker of millet beer' (kònjó nó:-)

5.1.6 Compounds with yî: 'child' and nijâ or nàr 'fruit'

The noun yî: 'child' (irregular plural yǐtègê 'children') can be possessed, generally in the sense 'son or daughter (of someone)'. In cases like ámbérĭ yĭ: 'chief's child', if the possessor is understood to be generic ('a/any chief's child') the construction fits the form and meaning of possessor-type compounds (§5.1.4).

An important compound of the $[\hat{\mathbf{n}} \ \bar{\mathbf{n}}]$ tone pattern (§5.1.2) is $b\hat{\mathbf{a}}:-y\hat{\imath}$: 'child of the same agnatic family (e.g. offspring of one's father's brother)'. This compound is itself usually possessed, as in $b\hat{\mathbf{a}}:-y\hat{\imath}$

In other compounds with this tone pattern, the initial is not a possessor, rather a noun that defines the category that the child belongs to: tà:libù-yi: 'child beggar (koranic-school pupil)', [kè-kèrⁿi]-yi: 'recently circumcised boy',

lèkòl-yî: 'school child (pupil in public school)'. In these cases the plural is with -yǐtègê. Similar compounds are used to denote the young of animals, e.g. bèr-yî: 'goat kid', ènjè-yî: 'chick'.

For 'girl' and 'boy', see §5.1.7, just below.

With plants, -yî: as compound final denotes young plants (e.g. saplings) rather than fruits or other products. Thus tùmà-yî: 'sapling (of tree)', tò:-yî: 'sprout' (tŏ: 'sown seeds'). My assistant accepted plurals with -yǐtègê.

For other inanimate referents, I can cite the pairs sèrî 'mound of excavated earth (at a burial)' and sèrì-yî: 'small pit dug for corpse inside larger burial pit'; kó-kóró 'bobbin (in loom)' and [kò-kòrò]-yî: 'thin wooden rod on which bobbin turns'; and béré 'stick, wood' and bèrè-yî: 'small stick, twig'. Again, my assistant accepted plurals with -yǐtègê.

For 'fruit of (plant)', the compound final is either njâ 'seed' (e.g. for nutlike fruits) or more often nàr 'fruit' (related to the verb nàr 'á- 'give birth, bear').

- (xx1) a. mɔ̃:rnɔ̃-njâ wild.date-seed 'fruit (date-like) of wild-date tree'
 - b. kùrò-nàrⁿá
 wild.grape-fruit
 'fruit of wild-grape tree'

5.1.7 'Man' ($-àr^n \acute{a}$, $\grave{a}r^n \acute{a}$ -), 'woman' ($-y \check{a}$:, $y \grave{a}$:-, $y \grave{\epsilon}$ -)

The adjectives used e.g. after terms for animal species, are árⁿâ 'male' and yă: 'female'.

yǎ-ŋ 'woman' (plural yǎ:) takes the form yǎ: before numerals (from '2' up) and yà: before modifying adjectives: yà: tùmâ 'one woman' ('1' is a modifying adjective), yà: èsî 'pretty woman', yà: kó: 'unmarried woman', yǎ: wǒy 'two women'. yà- with short vowel occurs as a compound initial in e.g. yà-gùrô 'young adult woman', yà-wô: 'woman who has just given birth'. 'Girl' is yǎ-ỳn (plural yà-tègê).

árⁿâ 'man' (Sg=Pl) is generally regular: árⁿâ wŏy 'two men', àrⁿà dùgî 'fat man'. 'Boy' is árⁿâ-yⁿ.

5.1.8 Compounds with bàná 'owner'

The regular 'owner of X' construction is a possessive-type compound with final báŋà or bànà 'owner(s)', the tones depending on whether the initial ends in a high or low tone. Simple examples are ńdô bàŋà 'house owner' and nàŋá báŋà 'cow owner'. The unpossessed form is bàŋá 'owner, master'.

The initial is a NP in form and may be expanded (e.g. with a determiner), as in [nànà wǒn] bánà 'the owner of this cow'.

The construction with bàŋá as final may be used to define a person or object on the basis of a major characteristic. The construction therefore competes with the Characteristic derivation with suffix -gî (§4.xxx). An example is gúmjɛ́ báŋà 'person with curved (hunched) back', based on the noun gúmjɛ́ 'curved back'. Other examples include kiyá báŋà 'hair-owner' (i.e. 'hairy person'), yăr báŋà 'sky-owner' (i.e. 'lightning jolt'), lèwɛ́ báŋà 'owner of sacrificial altar' (i.e. member of a founding family of the village), isê bàŋà 'village-owner' (i.e. aboriginal, old-stock), and tùrá:bǐ bàŋà 'divination owner' (i.e. 'Muslim fortune teller'(,

5.1.9 Loose and tight compounds with de: ('authentic', 'entire')

Adjective dé: 'authentic, prototypical' is distinct tonally from dě: 'mother', but high-toned adjective dé: occurs with animal terms in e.g. nàŋà dé: 'cow that has calved at least once' (nàŋá 'cow'). The senses 'mother' and 'authentic' or 'entire' are associated in other Dogon languages. Cf. also dè-dé: 'papa!', a resepctful address form, considered by my assistant to be derived from dèré 'elder sibling'.

dé: was recorded as an adjective specifying the prototypical member of a multi-member category in connection with the taxon 5:sò, which denotes a group of shrubs with edible fruits of the genus *Grewia* (excluding *G. bicolor*). Each species can be individually denoted by adding an adjective or compound final (5:sò kóró-kàrá *G. flavescens*, 5:sò jáŋmá *G. tenax*, 5:sò ògò-béré *G. villosa*). I recorded 5:sò dé: 'prototypical Grewia' for *G. lasiodiscus*.

kùmbǐ dé: 'prototypical (broad-leaved) fig tree' is used, as a synonym for kùmbǐ pîrî 'white (broad-leaved) fig tree', for the mountain fig *Ficus abutilifolia*. The other fig called kúmbǐ is *Ficus platyphylla*, which can be specified as kùmbǐ bár¹¹í 'red (broad-leaved) fig tree'.

A non-prototypical variant of a species may alternatively be named after the prototypical species, with the adjective să:ndâ, which we might gloss as 'false' (though 'second-string' as used in baseball is closer to the Nanga sense). My only examples are these: kùrô 'wild-grape tree' (*Lannea microcarpa*) and kùrò

să:ndà (*Lannea acida*); wèrè-wérê 'swift or swallow' (focally a large swift) and wèrè-wèrè să:ndà 'small swift or swallow'.

dé: is also used productively in the sense 'entire (plant)', distinguishing this from a fruit or other part that is also denoted by the same name. Thus màngòrò dé: 'mango tree' (măngórò).

5.1.10 Natural-species compounds with medial linking element

I did not observe any instances of flora-fauna or other nouns of the type X-L-X with an iterated segment X flanking a linking element like Jamsay -na:- or Tommo-So -ma:-.

5.1.11 Instrumental relative compounds ('oil for rubbing')

In this construction, the noun is head of a relative clause (with Imperfective participle) describing the function.

```
(xx1) a. ni: nô:-mi
water drink-Ppl.Impf
'water for drinking'
```

b. ni: diyé-mi water bathe-Ppl.Impf 'water for bathing'

5.1.12 Other phrasal compounds

tàgà [mó:tì yâ:-tà] denotes a plastic man's shoe with raised top. tàgá 'shoe' has as compound-like modifier a Fulfulbe phrase '(it) doesn't go to Mopti', which also occurs in similar terms for this shoe in other Dogon languages (Bankan Tey, Najamba, Tommo So).

A lightweight woman's plastic shoe is called jènèbà-[yéré-yéré], which is said to mean 'Diénéba [woman's name] is shaking' in Bambara.

Such expressions based on VPs or clauses, in Nanga or borrowed from other languages, are not common overall.

5.2 Adjectival compounds

5.2.1 Bahuvrihi ("Blackbeard") compounds (\bar{n} \hat{a})

The initial is a noun that denotes an attribute of the referent. The final is an adjective or cardinal numeral that describes or quantifies the attribute. The compound as a whole may be used as a modifying adjective, or absolutely as a stand-alone noun.

The initial is a noun with its regular tones. The final has {HL} contour, expressed as <HL>, HL, or HHL depending on syllable count. If the initial ends in a high tone element, the tonal pattern of the bahuvrihi is the same as for possessive-type compounds. However, in the bahuvrihi, the final has {HL} contour even if the initial ends in a low-tone element.

I transcribe bahuvrihis as hyphenated compounds.

5.2.1.1 With adjectival compound final

Examples are in (xx1). The initial, which often denotes a body part, has its regular tones. The adjectival final has {HL} contour; for the location of the tone break cf. §3.7.3.2. The regular form of the adjective showing the lexical tone is given in parentheses.

```
(xx1) bèndé-dúgî
kû:-dúgî
'big-bellied' (dùgî)
kû:-mé:njè
'small-headed' (mè:njé)
sègé-mé:njè
'skin-and-bones' (lit. "skeleton-thin") (mè:njé)
gèsî-búrî
kîr"ê-sî:
kû:-púrúgǐ
'with an off-white head' (púrúgî)
```

5.2.1.2 With numeral compound final

Examples are in (xx1).

```
      (xx1)
      kû:-wôy
      'two-headed' (wŏy)

      kû:-tá:nđi
      'three-headed' (tà:nđi:)

      nă:-nîmî
      'fīve-armed' (nìmǐ:)

      nă:-kúrê
      'six-armed' (kúrê)
```

5.2.1.3 Alternative bahuvrihi with -ni suffix

I also recorded a minor compound construction with bahuvrihi-like sense but with a different form, including a terminal suffix -ni following the adjective. The only two examples I have are in (xx2). Each shows a vowel-length and tonal change vis-à-vis the normal form of the adjective. As the glosses suggest, these belong to a slangy register.

```
(xx2) dúmbó-kèmběy-ni 'having skinny buttocks' (ké:mbé 'slender [person]')
dúmbó-tàsirây-ni 'having flat buttocks' (tàsirî 'flat and small')
```

5.2.2 Compounds of 5:- 'very' plus adjective

The initial element 5:- may be used with a following adjective, in low-toned form, as one way of emphasizing the extent of the quality. It is used in predicates with the 'it is' clitic (my assistant balked at suggested examples as part of a nonpredicative NP). Examples: 5:-dùgù=\infty 'he/she is very fat', 5:-dùgù=\infty 'you-Sg are very fat'; 5:-bàr^n\u00fc=\infty 'it is very red'.

6 Noun Phrase structure

6.1 Organization of NP constituents

6.1.1 Linear order

The basic ordering of elements within NPs is (xx1).

- (xx1) Order within NP (first approximation)
 - a. Definite demonstrative kú 'that (same) ...'
 - b. prenominal possessor
 - b₁. possessor NP
 - b₂. pronominal possessor (with kin terms)
 - c. noun
 - d. modifying adjective(s)
 - e. pronominal possessor (including classifier /kô/ or /yê/)
 - f. cardinal numeral or distributive (less often, a cardinal numeral precedes (e), see (xx6), below)
 - g. determiners
 - g₁. deictic demonstrative pronoun 'this/that'
 - g₂. Definite morpheme
 - h. non-numeral quantifiers universal quantifier 'all' (kéréw) distributive quantifier 'each' (kámâ, kêw)

Examples are in (xx2). After the free translation of each example, an indication of which positions in (xx1) are filled is given in curly brackets.

(xx2) a. kú á:mádù ndò Dem Amadou house.L

```
'that (same) house of Amadou's'
                                                        {abc }
b. kú
                 ú
                               lési
                                            nè
                 2SgP
                                            Def.AnSg.L
    Dem
                               uncle.HL
    'that (same) uncle of yours-Sg'
                                                        {abc g }
c. nèr<sup>n</sup>i
                 tà:ndì
                               bû:
                               Def.AnPl
    dog.L
                 three.L
    'the three dogs'
                                                          __c__fg_}
d. nèr<sup>n</sup>i
                 tà:ndì
                               wê:
                 three.L
    dog.L
                               Dem.AnPl
    'these three dogs'
                                                           c fg }
e. [pèrgè
                 bû:
                             kéréw]
                                            ńné-èr-à
    [sheep.L
                 Def.AnPl
                             all]
                                            go-Perf-3PIS
    'All the sheep went.'
                                                          __c__gh}
f. ńdó
                                    tà:nđi:
                                               kéréw
              ſú
                      yê]
    house
              [2Sg
                    Poss.InanPl] three.L
                                               all
    'all three of your-Sg houses'
                                                          c ef h}
g. ndò
               òwó
                            Γú
                                     gŝ]
                                                        gù
    house.L
               big
                            [2Sg
                                     Poss.InanSg]
                                                        Def.InanSg.L
    'your-Sg big house'
                                                        { cde g }
h
   àdò
              òwó
                      ſú
                              yê]
                                              gà:rè
                             Poss.InanPl]
    house.L big
                      [2Sg
                                              eight.L
                                                        Def.InanPl
    'your-Sg eight big houses'
                                                        {__cdefg_}
```

kámâ 'each' is somewhat problematic in terms of ordering. It is usually combined with a simple noun: 'each house' (ndò kámâ), 'each person' (nù kámâ). It does not co-occur with the universal quantifier 'all' for logical reasons.

However, kámâ may follow a cardinal numeral that has summative (rather than distributive) sense. Therefore (xx3.a) means 'each of the three has two women' rather than 'each (group of) three has two women'. My assistant had difficulty with elicitation of distributive examples, but in (xx3.b) he eventually settled on a construction with cêw, a less common 'each/every' expression that is syntactically adverbial (cf. Jamsay cêw).

$$(xx3)$$
 a. $[ar^na]$ $ta:ndi$ $kama$ $[ya:$ $woy]$ $so-\varnothing$

[man.L **three**.L each] [woman two] have-3SgS 'Each of the three men has two women.'

b. [árⁿâ tà:ndǐ: kêw] kîlô: tùmâ béré-m-è [man.L three.L each] kilo one get-Impf-3PlS 'Each/Every (group of) three men gets one kilo.'

kámâ may co-occur with a possessor, even a postnominal possessor with a possessive classifier (xx4).

It may also follow a determiner (xx5).

Numerals sometimes precede postnominal possessors (xx6.a). However, care should be taken to distinguish this NP construction from a clause with an unquantified possessed NP as subject and with a numeral as predicate (xx6.c).

(xx6) a. pèrgé kúrê Γú yê] Poss.AnSg] sheep six [2SgP 'your six sheep' b. pèrgé kúrê yε̂] Poss.AnSg] sheep [2SgP six [=(a)]c. [pèrgé ú yê] kúrê≡yὲ 2SgP [sheep Poss.AnPl] six≡it.is.3PlS 'Your sheep are six (in number)'.

6.1.2 Headless NPs (absolute function of demonstratives, etc.)

Deictic demonstrative pronouns (§4.xxx) and universal quantifiers ('all', §6.8.1), can be used absolutely.

- (xx1) a. ŋgú kó:-ŋò-yⁿ
 Dem.AnSg eat.meal-ImpfNeg-1SgS
 'I won't eat that.'
 - kéréw k5: jὲ-∅
 all eat.meal RecPf-3SgS
 'He/She ate everything.'

6.1.3 Detachability (in relatives)

Possessors (preceding or following the head noun), modifying adjectives, and cardinal numerals remain with the NP when it functions as head NP of a relative. Determiners (demonstrative pronouns, Definite morphemes) and non-numeral quantifiers follow the verbal participle and may therefore be separated from the head NP by various other constituents.

6.1.4 Internal bracketing and tone-dropping

Within a NP without a prenominal possessor, tone-dropping is forced onto the preceding word X (noun, adjective, cardinal numeral, or distributive quantifier) in any of the sequences in in (xxx). (NPs with a possessor are covered below.)

(xx1) Tone-Dropping within NP

- a. [X + adjective] ("adjectives" includes ordinals)
- b. [X + káma 'each, any']
- c. [X + postnominal demonstrative pronoun]
- d. [X + Definite morpheme]

Tone-dropping is indexed by "L" after the gloss in the interlinear. Examples: ndó 'house' drops tones to ndò before a modifying adjective in (xx2.a), a demonstrative pronoun in (xx2.b), a Definite morpheme in (xx2.c), and distributive quantifier kámâ 'each' in (xx2.d).

```
[house.L good] Exist have-1SgS 'I have a good house.'
```

- b. ndo ngú
 [house.L Dem.InanSg
 'this house'
- c. ndo gú
 house.L Def.InanSg
 'the house'
- d. ndò kámâ
 house.L each
 'each house'

Recursive tone-dropping is seen in (xx3), where 'red' induces tone-dropping on 'house', and is in turn tone-dropped itself under the influence of the following 'good'. This seems to be the most reasonable analysis, though one could alternatively argue that the final adjective 'good' simultaneously tone-drops the preceding adjective and (at one remove) the noun.

```
(xx3) [[ndò bàr"i] èsi] yá sò-y
[[house.L red.L] good] Exist have-1SgS
'I have a good red house.' (ndô, bár"i)
```

Recursion is also observed in (xx4), where kámâ 'each' forces tone-dropping on the demonstrative pronoun wê: 'these', which has already induced tone-dropping on 'children'.

```
(xx4) [yǐ-tègè wè:] kámâ
[child-Pl.L DemAnPl.L] each
'each of these children' (yǐtègê, wê:)
```

Long-distance tone-dropping (where an element induces simultaneous tone-dropping on two other words), is discussed below.

Tone-dropping **does not occur** in the combinations in (xx5) within an NP.

- (xx5) No Tone-Dropping
 - a. [X + cardinal numeral]
 - b. [X + universal quantifier kéréw 'all']
 - c. [X + distributive adverbial quantifier kêw 'each']

d. [X + postnominal pronominal possessor]

Examples are in (xx6). In (xx6.b-c), the relevant point is that the demonstrative \acute{y} retains its tone.

- (xx6) a. ndó tà:ndi: house three 'three houses'
 - b. ndò yey kéréw house.L Dem.InanPl all 'all of these houses'
 - c. ndò yěy kêw house.L Dem.InanPl each 'each of these houses'
 - d. ndô kô: house 1SgP.InanSg 'my house'

This raises the question of what happens when a sequence of tonally independent words like those in (xx6) is itself followed by a word that forces tone-dropping, such as a determiner. kéréw 'all' cannot be followed by anything within the NP, but numerals and possessors can.

It turns out that two outputs types occur, depending on the precise combination. When **[noun + numeral]** (without a possessor) is followed by a determiner, **both the noun and the numeral drop tones**. The determiner has its normal pronunciation including a high-tone element (xx7).

- - b. yà: wòy bû:
 woman.L two.L Def.AnPl
 'the two women' (yă:, wŏy)
 - c. yà: wòy wê:
 woman.L two.L
 'these two women'

However, the combination of a **noun with a possessor** (in either order) forms a **tonosyntactic island** that is impervious to tone-dropping induced by a following determiner. Instead, the sequence [Poss noun] or [noun Poss] has the same tone it would have without a following determiner. Instead, **the determiner drops its tones**. Animate Plural bû: usually shortens its vowel and is heard as bù, though long-voweled bù: is also possible.

The tonological-island pattern is observed when the possessed noun is followed by a determiner (xx8).

```
ńdô
(xx8)
        a.
                        yε̃:
                                                  Def.InanPl.L
                        1SgP.PossInanPl
            'my houses' (= 'the houses of mine') (\acute{y})
        b. ńdô
                        k3:
                                                  gù
                        1SgP.PossInanSg
                                                 Def.InanSg.L
            'my house' (= 'the house of mine' (gú)
        c. ńdô
                        ú
                                    gŝ
            house
                        2SgP
                                    PossInanSg
                                                     Dem.InanSg.L
            'this house of yours-Sg' (ngú)
        d. ú
                        lés<sub>i</sub>
            2SgP
                        uncle.HL
                                         Def.AnSg.L
            'this uncle of yours-Sg' (nέ)
                                                          (or: ... bù:)
        e.
            ú
                        lési
                                         hiì
            2SgP
                        uncle.HL
                                         Def.AnPl.L
```

'these uncles of yours-Sg' (bû:)

In (xx8.a-c), ńdô 'house' has its lexical tones. In (xx8.d), the kin term lèsî 'uncle' has the {HL} tone imposed by the preceding possessor. Throughout (xx8), the final determiner (Definite morpheme or demonstrative pronoun) has no influence on the tones of the preceding words, and in fact own tones (compare ý, gú, ngú, and né after unpossessed nouns). This tone-dropping can be understood as a **rightward extension of the final low tone** of the overlaid {HL} or all-low tone contour on the possessed noun with preposed possessor in (xx8.d-e), or that of the similar {HL} or all-low tone contour on the possessive classifier /yê/ or /gô ~ kô/ in (xx8.a-c). Therefore, for tonological purposes, one could bracket (xxb.c) along the lines of [your [thing.HL determiner.L]], where your has tonological scope over the sequence of thing (possessive classifier) and the determiner. One could likewise bracket (xx8.e) as [your [uncle.HL determiner.L]].

The tonological-island effect is also observable in the combination of a **possessed noun with a modifying adjective**. While one may surmise that the [noun adjective] sequence becomes [noun.L adjective] by tone-dropping internal to the core NP, this is erased by the possessor. When the possessor precedes the core NP, the output is [possessor [noun.(H)L adjective.L]], where the possessor imposes either {HL} or all-low tone on the noun (depending on whether the possessor ends in a high or a low tone), and the low-tone element extends rightward through to the end of the adjective (xx9).

```
(xx9)
       a.
                      lési
                                    mòsì
           2SgP
                      uncle.HL
                                    bad.L
            'your-Sg bad uncle' (mòsi)
        b.
          û:
                      lèsi
                                    kàndà
            2PIP
                      uncle.L
                                    new.L
            'your-Pl new uncle' (kándà)
```

When the possessor follows the possessed noun, it comes at the end of the core NP (including the adjective), and there is no tonal interaction between the possessor (and its possessive classifier) and the core NP (xx10).

```
(xx10) a. lèsì mòsî yê:
uncle.L bad 1SgP.PossAnSg
'my bad uncle'

b. [nìdò owó] [ú gô]
[house.L big] [2SgP Poss.InanSg]
'your-Sg big house'
```

While determiners and modifying adjectives cannot force tonal changes on a preceding possessed noun, **distributive quantifier** kámâ 'each' can. Therefore kámâ forces double tone-dropping not only on a preceding nounnumeral sequence (xx11.a), but also on a preceding noun with pronominal possessor, in either possessor-possessed order (xx11.b) or possessed-[possessor-classifier] order (xx11.c).

```
(xx11) a.
           [nù
                       tà:ndi
                                   kámâ]
                                              [kîlô:
                                                         tùmâ-tùmâ]
            [person.L three.L
                                   each]
                                              [kilo
                                                         one-one]
            'one kilo for each three persons'
        b. [ù
                      lèsi]
                                   kámâ
            [2SgP.L uncle.L]
                                   each
            'each uncle of yours-Sg' (= 'each of your uncles') (ú lési)
```

```
c. [nndo [u go]] káma

[house.L [2SgP.L Poss.InanSg.L] each

'each house of yours-Sg' (= 'each of your houses') (nndo [u go])
```

When the prenominal possessor is a nonpronominal NP, my assistant switched from kámâ to the less common, but syntactically (and tonologically) less problematic, adverbial $k\hat{\epsilon}w$, which has no tonal interaction with the preceding string (xx13).

```
(xx13) a. [yǎ-ŋ lésǐ] kêw

[woman-Sg uncle.HL] each

'each uncle of a woman'
```

b. [àrnà wŏŋ] lési kêw [man.L Dem.AnSg] uncle.HL each 'each uncle of this man'

6.2 Possessives

6.2.1 Alienable possession (P Q.L)

For special features of kin terms (with pronominal possessor), see the immediately following section on inalienable possession. In this section I cover the productive possessed NP construction applicable to the great majority of possessed nouns.

6.2.1.1 Nonpronominal NP as prenominal possessor

When the possessor is a **nonpronominal NP** (i.e. other than a simple pronominal with no additional quantifier or determiner), it precedes the possessed noun. There is **no overt genitive morpheme** between the possessor and the possessed noun. The possessed noun has either of two overlaid tone contours.

If the possessor **ends in a low tone** (including falling <HL> or bell-shaped <LHL>), the possessed noun is **all-low toned**, indicated by ".L" in the interlinear (xx1).

(xx1) a. sùmăylâ ndò Soumaila house.L

```
'Soumaila's house' (ńdô)
b. sùmăylâ pèrgè
Soumaila sheep.L
    'Soumaila's sheep-Sg' (pèrgé)
c. [bă: yề:] sòm
    [father Poss.An.1SgP] horse.L
    'my father's horse'. (sŏm)
```

If the possessor **ends in a high tone**, the possessed noun has a **{HL} tone contour**. The tone break is near the right edge of the stem but does not obliterate the L-tone element: \hat{Cv} :, \hat{CvCv} (with final falling tone), \hat{CvCv} , \hat{CvCv} , \hat{CvCv} , \hat{CvCv} , \hat{cvCv} , \hat{cvCv} , etc. This contour is indicated by ".HL" in the interlinear. Examples are in (xx2), with the lexical form of the noun shown in parentheses after the gloss.

```
(xx2) {HL}-toned possessed noun after final-H-toned possessor
```

```
a. possessor is yǎ-ŋ 'a woman'
                           'a woman's horse' (sŏm)
   yǎ-ŋ
                sôm
                kásî
                           'a woman's calabash' (kòsì)
   yǎ-η
                           'a woman's ax' (gúlá:ri)
   yǎ-η
                gúlá:ri
   yǎ-ŋ
                túngúri
                           'a woman's stool' (túngúri)
                           'a woman's de-braiding tool' (gùsìrî)
                gúsîri
   yǎ-ŋ
                bîsîyémì
                           'a woman's acacia' (bisiyémi)
   yǎ-ŋ
b. possessor is yă: wŏy 'two women'
   [yǎ: wŏy]
                           'a horse of two women'
                sôm
c. possessor is yà: nέ 'the woman'
                           'the woman's horse'
   [yà: né]
                sôm
d. possessor is và: wǒn 'this woman'
   [yà: wŏŋ]
                sôm
                           'this woman's horse'
d. possessor is yă-ŋ kéréw 'every woman'
    [yǎ-ŋ kéréw] sôm
                           'every woman's horse'
```

As shown especially by the longer possessed nouns in (xx2.a), the tone break in {HL} occurs close to the right edge of the stem. However, most quadrisyllabic noun stems are treated phonologically as compounds, unlike 'acacia' in (xx2.a), whether or not they actually originated as compounds. This

means that the {HL} overlay is completed on the compound initial, with the compound final then continuing the L-tone (xx3). CvCv- initials are realized as $C\dot{v}C\dot{v}$ - instead of $C\dot{v}C\dot{v}$ -, since a falling tone cannot occur in a medial position.

```
    (xx3) a. yă-ŋ lásì-kòrò 'a woman's perfume' (làsì-kórô)
    b. yă-ŋ súgò-pàpà: 'a woman's Albizzia tree' (sùgó-pàpâ:)
```

6.2.1.2 Tonal treatment of modifiers following the possessed noun

If the possessed noun is followed by a **modifying adjective**, tone-dropping applies to that modifier as well. In (xx3.a-b), both 'big' and 'house' undergo tone-dropping. Phonologically, the tone contour is first applied to the noun (so that the low-tone portion of the $\{HL\}$ contour is audible on the final syllable or mora of the noun), as for the compounds described in the preceding section. A light bisyllabic (CvCv, nCv) with $\{HL\}$ contour overlaid is realized as CvCv or nCv before an adjective. Then the low tone of $\{HL\}$ extends to the end of the adjective. If the noun and adjective had been treated as an undifferentiated syllable string, the tone break in the $\{HL\}$ contour would have occurred at the final syllable break in the adjective, giving e.g. the incorrect #ya-n [ndo owa] in (xxx.b).

```
(xx3) a. [sùmǎylâ [ǹdò ɔ̀wò]] àrnáná bù-\varnothing [Soumaila [house.L big.L] where? be-3SgS 'Where is Soumaila's big house?' (ṅdô, òwó)
```

```
    b. [yă-ŋ [ńdò ɔwò]] àrnaŋa bù-∅
    [woman [house.L big.L]] where? be-3SgS
    'Where is a woman's big house?' (ńdô, ɔwó)
```

Tone-dropping also applies to a **cardinal numeral** following the possessed noun (or noun plus adjective). Therefore **kúrê** 'six' drops to **kùrê** in (xxx.a-b). As we will see later (end of §6.2.2), a prenominal pronoun possessor (before an inalienably possessed noun) does not cause tone-dropping on a numeral.

```
(xxx) a. [sùmǎylâ ndò kùrè] àrnáná bù-Ø [Soumaila house.Lsix.L] where? be-3SgS 'Where are Soumaila's six houses?' (ndô, kúrê)
```

```
b. [sùmǎylâ ndò òwò kùrè] àrnáná bù-Ø [Soumaila house.L big.L six.L] where? be-3SgS 'Where are Soumaila's six big houses?' (ndô, òwó, kúrê)
```

A **Definite** morpheme, **demonstrative** pronoun, or non-numeral **quantifier** ('all') at the end of the possessed NP is **not tone-dropped** under the influence of a prenominal possessor (xxx.a-c). (In the case of kéréw 'all', it is debatable whether the quantifier is in the NP or is a postnominal adverbial.)

- (xxx) a. [sùmǎylâ ndò kú] Èsì≡ndó-∅ [Soumaila house.L Def.InanSg] good≡not.be-3SgS 'The house of Soumaila's is no good.'
 - b. [sùmǎylâ ndò ngú] èsì≡ndó-∅
 [Soumaila house.L this.InanSg] good≡not.be-3SgS
 'This house of Soumaila's is no good.'
 - c. [sùmǎylà ndò kéréw] Èsì≡ndó-∅ [Soumaila house.L all] good=not.be-3SgS 'Every house of Soumaila's is no good.'

6.2.1.3 Pronoun plus classifier as postnominal possessor

A **pronominal possessor** is expressed by juxtaposing the possessed noun (in its regular form) with a following pronominal possessor. For **3Sg** (including Inanimate Singular), there is a special invariant possessor morpheme $/n\delta$ /. For all other pronominal categories, the postnominal possessor form consists of a pronominal morpheme plus a **possessive classifier** that originated as a generic noun. Thus 'your house' is expressed as 'house [you(r) thing]', essentially an **appositional** construction with the pronominal attached directly to the generic noun 'thing' that now functions as a classifier agreeing with 'house'. There are two such classifiers, one (gô) used for inanimate singulars, and the other (yê) used for everything else: animate singular, animate plural, and inanimate plural.

For **inanimate singulars**, the generic noun is $\frac{g}{3}$ (varying with $\frac{k}{3}$) which is related to the noun $\frac{k}{3}$ 'thing'.

The 1Sg form fuses the pronominal possessor with the generic noun, resulting in $k\tilde{\mathfrak{d}}$: (with consistently voiceless /k/, and with <LHL> tone). This suggests an original form of the general type $*\tilde{\mathfrak{d}}$ $k\hat{\mathfrak{d}}$:, where *x was C-final (favoring preservation of voiceless /k/) and had a L-tone (or R-tone) that ended up as the initial L tone segment in <LHL>. The usual 1Sg pronoun is $\check{\mathfrak{t}}$:", which is close enough if we assume an original stem-final nasal consonant ($*\check{\mathfrak{t}}$). The other combinations with this classifier were regularly heard with /g/, but in elicitation I also heard variants with /k/. We get $/g\hat{\mathfrak{d}}/$ (after H-tone) or $/g\hat{\mathfrak{d}}/$ (after F-tone). The paradigm is (xx1).

(xx1) Pronominal Possessor (inanimate possessed noun)

category	possessor form
1Sg	kð:
1Pl	î: gò
2Sg	ú gô
2Pl	û: gò
3Sg Inan	 nò (but predicative ńné gô≡ŋ 'is his/hers') nò (but predicative kú gô≡ŋ 'is its')
3Pl	bû: gò
InanPl	kû: gò
Refl/LogoSg	á gô
Refl/LogoPl	â: gò

With 'house', for example, we get forms like those in (xx2.a-c). The predicative form for 3Sg possessor is seen in (xx2.d). The noun /k5/ 'thing' may also be possessed, like any other inanimate noun (xx2.e).

- (xx2) a. ndô kɔ̃: house 1SgP.Inan 'my house'
 - b. ńdô ú gô house 2Sg Poss.Inan 'your-Sg house'
 - c ńdô nò house 3SgPoss 'his/her house'
 - d. [nndo gú] [nnε kô]=ŋ
 [house.L Def.InanSg] [3Sg Poss.Inan]=it.is.3SgS
 'This house is his/hers.'
 - e. k5 ú gô thing 2SgS Poss.Inan 'your-Sg thing'

When the possessed noun is **inanimate plural**, the generic noun $g\hat{o}$ is either directly pluralized by adding Definite Inanimate Plural \hat{y} (in L-toned form), or replaced by another generic noun $y\hat{\epsilon}$. The first alternative is exemplified in (xx3).

```
(xx3) a. <a href="mailto:nd6">nd6</a> kɔ̃: ỳ house 1SgP.Poss.InanSg Def.InanPl 'my houses'

b. <a href="mailto:nd6">nd6</a> î: gɔ̀ ỳ house 1Pl Poss.InanSg Def.InanPl 'our houses'
```

The alternative, whereby generic $g\hat{\mathfrak{o}}$ is **replaced** (for inanimate plural possessed noun) by $y\hat{\mathfrak{e}}$, is interesting since the same generic $y\hat{\mathfrak{e}}$ is used for **animate** possessed noun (regardless of number). For first and second person possessor, and for 3Pl and InanPl possessor, the two paradigms are identical (xx4).

(xx4) Pronominal Possessor (inanimate plural or animate possessed noun)

```
possessor form
category
1Sg
                yε̃:
1P1
                î: yè
2Sg
                ú yê
2P1
                û: yè
                nò (but predicative ńné yê≡ỳ)
3Sg
3P1
                bû: yè
                nò (but predicative kú yê≡ỳ)
InanSg
InanPl
                kû: yè
Refl/LogoSg
                á yê
Refl/LogoPl
                â: yè
```

Inanimate plural possessed nouns are exemplified in (xx5).

```
(xx5) a. ndô yê:
house 1SgP.Poss.InanPl
'my houses'
```

```
b. ndô ú yê house 2Sg Poss.InanPl 'your-Sg houses'
```

c. ńdô î: yè house 1Pl Poss.InanPl 'our houses'

Animate singular possessed nouns are in (xx6).

```
(xx6) a. pèrgé yê:
sheep 1SgP.Poss.AnSg
'my sheep-Sg'
```

- b. kóróji ú yê family 2SgP Poss.AnSg 'your-Sg family'
- c. gùndá î: yè slave 1PIP Poss.An 'our slave'

The situation for 3Sg (and Inanimate Sg) possessor is more complex. With an animate singular possessed noun, the same $/n\delta$ / seen above (with inanimate singular possessed noun) is used postnominally. The expected, fully transparent predicative forms are also used. For inanimate plural possessed noun, 3Sg possessor can again be expressed by $/n\delta$ /, meaning that there is no distinction between singular and plural possessed noun. The distinction can be expressed by a following determiner, such as Definite (Inanimate Singular gú, Inanimate Plural ý, in low-toned forms): ńdô nò gù 'the house of his/hers', ńdô nò ỳ 'the house of his/hers'.

While nò is the usual 3Sg/Inanimate possessor form, it is also possible to revert to the more generally productive pattern for pronominal possessors, using 3Sg pronoun ńné as the possessor, with a following classifier. (My assistant rejected comparable examples with Inanimate Singular pronoun kú.) Thus ńné gô 'his/her' for inanimate singular possessed noun, and ńné yê for inanimate plural (as well as all animates).

```
(xx7) a. usual invariant 3Sg/Inanimate possessor
```

3Sg nò (but predicative ńnε yε̂≡ὴ)

```
InanSg nò (but predicative kú yε≡ὴ)
```

b. alternative 3Sg possessor form with classifier

```
3Sg with inanimate singular nné gô
3Sg with inanimate plural nné yê
```

An example with animate singular possessed noun is (xx8.a). Of course the reading with inanimate possessor and animate possessed noun is rare. That nò, like other postnominal pronominal possessors (those with classifiers), can follow a modifying adjective, is shown by (xx8.b). Predicative counterparts, distinguishing animate 3Sg possessor (common) from inanimate singular possessor (uncommon) are in (xx8.c-d).

- (xx8) a. pèrgé nò sheep 3SgP.AnSg 'his/her sheep-Sg' = 'its sheep'
 - b. pèrgè dùgî nò
 sheep.L big 3SgP.AnSg
 'his/her big sheep-Sg' = 'its big sheep'
 - c. [pèrgè wŏŋ] [ńné yê]=ŋ̀ [sheep.L Dem.AnSg] [3Sg Poss.AnSg]=it.is.3SgS 'This sheep is his/hers.'
 - d. [pèrgè wŏŋ] [kú yê]=ŋ̂
 [sheep.L Dem.AnSg] [InanSgPoss.AnSg]=it.is.3SgS
 'This sheep is its (= belongs to it).'

Examples with inanimate plural possessed nouns are in (xxx). (xxx.a) is only good for animate possessor, while (xxx.b) can have animate or inanimate possessor. Predicative forms are in (xxx.c-d).

- (xxx) a. ndô nné yê house 3Sg Poss.InanPl 'his/her houses'
 - b. ndó nòyn
 house InanSgP
 'his/her houses = 'its houses'

```
b. [nndo ngú-yè] [nné yê]-n

[sheep.L DemInanPl] [3Sg Poss.AnSg]-be.3SgS

'This sheep is his/hers.'
```

If the possessed noun is animate plural, the same forms as for animate singular are used.

The **grammatical number** of the possessed noun may be expressed using a Definite morpheme or a demonstrative pronoun, which distinguish singular from plural. In this combination, the determiner is low-toned, carrying over the final tone of the possessive morpheme. The use of a number-specified determiner is especially useful for animate possessed nouns, which use the same possessive morpheme $y\hat{\epsilon}$ for singular and plural.

- - b. [pèrgé yề: wòŋ] gùró j-à [sheep 1SgPoss.An] this.AnSg.L] steal RecPf-3PlS 'They stole this sheep of mine'. (wŏŋ)
 - c. [pèrgé yè: bù:] gùró j-à [sheep 1SgPoss.An] Def.AnPl.L] steal RecPf-3PlS 'They stole my sheep-Pl'. (bû:)
 - d. [pèrgé yề: wè:] gùró j-à [sheep 1SgPoss.An] thisAnPl.L] steal RecPf-3PlS 'They stole my sheep-Pl'. (bû:)

6.2.2 Inalienable possession

6.2.2.1 Kin terms and similar relationship terms

Kin terms have the same possessed NP construction as do alienable nouns when the possessor is a **nonpronominal NP**. As with alienable possession, the possessor precedes the possessed noun, which has overlaid {HL} or all-low tone depending on whether the possessor ends in a high or low tone.

```
(xxx) a. sùmǎylà bà:
Soumaila father.L
'Soumaila's father' (bǎ:)
```

```
b. ámbéri dè:
    chief mother.L
    '(a/the) chief's mother' (dě:)
c. [àmbèri né] dê:
    [chief.L Def.AnSg] mother.HL
    'the chief's mother'
```

As we have seen, with alienable nouns, pronominal possessors are expressed in a distinctive construction involving a postnominal possessive classifier ('house you(r) thing' = 'your house'). With inalienables nouns (i.e. kin terms) as the possessed noun, however, pronominal possessors (other than 1Sg and 3Sg/Inanimate, on which see below) have the same prenominal position as do nonpronominal possessors. The possessed noun, as usual, has overlaid {HL} contour if the pronoun ends in a high tone (2Sg ú, Logophoric Sg á), and all-low contour if the pronoun ends in a low-tone element (1Pl î:, 2Pl û:, 3Pl bû:, Logophoric Pl â:).

For 3Sg (including Inanimate) and 1Sg possessor, there is no difference between alienable and inalienable possessor. 3Sg possessor is expressed by the invariant postnominal morpheme no, and 1Sg possessor is expressed by the the relevant 1Sg form of the postnominal possessive classifier, which for kin terms (which of course are animate) is always yes: in both singular and plural.

The paradigm of ba: 'father' is therefore (xx1).

```
(xx1)
                        'X's father'
            category
        a. prenominal possessor ending in high tone, {HL} contour
            2Sg
                        ú bâ:
            LogoSg
                        á bâ:
        b. prenominal possessor ending in low tone, all-low contour
            1P1
                        î: bà:
            2P1
                        û: bà:
            3P1
                        bû: bà:
            LogoPl
                        â: bà:
        c. postnominal possessorwith possessive classifier, lexical tone
            1Sg
                        bă: ỹe:
        d. special 3Sg postnominal morpheme, lexical tone
                        bă: nò
            3Sg
```

Further examples of the full {HL} contour with kin terms: dê: 'mother' (unpossessed dě:), dérè 'elder sibling' (unpossessed dèré), lógò 'husband' (lògó), ómîrⁿi 'parent-in-law' (òmìrⁿi).

Aside from the obvious kin terms, some other relationship nouns are treated in this fashion. Examples: tiyá 'friend', tògĭrá 'namesake' (anyone with the same personal name), tǔ: 'agemate', kàdàgá 'agemate group', àndirí 'rival, competitor'. The forms with {HL} possessed-noun contour are tîyà, tógîrà, tû:, kádágà, and ándîrĭ. The Reciprocal pronoun tǔ: is related to tǔ: 'agemate' (§18.3).

'Woman' (even in the sense 'wife') and 'child' (even in the kinship sense) are regular alienable nouns, so pronominal possessors are postnominal: yă: ú yê 'your-Sg wife', yî: ú yê 'your-Sg child'

6.2.2.2 No tone-dropping of numeral after pronominally possessed kin term

Like nonpronominal possessors, preposed pronominal possessors impose an {HL} or (if ending in a low tone) all-low tone contour not only on the following possessed noun, but also on a following modifying adjective (which is therefore always low-toned). Thus [ú dérè mòsì] 'your-Sg bad (older) brother', where mòsì 'bad' drops tones just as it does in .e.g [[àrnà né] dérè mòsì] 'this man's bad (older) brother'.

However, there is a difference between pronominal and nonpronominal possessors with respect to tone-dropping of a numeral following the possessed noun. In (xx1.a-b), the nonprominal possessor NP imposes the all-low (xxx.a) or {HL} (xx1.b) contour on the possessed NP, extending the low-toned portion to the numeral (kúrê 'six'). In (xx1.c), by contrast, kúrê keeps its lexical tones after a noun with pronominal possessor.

- (xx1) a. sùmǎylâ dèRè kùrè S older.sibling.L six.L 'Soumayla's six older brothers'
 - b. [àrnà wǒn] dérè kùrè [man.L Dem.AnSg] older.sibling.HL six.L 'this man's six older brothers'
 - c. ú dérè kúrê 2SgP elder.sibling.HL six 'your-Sg six older brothers'.

6.3 Noun plus adjective

6.3.1 Noun plus regular adjective

Modifying adjectives (including ordinals) follow the noun. In the absence of a possessor, an adjective induces tone-dropping on the noun. Thus ndô 'house' drops its tones to ndô in (xx1.b-d).

```
(xx1) a. ńdô '(a) house'
b. 'ndò èsî '(a) good house'
c. 'ndò pîrî '(a) white house'
d. 'ndò kǐyá '(the) first house'
```

6.3.2 Adjective gàmbi 'certain (ones)'

The adjective gàmbi is most often used in the plural form gàmbi-yê, meaning 'certain (ones)', i.e. a subset (with at least two members) of a larger collectivity. The singular form gàmbi '(a) certain (one)' is also possible but less common in connection with countable entities.

```
(xx1) a. nǔ:
   b. nù gàmbî
   c. nù gàmbî-yê
   '(a) person'; 'people'
   'a certain person'
   'certain people'

(xx2) a. kúr³ô
   b. kùr³ò gàmbî
   c. kùr³ò gàmbî-yê
   'a certain stone'
   'certain stone'
```

A typical discourse context for gàmbi is a parallel construction contrasting an eventuality involving one subgroup with a distinct or polar opposite eventuality involving the complementary subgroup (xx3). The noun may or may not be repeated with the second occurrence of gàmbi.

```
(xx3)
       [nù
                        gàmbi]
                                        wárà: n-s-è,
                                       do.farm.work-Prog-3PIS,
        [person.L
                        certain]
                        gàmbi]
                                        wórî
                                                    dògó-tì-yà
        [nù
        [person.L
                                       farming
                        certain]
                                                    leave-Perf1b-3PIS
        'Some people are (still) farming, (while) some (others) have given up
        farming.'
```

With mass nouns like 'sugar', gàmbi means 'some of (X)', denoting a portion of a larger quantity (xx4).

```
(xx4) [sùkòrò gàmbî] kò:-y∴ ndè, gàmbî àgî-yé-m-ì: [sugar.L some] eat.Perf.L-1PIS if, some hold-MP-Impf-1PIS 'We'll consume some of the sugar, and we'll keep some (= the rest).'
```

6.3.3 Expansions of adjective

6.3.3.1 Adjective sequences

An adjective may follow a sequence of a noun and one or more adjectives. In this case, the final adjective has its lexical tones, but all preceding words in the core NP are tone-dropped.

```
(xx1) a. nèr<sup>n</sup>i dùgi mòsi dog.L big.L nasty 'a big vicious dog' (nèr<sup>n</sup>i, dùgi)
```

b. yèbùmbà bàrⁿi gùró snake.L red.L long 'a long red (= brown) snake' (yèbúmbâ, bárⁿi)

6.3.3.2 'Good to eat'

In the most transparent construction, the NP in question is subject of an adjectival predicate ('be good' etc.), and the activity type is expressed by the locative form of a verbal noun. The adjectival predicate agrees with the subject (xx1.a-b.).

```
b. yì-tègê [kúwó-ndé gá] érî-yè children [eat.meat-VblN in] be.good-Pl 'Children are good to eat.'
```

A variation on this is with a form of the verb equivalent to the Imperfective (without any further pronominal suffixation) instead of the verbal noun.

winged.termite eat.meat-Impf in be.good 'Winged termites are good to eat.'

These constructions are distinct from one where the verbal noun is the subject of the adjectival predicate (xx3). Here there is no agreement in the predicate with the subject of the verbal-noun complement ('children').

(xx3) [yǐ-tègê ŋîrʰé-ndé] érû [children look-VblN] be.good 'It's good to look at children.'

6.4 Noun plus cardinal numeral

Cardinal numerals follow a noun and any modifying adjectives, which together constitute the core NP. In the absence of a possessor, the numeral has its lexical tones, and it has no tonal effect on the words in the core NP.

(xx1) a. ńdô 'house' 'two houses'
b. ndô owó 'a large house' 'two large houses'

The numeral remains with the core NP (and a possessor if present) when it functions as head NP of a relative.

6.5 Noun plus determiner

6.5.1 Prenominal kú 'the (afore-mentioned)'

kú before a noun may be used as a discourse-definite determiner, somewhat more emphatic than the usual postnominal Definite morphemes (which may cooccur with it). In form, kú is indistinguishable from the Inanimate Singular possessor pronominal kú 'its'. It also has the same tonal treatment as the pronominal possessors. The possessed noun has $\{HL\}$ tone contour overlaid. The examples in (xx1) include postnominal Definite morphemes, which in the absence of kú force tone-dropping on the preceding noun. Determiners such as Definite morphemes have their regular tones (i.e. they are not themselves subject to tone-dropping).

```
kúr<sup>n</sup>ô
                                        'stone'
(xx1) a.
              kùr<sup>n</sup>ò gú
                                        'the stone'
              kú kúr<sup>n</sup>ò (gú)
                                       'that (same) stone'
         b. kùr<sup>n</sup>ò ý
                                        'the stones'
              kú kúr<sup>n</sup>ò (ý)
                                        'those (same) stones'
         c. nů:
                                        'person'
              nù: nέ
                                        'the person'
                                        'that (same) person'
              kú nû: né
         d. nù: bû:
                                        'the people'
                                        'those (same) people'
              kú nû: bû:
```

An example with a postnominal demonstrative pronoun is kú kúrⁿò ŋgú 'this (same) stone', with the Inanimate Singular demonstrative ṅgú.

Although kú behaves much like a possessor syntactically, it can **co-occur** with a true possessor morpheme. In (xx2.a), the possessor is a pronoun, and therefore it (along with its possessive classifier) follows the noun. The low tone of the final Definite morpheme is attributable to ú gô, as is suggested by the bracketing. In (xx2.b), kú precedes (and has broad semantic scope over) a sequence beginning in an indefinite possessor NP ('woman'). kú here has no effect on this possessor NP (or on any other word). This construction is distinct from that in (xx2.c), where kú has local scope over the immediately following possessor (again 'woman'), and therefore forces the {HL} possessed-noun contour on this noun. My assistant accepted (xx2.d) as grammatical, though awkward; it combines two occurrences of Definite kú, with broad and narrow semantic scope, respectively.

```
(xx2) a. [kú
           [Def house.HL] [[1SgP.Poss.InanSg Def.InanSg] Def.InanSg]
           'that (same) house of yours-Sg'
       b. kú
                                                    gú]
                      [yǎ-ŋ
                      [woman-Sg
                                                    Def.InanSg]
           Def
                                     house.HL
           'that (same) house of (a/the) woman'
       c.
          [kú
                                         'ndò
                                                    gú
                      yâ-ŋ]
                                         house.L
                                                    Def.InanSg
           Def
                      woman-Sg.HL]
           '(a/the) house of that (same) woman'
       d. kú
                 [[kú
                        yâ-ŋ]
                                         'ndò
                                                    gúl
           Def [Def
                        woman-Sg.HL]
                                         house.L
                                                    Def.InanSg]
```

'that (same) house of that (same) woman'

6.5.2 Postnominal demonstrative pronouns

A demonstrative pronoun ($\S4.4.1.1$) follows the core NP and any cardinal numeral that may be present. In the absence of a possessor, the demonstrative retains its own tones, but forces tone-dropping on the final word of the core NP (the nonfinal words in the core NP are already tone-dropped), and (simultaneously) on the numeral. The illustrative examples in (xx1) involve the inanimate demonstrative pronouns, singular $\eta g u$ and plural yey.

(xx1)	a.	ńdô ndò ngú	'house' 'this house'
	b.	ndò òwó ndò òwò ngú	'(a) big house' 'this big house'
	c.	ńdô wŏy ǹdò wòy yĕy	'two houses' 'these two houses'
	d.	[ǹdò ɔ̀wɔ́] wŏy [ǹdò ɔ̀wɔ̀] wòy yĕy	'two big houses' 'these two big houses'

When a NP containing a demonstrative functions as head NP of a relative, the demonstrative pronoun is separated from the core NP and the numeral, and is placed after the verbal participle (§14.xxx).

6.5.3 Definite morpheme plus noun

The Definite morphemes (§4.4.1) function syntactically in the same fashion as the demonstrative pronouns. They induce tone-dropping on the final word of a preceding core NP, and (simultaneously) on any cardinal numeral.

(xx1)	a.	ńdô ndò gú	'house' 'the house'
	b.	ndò dwó ndò dwd gú	'(a) big house' 'the big house'
	c.	ńdô wŏy ǹdò wòy ý	'two houses' 'the two houses'

```
d. [ndò owo] woy 'two big houses' 
[ndò owo] woy ý 'the two big houses'
```

Like the demonstrative pronouns, Definite morphemes are separated from a NP that functions as head of a relative, and are placed after the verbal participle (§14.xxx).

6.6 Universal and distributive quantifiers

```
6.6.1 'All' (kéréw, sóy)
```

For 'all' (universal quantifier), kéréw is added at the end of the NP. With inanimates, agreement is 3Sg rather than 3Pl (xxx.a). With animates, agreement is plural. kéréw may be combined directly with an independent pronoun (xxx.c), with appropriate person agreement. kéréw can also be used absolutely, in the sense 'everything' or 'everyone'.

- (xxx) a. [ndò ý kéréw] yègé-èrè-∅ [house.L Def.InanPl] all] fall-Perf1a-3SgS 'All of the houses fell (=collapsed).'
 - b. [yǐ-tègè bû: kéréw] yá b-è [child-Pl.L Def.AnPl all] Exist be-3PlS 'All of the children are present.'
 - c. [1: kéréw] yá bù-y:...
 [1Pl all] Exist be-1PlS
 'We are all present.'
 - d. kéréw ńné-èr-à all go-Perf1a-3PIS 'Everyone went.'
 - e. kéréw kó:-fi-yà all eat-Perf1b-3PIS 'They ate everything.'

It is also possible to use the more emphatic sóy in the same NP-final position (xxx.a), and absolutely (xxx.b). However, sóy can also be used phrase-initially, and in this case it may co-occur with kéréw (xxx.c).

- (xxx) a. [ndò ý sóy] nàmé-èrè-∅ [house.L Def.InanPl all] be.ruined-Perf-3SgS 'All of the houses were ruined.'
 - d. sóy ńné-èr-à all go-Perf-3PIS 'Everyone went.'
 - c. [sóy yǐ-tègè bû: kéréw] yá b-è [all child-Pl.L Def.AnPl all] Exist be-3PlS 'All of the children are present.'

For kéréw and sóy with negation, see §6.xxx, below. For '(not) at all' emphatics, see §19.xxx.

6.6.2 'Each' (kámâ)

Distributive 'each' is kámâ. It forces tone-dropping on the preceding noun: ńdó 'house', 'ndò kámâ 'each house'. Agreement is 3Sg. It cannot be directly attached to a personal pronoun, but an independent pronoun with nonsingular reference may precede it appositionally. In this case, agreement is 3Sg regardless of the category of the pronoun (xxx.b).

- (xxx) a. [nù kámâ] [pèrgé wŏy-wŏy] ńdî-m-ĭ:y
 [person.L each] [sheep two-two] give-Impf-1PlS
 'We will give two sheep (apiece) to each person.'
 - b. [nù kámâ]
 [person.L each]
 [pèrgè tùmâ-tùmâ ǹjî-ŋ ńdî-ṅ
 [sheep one-one] 1SgO give-Impf.3SgS
 'Each person will give me one sheep.'
 - b. î: [nù kámâ]

 1PIS [person.L each]

 [á gò] éwé-ŋ

 [Refl Poss.InanSg] pay-Impf.3SgS

 'Each of us will pay for his own (portion).'

The expression kò kámâ 'each thing' is commonly used under negation to mean '(not) anything', i.e. 'nothing'. It is a high-frequency expression and is often heard as [kòkómà] with partial forward vocalic assimilation.

6.6.3 Universal and distributive quantifiers with negation

Of the two 'all' particles (universal quantifiers) described above (§6.xxx), the more emphatic sóy is most likely to have wide scope containing a negation. It therefore functions to emphasize the negation (xxx).

```
(xxx) a. sóy kó:-ŋò-w-nde, tù-túwɛ́-m̀-w all eat-Perf.Neg-1SgS, Rdp-die-Impf-2SgS 'If you-Sg don't eat at all, you will die.'
```

The more basic and unmarked universal quantifier kéréw is more usual in narrow-scope contexts, i.e. where the negation has wide scope (xxx).

(xxx)[yú: kà-kã: gàmbi kúwó ń, a. î: gà], [millet 1Pl Poss.InanSg] Rdp-grasshopper some eat and.SS gàmbí dòg-ò, [kéréw kùwò-ndú gày leave.Perf.L-3PIS, [all Topic] eat-**Perf.Neg**.3Pl 'The locusts ate some of our millet and left some; they did not eat all (of it).'

6.6.4 Universal quantifier combined with a numeral

6.7 Accusative -n

The Accusative suffix (or clitic) -ŋ can be added to nouns and pronouns in direct object function. It is more common with animate NPs and pronouns, but is also possible with inanimates. It occurs at most once in a multi-word NP.

It usually reduces phonetically to nasalization and perhaps very slight lengthening of a preceding vowel, though I normalize transcription as -ŋ. When followed by a vowel-initial word, we might expect a loud and clear [ŋ], but the -ń in pèrgé-ń 'sheep-Acc' has the same phonetic reductions in e.g. pèrgé-ń ay-só-ý 'I picked up a sheep' as it does before a consonant or prepausally. The suffix is often **entirely inaudible** in contexts where it is allowed. Inaudibility is common when the final syllable of the form is already nasal (kúrⁿô 'stone', árⁿâ 'man', nàmâ 'meat'), but inaudibility is also fairly common even in more

benign phonological environments. When there is no audible cue, it is difficult to determine whether the suffix is structurally absent (i.e. is optional), or has merely been reduced to zero phonetically. My practice is to omit it in transcriptions unless it has at least some audible trace (such as nasalization). However, (animate) pronouns seem to have (underlyingly) obligatory -ŋ in the relevant synctactic environments, and in the case of 1Sg njî-ŋ even the phonetic disappearance of -ŋ is recoverable from the change in pronominal allomorph (compare regular 1Sg pronoun i:n).

The suffix has **no intrinsic tone**, so the final tone of the noun or pronoun is extended to the suffix. (xx1.a) show $-\acute{\eta}$ after a simple animate noun stem. The pronominal version in (xx1.b) also shows $-\acute{\eta}$. A prenominal possessor is compatible with $-\acute{\eta}$ on the possessed noun (xx1.c).

- (xx1) a. pèrgé-ń sémé-só-ý sheep-Acc slaughter-Perf2-1SgS 'I slaughtered a sheep.'
 - b. ńné-ń sémé-só-ý AnSg-Acc slaughter-Perf2-1SgS 'I slaughtered it (animate).'
 - c. [á bâ:-ŋ] yĭ:-só-Ø [ReflSgP father-Acc] see-Perf2-3SgS 'She saw her (own) father.'
 - d. béré-ń késé-só-ý stick-Acc cut-Perf2-1SgS 'I cut a stick.'

If a noun is followed by a modifying adjective, Accusative - η (if present at all) is attached to the adjective (xx2.a-b). The same is true when the noun is followed by a postnominal possessor (pronominal plus possessive classifier) (xx2.c). The noun cannot also take - η in these combinations.

- (xx2) a. [pèrgè èwré-ń] sémé-só-ý slaughter-Perf2-1SgS 'I slaughtered a small sheep.'
 - b. [àrnà gàwá-ń] súyó-só-ý [man.L tall-Acc] hit-Perf2-1SgS 'I hit a tall man.'
 - c. [pèrgé á yê-ŋ] sémé-só-Ø [sheep ReflSgP Poss.AnSg-Acc] slaughter-Perf2-3SgS

'He slaughtered his (own) sheep-Sg.'

When a numeral greater than 'one', the universal quantifier kéréw 'all', distributive quantifier kêw 'each', or săy 'only' follows an animate noun, Accusative -n is suffixed to the noun (not to the quantifier) (xx3).

- (xx3) a. [pèrgé-ý pê:r] sémé-só-ý [sheep-Acc ten] slaughter-Perf2-1SgS 'I slaughtered ten sheep.' (pé:rù)
 - b. [pèrgé-ŋ kéréw] sémé-só-ý slaughter-Perf2-1SgS 'I slaughtered all the sheep.'
 - c. [pèrgé-ŋ kêw] [ná wŏy-wŏy] sùyò-ỳ [sheep-Acc each] [time two-two] hit-Perf2-1SgS 'I hit each sheep twice.'

The more common distributive quantifer kámâ is treated like a modifying adjective (in this as in other respects), so $-\eta$ in this case is added to the quantifier (xx4.a). The numeral 'one' is also an adjective syntactically and takes -N directly (xx4.b).

- (xx4) a. [pèrgé-ý sǎy] sèmè-ỳ [sheep-Acc only] slaughter.Perf.L-1SgS 'I slaughtered only a sheep.'
 - b. [pèrgè kámâ-ŋ] [ná wŏy-wŏy] sùyò-ỳ [sheep.L each-Acc] [time two-two] hit-Perf2-1SgS 'I hit each sheep twice.'

Determiners are the only postnominal modifiers that Accusative $-\eta$ can attach to. In this case, the noun (which is low-toned) does not also show $-\eta$ (xx5).

- - b. [pèrgè èwrè né-ń] sémé-só-ý [sheep.L small.L Def.AnSg-Acc] slaughter-Perf2-1SgS 'I slaughtered the small sheep-Sg.'
 - c. [kùrⁿò gú-ή] gùyó-só-ý

```
[stone.L Def.InanSg-Acc] break.up-Perf2-1SgS 'I broke up the rock.'
```

d. [ndò ngú-n] éwé-só-y [house.L Dem.InanSg-Acc] buy-Perf2-1SgS 'I bought this house.'

However, the determiners that **end in a consonant**, namely demonstrative pronouns won 'this (animate)' and yey 'these' (inanimate), and Definite Plural \acute{y} , do not occur with Accusative - η (or at least do not allow it to be audibly expressed).

Accusative -ŋ does not occur in my data with pro forma cognate-nominal objects (§xxx), like kò-kósô 'coughing' in kò-kósô kósó-só-ý 'I coughed (a cough)'.

A difficulty in parsing texts is that Accusative $-\eta$ is phonologically indistinguishable from $=\eta$, the 3Sg animate form of the 'it is' clitic, which is commonly added to nouns and NPs ($\S xxx$).

7 Coordination

7.1 Conjunction

7.1.1 NP conjunction ('X and Y') with yò

The basic 'and' conjunction is yò, following both coordinands (xxx).

7.1.1.1 Ordering of coordinands

My assistant indicated that the coordinands can be ordered either way. He did, however, express a preference for ordering pronouns before names or other NPs. In combinations of two pronouns, he preferred ordering third person before second person ('he/she and you') and second before first ('you and I').

7.1.1.2 Conjunction with final quantifier

kéréw 'all' may appear at the end of a conjoined NP.

7.1.2 "Conjunction" of verbs or VP's

The conjoined NP type [X yò Y yò] is not used with verbs, VP's, or clauses (except when quoted and treated as NPs, as in 'X said ["Y" and "Z"]').

For the various ways to chain together verbs, VP's, and clauses, see Chapter 15.

7.2 Disjunction

There is no sharp distinction between disjunctive 'X or Y' in indicative contexts, and polar (including yes/no) interrogatives (§13.2.1).

```
7.2.1 'Or' (mà)
```

The simplest disjunctive phrase is of the type [X mà Y mà], as in (xx1.a). This is directly comparable to the conjunction type [X yò Y yò] described above. However, there is an asymmetry in the disjunctive construction, in that the first but not the second mà is subject to intonational prolongation, the effect of which is to make it difficult to bracket this mà with the first coordinand. Furthermore, the second mà is often omitted, as in (xx1.b).

```
(xx1) a. àsú⇒ [pèrgé mà⇒ bérî mà] sémé-m-i∴
always [sheep or goat or] slaughter-Impf-1PlS
'We slaughter (either) a sheep or a goat always (= every day).'
```

```
    b. [nù bû: [nàŋá tà:ndǐ: mà⇒ nŏy¹] [person.L Def.AnPl] [cow three or four] màrà-g-à be.lost-Caus-Perf.L-3PlS 'The people lost three or four cows.'
```

7.2.2 Clause-level disjunction

It is particularly difficult to keep clausal disjunction (which by definition allows two or more possibilities) distinct from polar interrogation. The examples in (xx1) were elicited as disjunctions. They show a single occurrence of $ma \Rightarrow$ 'or' between the two clauses, which is also possible with interrogatives ('will we go to the market, or will we die?'). In the indicative disjunctive sense, my assistant showed a distinct **preference for the unsuffixed Perfective** as opposed to the Imperfective for future as well as past time frames. Stative verbs are also possible (xx1.c).

```
(xx1) a. \dot{\epsilon}:\eta\hat{i} [\dot{\epsilon}w\dot{\epsilon} gá] \dot{n}n\dot{\epsilon}-y... m\dot{a}\Rightarrow t\dot{u}w\dot{\epsilon}-y...
```

tomorrow [market in] go.Perf.L-1PIS or die.Perf.L-1PIS 'Tomorrow, (either) we'll go to the market, or we'll die.' (lit. "... we'll have gone ... or we'll have died")

- b. bàmàkó nnè-w mà⇒ ngá bè:-w
 B go.Perf.L-2SgS or here stay.Perf.L-2SgS
 'You-Sg will go to Bamako or you will stay here.' (but e.g. you won't go to Mopti)
- c. [ô: gó] nnè-Ø mà ni-niyà-Ø [field.L in] go.Perf.L-3SgS of Rdp-sleep.Stat-3SgS 'Either he has gone to the field, or he's sleeping.' (since he doesn't answer our knock at the door)

My assistant rejected disjunctions including morphological imperatives (either two imperatives, or a combination of some other verb form with a final imperative). (xx1.b) can be used as a rough pragmatic equivalent to a disjunction of two imperatives.

8 Postpositions and adverbials

8.1 Dative and instrumental

8.1.1 Dative bay ('for', 'at the place of')

This postposition appears in two tonal variants, báy and bày, carrying over the final tone from the preceding morpheme.

- - b. kě:rè [á:mádù bày] ńdî-só-ý money [Amadou **Dat**] give-Perf1b-1SgS 'I gave the money to Amadou.'
 - c. [[ú bâ:] bày] dámâ [[2SgP father] **Dat**] speak.Imprt 'Speak-2Sg to your father!'
 - d. nî: [lò:sî báy] ndî-y water [visitor **Dat**] give.Perf.L-1SgS 'I gave some water to a visitor.'

With 'give' and 'show', datives are optionally replaced by accusatives, but 'say' requires that the addressee (if specified overtly) be in dative form.

This postposition can also be used like French *chez* X 'at the place/house of X'.

Pronominal datives are repeated in (xx3) from §4.3.xxx. Except for the first singular, the forms are transparent. 1Sg bàrⁿi arguably begins with a variant of

bay followed by a 1Sg element (reversing the usual order), but the morphology here is messy.

(xx3) Pronominal Datives

	category	dative
a.	1Sg 1Pl	bàr ⁿ î î: bày
b.	2Sg 2Pl	ú báy û: bày
c.	3Sg 3Pl	ńné báy bû: bày
d.	InanSg InanPl	kú báy kû: bày
e.	LogoSg LogoPl	á báy â: bày

Further examples of pronominal datives are in (xx4).

- (xx4) a. bàrⁿî njènà:rⁿî dàmà-rⁿî- Ø

 1SgDat nothing speak-Perf.Neg-3SgS

 'He/She said nothing to me.'
 - b. tégî [ú báy] dámá-m-∅
 truth [2Sg Dat] speak-Impf-1SgS
 'I will tell you-Sg the truth.'

8.1.2 Instrumental yànà (ỳnà)

The basic instrumental postposition is low-toned yànà, optionally elided after a vowel to ỳnà. It is distinct from /yana/ (atonal, hence either yáná or yànà) 'also, even', (§19.1.3).

(xx1) a. nàmâ [pòrî yàŋà] késé-só-∅ meat [knife with] cut-Perf2-3SgS 'He/She cut the meat with a knife.'

- b. [nè:mî yànà] nènî néngîrè-m-è [salt with] sauce cook.sauce-Impf-3PIS 'They cook the sauce with salt.'
- c. [kúrⁿô yàŋà] ńné-ŋ sùyð-y [stone with] 3Sg-Acc hit.Perf.L-1SgS 'I hit him/her with a stone.'

8.2 Locational postpositions

8.2.1 Locative, allative, and ablative functions

Simple locative adverbials, including locative PPs, can be translated as (stationary) locatives, as allatives, or as ablatives ('in/at X', 'to X', 'from X'). Directional senses (allative and ablative) are expressed by other words, primarily motion verbs. Ablative is expressed by gŏ: 'go out, exit', which can be chained to other verbs. Any other verb of motion or locomotion ('go', 'run') can be used with a locative adverbial or place name to express the allative. For examples see

8.2.2 Simple and complex PPs

The complex PPs are parallel to English 'in front of X' or the like. In Nanga they involve a possessed noun expressing the spatial relation ('over', 'under', 'front', etc.). This possessed spatial noun is then followed by the simple Locative postposition ga (§8.3.3). Since the spatial noun in question is possessed, it has its regular tone before a pronominal possessor, and a basic HL tone contour following a possessor noun, becoming all-L when the possessor noun itself ends in a L- (or F-) tone. The spatial noun may therefore have three distinct tone patterns: lexical tone, HL contour, or all-L contour. In practice, the HL contour has generalized to become the lexical (as well as possessed) tone.

8.2.3 'In, on' (ga, ηa , go, ηo , go, ηo)

The basic locative postposition has allomorphs {ga ŋa go ŋo go ŋo}. It is used chiefly with inanimate NP complements, but not with toponyms, which instead have a clitic =yè that may reduce to just a tone element (§8.2.4, below). The opposition of a general inanimate category associated with a morpheme KO (K a yelar, O a back or low yowel) to a toponym category associated with a

morpheme YE can be connected with Najamba nominal morphology, where a final E vocalism $\{e \ \epsilon\}$ is associated with nouns denoting topographic features (and liquids), while the more common final O vocalism $\{o \ o \ a\}$ is found with other inanimate nouns.

The NP preceding Locative {ga ŋa go ŋo go ŋo} has its regular tones (with exceptions noted below), and the **final tone spreads** to the postposition. There is a partial phonological basis for the allomorphs, but some nouns have unpredictable allomorph choices and there is some variation in my data. The following is a summary of what seem to me to be the basic patterns.

Segmentally, one common allomorph of the postposition is ga. It is realized tonally as either gà or gá depending on the final tone of the noun (xx1). The stems with this allomorph generally have vowels from the set $\{a \ \epsilon \ i\}$ and do not have a final nasal syllable.

(xx1)	gloss	noun	'in a(n) X'	
	'shop'	bìtîkî	bìtîkî gà	(syncopated: [bifikgà])
	'pond; shoe'	tàgá	tàgá gá	
	'cousinhood'	màgî	màgî gá	
	'courtyard'	dámbî	dámbî gá	
	'market'	έwέ	éwé gá	
	'afternoon'	dèndè-sî	dèndè-sî gà	
	'leaves'	úwâ	úwâ gà	
	'goat(s)'	bérî	bêr gà	

The /g/ of the postposition is **nasalized** to /ŋ/ after a nasalized vowel or after a syllable of the shape NV with a nasal (or nasalized sonorant) followed by an oral vowel. A nasal earlier in the word does not have this effect; see cousinhood' in (xx1), above. Examples showing the postposition in the form /ŋa/ are in (xx2).

(xx2)	gloss	noun	'in a(n) X'
	'shed'	tă: ⁿ	tǎ: ⁿ ŋá
	'man(-hood)'	ár ⁿ â	ár ⁿ â ŋà
	'word'	dàmá	dàmá ŋá
	'chaff'	òy ⁿ ó	òy ⁿ ó ŋó
	'tree'	tùmá	tùmá ŋá
	'beam'	gâ: ⁿ	gâ: ⁿ ŋà
	'cow'	nàŋá	nàŋá ŋá
	'meal'	nă:	ກă: ŋá
	'water'	nî:	nî: ŋś

The vowel of the postposition is often **rounded** to /o/ or /o/. In many cases this is by **assimilation** to the final vowel of the preceding NP. If the final vowel is /o/, the postposition has /o/ (xx3.a). If the final vowel is /o/ or /u/ (the latter is rare at the end of nonmonosylalbic noun stems), the postposition has /o/ (xx3.b). Stems ending in a sequence like ...oCi and ...oCi may appear as ...oCu and ...oCu, respectively, before the postposition, and the /o/ or /o/ may carry over into the postposition. My assistant inconsistently distinguished the /no/ and /no/ variants, tending to pronounce /no/ even in vocalic environments where /no/ is expected. Some but not all nouns containing a nasal require /o/ in the postposition even without a rounded vowel in the noun itself (xx3.e). Stems with /e/ also favor /o/ in the postposition (xx3.f).

(xx3)	gloss	noun	'in a(n) X'
	a. after /ɔ/ 'immature pod' 'wide pond' 'mountain'	gǒ:njô tàgà òwó tórô	gǒ:njô gò tàgà òwó gó tórô gò
	b. after /o/ or /u/	ò:-sóró ŏ: gálô dómór ⁿ ó kû:	ò:-sóró gó ò: gó gálô gò dómór ⁿ ó ŋó kú: gó
	c. after oi 'toilet area'	sùgò-gòmî	sùgò-gòmú ŋɔ́
	d. after oi or ui 'road' 'mountain' 'rear' 'skin' 'forest'	ósî tòró-kúndî tùndî gùsî úndî	ósú gó tòrò-kúndú gó túndú gó gùsú gó úndù gò
	e. 'meat' 'speech' 'milk' 'water'	nàmâ dàmá émê nî:	nàmâ ŋò dàmá ŋó émê ŋò nî: ŋó

```
f. 'speech' tégî tégî gò (syncopated [têg:ò])
'belly' bèndé bèndé gó
'eye' gìré giré giré gó
'sauce' nènî nènî nó (syncopated [něn:ó])
```

Combined with Nasalization-Spreading (see above), the rounding of the postposition vowel accounts for the allomorphs go, no, go, and no.

In several high-frequency combinations, a final low or falling tone on the noun **shifts to H-tone** before the postposition (which then itself acquires the H-tone by spreading). In the case of go: 'fire', the same process is at work, but since the input stem is a <LHL> monosyllable the output is <LH> (i.e. rising tone).

(xx4)	gloss	noun	'in a(n) X'
	a. final <hl> to H</hl>		
	'house'	ńdô	ńdó gó
	'road'	ósî	ósú gó
	'farm work'	wórî	wór gó
	'grass'	sàwâ	sàwá gá
	'head'	kû:	kú: gó
	b. final <lhl> to <i< td=""><td>LH></td><td></td></i<></lhl>	LH>	
	'fire'	gõ:	gŏ: gó

The shift of the final low-tone element in (xx4) is not a general phonological rule; several examples given above preserve a final low or falling tone before the postposition, which therefore itself has low tone.

The common expression 'in/to the field(s)' is ò: gó, with **low-toned** version of ŏ: 'field(s)'. The other example I know of is isè gó 'in/to (the) village' from isê 'village'. Again, there is no general phonological rule converting a risingtoned monosyllabic to a low-toned one, compare ĕ: gá 'in/to the well'.

Locative ga fuses with a preceding Definite kú as gá, presumably contracted from syncopated *k gá. Definite kú induces tone-dropping on the noun, and tone-dropping is observed in the fused forms, which are therefore always distinguishable from the indefinite locatives illustrated above. gá is invariant in form in this construction (there are no other allomorphs such as $/\eta$ á/ or /gó/).

(xx5)	gloss	noun	'the X'	'in the X'
	'hide bag'	nàkòmbó	nàkòmbò kú	nàkòmbò gá

'calabash'	kòsî	kòsù kú	kòsì gá
'water'	nî:	nì: kú	nì: gá
'shop'	bitîki	bìtìk kú	bìtìk gá
'house'	ńdô	ndò kú	ndò gá
'dense forest'	únďi	ùndù kú	ùndi gá
'river'	nì:-bá:	nì:-bà: kú	nì:-bà: gá
'field'	ŏ:	ò: kú	ò: gá
'seeds'	tŏ:	tò: kú	tò: gá
'shed'	tă: ⁿ	tà:n kú	tà: ⁿ gá

The Inanimate Plural Definite is \circ (with tone-dropping on the noun), see. The Locative postposition takes L-toned form \circ after this, regardless of the vocalism of the noun.

(xx6)	gloss	noun	'the X's'	'in the X's'
	'stick'	béré	bèrè ý	bèrè ý gò
	'house'	ńdô	ndò ý	ndò ý gò
	'field'	ŏ:	ò: ý	ò: ý gò
	'market'	έwέ	èwè ý	èwè ý gò
	'writing	tòŋś	tòŋò ý	tàŋà ý gò
	'damage'	námâ	nàmà ý	nàmà ý gò

A similar fusion occurs with the Inanimate Singular **demonstrative pronoun** ngú, which has locative form ngá. As with the fusion of Definite kú and the Locative into gá, the fused form ngá requires tone-dropping on the preceding noun: ndò ngá 'in this house'. The plural is ndò yĕy gò 'in these houses'.

Without a preceding noun, $\eta g a$ functions as the regular 'here' demonstrative adverb ($\S 4.xxx$).

Animate nouns are not commonly followed by the Locative postposition, but the combination is attested. Most often it involves a plural noun, in partitive or spatial sense. The combinations are phonologically regular with no fusion. My assistant pronounced them as follows: Definite Animate Singular $n\epsilon$ $n\delta$, Definite Animate Plural $b\hat{u}$: $g\delta$ (xx7), Animate Singular demonstrative $v\delta$, Animate Plural demonstrative $v\delta$.

```
(xx7) [[àr<sup>n</sup>à bû:] gò], [nù tùmâ sǎy] ésû [[man.L Def.AnPl] in] [person.L one only] be.good 'In (= among) those men, only one is any good.'
```

In the examples given in this section so far, the postposition can usually be glossed with 'in', since the reference entity is (potentially) a container or a zone with an interior. The postposition can also mean 'on X' where X is an entity with an upper or lateral surface ('rock', 'wall'). Examples (with the definite form of the postposition) are in (xx8).

(xx8)	gloss	noun	'on the X'
	'rocky area' 'rock'	pàpàgìr í kúr ⁿ ô	pàpàgiri gá kùr ⁿ ò gá
	'wall'	ndò-tùndî	ndò-tùndì gá

The 'on X' use of the postposition competes with fuller expressions of the type 'on the head of X' described below ($\S 8.xxx$).

The locative, or its Definite form $g\acute{a}$, may also be used with NPs denoting periods of time, such as seasons (xx9).

```
    (xx9) gloss noun 'in X' 'in the X'
    'rainy season' gèr<sup>n</sup>é gèr<sup>n</sup>é ηό gèr<sup>n</sup>è gá 'hot season' ùsiyè-bár<sup>n</sup>â ùsiye-bár<sup>n</sup>à nà ùsiyè-bàr<sup>n</sup>à gá
```

However, time-of-day expressions such as 'at night' are expressed simply as nouns ('night'). These nouns are interpreted adverbially unless the syntax points in another direction.

```
(xxx) dè:ndê bîré-m-iy∴
night work-Impf-1PlS
'We work at night.'
```

The precise gloss of the Locative in context depends not only on the 'in/on' distinction (container versus surface), but also on the verb. As in the other languages of the zone, verbs rather than postpositions distinguish (static) locative from allative and ablative senses.

```
\begin{array}{cccc} (xxx) & a. & [\grave{n}d\grave{o} & g\acute{a}] & n\acute{n}y\acute{e}-\grave{m}- \not \bigcirc \\ & & [house.L & Def.InanSg.\textbf{Loc}] & sleep-Impf-1SgS \\ & & `I & will sleep in the house.' (\acute{n}d\^{o}) \end{array}
```

```
b. [éwé gá] ńnî-m-∅

[market Loc] go-Impf-1Sg

'I am going to the market.'
```

```
c. [ósú gó] gô:

[road Loc] go.out.Imprt

'Get-2Sg out of the road!'
```

tùyà-gá 'bunch (unit of items for sale)' is a case where an original locative PP has become frozen into a noun. It can be followed by a numeral or other modifier: tùyà-gá wǒy 'two bunches', tùyà-gà òwó 'large bunch'. The composite origin of tùyà-gá is suggested by comparison with verb túy 'put down'.

8.2.4 Locative clitic =yè (or final tone change) with place names

There are no Jamsay-style tonal locatives (locative forms of nouns marked only by a tonal change). However, there is a locative (hence also allative and ablative) form of **place names** that is expressed by a final clitic, which in some cases **reduces to a tonal change**. The forms of the clitic are in (xx1).

```
(xx1) form context

= y \grave{e} \qquad \qquad \text{default}
= y^n \grave{e} \qquad \qquad \text{after nasalized syllable}
(:`) = \varnothing \text{ (final L-tone)} \qquad \text{after some vowels}
```

The default allomorph, and the one that I take to be phonologically basic, is =yè. It is required after /i/ in a non-nasal syllable, and it occurs after other vowels in some place names (xx2.a). If the preceding syllable is nasalized, the variant =y°è is obligatory (xx2.b). After long (i.e. quadrisyllabic) place names, =yè tends to contract with a stem-final vowel, resulting in either a long or a short vowel with falling tone, although the full pronunciation is also possible (xx2.c).

(xx2)		location	Nanga term	locational form
	a.	Mopti Wakara Perge Pergesa	mó:ti wàgári pèrgé pègèsá	mó:tî≡yè wàgárî≡yè pèrgé≡yè pègèsá≡yè
	b.	Douentza Boni	dúwánsár ⁿ î bó:nî	dúwánsár ⁿ î=y ⁿ ὲ bó:nî=y ⁿ ὲ

	Ben Tey	bě:r ^m i	bě:r ⁿ ì≡y ⁿ è
	Soro-ni	sórònî	sórònî≡y ⁿ È
c.	Namakoro	námbákòré	námbákòr=ê: ~ námbákòré=yè
	Bandiagara	bànàgàrá	bànàgàr≡ê: ~ bànàgàré≡yè
	Bamako	bàmàkó	bàmàkô:≡∅ ~ bàmàkó≡yè
	Dianwely	jáw ⁿ lé	jáw ⁿ lê≡∅ ~ jáw ⁿ lê:≡∅
	Kono	kό:r ⁿ έ	kó:r ⁿ ê≡Ø
	Anda	á:ndé	á:nd̂ε≡∅

This clitic is used with place names as locations, destinations, or points of departure.

(xx3) a. bàmàkó≡yè nnè-tà:-rî-y
Bamako≡in go-ExpPf-Perf.Neg-1SgS
'I have never gone to Bamako.'

b. bàmàkó≡yè tùwè-∅ Bamako≡in die-3SgS 'I have never gone to Bamako.'

c. bàmàk5≡yè gò-y Bamako≡in go.out-1SgS 'I left Bamako.'

8.2.5 'On (the head of) X' ([X kû:] gò)

The concept 'on X' is expressed as $[[X \, k\hat{u}:] \, g\hat{a}]$ or $[[X \, k\hat{u}:] \, g\hat{a}]$, depending on whether the NP (X) ends in a high or low tone.. It is based on possessed forms of $k\hat{u}$: 'head' followed by the basic Locative postposition /ga/. In some cases, when X is a human or animal, the literal sense 'on the head of X' may be valid.

(xxx)	gloss	noun	'on X'
	'stool'	túŋgúrî	[túŋgúri kû:] gò
	'box	góŋgó	[góŋgó kû:] gò
	'cart'	wògòtórô	[wògòtórô kù:] gò

With a pronoun, we get the usual forms for pronominal possessor, after the noun in its regular tonal form (xxx).

```
(xxx) a. kû: kô:
head 1SgP.Poss.InanSg
'my head'
```

```
b. kû: kɔ̃: gɔ̀
head 1SgP.Poss.InanSg in
'on me' (or: 'on my head')
```

8.2.6 'Next to, beside X' ([X kéri] gà)

This concept is expressed using a possessed form of kérî 'side' followed by Locative ga. When X is a nonpronominal NP, the combination is heard as [[X kéri] gà] or [[X kéri] gà], depending on whether X ends in a high or low tone. The Locative postposition is optional after a pronominal possessor (xx1.b).

```
(xx1) a. [[ndò gú] kéri] gà
[[house.L Def.InanSg] side.HL] in
'beside the house'

b. [kéri kɔ:] (gò)
[side 1SgP.Poss.InanSg] (in)
'beside me'
```

This complex postposition specifies that the topical entity is next to the reference point, as with two persons sitting right next to each other. For a looser spatial connection, as when the two persons are in the same setting but not directly next to each other, [[X dósù] gà] 'under X' can be used.

8.2.7 'In front of' ([X gîrè] gà)

With nonpronominal complement, 'in front of X' is expressed by a possessed form of girê 'front' (cf. giré 'eye'), plus Locative postposition ga. When X is a nonpronominal NP, the result is [[X girè] gà] or [[X girè] gà] depending on the final tone of X. The final /gà/ is optional after a pronominal possessor, which follows girê in its lexical form (xx1.b).

```
b. [girê kɔ̃:] (gà)
[front 1SgP.Poss.InanSg] (in)
'in front of me'
```

c. [ndô girè] gà
[house front.L] in
'in front of (a/the) house'

The corresponding adverb, with no overt reference NP, is girê gò 'ahead, in front'.

8.2.8 'Behind/after X' ([X túndù] gò), 'about'

'About (=concerning) X' is expressed with X as possessor of a verbal noun (or similar nominal), when the verb is 'speak'. With other verbs, we get [[X túndù] gò] 'after X', cf. noun tùndî 'back, rear' and adverbial PP [tùndú gó] 'afterwards'.

```
(xxx) a. [émê dàmà] dámá-m-ĭ∴

[milk talk.L] speak-Impf-1PIS

'We will speak about milk.' (lit. "... speak some talk of milk")
```

```
b. [[émê tùndù] gò] mă:ndi=bé-y
[[milk rear.L in] think.Perf=Past-1SgS
'I thought about milk.'
```

```
c. [[émê tùndù] gò] nùŋú≡b-á

[[milk rear.L] in] sing≡Past-3PlS

'They sang about milk.'
```

8.2.9 'Over X' ([X témbè] gà), 'under X' ([X dósù] gò)

There are complex postpositions with témbè 'top' and dósî 'bottom, underneath', in possessed form with following Locative postposition (gà, gò). The forms with preposed X are [[X témbè] gà] or [[X tèmbè] gà], and [[X dósù] gò] or [[X dòsù] gò], with the tone depending on whether X ends in a high or low tone. As in other such complex postpositions, the Locative postposition is optionally omitted after a pronominal possessor.

```
(xx1) a. [ńdô tèmbè gà] bù-∅
```

[house **over.L** in] be-3SgS 'It is over the house.'

- b. témbè ú gô (gò)
 over 2SgP Poss.InanSg (in)
 'over you-Sg'
- a. [ndo gú] dósù gò [house.L Def] under.HL in 'under the house'
- b. [tùmá dósù gò] bìyò-∅
 [tree under.HL in] lie.down.Stat-3SgS
 'He/She is lying down under a tree'
- c. mð:rⁿô [[dósù kồ:] (gô)] bèr-à wild.date [[**under.HL** 1SgP] (in)] find.Perf.L-3PlS 'They found some wild dates under me.'

Without an explicit reference NP, we get pure adverbials témbe gà 'up above, overhead' and dósú gó 'down below, at the bottom, underneath'.

'Under X' can also be stretched to mean 'beside, in the vicinity of (someone)'.

8.2.10 'Between' ([X bérè-kèndè] gá)

'Between' or 'among/amidst' is expressed by a complex postposition containing the noun bèrè-kéndè 'middle'. The postposition, with noun X, appears as [[X bérè-kèndè] gà] or [[X bèrè-kèndè] gà] depending on whether X ends in a high or low tone. The first of these (i.e. after a final high tone) can also be pronounced [[X bérékéndè] gà] with the break in the {HL} tone contour delayed until the beginning of the final syllable of /berekendé/. The difference between [[X bérè-kèndè] gà] and [[X bérékéndè] gà] is attributable to the ambiguous status of /bérékéndè/ as compound béré-kéndè or as an uncompounded quadrisyllabic noun, since the application of {HL} tone contours is sensitive to this distinction.

This complex postposition may take either a simple NP (with plural reference) or a conjunction as its complement.

'between/among us'

```
b. [[[ú yò] [ǐ:<sup>n</sup> yò]] bèrè-kèndè] gà

[[[2Sg and] [1Sg and]] middle.L] in

'between you-Sg and me'
```

- c. wògòtórô [[[[á:ndɛ yò] [dúwánsárⁿî yò]] donkey.cart [[Anda and Douentza and] bèrè-kèndè] gà] nàmè-Ø middle.L in] be.ruined.Perf.L-3SgS
- d. [isè kùrè ý] bérè-kèndè] gà [village.L six.L Def.InanPl] middle.HL] in 'between/among the six villages'

8.2.11 'From X to Y'

A complete locational 'from X to Y' expression requires two clauses, the first including gŏ: 'go out, leave' to convey an ablative sense. bă: 'since, all the way from' may be added (xx1).

```
[[dúwánsár<sup>n</sup>i≡y<sup>n</sup>ê
(xxx)
                                bă:]
                                                      gŏ:
                                                                 ή]
        [[D=Loc
                                all.the.way.from
                                                      go.out
                                                                 and.SS]
        [lòsó
                              mó:tî≡vè
                                                 ć-nń
                    gá]
        [foot
                    Loc
                              M≡Loc
                                                 go.Perf.L-3PIS
         'They walked on foot from Douentza to Mopti.'
```

If the distance to the endpoint is emphasized, hálì 'all the way to' can be used; cf. §19.2.1.

8.3 Purposive-Causal dèrⁿí

This postposition has invariant segmental and tonal form; it does not behave like a possessed noun or have any other phonological interaction with the preceding complement NP.

The postposition occurs in a range of purposive and causal senses. In (xx1.a-b), the purposive sense is prospective, while the causal sense in (xx1.c) and arguably that in the high-frequency phrase in (xx1.d) is retrospective.

- - b. [kě:rê dèrⁿi] núŋi-ŋ̀ [money **for**] sing-Impf.3SgS 'He/She sings for money.'
 - c. [[bònđi dùgù gú] dèrⁿi] yòg-ò [[rain.L big.L Def.InanSg for] flee.Perf.L-3PlS 'They fled because of the great (= heavy) rain.'
 - d. [nù bû:] [jěnjê dèrⁿî] njî bà:r-à
 [person.L Def.AnPl] [God **for**] 1SgO help.Perf.L-3PlS
 'The people helped me on account of (=for the sake of) God.'

8.4 Other adverbials (or equivalents)

8.4.1 Similarity (mayⁿ 'like')

The particle may^n 'like' follows the argument that is has scope over (xx1). The final tone of the preceding word spreads into the particle.

- $\begin{array}{cccc} (xx1) & a. & [\acute{n}n\acute{e} & m\acute{a}y^n] & b\grave{u}\text{-}y \\ & [3Sg & like] & be\text{-}1SgS \\ & `I \text{ am like him/her.'} \end{array}$
 - b. [yǎ-ŋ máyⁿ] dámá-m̀-w [woman-Sg like] speak-Impf-2SgS 'You-Sg talk like a woman.'
 - c. [árⁿâ màyⁿ] bîrέ-m-Ø
 [man like] work-Impf-1Sg
 'I work like a man.'

In the common phrase kú màyⁿ 'thus, like that (inanimate)', which is also often used in narrative ('that being the situation, ...'), the tone of màyⁿ is unexpectedly low. Contrast the regular high tone in e.g. ngú máyⁿ 'like this'.

8.4.2 Extent ('a lot', 'a little')

The adverb <code>\exi(si) => 'a lot'</code> can be used in a wide range of senses, ranging from **quantity** ('a lot' in the sense 'a large amount') to **frequency or intensity** of an event type ('a lot' as synonym of 'greatly' or 'frequently'). It is normally preverbal, but its position with respect to e.g. object nouns is variable, and it does not behave like an adjective (for example, it does not force tone-dropping on a preceding noun).

```
(xx1) a. birá Èsî⇒ biré-ŋ̀
work(noun) a.lot work-Impf.3Sg
'He/She works a lot.'
```

- b. èsî⇒ bîrá bîré-ŋ̀ [= (a)]
- c. èsí isê-ŋ nd-à
 a.lot village-Acc give.Perf.L-3PlS
 'They gave a lot (=a large amount) to the village.'

In the sense of 'large amount' ('many', 'much'), other devices are also available. As predicate, there are several options (xx2).

- (xx2) a. $\delta: nd\delta$ $\delta: si \Rightarrow$ $bu-\emptyset$ honey a.lot be-3Sg 'There is a lot of honey.'
 - b. ò:ndó jŏ-èrè-∅ honey be.much-Perf1a-3SgS "There is a lot of honey.' (lit. "Honey has become abundant.")
 - c. ò:ndó sámá-èrè-Ø honey be.common-Perf1a-3SgS 'There is a lot of honey.' (lit. "Honey has become common.")
 - d. ò:ndó lógó-èrè-∅ honey be.excessive-Perf1a-3SgS

'There is a whole lot of honey.' (or: 'There is too much honey')

e. ò:ndó dăyⁿ sò-ndó-∅ honey limit(n) have-Neg-3SgS "There is a whole lot of honey." (lit. "The honey has no end.")

(xx2.a) is semantically unremarkable, merely stating a current abundance. (xx2.b-c) imply that the element in question was less common previously. jŏ: 'become much/many' is otherwise semantically neutral, while sámá- 'become common' implies a loss of value due to abundance. (xx2.d-e) are more emphatic in nature; (xx2.d) tends to be pejorative.

(xx3) illustrates modifying function. In (xx3.a), $\grave{\epsilon}s1\Rightarrow$ is still an adverb syntactically, and has no tonal effect on 'sheep'. In (xx3.b), $\flat\acute{o}$: is a regular modifying adjective, and forces 'sheep' to drop tones. There is no significant difference in meaning.

```
(xx3) a. pèrgé èsî⇒ tùw-à
sheep a.lot die.Perf.L-3PIS
'A lot of sheep died.'
```

```
b. [pèrgè jó:] tùw-à [sheep.L a.lot] die.Perf.L-3PIS 'Many sheep died.'
```

Antonyms ('a little' or 'slightly') are expressed in either adverbial or nominal function by the adverbials dágáy (variant dákáy) and démi, and in nominal (but not adverbial) function also by the adjective èwré 'small' used as a noun. démi is best glossed 'somewhat', and suggests that the amount is adequate though not especially big, while dágáy is more emphatic, may be glossed 'slightly' when adverbial, and is more likely than démi to occur with say 'only'.

- (xxx) a. á:mádù dágáy (sǎy) ǹd-à
 A a.little (only) give.Perf.L-3PlS
 'They gave (only) a little to Amadou.'
 - b. á:mádù démì⇒ nd-à
 A somewhat give.Perf.L-3PIS
 'They gave a little to Amadou.' (suggests adequacy)
 - c. á:mádù èwré nd-à A small give.Perf.L-3PIS

```
[=(a)]
```

- d. dágáy ĭró-∅
 a.little be.better-3SgS
 'It (e.g. illness) is a little (= slightly) better.'
 (i.e. 'I'm feeling better')
- e. démi⇒ iró-∅ somewhat be.better-3SgS 'It (e.g. illness) is a somewhat better.'
- 8.4.3 Specificity
- 8.4.3.1 'Approximately'

bă: can be used to indicate approximateness of a numerical value.

```
(xxx) [pèrgé pé-nimi:<sup>n</sup> bă:] màrá-s-é
[sheep ten-five about] be.lost-Perf2-3PlS
'Around fifty sheep were lost (=died).'
```

An alternative is to use mayⁿ 'like', as in 'I will buy like (= somewhere around) fifty sheep'.

```
8.4.3.2 'Exactly' (lék, cók, já:dǐ, té=>)
```

lék is an interjection-like intensifier that can be used with numerals (xx1).

```
(xxx) [pèrgé pé-nimi: lék] màrá-s-é [sheep ten-five exactly] be.lost-Perf2-3PlS 'Exactly fifty sheep were lost (=died).'
```

cók is another interjection-like particle used to indicate 'exactly identical' (on some measure, usually height). It can also be used to indicate that all members of a group are present.

To confirm the truth or correctness of what an interlocutor has just said, the particle já:dì 'exactly' (i.e. 'you're exactly right') is used.

Adverbial té=> can be used with time expressions: midî té=> 'at noon sharp'. Its reduplication té:-té: is also in use in similar contexts. See also the following subsection.

8.4.3.3 'Specifically' (té=>)

Adverbial particle té=> is also used in the sense 'specifically'.

```
(xx1) ú-ỳ té=> ndĩ-∅
2Sg-Acc specifically give.Perf.L-3SgS
'He/She gave (it) specifically to you-Sg.'
```

8.4.4 Evaluation

8.4.4.1 'Well' and 'badly'

The preferred construction equivalent to English evaluative manner adverbials is one with a complement NP (often containing a cognate nominal or similar noun) plus an adjective.

```
(xx1) a. [bǐrà Èsî] bîré-m

[work(noun).L good] work-Impf-1SgS

'I do good work.' (= 'I work well.')
```

```
b. [birà mòsi] biré-m

[work(noun).L bad] work-Impf-1SgS

'I do bad work.' (= 'I work badly.')
```

8.4.4.2 'Proper, right'

The adverbial jâ:", generally predicative with following 'it is' clitic, characterizes an action or behavior pattern as normal or proper (following social norms), or as contextually appropriate. This adverbial is also found in Jamsay.

```
(xx1) a. jâ:<sup>n</sup>≡∅
proper≡it.is
'It's proper (right, normal, appropriate).'
```

```
b. [kàr<sup>n</sup>à ú kàr<sup>n</sup>à-mù gú] jà:<sup>n</sup>=ndó-∅
[doing.L 2SgS do-Ppl.Impf Def] proper=not.be-3SgS
'What you are doing is not right.'
```

8.4.5 Manner

The equivalents of simple English manner adverbials (-ly) can be constructed using NP complements (with an adjective) as in (xx1.a). Or an adverbial PP may be used if there is a suitable noun available, like paná 'strength, force' in (xx1.b).

```
(xx1) a.
           [bìrà
                                       bîrέ-m̀
                               ógù]
           [work(noun).L
                               fast]
                                       work-Impf-1SgS
            'I do fast work.' (= 'I work fast.')
        b. [pàŋá
                       ηá]
                                 ìί
                                           dàmbi-Ø
           [strength
                       in]
                                 1SgO
                                           push.Perf-3SgS
            'He/She pushed me hard (with force).'
```

The noun meaning 'manner' is dăyⁿ. For manner adverbials, see §xxx.

8.4.6 Spatiotemporal adverbials

8.4.6.1 Temporal adverbs

Some of the major temporal adverbs are in (xx1). Note the tonal distinction between 1yé 'today' and 1yê 'again' (xx1.a). niŋèyn 'now' and £ndɛy 'day after tomorrow' were heard with HL tone pattern.

```
'today; nowadays'
(xx1) a. iyé
                                         'again'
            îyê
                                         'yesterday; formerly, in the old days'
            yéŋir"i:
                                         'day before yesterday'
            îyé tà:ndì:
            nînèyn
                                         'now'
        b. έ:ηî
                                         'tomorrow; in the future'
            έndèy
                                         'day after tomorrow'
                                         'second day after tomorrow' (third from
            èndèy túndèy
                                         'third day after tomorrow' (fourth from
            lég tàrà
                                        today)
                                        'fourth day after tomorrow' (fifth from
            júgú
                                        today)
                                        'fifth day after tomorrow' (sixth from
            júgú-jàgú
                                        tomorrow)
        c. gšỳ<sup>n</sup>
                                        'last year'
```

```
àr<sup>n</sup>à kámâ 'next year'
núy<sup>n</sup>áy<sup>n</sup> 'this year'
```

8.4.6.2 'First' (kǐyá)

Adverbial 'first' ('firstly', 'at first') is expressed by kiyá, which is identical to the ordinal adjective 'first' (as in English).

(xx1) bîrá kiyá bìrè gáy, nă: k5:-m-i: work(noun) first work.L and.then.SS, meal eat.meal-Impf-1PlS 'We'll work first, then we'll eat.'

8.4.6.3 Spatial adverbs

```
(xx1) a. témbè
                                      'above, top, summit'
                                      'below, bottom, down'
            dósî
        b. ùsì-[túm-nó], dû: gìrè
                                       'east'
                                       'west'
            ùsì-[yégî-nó]
                                      'south'
            tèŋi-dágá
            dù:-dágá
                                       'north'
                                      'going backward, in reverse'
        c. tùn-túndi⇒
            tùndú gó
                                      'in the rear'
                                      'forward; in front' cf. giré 'eye'
            gîré gó
```

ùsì-[túm-nó] 'east' and ùsì-[yégî-nó] 'west' are compounds containing ùsî 'sun' and nominals based on the verbs túmbó '(sun) rise' and yègé 'fall; (sun) set', respectively. The term for 'south' is based on the location of the Tengou (southern Dogon group).

'Left hand' is nà: bàriyé, while 'right hand' is nà: nàr"iyné (with nă: 'hand').

8.4.7 Adverbials

Like the other Dogon languages, Nanga has many **adverbials**, which I define syntactically as forms that cannot be directly inflected but that can be made predicative by adding an auxiliary. They are generally expressive, i.e. semantically colorful, and they are similar to what are often referred to (in other

languages) as mimetics or ideophones. These latter terms seem inadequate for various reasons.

Some adverbials have adjective-like senses, but they differ morphosyntactically from adjectives. Adverbials do not form parts of NPs and (therefore) have no tonal interactions with nouns or other NP words. Like adjectives, they can also function as predicates, but the form of the predication is different. Their phonological form is usually distinctive, with full-stem iteration or final intonational prolongation (but usually not both).

In positive predicates, the adverbial is followed by either quasi-verb bù- 'be (somewhere)' for stative and durative senses. The negative form of bù- is ngó- 'not be (somewhere), be absent'. Thus jen= bù- \emptyset 'it is tilting', jen= ngó- 'it is not tilting'. Adverbials with more active senses are made predicative by adding regular transitive verb ngó- 'do', which has a fuller range of aspect-negation forms. Adverbials like ngó- 'it flapping' can take either: ngó- 'it flapped in the wind (perhaps briefly)', ngó- 'it is flapping in the wind'.

Some expressive adverbials, like yàl-yàl, are **entirely low-toned** lexically. There are no lexically {L}-toned nouns, adjectives, numerals, or verbs.

Many adjectives can be made into adverbials, with little difference in sense, by adding a final $-\hat{i} = >$. Thus èsî 'good', adjectival predicate èsú 'it is good', adverbial ès $-\hat{i} = >$, adverbial predicate ès $-\hat{i} = >$ bù $-\emptyset$.

8.4.7.1 Expressive adverbials

Examples of expressive adverbials without stem iteration are in (xx2). Only a handful, such as those in (xx2.c), are parts of larger word families.

```
(xx2)
             form
                                                           related form
                                 gloss
        a. high-toned, final prolongation
           {H}-toned
                                 'gaping (hole)'
             pó⇒
             pá<sup>n</sup>⇒
                                 'wide open (doorway)'
             kárás⇒
                                 'face to face'
           {LH}-toned
             kǎy<sup>n</sup>⇒
                                 'wide open (eyes)'
                                                           jènî 'tilt'
             jèηî⇒
                                 'tilting'
                                 'with head tilting'
             màmí⇒
                                 'ajar (door)'
             èndiri⇒
                                 'rickety, shaky, poorly encased'
             gògìrí⇒
           \{L\}-toned
             lòm⇒
                                 'froth forming'
```

```
'towering, lofty'
    dim⇒
                      'straight-nosed'
    sèm⇒
    yòw⇒
                      'slightly open (mouth)'
                      'slightly open (eyes)'
    yèw⇒
                      'fat (woman, cow)'
    taru = >
                      'motionless'
    gènjèy⇒
b. no final prolongation
  {H}-toned
                      '(head) long and bending forward'
    sómógó
  \{L\}-toned
   nàm
                      'brief shower (rain)'
   jùŋày<sup>n</sup>
                      'fat and clumsy'
c. CuCuCôy
    lùgùsôy
                      'chubby, puffy'
                                               lùgùsî (adjective)
    yùgùsôy
                      'woolly, disheveled'
                                               yùgùsî 'velvet'
```

The intonational prolongation applies to the final segment. Therefore in e.g. sem > 'straight-nosed', the m but not the vowel is prolonged.

Examples of iterated adverbials involving "nonsense" stems (i.e. those whose uniterated form do not occur) are in (xx3).

```
related form
(xx3)
              form
                                   gloss
         a. high-toned
             néy<sup>n</sup>-néy<sup>n</sup>
só:<sup>n</sup>-só:<sup>n</sup>
                                   'drizzle (fairly light rain)'
                                   'all together'
             ném-ném
                                   'drizzle (very light rain)'
              yál-yál
                                   'flapping'
                                   'shiny new'
              pár-párú
                                   'suddenly encountering'
              pèré-pàrá
                                   'runty and weak'
              yigisé-yigisé
         b. low-toned
                                   'swaying'
             jà:-jà:
         c. complex tone contour
                                   'going backward'
                                                               tùndú 'rear'
              tùn(đi)-túnđi⇒
              dòndi-dòndi⇒
                                   'almost alongside'
```

8.4.7.2 Adjectival intensifiers

Like the other Dogon languages, Nanga has an abundance of uninflectable adverbials used as intensifiers, primarily for adjectival but also for a few verblike senses. Compare English *brand new*, *blind as a bat*, *X stopped still (in his tracks)*, and the like. In most cases the intensifer has no phonological relationship to the semantically associated word(s). There is no sharp difference between intensifiers and (other) expressive adverbials.

Intensifiers (mostly adjectival in meaning) are in (xx1). For the glosses, supply 'very' or the like for the sense of the intensifier. If there is no associated word that commonly co-occurs with the intensifier, the semantically closest word is given in parentheses. In <code>ese-[téw-téw]</code> and <code>bùde-[téw-téw]</code> (xx1.a), there is (unusually) a {L}-toned compound initial corresponding to the associated word.

Vocalic sound symbolism is seen in the alternation of lèré-lèré 'cleaned up' with lòró-lòró 'clean-shaven head'.

Because of the frequency of collocations (associated word followed by intensifier), it is possible for a single form to serve as intensifier for unrelated senses, see kát-kát in (xx1.a). It is also possible for a basic sense to have multiple intensifiers, normally used by different speakers, see kát-kát, kúsú-kúsú, and kúrúŋ-kúrúŋ for 'black' in (xx1.a).

The data in (xx1) are organized by the phonological form of the intensifier.

intensifier (xx1)gloss associated word(s) a. iteration (no tonal or vocalic change) one iteration $s\mathbf{\hat{3}}^{n}$ - $s\mathbf{\hat{3}}^{n}$ 'newborn' bà-bàr^{ri} 'newborn baby' púl-púl '(brand) new' kándà 'new' bóm-bóm 'stout' dùgi 'big' dîm-dîm 'straight' dém = > 'straight' kèndè bárⁿi 'anger' dóŋ-dóŋ 'furious' sél-sél 'long; tall' gùró 'long' pòró- 'be tight-fitting' gén-gén 'tight-fitting' kúy-kúy 'stocky (person)' démbiré 'stout' kéy-kéy 'hard' mă: 'hard, dry' 'hard, dry' mă: 'hard, dry' káy-káy 'unripe' kè:sî 'unripe' péy-péy kóyⁿ-kóyⁿ 'emaciated' kó:mbi-yé- 'be lean' séyⁿ-séyⁿ kέ:mbέ 'slender' 'slender (person)' gáyⁿ-gáyⁿ 'tight; crowded' č: 'tight' jáyⁿ-jáyⁿ 'uncooked (red)' bárⁿi 'red' táyⁿ-táyⁿ 'sweet' érî 'sweet'

```
táy<sup>n</sup>-táy<sup>n</sup>
                        'salty'
                                                 párî 'salty'
    dúy<sup>n</sup>-dùy<sup>n</sup>
                                                 bár<sup>n</sup>i 'red'
                        'red'
                                                 sòy 'hot weather'
    táw-táw
                        'hot (weather)'
                                                 ógî 'hot; fast'
                        'hot (object)
    jáw-jáw
                        'fast'
                                                 ógî 'hot; fast'
    lâw-lâw
                        'lightweight'
    jέw-jέw
                                                 ér<sup>m</sup>i 'lightweight'
    èsè-[téw-téw]
                        'unfertilized'
                                                 έsέ- 'be unfertilized'
                                                 ây 'bland'
                        'bland'
    èsè-[téw-téw]
                        'fine (powdery)'
                                                bùté 'fine, powdery'
    bùdè-[téw-téw]
    bùdè-[téw-téw]
                        'supple'
                                                bùté 'supple'
                        'full'
                                                 bá: 'full'
    pép-pép
    ték-ték
                        'standing straight'
                                                i:-yi- 'stand, stop'
                                                 òmbi 'rotten'
                        'rotten'
    kát-kát
    kát-kát
                        'bitter'
                                                 gári 'bitter'
                                                jémi 'black'
                        'black'
    kát-kát
    pár-párú
                        'shiny new'
                                                kándà 'new'
                        'lost'
                                                 màrá- 'be lost'
    pîsî-pîsî
                                                kóngò 'dust'
    sérî-sérî
                        'dusty'
    kúsú-kúsú
                        'black'
                                                jémî 'black'
                                                bèmbi- 'glare (at)'
                        'glaring (at)'
    kúsú-kúsú
    yágá-yágá
                        'soft'
                                                 búrî 'soft'
    búdé-búdé
                        'fine (powdery)'
                                                 bùté 'fine, powdery'
                                                bùté 'supple'
    búdé-búdé
                        'supple'
    púlá-púlá
                        'hot (object)
                                                 ógî 'hot; fast'
                                                gòyó '(the) cold'
                        'cold (weather)'
    péré-péré
    pàsá-pàsá
                        'white'
                                                pîrî 'white'
                        'moon shining'
                                                 wà: pîrî 'moonlight'
    tègé-tègé
    lèré-lèré
                        'cleaned up'
                                                 έsέ- 'be clean'
                        'clean-shaven head'
                                                ká:- 'shave'
    lòró-lòró
                        'black'
                                                jémí 'black'
    kúrúŋ-kúrúŋ
    gérén-gérén
                        'inflated'
                                                pîrîyé- 'be inflated'
    kórógó-kórógó
                        'loose-fitting'
                                                 kórógó 'loose-fitting'
                        'fraying'
                                                 gùsúró- 'fray'
    gùsùró-gùsùró
  more than one iteration
    dón dón dón
                        'pouting'
                                                nž: sómó- 'pout'
b. iteration (with vocalic and/or tonal change)
  tonal change only
    táy-tày
                        'used up'
                                                 dimé- 'be used up'
  tonal and vocalic change (high vowel to a)
    yùgùsì-yágisim
                        'very woolly'
                                                 yúgúsî 'furry, woolly'
    birgi-bárgi
                        'junk (in disorder)'
                                                 ñàmà-ñàmá 'junk'
c. intonational prolongation
```

simple

```
dám = >
                          'blind'
                                                    girè-mbi 'blind'
    dim = >
                          'stout'
                                                    dùgi 'thick'
    p\acute{o}m = >
                                                    dùgi 'big'
                          'enormous'
    s\check{\epsilon}w^n = >
                                                    èwré 'small'
                          'tiny (eye)
    táy^n = >
                                                    sir<sup>n</sup>ε- 'be full, satisfied'
                          'full (eating)'
    k\check{a}y^n = >
                          'oversized (eye)'
                                                    (dùgi 'big')
    k\check{\epsilon}y^n = >
                          'tiny (moon, eye)'
                                                    (èwré 'small')
    k\check{\epsilon}w^n = >
                                                    (èwré 'small')
                          'tiny (moon)'
    pútúm = >
                          'flowery'
                                                    pùr<sup>n</sup>ó 'flower'
    pútúm = >
                          'foggy, hazy'
                                                    súdi 'haze'
    l erew = >
                          'everything'
                                                    kéréw 'everything'
    d\tilde{u}-d\tilde{u}y^n = >
                          'red'
                                                    bár<sup>n</sup>i 'red'
  apparently compound
    késé-kéréy = >
                                                    mă: 'hard, dry'
                           dry'
    lèrè-gèrèw = >
                          'everything'
                                                    kéréw 'everything'
d. final reduplication
                          'well-branched'
                                                    jănmi- 'ramify'
    îsásâ:
    wúsúsú
                          'long'
                                                    gùró 'long'
                          'sweet (abstract)'
    èrélélé
                                                    érî 'sweet'
    dimámámá
                          'stout'
                                                    dùgî 'thick', cf. dîm = >
    pàrálálá
                          'sour'
                                                    párî 'sour'
                          'solid (no holes)'
    màr<sup>n</sup>ánáná
                                                    (dén 'hard, stiff')
    pèsésésé
                          'cold (object)'
                                                    támî 'cold'
                          'point of light'
                                                    (ésê 'light')
    pòsósósó
                          'point of light'
    pàsásásá
                                                    (ésê 'light')
                          'heavy'
    dùsúsúsú
                                                    dúsî 'heavy'
                                                    bár<sup>n</sup>i 'red'
    bùrúndúndú
                          'red'
                                                    òmbi 'rotten'
    gòmómómó
                          'rotten smelling'
e. other
                          'pouring out'
                                                    ti:ri- 'pour'
    mâ:
                                                    bá: 'full'
                          'full'
    tέp
                                                    tùmâ 'one'
                          'sole, only (one)'
    déndè
                                                    tùmâ 'one'
    lék
                          'sole, only (one)'
    kέk
                          'completely'
                                                    (kéréw 'everything')
    kèmìr<sup>n</sup>êy<sup>n</sup>
                          'tiny (eye)
                                                    (èwré 'small')
                          'short'
                                                    dénî 'short'
    kédégéy
```

8.4.7.3 'Straight' ($d\acute{e}m = >$)

 $d\acute{e}m = >$ is the basic adverb for 'straight' in the sense of a direct trajectory (not the absence of crookedness in e.g. a stick). The /m/ is prolonged intonationally.

```
(xx1) mó:ti dém = > ńné-m-i:

Mopti straight go-Impf-1PIS

'We'll go straight (= directly) to Mopti.'
```

Iterated dém-dém can be used in the sense 'straight ahead' without a NP complement (French *tout droit*). As usual in iterations, there is no intonational prolongation.

```
(xx2) dém-dém ńnô
straight go.Imprt
'Go-2Sg straight (ahead)!.'
```

```
8.4.7.4 'Apart, separate' (\text{d\'ey}^n = >)
```

The adverbial $d\acute{e}y^n = >$ is used in parallel constructions of the type 'X is apart, Y is apart' (meaning 'X and Y are separated or distinct').

```
(xx1)
                                           d\acute{e}y^n = >
                                                          b-è,
         [pèrgè
                        bû:]
         [sheep.L
                        Def.AnPl]
                                           apart
                                                          be-3Pl,
                                           d\acute{e}v^n = >
         [b\ear
                        bû:]
                                                          b-è
         [goat.L
                        Def.AnPl]
                                           apart
                                                          be-3Pl
         'The sheep-Pl and the goats are apart (= separated or distinct).'
```

The iterated form $d\acute{e}y^n$ - $d\acute{e}y^n$ occurs in examples where the parallelistic phrasing is absent (xx2).

```
(xx2) [[pèrgé yò] [bérî yò]] déyn-déyn kúrnú-m-iy.:

[[sheep and] [goat and] separated put-Impf-1PIS

'We'll put sheep and goats in separate spots.'
```

```
8.4.7.5 'Always' (àsú⇒), 'never' (à:bádá)
```

The adverbial 'always' is $\grave{a}s\acute{u}\Longrightarrow$ (also found in Ben Tey and Najamba) (xx1.a). It might be etymologically connected with $\grave{u}s\acute{u}$ 'day'. The usual 'never' particle is the regionally widespread $\grave{a}:b\acute{a}d\acute{a}$ (ultimately from Arabic), which occurs in combination with a negated predicate (xx1.b).

```
(xx1) a. àsú⇒ sigórò [íníc báy] éwé-mì-Ø always sugar [3Sg Dat] buy-Impf-1SgS 'I always buy sugar from him (= at his store).'
```

b. à:bádá sîgórð [ńné báy] éwé-ŋð-y never sugar [3Sg Dat] buy-ImpfNeg-1SgS 'I never buy sugar from him (= at his store).'

8.4.7.6 'Exclusively, together' (s5: "-s5:")

The adverb s5:n-s5:n is used, for example, to indicate that a group is seated together (in a bus or concert). The context suggests both togetherness of the group and the exclusion of others from the zone occupied by the group.

(xx1) [kàdàgá yɛ̃:] sɔ́:n-sɔ́:n nné-m-ī: [agemate 1SgP.AnPl] together go-Impf-1PlS 'Only I and my agemates will go.'

In many cases, 'together' is translated indirectly by a verb-chain ('assemble and work'), by a PP ('beside each other'), or by a numeral in the subject NP ('we two work'). See also the construction with bendey (§18.3.2).

8.4.7.7 'All, entirely' (kéréw, sóy)

The usual 'all, entirely' adverb is kéréw (xx2). This is also the most common universal quantifier ('all X'). A less common form with similar sense is sóy (xx1.c). An intensifier for 'all, entirely' is tăy or its iteration tăy-tăy. These adverbs follow NPs (with anything from human to inanimate reference) but have no tonal interactions with them.

- (xx1) a. kéréw ńné-èr-à all go-Perf1a-3PIS 'They all went (away).'
 - b. sigórò kéréw dimé-èrè-Ø sugar all be.finished-Perf1a-3SgS 'The sugar is all used up.'
 - c. [àrnà-y tà:ndî: bû:] kéréw ńné-èr-à [man-child.L three.L Def.Pl] all go-Perf1a-3PlS 'The three boys all went (away).'
 - d. sigórò sóy dimé-èrè-Ø sugar all be.finished-Perf1a-3SgS

'The sugar is all used up.'

e. sigórò tăy dimé-èrè-∅ sugar all.Intens be.finished-Perf1a-3SgS 'The sugar is used up (to the last grain).'

Another adverb, lèrew, is not commonly used with human referents. Its characteristic context is exemplified by e.g. 'they swept the courtyard completely (= thoroughly)'. However, it was accepted by my assistant as an alternative to the 'all, entirely' adverbials in the 'sugar' examples in (xx1).

8.4.8 Derived iterated adverbials

8.4.8.1 Distributive adverbial iteration

Any numeral, or the interrogative à:ŋgǎy 'how many?', can be iterated to form a distributive adverb with meanings like 'six at a time', 'six apiece', or 'six by six'. Such phrases can be used, among other things, to specify the price per unit of a commodity for sale.

- (xxx) a. tùyà-gá à:ŋgǎy-à:ŋgǎy má bunch how.many?-how.many? Q 'How much per bunch (unit of sale)?'
 - b. pèrî-yěy pèrî-yěy ten-two ten-two 'Twenty (riyals, = 100 francs CFA) each'

8.4.8.2 'Scattered, here and there' (kân-kân)

This adverb, also found in Jamsay (and with phonological variation in other nearby Dogon languages) indicates irregular and sparse distribution.

(xx1) tŏ: kân-kân tɛ́:-só-∅ seeds here.and.there sprout-Perf2-3SgS 'The (planted) seeds have sprouted here and there.'

9 Verbal derivation

The productive suffixal derivations (stem to stem) for verbs are the Reversive ('un-...') and the Causative. There are a fair number of verbs with segmentable Mediopassive and (contrasting) Transitive endings. Adjectives have corresponding intransitive (Inchoative) and transitive (Factitive) verb forms, but these are not directly formed from the adjective by adding a suffix.

9.1 Reversive verbs (-r1-)

The Reversive suffix is -ri- (rarely -ré-). It is common in verb pairs like 'shut/open' and 'cover/uncover' that denote complementary actions, one of which reverses or undoes the other. (Note that 'shut' is more basic than 'open'.) The Reversive is often transitive ('X opened the door') but may also be used intransitively ('the door opened').

The Reversive is often chained with a following intransitive gó- 'exit' or transitive gò-ndó- 'take out, remove', which helps to clarify the valency, as in níndí-r¹¹ gó-èrè as an alternative to níndí-r¹-èrè 'it became untangled', and níndí-r¹ gò-ndó-tì as an alternative to níndí-r¹-tì 'he/she untangled (it)'. The chain construction also makes it unambiguous that a reversive sense is intended (some reversives are homophonous with nonreversive transitives). For verbs that have no morphological Reversive, the chain construction can be used as a periphrastic reversive.

A full list of attested Reversives is in (xx1). The input must be mono- or bisyllabic, so the Reversive is bi- or trisyllabic. The Reversive usually respects tonal patterns, and restrictions on vowel sequences, for underived trisyllabic verbs (this is not true of all derivational suffixes). This can be seen most clearly in (xx1.a-b), where merely adding -rî- to the input stem would produce an incorrect vowel sequence like a...a...i. So the outputs shift to acceptable vowel sequences like a...i...i. In (xx1.c), the input is already an i-final stem, so no observable change is needed in the vocalism when -rî- is added. In (xx1.d), the initial stem vowel is shortened. In (xx1.e), a Reversive of the underlying shape /CvCv-rî-/ has undergone Syncope, resulting in a CvC-rî- output with R-toned initial syllable. (xx1.f) illustrates monosyllabic inputs.

(xx1) input gloss reversive gloss

```
a. non-high vocalism adjusted to acceptable trisyllabic pattern
    kámá-
                   'crumple'
                                      kámî-r<sup>n</sup>î-
                                                           'uncrumple'
                   'lock'
                                                           'unlock'
    dàgá-
                                      dàgî-rî-
                   'nail'
    pέgέ-
                                      pégî-rî-
                                                           'remove (nail)'
b. like (a) but medial /i/ syncopated
                   'braid (rope)'
                                                           'unbraid (rope)'
    óy<sup>n</sup>́σ-
                                      óy<sup>n</sup>-r<sup>n</sup>1-
c. underived stem already ends in /i/
                   'surround'
                                      gòni-r<sup>n</sup>i-
                                                           'un-surround'
    gòni-
                   'bury'
                                                           'disinter'
    bèsi-
                                      bèsî-rî-
                   'immobilize'
                                                           'allow to move'
    gisî-
                                      gisî-ri-
                   'tie'
                                      págî-rî-
                                                           'untie'
    págî-
                                      légî-rî-
                   'insert'
                                                          'remove inserted
    légî-
                                                          item'
    ténî-
                   'hobble'
                                      ténî-r<sup>n</sup>î-
                                                           'unhobble'
    nîndî-
                   'tangle'
                                      nîndî-rî-
                                                           'untangle'
    tîmbî-
                   'put lid on'
                                      tîmbî-rî-
                                                           'take lid off'
    nàmbi-
                   'step on'
                                      nàmbi-ri-
                                                          'remove foot from'
                   'cover (person)' yèmbî-rî-
                                                           'uncover (person)'
    yèmbî-
                   (yèmbi-ri- also used as synonym of yèmbi-)
                   'press to wall'
                                      pémbi-ri-
                                                           'release (sth
    pémbi-
                                                          pressed to wall)'
                   'roll up (pants)'
    kéndî-
                                      kéndî-rî-
                                                           'unroll (pants)'
                                                           'unfold'
    mèndi-
                   'fold'
                                      mèndî-ri-
                                                           'bounce back'
    nómbî-
                   'sag'
                                      nómbi-ri-
d. vowel shortened
    kó:ndî-
                   'bend'
                                      kóndî-rî-
                                                           'unbend'
e. Reversive with RH tone after syncope
    dèwi- [děw] 'cover (object)' děw-ri-
                                                           'uncover (object)'
f. monosyllabic stems
                   'shut'
                                      pî:"-r"î-
                                                           'open'
    pi:n-
                   'fence in'
    jă:-
                                      jă:-rî-
                                                           'un-fence'
                   'tie (knot)'
                                      mɔ̃:-rñi-
                                                           'untie (knot)'
    mš:-
    lá:-
                   'braid (rope)'
                                      lá:-rî-
                                                           'unbraid (rope)'
    tó:-
                   'roll turban'
                                      tó:-rî-
                                                           'unroll turban'
```

In (xx2), we get a suffix allomorph -ré- or -ró- with upper-mid-height vowel instead of -rî-. These are regular adjustments to the vowel-sequence constraints on trisyllabics (§10.xxx). In (xx2.a), the initial syllable has /o/, which requires a matching /o/ rather than /i/ in the third syllable. In (xx2.b), the initial syllable has /i/, which is compatible with either i...i...i or i...i...e trisyllabic sequences, and the latter is decisively favored by the input bisyllabic sequence i...e.

```
(xx2) a. nóŋgî-yé- 'be caught in tree' nóŋgú-ró- 'be un-caught'

b. wĭré- 'go into coma' wĭllî-ré- 'come to (= recover)'

dĭsé- 'prop up' dĭsî-ré- 'remove a prop from'
```

(xx2.a) also illustrates the pattern whereby a Mediopassive suffix -yv- is dropped when Reversive -ri- is added (there are no quadrisyllabic or longer Reversives). Another example of this is (xx3).

```
(xx3) pémbi-yi- 'put on a wrap' pémbi-ri- 'take off wrap' [pémbi-ri- also 'put a wrap on (sb)']
```

The phonologically most difficult Reversives are those in (xx4). When -riis added to a stem of the shape (C)vrv-, with oral /r/ or /rⁿ/, either of two patterns is observed. In (xx4.a), after syncope, the expected rhotic cluster shifts to /ll/. In (xx4.b), there is some ambiguity as to the morphemic composition. In one analysis, Reversive -ri- is added to a variant form of the stem with /ll/ instead of /r/ (a kind of dissimilation to the suffixal rhotic). In the other analysis, the variant stem with /ll/ already contains the Reversive morpheme, and the final suffix is Transitive -rv-, which is elsewhere often paired with Mediopassive -yv-. In (xx4.c), because of the initial nasal, there are optional pronunciations with /n/ or /nd/ (perhaps via *nn) instead of /ll/. My assistant struggled with several of these forms in elicitation.

```
'be unhooked'
                         'be hooked'
(xx4) a. kóri-yi-
                                          kól-lî (gŏ:-)
             gàr<sup>n</sup>i-
                         'put in'
                                          găl-lî-
                                                                    'take out'
             tárî-yî-
                         'be affixed'
                                          tál-li-vi-
                                                                    'affixed item come
                                                                    off"
                         'forget'
                                                                    'remember'
             ìrέ-
                                          illi- (variant)
                         'hook, hang'
                                                                    'unhook'
         b. kóri-
                                          kólli-ri-
             ìrέ-
                         'forget'
                                          illî-rî- (variant)
                                                                    'remember'
```

```
tárî-
                'affix'
                                  tál-lî-rî-
                                                             'remove affixed
                                                             item'
                'seal up'
                                                             'unseal'
c. màr<sup>n</sup>i-
                                  măl-lî-rî- ~ mànî-r<sup>n</sup>î-
                'become lost' màndi-ri (gŏ:-)
                                                             'lost item be
    màrá-
                                                             found'
                                  [= màlli-ri (gŏ:-)]
                                  màlli-ri-
                                                             'recover lost item'
```

Some synchronically unsegmentable trisyllabic stems ending in -rî may have originated as reversives. Since -rv- is also a minor Transitive or Causative suffix, caution should be exercised here. One example is nángîri- 'remember' (synonym of ĭllî-rî-), which is synchronically isolated but corresponds to Jamsay náná-rná- 'remember', reversive of Jamsay náná- 'forget'.

Representative AN forms of two trisyllabic reversive verb stems (i.e. from bisyllabic inputs) are in (xx5). AN stems of bisyllabic reversives are in (xx6).

(xx5)		'untie'	'unhook'
	bare stem	pág î -rî	kóllî-rî
	Imprt	pág í -rà	kállá-rà
	Imprt.Neg	págî-rî-ndà:	kóllî-rî-ndà:
	Perf1b	págî-rî-tì-	kóllî-rî-tĭ-
	PerfNeg	pàgì-rà-rí-	kòllò-rò-rî-
	Impf	pà-pági-rá-m-	kò-kólló-ró-m-
	ImpfNeg	pági-rá-ŋò:-	kólló-ró-ŋò:-
(xx6)		'open'	'uncover (object)'
(xx6)	bare stem	'open' pî:n-rnî	'uncover (object)' dĕw-rî
(xx6)	bare stem Imprt	•	
(xx6)		pî: ⁿ -r ⁿ î	děw-rî
(xx6)	Imprt	pî: ⁿ -r ⁿ î pî: ⁿ -r ⁿ à	dĕw-rî dĕw-râ
(xx6)	Imprt Imprt.Neg	pî: ⁿ -r ⁿ î pî: ⁿ -r ⁿ à pî: ⁿ rî-ndà:	děw-rî děw-râ děw-rî-ndà:
(xx6)	Imprt Imprt.Neg Perf	pî: ⁿ -r ⁿ î pî: ⁿ -r ⁿ à pî: ⁿ rî-ndà: pî: ⁿ -r ⁿ î-tĭ-	děw-rî děw-râ děw-rî-ndà: děw-rî-tì-

9.2 Deverbal causative verbs

9.2.1 Productive causative with suffix -mi-

The productive Causative suffix added to verb inputs is -mi. It preserves the $\{H\}$ or $\{LH\}$ tone contour of the input (xx1).

(xx1) Causatives with -mi- (input verb ends in non-high vowel)

input	gloss	causative	gloss
a. {H}-toned kóyó kúwó- ńné- tómbó- kó:- sîr ⁿ é- éw-yé- péré- pîriyé-	'weep' 'eat (meat)' 'go' 'jump' 'eat (meal)' 'be full (sated)' 'sit down' 'jump off' 'be inflated'	kóyó-mî- kúwó-mî- ńnέ-mî- tómbó-mî- kó:-mî- sîr ⁿ é-mî- éw-yé-mî- péré-mî- pîrîyé-mî-	'make weep' 'give meat to' 'allow to go' (with /ε/) 'make jump' 'feed, nourish' 'make full (sated)' 'make sit' 'make jump off' 'inflate'
b. {LH}-toned jùgó-	'know'	jùgó-mî-	'in form'
c. monosyllabio dă:- nă:-	c Cv:- verbs 'arrive' 'drink'	dŏ:-mî- nŏ:-mî-	'cause to arrive' 'give drink to'

As the examples in (xx1) show, -mi- does not force changes in the vowel qualities of the preceding stem, provided that the stem ends in a non-high vowel (i.e. in anything but /i/). In other words, although most of the causatives are trisyllabic or longer, the constraints on vowel sequences that apply to underived trisyllabics (and longer stems) do not apply to causative derivatives.

However, if the input stem ends in /i/, this vowel must be changed before -mî-, as shown in (xx2). If there is a preceding non-high vowel, which in practice is almost always from the set $\{a \in 5\}$, this vowel is copied on the final vowel of the input stem before -mî- (xx1.a). If the stem has no non-high vowel, the default for the stem-final vowel is /ɔ/ after /u/, and /ɛ/ after /i/ (xx1.b). I have also recorded /e/ (as an optional variant of /ɛ/) after initial-syllable /i/ under the influence of a preceding nasal (xx1.c).

(xx1) Causatives with -mi- (input verb ends in /i/)

```
input
                 gloss
                                      causative
                                                      gloss
a. input has preceding non-high vowel \{a \in a\}
    bă:rî-
                 'help'
                                      bă:rá-mî-
                                                      'make help'
                                                      'make cover'
    dèwî-
                 'cover'
                                      dèwé-mi-
    sź:rî-
                 'creak'
                                      só:ró-mî-
                                                      'make creak'
b. input has preceding {u i} only
    núy<sup>n</sup>i-
                 'enter'
                                      núy<sup>n</sup>5-mî-
                                                      'make enter'
    pú:ri-
                 'frisk'
                                      pú:ró-mî-
                                                      'make frisk'
                                                      'make cover'
    tîmbî-
                 'cover'
                                      tîmbέ-mî-
    pîy<sup>n</sup>î-
                 'shut'
                                      pîy<sup>n</sup>έ-mî-
                                                      'make shut'
    jinjî-
                 'make noise'
                                      ĵinjε-mî-
                                                      'cause to make noise'
c. input has preceding /i/ only, nasal allows variant /e/
    tîŋî-
              'speak'
                               tiné-mi- ~ tiné-mi-
                                                          'make speak'
    jiŋi-
             'ride double'
                               ĵiŋé-mî- ~ ĵiŋέ-mî-
                                                         'have ride double'
d. after Mediopassive -yi-
    tágî-yî-
                 'put on shoes'
                                      tágî-yé-mî-
                                                      'put shoes on (sb)'
```

The causative stem has a characteristic conjugation (xx2). The vocalism of the suffix combinations is independent of that of the preceding input stem. For example, kóyó-mi- 'cause to weep' has the same suffixal forms as núyⁿó-mi- 'cause to go in' (xx2), e.g. Imperative kóyó-mò.

(xx2) Paradigm of Causative ('cause to go in')

```
núy<sup>n</sup>ó-mî
                                bare stem
núy<sup>n</sup> 5-mi-ndέ
                                Verbal Noun
nùy<sup>n</sup>ò-mì-
                                unsuffixed Perfective (3Pl y-à:)
núy<sup>n</sup>5-m-tì-
                                Perfective-1b
núy<sup>n</sup>ó-m-só-
                                Perfective-2
núy<sup>n</sup>ó-m-jè-
                                Recent Perfect
núy<sup>n</sup>ó-mi tá:-só-
                                Experiential Perfect
                                Perfective Negative
nùy<sup>n</sup>ò-mè-r<sup>n</sup>i-
nù-núy<sup>n</sup>ó-mέ-m-
                                reduplicated Imperfective
nù-núy<sup>n</sup>ó-mê:-sò-
                                reduplicated Progressive
núy<sup>n</sup>ó-m-ŋò:-
                                Imperfective Negative
```

```
núy<sup>n</sup>5-mò Imperative
núy<sup>n</sup>5-m-ndà: Imperative Negative
núy<sup>n</sup>5-mé-mày<sup>n</sup> Hortative
```

9.2.2 Minor causative suffixes ($-gi \sim -\eta i$ -, $-nd\acute{e} \sim -nd\acute{i}$ -)

Transitive -ri- functions as a causative in some but not all cases, see §9.2.3, below.

A suffix $-g\hat{i} \sim -\eta i$ - occurs with a few stems, mainly in causative function. All clear examples known to me are in (xx1). The variant $-\eta \hat{i}$ - occurs after nasal syllables. Note that $-g\hat{i}$ - does not alter the vocalism of the preceding stem (there is no conversion to canonical trisyllabic vowel-sequence patterns).

(xx1) $-gi \sim -\eta i$ - in causative and other functions

```
a. causative
 -gi- after nonnasal syllable
                '(sth) crumble'
    pásá-
                                     pósó-gí-
                                                   'crumble (sth)'
                'become lost'
                                                   'get rid of'
    màrá-
                                     màrá-gî-
    wòró-
                'cave in'
                                     wòró-gi-
                                                   'demolish'
    wùró-
                '(sb) wake up'
                                     wùró-gî-
                                                   'wake (sb) up'
    párá-
                '(sth) snap'
                                     párá-gî-
                                                   'snap (sth)'
 -ni- after nasal syllable
    kúmó-
                '(bone) break)'
                                     kúmó-nî-
                                                   'break (sth long)'
    ñàmá-
                'malfunction'
                                     ñàmá-ŋî-
                                                   'damage, waste'
b. transitivizing (causative-like)
                'be narrow, tight'
                                     émbî-gî-
                                                   'hold (sth) in armpit
    έmbî-
                                                   (by squeezing)'
c. intensive
    kárá-
                'incise'
                                     kárá-gî-
                                                   'rip, tear'
```

A suffix -nde- is attested in the archaic and irregular si-ndé- 'take/bring down', from intransitive sigé- 'go down'. It is probably related to a minor Factitive suffix -ndi- that is attested with two adjectival verbs (bă:-ndi- 'fill, make full' and é:-ndi- 'tighten'), see §9.5, below.

9.3 Passive and Transitive

9.3.1 Mediopassive -yi- and Transitive -ri-

Alternations of Mediopassive (MP) -yi- and Transitive (Tr) -ri- occur chiefly with verbs of stance (xx1.a), putting on or wearing garments (xx1.b), and holding (xx1.c). The examples given are representative rather than exhaustive.

For the stance verbs, the mediopassive denotes taking a position, while the form with Transitive -ri- is causative. For the verbs of wearing and holding, the mediopassive is syntactically transitive but denotes an action that results in the agent being in a state (of wearing or holding something). In this case, the form with Transitive -ri- denotes the act of placing an object in such a way that another person wears or holds it.

```
(xx1)
             MP
                           gloss
                                                    stem
                                                                 gloss
         a. stance
             biyé-
                           'lie down'
                                                    bi:-ré-
                                                                 'have lie down, put
                                                                 to sleep'
                           'sit down'
             éw-yé-
                                                    éw-ré-
                                                                 'have sit, seat'
             î:-yî-
                           'stand up, stop'
                                                    î:-rî-
                                                                 'stop (sth)'
                           'kneel'
             túŋi-y<sup>n</sup>i-
                                                    túŋî-r<sup>n</sup>î-
                                                                  'cause to kneel'
         b. wearing clothes
             tágî-yî-
                           'put one's shoes on'
                                                    tágî-rî-
                                                                  'put shoes on (sb)'
             dòmbi-yé-
                           'roll on turban'
                                                    dòmbó-ró-
                                                                  'put turban on (sb)'
                           'put one's hat on'
             ków-yé-
                                                    ków-ró-
                                                                  'put hat on (sb)'
             págî-yî-
                           'tie one's belt on'
                                                    págî-rî-
                                                                 'tie belt on (sb)'
                    (cf. pági- 'tie')
         c. carrying/holding
             bàmbi-yi-
                           'carry on back'
                                                    bàmbi-ri-
                                                                  'put on (sb's) back'
             dùyi-
                           'carry on head'
                                                    dǔ:-rî-
                                                                  'put on (sb's) head'
             kómbî-yî-
                           'cling to'
                                                    kómbi-ri-
                                                                  'cause to cling'
         d. other
             nî: điyé-
                           'bathe'
                                                    nî: đi:-ré-
                                                                 'bathe (sb)'
                    (with ni: 'water')
             ńdέ-
                           'go up'
                                                    ńdî-rî-
                                                                  'take up'
                           'be separated'
                                                                  'separate [tr]'
             káwá-
                                                    káw-rî-
                           'become wet'
                                                                 'make wet'
             témbi-yi-
                                                    témbi-ri-
             ságî-yî-
                           'expand [intr]'
                                                    ságî-rî-
                                                                 'expand (sth)'
             dàwi-yi-
                           'hide [intr]'
                                                    dăw-rî-
                                                                 'hide (sth)'
                                               (</dawi-ri-/)
```

tîyé- 'be spilled' tî:-rî- 'spill [tr]'

The /e/ in -yé- and -ré- in some of these pairs is due to the vocalism of the stem, which contains {o e} and therefore triggers adjustments that reflect constraints on vowel sequences in trisyllabic stems.

Segmentability of Mediopassive -yi- is difficult in the cases of biyé- 'lie down' (xx1.a), and diyé- 'bathe' and tiyé- 'be spilled' (xx1.d) because of their initial short-voweled syllables. The corresponding transitives have a phonetic long [i:] that could easily be (re-)interpreted by native speakers as reflecting /iy/ syncopated from /iyi/, i.e. biy-ri- and diy-ri-. In this interpretation, -ri- is still segmentable as a Causative suffix allomorph, but the stems can be taken as biyé- and diyé- without segmentation. I therefore transcribe these stems without hyphens. A similarly problematic case is dùyi- 'carry on head' (xx1.c), but here the transitive is dù:-ri-.

In some of these cases, the transitive form shown competes with a Causative with suffix -mi- added to the mediopassive form, e.g. éw-yé-mi- 'cause to sit'. This suggests that the -yi-/-ri- alternation is not quite as productive as in e.g. Najamba.

Of the two verbs with the general sense 'put on clothes, get dressed', one has the expected alternation: dúŋî dùŋî-yî- 'get dressed' versus dúŋî dùŋî-rʰi- 'dress (someone)' (shown with cognate nominal). The alternative for 'get dressed' is yùrî kúrʰi-yʰi- (with yùrî 'fabric, garment'), but the expected transitive #kúrʰi-rʰi- does not occur. Instead, we get just yùrî kúrʰi- 'dress (someone)'. This suggests that -ri- is not readily added to stems with a rhotic in the final syllable.

9.3.2 Passive suffix -yé

The Passive $-y\hat{\epsilon}$ allows, but does not require, the presupposition of an **unexpressed agent**. The preceding stem shifts to **H-tone** (if not already H-toned). The suffix itself is not affected by phonological features of the stem; for example, there is no Nasalization-Spreading, and no harmonizing of vowels.

The passive was readily elicited from a range of transitive inputs (xx1).

(xx1)	input verb	gloss	passive	gloss
	a. input {H}-toned	l		
	kár ⁿ i-	'do'	kár ⁿ i-yé	'be done'
	tîŋî-	'speak'	tîŋ-yέ	'be said'
	tóŋî-	'write'	táŋ-yέ	'be written'

káwrá- kémé-	'disperse' 'build'	káwrî-yé kémé-yé	'be dispersed' 'be build'
b. input {LH}-tor	ned		
yĭ:-	'see'	yi̇̀:-yέ	'be seen'
gùró-	'rob'	gúró-yé	'be robbed/stolen'
gòndó-	'take out'	góndó-yέ	'be taken out'
no‡:-	'drink'	nό:-yέ	'(liquid) be drunk'

The Passive form is **adjectival** in form; more specifically, it is a special use of a participial form (as we will see below). It can function as a modifying adjective, either with a semantically light stem filling the noun slot (xx2.a), or with a more typical common noun (xx2.b).

```
(xx2) a. [kò gúró-yɛ́] ɛ́wɛ́-ŋò-yn [thing.L steal-Pass] buy-ImpfNeg-1SgS 'I don't/won't buy anything stolen.'
```

```
    b. [mòbìfi gúró-yέ] έwέ-ŋò-y<sup>n</sup>
    [vehicle.L steal-Pass] buy-ImpfNeg-1SgS
    'I don't/won't buy a stolen vehicle.'
```

The passive form is, however, often **predicative**, in which case it is followed by the appropriate conjugated form of the **'it is' cltiic** (§11.xxx). For example, ngú gúró-yé=w 'this is/was stolen' can be said while pointing to an object of dubious provenance found in a thieve's den. It can be negated with the Stative Negative =ndŏ:, as in ngú gúró-yé-w=ndŏ: 'this is/was not stolen'.

The third plural forms end in $-y \in \emptyset$ instead of $-y \in y \in \emptyset$. Some further examples are in (xx3). See also §17.2.2.2.

- (xx3) a. kárⁿî-y€≡ẃ do-Pass≡it.is.InanSgS 'It is (= has already been) done.'
 - b. tóŋ-yέ=∅ write-Pass=it.is.InanPIS
 'They (e.g. books) are (= have already) been written.'
 - c. gùrîgúró-yέ≡m-1:robbery rob-Pass≡it.is-1SgS'I have been robbed.'

That we are dealing with a participial form, and therefore with a kind of relative construction, is suggested by the fact that the subject (when overtly expressed) undergoes tone-dropping. This is consistent with the tone-dropping on relative head NPs. Examples are in (xx4).

```
a. pèrgè sémé-yé≡ή sheep.L slaughter-Pasś≡it.is.3SgS 'A/the sheep-Sg is (= has already been) slaughtered.' (pèrgé)
b. àrnà gùrî gúró-yé≡ή man.L robbery rob-Pass≡be.3SgS 'A/the man has been robbed.' (árnâ)
c. tòŋòtóŋ-yé≡ẃ writing.L write-Pass-be.InanS 'A/the book is (= has already been) written.' (tòŋó)
```

However, the passive-predicate construction differs from the true relative construction in that the **subject NP may end in a determiner or a non-numeral quantifier** (in relatives, such elements must appear after the verbal participle).

- (xx5) a.. [pèrgè bû:] sémé-yé≡∅ [sheep.L **Def.AnPl**] slaughter-Pass=it.is.3PlS 'The sheep-Pl are (= have already been) slaughtered.'
 - b. [àrⁿà wŏŋ] gùrî gúró-y€≡ý [man.L **Dem**] robbery rob-Pass≡it.is.3SgS 'This man has (already) been robbed.'
 - c. [árⁿâ gá:rɛ̀] gùrî gúró-yɛ́≡∅ [man.L eight] robbery rob-Pass≡it.is.3PlS 'Eight men have been robbed.'

Negative predicative forms of the passive can be formed by adding an inflected form of the Negative 'it is not' clitic =ndŏ: (\S xxx) to the Passive suffix (xx6).

```
(xx6) tɔ́ŋɔ́-yɛ́=ẃ=ndŏ:-Ø
write-Pass=it.is.InanSgS=it.is.not-3SgS
'It has not been written.'
```

Alternatively, $-y\dot{\epsilon}=$ plus the inflected form of the (positive) 'it is' clitic is added to the Perfective Negative verb form with $-r\dot{i}-$ (xx7). In the case of 3Pl subject, both inanimate and animate, the 3Pl Perfective Negative $-nd\dot{u}-$ is the basis for the form, which is therefore doubly conjugated. $-nd\dot{u}-$ is pronounced $-nd\dot{i}-$ before $-y\dot{\epsilon}=$.

- (xxx) a. tòŋò-rⁿi-yέ≡ẃ write-Perf.Neg-Pass≡it.is.InanS 'It has not been written.'
 - b. gùrò-rî-yé≡m-îy rob-Perf.Neg-Pass≡it.is-1SgS 'I have not been robbed.'
 - c. gùrò-ndî-yέ≡∅
 rob-Perf.Neg.3PIS-Pass≡it.is.3PIS
 'They have not been robbed.'
 - d. tòŋò-ndî-yέ≡∅
 write-Perf.Neg.3PIS-Pass≡it.is.3PIS
 'They have not been written.'

Both positive and negative predicative forms of the passive may combine with Past clitic $=b\varepsilon$ -. For the forms and examples, see §10.5.1.xxx.

9.3.3 Minor passive suffix -mi-

One unproductive type is a morphological mimic of the causative, using the same suffix -mî-. The two known examples are in (xx1). The best gloss is of the type 'be VERB-able'.

```
(xx1) a. témbé- 'encounter' témbé-mî- 'be found (often), be findable' b. bèré- 'obtain' bèré-mî- 'be gotten (often), be available'
```

Examples: reduplicated imperfective tè-témbé-mé-ἢ 'it is found (= possible to find, available)', imperfective negative béré-m-ŋò:-Ø 'it isn't available, it can't be gotten'.

9.4 Ambi-valent verbs without suffixal derivation

Ambi-valent verbs, i.e. those used both transitively and intransitively, are not typical of Nanga.

9.5 Deadjectival inchoative and factitive verbs

In the tables in this section, the adjective is first given in its usual modifying form (it should be recalled, though, that many nonmonosyllabic adjectives with final /i/ in this form also have a predicative form with final /u/). The adjective is followed by the inchoative verb ('become ADJ') and the factitive verb ('make [something] ADJ'). The factitive is the causative of the inchoative, and usually ends in Causative -mî- (there are also a few cases with -ndî-). The inchoative is therefore the lexically basic verb.

In the first set of examples, the inchoative verb has **no segmentable derivational suffix**. The inchoative verb and the adjective are semi-autonomous members of the same word-family, rather than one being directly derived from the other. The phonological form of the verb respects the usual constraints on verb-stem shapes regarding tone and vocalism. The verb is {LH} toned if it begins with a voiced stop, and high-toned if it begins with a voiceless stop. These constraints do not apply to the adjective.

```
factitive
(xx1)
             gloss
                                     adj
                                                    inchoative
         a. Factitive -ndi-
             'full'
                                     bá:
                                                    bă:-
                                                                      bă:-ndî-
             'tight (screw)'
                                     ě:
                                                    έ:-
                                                                      έ:-ndî-
         b. Factitive with Causative -mî-, inchoative ends in non-high vowel
             'tight (rope)'
                                     ě:
                                                    έ:-
                                                                      έ:-mî-
             'old'
                                                    pέ:-
                                                                      pέ:-mî-
                                     pĚ:
             'dry, hardened'
                                                                      mă:-mî-
                                     mă:
                                                    mă:-
             'ripe (crop); fresh'
                                     îrέ
                                                    îré-
                                                                      îré-mî-
             'half-ripe'
                                     àmá
                                                    ámá-
                                                                      ámá-mî-
             'plump'
                                                                      ámá-mî-
                                     àmî
                                                    ámá-
                                                                      bàrná-mî-
             'red'
                                     bár<sup>n</sup>î
                                                    bàr<sup>n</sup>á-
             'empty, bare'
                                     kóró
                                                    kóró-
                                                                      kóró-mî-
             'weak, diluted'
                                     sèré
                                                    séré-
                                                                      séré-mi-
             'ripe (fruit)'
                                                    bòró-
                                                                      bòró-mi-
                                     bòró
             'black'
                                     įέmî
                                                    jèmé-
                                                                      jèmé-mî-
```

c. Factitive with Causative -mi-, inchoative ends in /i/

'curved'	gòndî	gòndi-	gòndó-mî-
'narrow'	èmbî	έmbî-	έmbέ-mî-
'tilted'	jèŋî	jèŋî-	jèné-mî-

In (xx1.c), the final /i/ of the inchoative is replaced, before the Causative suffix, by a non-high vowel copied from the initial syllable.

In the remaining forms, the inchoative verb has a **derivational suffix**. One could therefore argue that the inchoative is formed by adding an Inchoative suffix directly to the adjective. To make this work, the adjectival stem would have to be reshaped to fit the constraints on the phonolological form of verbs, in addition to the simple addition of the derivational suffix.

In (xx2), the Inchoative suffix is -yî-, with variant -yé- after verbs with $\{e \ o\}$ vowel. This suffix requires the preceding stem to end in /i/. The -yî-variant shifts to -yé- before the Causative suffix. The /y/ is regularly nasalized to /yⁿ/ after a nasal syllable. This inchoative formation is found with stems that have an initial long vowel, i.e. of shape Cv:C(C)v-, and with the two basic adjectives denoting temperatures.

(xx2) Inchoative -yi-/-yé-

gloss	adj	inchoative	factitive
a. Cv:C(C)v-			
'thin'	ké:mbé	ké:mbî-yî-	ké:mbî-yé-mî-
'skinny'	kó:mbó	kó:mbî-yé-	kó:mbî-yé-mî-
'young, adolescent'	sŏ:rô	só:rî-yé-	só:rî-yé-mî-
'easy, cheap'	nà:r ⁿ á	ná:r ⁿ i-y ⁿ i-	ná:r ⁿ i-y ⁿ έ-mi-
'unripe, raw'	kè:sî	ké:sî-yé-	ké:sî-yé-mî-
b. temperature adjectives	S		
'cold'	támî	támî-yî-	támî-yé-mî-
'hot'	ógî	ógî-yî-	ógî-yέ-mî-

The most common and productive Inchoative suffix, however, is -ndiyé-. The /e/ vowel is stable, occurring with any stem vocalism. The comments at the beginning of each group of examples in (xx3) suggest how one might derive the vocalism of the inchoative verb from that of the adjective.

The tone contour of the inchoative can in most cases also be predicted, but this is generally based on the initial consonant, rather than on the lexical tones of the adjective. The inchoatives beginning in a voiceless stop, /s/, or zero (vowel-initial) have all-high tone contour, while those beginning in a voiced stop have {LH}. This leaves sonorant-initial stems, which in principle allow either tone contour in verbs, and might therefore clarify whether the adjective

and the inchoative share a lexical tone contour. Here I find two cases of {LH} in both the adjective and the inchoative (wagá 'distant', mɔsi 'bad/ugly'), one with {HL} in the adjective and all-high in the verb (már"i 'hard'), one with {LHL} adjective and {LH} in the verb (yágî 'coarse'); and one with {LHL} adjective and all-high verb (nɔmî 'difficult/costly; this adjective has a predicative form nɔmú with {LH} contour). These data are rather equivocal.

(xx3) Inchoative -ndîyé-

```
a. stem ends in \{\varepsilon \circ a\}, vocalism stable
                                      wàgá-ndiyé-
    'distant'
                          wàgá
                                                       wàgá-ndíyé-mí-
    'fat; numerous'
                          òwó
                                      ówó-ndivé-
                                                       ówó-ndîyé-mî-
    'small, young'
                                      éwré-ndîyé-
                                                       éwré-ndiyé-mi-
                          èwrέ
    'long, tall'
                                      gùró-ndiyé-
                                                       gùró-ndíyé-mí-
                          gùró
b. CvC stem with non-high vowel extended to CvCv- by copying vowel
    'spacious'
                                      gàwá-ndíyé-
                                                       gàwá-ndîyé-mî-
                          gâw
    'short'
                          dêη
                                      dèné-ndîyé-
                                                       dèné-ndiyé-mi-
c. stem has u...i (but predicative u...u) becoming u...u, stem CvCv
    'big, fat'
                          dùgi
                                      dùgú-ndiyé-
                                                       dùgú-ndiyé-mi-
    'heavy'
                          dúsî
                                      dùsú-ndívé-
                                                       dùsú-ndîvé-mî-
    'soft (skin)'
                          búri
                                      bùrú-ndiyé-
                                                       bùrú-ndiyé-mi-
d. stem has u...i becoming u...e, stem CvCCv
    'coarse'
                                      kúnjé-ndíyé-
                                                       kúnjé-ndíyé-mí-
e. stem has i...i, vocalism stable
    'white'
                                      pîrî-ndîyé-
                                                       pîrî-ndîyé-mî-
f. Ci: stem extended to Cive-
    'pointed'
                                      sîyé-ndîyé-
                                                       sîyé-ndîyé-mî-
g. stem ends in /i/ that is replaced by copy of non-high first vowel
    'bitter'
                                                       gàrá-ndíyé-mí-
                                      gàrá-ndiyé-
                          gárî
    'salty, sour'
                          párî
                                      párá-ndiyé-
                                                       párá-ndíyé-mí-
    'half-bitter'
                          ásî
                                      ásá-ndivé-
                                                       ásá-ndíyé-mí-
    'hard'
                                     már<sup>n</sup>á-ndíyé-
                                                       már<sup>n</sup>á-ndíyé-mí-
                          már<sup>m</sup>i
    'coarse'
                                      yàgá-ndíyé-
                          yágî
                                                       yàgá-ndiyé-mi-
    'no good (garbage)'
                          gòmî
                                      gòmó-ndiyé-
                                                       gòmó-ndíyé-mí-
    'somewhat rotten'
                          gómî
                                      gòmó-ndiyé-
                                                       gòmó-ndíyé-mí-
    'bad, ugly'
                          mòsi
                                      mòsó-ndiyé-
                                                       mòsó-ndiyé-mi-
    'difficult, costly'
                          nòmî
                                     nómó-ndíyé-
                                                       nómó-ndíyé-mí-
    'hot, fast'
                                      ógó-ndivé-
                                                       ógó-ndívé-mí-
                          ógî
    'deep'
                          sóŋî
                                      sónó-ndíyé-
                                                       sónó-ndiyé-mi-
```

'smooth, sleek'	órî	óró-ndiyé-	óró-ndîyé-mî-
'good'	èsî	ésé-ndîyé-	ésé-ndiyé-mi-
'sweet; sharp'	έrî	éré-ndîyé-	éré-ndiyé-mi-
'lightweight''	έr ⁿ i	έr ⁿ έ-ndiyé-	ér ⁿ é-ndiyé-mi-

There are also two cases known to me of **Inchoative -rí-**. The adjective yágî 'coarse' has been given above with its inchoative yàgá-ndîyé- 'become coarse', but it also has a more specialized verb yàgí-rí- '(skin) be itchy'. The adjective ógî is also given above, in the sense 'hot', with inchoative ógî-yî-, and in the senses 'hot/fast', with inchoative ógó-ndîyé-. However, for 'become fast(er)' (i.e. 'speed up'), there is another verb, ógî-rî-.

Examples of adjectives (or adjective-like compound finals) that **do not have** an associated inchoative verb are in (xx4). For 'other' the problem is logical ('become other' makes little sense except in a postmodern context). 'Become new' is not much better. The other terms in (xx4) are noun-like.

(xxx)	gloss	adjective	inchoative/factitive
	'other'	bèndi	_
	'new'	kándà	_
	'young adult'	sátárá	_
	'living'	úmá	_
	'runty'	kèdé	_

9.6 Denominal verbs

There is no productive denominal verbalization. Some cases of verb-noun relationships where the noun is arguably lexically basic are listed in (xx1). We seem to have a -ri- suffix in (xx1.a), -giyé- in (xx1.b), and -yé- in (xx1.c). The -mi- ending in (xx1.d) may be a special case of the Causative suffix -mi-. The example in (xx1.e) involves just a final-vowel change (the stem is borrowed from Fulfulde).

(xx1)	noun	gloss	verb	gloss
a.	dû: tigâ	'load' 'family name'	dŭ:-rî- tîgî-rî-	'load (e.g. cart)' '(griot) chant the ancestry of (sb)'
b.	únďi	'forest'	úndú-giyé-	'(zone) become dense (e.g.after rains)'

'(a) share' kérî-yé-'share, divide up' kùrî c. pó:-mî-'greet' d. pŏ: (greeting) 'soil, make dirty' lisigirⁿé 'filth' lîsîgîrⁿέ-mîlîsîgîrⁿî-yⁿî-'become dirty, soiled' 'injury' 'injure, wound bármà bármî-(someone)'

9.7 Obscure verb-verb relationships

Listed without comment.

(xx1) verb gloss related verb gloss

ná:
'spend night' ná:-mî
'pàmá'malfunction' pàmá-ŋî'do harm to, ruin'

10 Verbal inflection

10.1 Inflection of regular indicative verbs

Inflected indicative verbs are normally followed by an aspect-negation (AN) suffix, then an optional Past suffix, then a pronominal-subject suffix (including 3Sg zero). There is an unsuffixed Perfective paradigm with no audible AN suffix preceding the pronominal.

Tense as such is not marked in the basic AN system, but a conjugatable Past clitic may be added to several AN suffixes.

The modal categories (Imperative, etc.) have distinctive morphology but are also suffixal, except that the singular imperative is marked by tonal and vocalic changes rather than by suffixation.

10.1.1 Suffixes versus chained verbs

As in other Dogon languages, in those categories where a nonzero AN morpheme follows the regular bare stem of the verb (with its regular tones), there is an issue as to whether the AN morpheme is a suffix or a chained verb. Experiential Perfect tá:-só- is unmistakably an auxiliary verb (with its own Perfective-2 suffix -só-). Recent Perfect jè- does not have a similar inflectional suffix in main clauses, but it is followed in relative clauses by Perfective participial -sè, and a preverbal subject pronominal (e.g. in nonsubject relatives) intervenes between the primary verb and jè- (§10.xxx). I write tá:-só- and jè- as separate words.

Perfective-1b -ti- and Perfective-2 -só- might also be analysed as constituting separate words, but the evidence is equivocal. In favor of this analysis is the fact that the preceding primary verb has its regular lexical tones and segments, as in chains. On the other hand, -ti- and -só- do not normally appear in relative clauses, which have Perfective Participial -sè instead. If -sè is identified as a variant of -só-, then there is no case for taking -só- as a separate word, since preverbal subject pronominals cannot intervene between the primary verb and -sè (§14.1.7.1).

It was possible to elicit relatives with -ti-, but only with some difficulty, and this construction did not occur in spontaneous data. My assistant did allow preverbal subject pronominals to intervene between the primary verb and -ti- in

most, but not all, of the elicited examples. In the absence of clear evidence, I write -ti- as a suffix, but a case can be made for a separate-word analysis.

10.1.2 Overview of categories

For regular (active) verbs in the basic indicative mood, there is a fundamental aspectual split between perfective and imperfective aspectual systems, though each system consists of more than one AN category. Both systems are also divided into positive and negative subsystems.

The aspectual split is neutralized in statives, which include Stative paradigms derived from otherwise active verbs (§10.4), the Progressive (§10.2.2.3), and various defective quasi-verbs like 'be (somewhere)' and 'have' (§11.2). Statives also have a special negative forms distinct from those of active verbs.

Modal categories marked in the morphology are imperative, (first person) hortative, and a kind of third-person hortative used in expressing wishes. The imperative and (first person) hortative have special negative forms.

10.1.3 Verb stem shapes

Verb stems (excluding inflectional suffixes) may have from one to three syllables. Some verbs are themselves derived from simpler verbs or from adjectives; for these derivations see Chapter 9. The present chapter disregards the internal structure of such derivatives.

In this section I discuss the basic lexical stem shapes, proceeding from monosyllabics to trisyllabics. The **basic stem** is used in bare-stem form (as in the complement of bèré- 'be able to', and in other verb-chains), and is also the basis for the Perfective and (with tonal overlays) the reduplicated Imperfective and the Future Negative. However, some other inflected forms (notably the Imperative and the Perfective Negative) induce phonological changes on verb stems.

10.1.3.1 Cv:- verb stems

The Cv:- verbs known to me are in (xx1). The subsections of (xx1) are organized by vowel quality. I initially thought that H-toned and R-toned stems could be distinguished even for monosyllabics, and early drafts of this grammar made this claim. Further study led me to conclude that the R-toned Cv:-

transcriptions were incorrect. However, vestiges of an original tone distinction in monosyllabics may survive in the third person hortative (§10.6.xxx).

I have no example of a Cu: verb; the verb 'go in' appears as núy (compare Jamsay nú:-, Ben Tey nú-, etc.). There is only one example of Ci:.

```
(xx1)
             H-toned {LH}
                                           gloss
        with /a/
                                            'uproot (peanuts)'
             á:-
             ká:-
                                            'shave'
             má:-
                                            'become dry'
                                            'spend night'
             ná:-
             pá:-
                                            'chip off'
             sá:-
                                            'reply'
                                            'uproot (with a tool)'
             sá:-
             sá:-
                                            'take (millet grain spikes) to pile'
                                            'strain'
             sá:-
             tá:-
                                            'shoot'
             tá:-
                                            '(ripening fruit) begin to turn color'
                        bă:-
                                            'be enough; be full'
                                            'learn'
                        dă:-
                                            'endure'
                        dă:-
                        dă:-
                                            'patch up'
                        gă:-
                                            'harvest (rice) with sickle'
                                            'be unaware'
                        gă:-
                                           'fence in (with thorn branches)'
                        jă:-
        with /an/
             tá:n-
                                           'avoid (taboo)'
             tá:n-
                                            'build shed (thatched shelter)'
             pá:n-
                                            'take (step)'
        with /o/
             kó:-
                                            'cover (e.g. box) with animal hide'
             kó:-
                                            'spit (wood)'
                                            'sew'
             kó:-
             pó:-
                                            'whistle'
                                            'replaster (mud wall)'
             pó:-
             só:-
                                            'dip'
             tó:-
                                            'hit hard (with stone)'
                        bŏ:-
                                            'sip'
                                           'wash (hands)'
                        dŏ:-
```

```
gŏ:-
                                   'go out'
                                   'catch, hold'
               wŏ:-
with /o/
    k5:-
               'eat (meal)'
               'tie (knot)'
    mó:-
               'drink'
    nó:-
    pá:-
               'pick (fruits)'
               'leach, let ferment'
    pá:-
    sá:-
               'peck at'
    tó:-
               'sow, plant (seeds)'
    tó:-
               'roll (turban)'
    tó:-
               'take out (daily rations)'
    tó:-
               '(milk) fill up (in udder)'
                                   'unsheathe'
               bă:-
                                   'arrive'
               dă:-
                                   'burn (on fire)'
               dă:-
                                   'jab'
               gš:-
               jš:-
                                   'pick out delicately'
               wă:-
                                   '(rain) fall'
with /e/
    ké:-
                                   '(grasshopper) bite off'
    té:-
                                   '(muddy water) become clear'
               bě:-
                                   'remain'
                                   'bring'
               jě:-
                                   'come'
               yě:-
with /e/
    έ:-
                                   'become tight'
    έ:-
                                   '(woman) marry (man)'
    tέ:-
                                   'lay out (mat)'
    pέ:-
                                   'break off'
                                   'get old'
    pέ:-
    kέ:-
                                   'shine'
                                   'cut out (leather sections)'
    kέ:-
                                   'trim (hair)'
    sέ:-
                                   'cut off end'
               bě:-
                                   'be tired'
               dě:-
               jě:-
                                   '(man) marry (woman)'
                                   (ńné jě:- 'go away, get out')
               jě:-
```

```
with /i/ (see also Ciy- stems below)
yĭ:- (yĭ-) 'see' (discussed separately below)
with /u/
[none]
```

For the verbs in (xx1) other than 'see', the long vowel is consistent across all AN suffixal categories, with the following exceptions. Those that take Perfective-1a -èrè- instead of Perfective-1b -tì- shorten the vowel, and may contract it with the initial $/\epsilon$ / of the suffix: gŏ:- 'go out', Perfective gŏ-èrè 'went out' (§10.xxx). In the Imperative, Ce: verbs shift to Co:, and Ce: verbs appear as Cea with a diphthong (§10.xxx).

```
10.1.3.2 vĭ:- 'see'
```

There only Ci:- verb is yǐ:- 'see', and it is somewhat irregular. Specifically, the Perfective Negative yè-rî- suggests a short-voweled variant /yǐ-/, and moreover suggests that the /i/ has affinities to the final short /i/ of nonmonosyllabic verbs like kár°i- 'do', which also drops to a non-high vowel before the Perfective Negative suffix (kàr°à-r°i-). The other cases where the vowel shifts from /i/ to /ɛ/ are certain Hortative forms. The Imperfective Negative nù-nj- (more common than the morphophonologically regular variant yǐ:-nj>:-) not only shows a similar short stem vowel, but shows other irregularities (low-toned stem followed by high-toned suffix, short suffixal vowel) that are shared only with nuy 'hear'. The other inflected forms are regular and are compatible with /yǐ:-/. The paradigm is in (xx1). The Perfective-1b is elicitable but uncommon, as the Perfective-2 is the common perfective of perception verbs.

(xx1) Paradigm of 'see'

yĭ: yî:-ndέ	bare stem Verbal Noun
yĭ:-	unsuffixed Perfective (3Pl y-à:)
yĭ:-tì-	Perfective-1b
yĭ:-só-	Perfective-2
yĭ:-jὲ-	Recent Perfect
yĭ: tá:-só-	Experiential Perfect
yè-rî-	Perfective Negative
yĭ-yî:-m-	reduplicated Imperfective
yì-yî:-sò-	reduplicated Progressive

```
nù-ŋó- ~ yǐ:-ŋò:- Imperfective Negative

ŷi: Imperative
ŷi-rá Imperative Negative
yɛ:-ndà: third-person Hortative Negative
yɛ:-mày<sup>n</sup> Hortative
```

10.1.3.3 *Cuy*(*i*)- and *Ciy*(*i*)- verbs

There is a difficulty in deciding whether the verbs in (xx1) are best represented as Cvy- (where "v" = a short high vowel /i/ or /u/) or as bisyllabic Cvyi-. In the cases where "v" is /i/ (xx1.c), the third possible representation is Ci:.

(xx1)		stem	gloss
	a.	núy ⁿ - nŭy ⁿ -	'go in' 'hear'
	b.	túy- ~ tîy-	'put down'
	c.	tiy- bĭy ⁿ -	'send' 'put (earth in hole)'

The analytical problem is due to the fact that a final /i/ syncopates or apocopates after a semivowel, so there is no surface contrast between Cuyi and Cuy, and no surface contrast between Ciyi, Ciy, and Ci:.

The best evidence for a bisyllabic representation is the Perfective Negative along with the Hortative. A final /i/ in a nonmonosyllabic stem shifts to a non-high vowel before Perfective Negative -ri- and Hortative -mày. The Perfective Negative forms of the verbs in question suggest /Cvyi-/. (xx2) shows the revised representations for the stem as well as the Perfective Negative form.

(xx2)		stem	PerfNeg	gloss
	a.	núy ⁿ î- nùy ⁿ î-	nùy ⁿ ò-r ⁿ î- nùy ⁿ ò-r ⁿ î-	ʻgo in' ʻhear'
	b.	túyî- ~ tîyî-	tùyò-ri- ~ tiyè-ri-	'put down'
	c.	tîyî- bĭy ⁿ î-	tìyè-rî- bìy ⁿ è-r ⁿ î-	'send' 'put (earth in hole)'

However, other forms of these verbs are compatible with monosyllabic Cvy- representations. It is possible that native speakers have a phonological analysis of the Perfective Negative and the Hortative (which are rather marked categories) that does not require positing a basic representation with final /i/. Possible direct evidence for Cvy- rather than Cvyi- is the Perfective-1a of 'go in', which is heard as núyⁿ-èrè-, not #núyⁿ1-èrè- (cf. dèw1-èrè- 'was covered').

10.1.3.4 nCv- and mCv- verbs

The four verbs in (xx1) are bisyllabic (and treated as such morphophonologically), but their first syllable consists just of /n/, followed by a homorganic consonant (/n/ or /d/). The initial /n/ may be high-toned (xx1.a) or low-toned (xx1.b). For some (historical) phonological background see §3.3.8.1.

(xx1)		stem	gloss	reduplicated Imperfective
	a.	ńné- ńdi-	'go' 'give'	ĭ-ini-m- ì-indi-m-
	b.	ìdέ-	'go up'	ì-indé-m-
	c.	(kìr ⁿ é) mbó-	'blow (nose)'	ù-úmbó-m-

For background on such initial nasal-stop and /nn/ clusters, see §3.3.8.1-2.

In the reduplicated forms in (xx1), a synchronic constraint requiring a vocalic nucleus in the reduplicant forces the stems in question to "grow" an initial /i, which is then copied in the reduplicant.

The 'go' verb has somewhat irregular stem-final vowel-quality alternations, in addition to the regular shift of final /e/ to /o/ in the Imperative and to /i/ in the Imperative Negative. We get /ɛ/ instead of /e/ in several forms, including those with a suffix containing the vowel /ɛ/ or /a/, but also in the unsuffixed Perfective and in the Perfective Negative (suffix /-ri-/). We get /i/ for expected /e/ in the positive and negative imperfective forms.

```
ńné-só-
                         Perfective-2
                         Experiential Perfect
    ńné tá:-só-
    ńné-mày<sup>n</sup>
                         Hortative
b. with /ɛ/
                         Verbal Noun
    ńnέ-ndέ
    'nnè-
                         unsuffixed Perfective (3Pl nn-ò)
    ńnέ-ὲrὲ-
                         Perfective-1a
                         Recent Perfect
    ńnέ-jè-
    nnε-rni-
                         Perfective Negative
c. with /i/
                         reduplicated Imperfective
   ì-nnî-m-
   ì-nnî:-sò-
                         reduplicated Progressive
                         Imperfective Negative
    ńnî-ŋò:-
    ńnî-r<sup>n</sup>á
                         Imperative Negative
d. with /o/
    ńnô
                         Imperative
```

10.1.3.5 Regular bisyllabic stems

All non-monosyllabic stems end in a short vowel. Bisyllabics are (C)vCv, (C)v:Cv, (C)vCCv, and rarely (C)v:CCv. The initial C position may be empty. Except in recent French or Fulfulde loanwords, there are tight restrictions on vocalism. For bisyllabics, there are two primary patterns, one where the second vowel is identical to or closely related to the first vowel, and another where the second vowel is /i/.

In the first pattern, the two vowels are either **identical non-high vowels**, hence $\{e...e, e...e, a...a, o...o, o...o\}$ (xx1.a) or have an **initial high vowel followed by a mid-height vowel** with the same roundness value, hence $\{i...e, i...e, u...o, u...o\}$ (xx1.b). There are therefore nine possible combinations of the seven vowel qualities, meaning that one cannot quite reduce the vocalisms to single autosegments.

```
(xx1) stem gloss

a. identical non-high vowels
bàyá- 'be cured'
téré- 'chop'
ké:ndé- 'do well'
kóyó- 'weep'
```

```
dògó- 'leave'

b. high vowel followed by non-high vowel
gùró- 'steal'
dùyó- 'insult'

'bǐndé- 'go back'
wĭsé- 'swing (arms)'
pîsé- 'spray'
tîmbé- 'lean on (cane)'
```

Verbal loans from Fulfulde normally end in $/\epsilon$ / regardless of other vowels in the stem, and unless further assimilated they often have vowel sequences in violation of the tight patterns illustrated in (xx2.a). The final $/\epsilon$ / is also typical of French loans (xx2.b), reflecting both an extention of the Fulfulde pattern to a wider range of loans, and the convenient fact that French verbs have several high-visibility forms ending in phonetic [e] or $[\epsilon]$ (written -er, -ez, -ait, etc.). A further source of non-canonical vowel sequences is the tendency for *iwv to shift to uwv, notably in the word for 'die', tiwé- or túwé- (xx4.c), where the usual reduplicated Imperfective tù-túwé-m- shows that the representation with /u/ now has the upper hand.

```
(xx2) stem gloss

a. súré-
pálé- 'pacify'
'box in'

b. gápé- 'win (match, election)'

c.. túwé- ~ tîwé- 'die'
```

The second major pattern for bisyllabic stem vowelism is with **final** /i/ (in some environments fluctuating with /u/). There are no clear restrictions on the quality of the preceding vowel, but $\{\epsilon \ o\}$ are more common than $\{e \ o\}$. The /i/ is subject to syncope/apocope after an unclustered sonorant, most systematically after a semivowel /w/ (dèwî- 'cover') or /y/ (î:-yî- 'stop').

```
(xx3) stem gloss

a. CvCi-
gàr<sup>n</sup>i- 'put'
jòŋi- 'cure'
bògi- 'perpetrate (crime)'
```

```
děw (</dèwi/)
                             'cover'
    dèni-
                             'tamp'
    dùŋi-
                             'put down'
                             'speak'
    tîŋî-
b. Cv:Ci-
    bă:rî-
                             'help'
                             'take away'
    gě:r<sup>n</sup>i-
                             'stop'
    î:-yî-
c. CvCCi-
                             'find'
    témbi-
                             'open wide'
    gòmbi-
    tîmbî-
                             'put lid on'
d. Cv:CCi-
    mɔ̃:ndi-
                             'gather'
```

(xx3) contains examples of **all seven vowel qualities** in the initial syllable. The contrast of timbi- 'put lid on' with timbé- 'lean on (cane)' shows how one cannot always predict the final vowel from the initial one.

10.1.3.6 Trisyllabic stems

Trisyllabic stems are often etymologically composite, but in individual cases it is difficult to demonstrate this in the absence of semantically and phonologically linked basic stems. As with bisyllabics, an important division is between stems ending in a non-high vowel and those ending in /i/.

(xx1) illustrates the vocalic patterns of verbs ending in a non-high vowel.

```
a. repeated {e o} vowels flanking medial high vowel

néngîyé-
bègîré-
yègîsé-
mònjúró-

b. (o...i...e)
dògîyé-
'look up at'
```

```
c. (o...o after syncope)

kómró- ~ kómúró- 'shell (peanuts)' (and other senses)

dŏmró- ~ dòmúró- 'shave around edges'

d. (initial high vowel, final mid-height vowel)

pîgîré- 'screw in'

wigîré- 'be dizzy'

lúgúró- 'ransack'

dùsúró- 'poke'
```

Patterns (xx1.a) and (xx1.d) are productive. In (xx1.a), an initial upper-midheight vowel {e o} occurs in the first and third syllables. The second syllable, in a metrically weak position, shows the corresponding rounded or unrounded high vowel. Cognate nominals sometimes show three identical vowels with no weakening of the medial, e.g. mónjórò 'dream (noun)' versus verb mònjúró-.

(xx1.b) and (xx1.c) are offshoots of (xx1.a). In (xx1.b), the pair of initial and final vowels is o...e rather than o...o or e...e, though at least the vowel-harmonic set $\{e \ o\}$ is maintained. This uncommon pattern is probably a reflection of frozen suffixal derivations, in this case with Mediopassive -y \acute{v} -.

In (xx1.c), the metrically weak /u/ between /m/ (alternating with /mb) and /r/ may syncopate to produce a CvCCv- stem.

In (xx1.d), the initial vowel is high $\{i u\}$ and the final vowel is upper-midheight, hence i...e or u...o (there are no attestations of i... ϵ or of u...o). The medial vowel is again, high, as in type (xx1.a).

We now consider trisvllabic verbs ending in /i/ (xx2) These are the trisyllabic counterparts of the bisyllabics ending in /i/ discussed above, above. In the morphologically uncomplicated examples (xx2.a), the initial syllable has any vowel quality other than /e/ or /o/. The middle syllable is basically /i/, but it is in a metrically weak position and is subject to sporadic rounding after a rounded vowel, especially /u/. When the final syllable begins with /y/, the resulting /...ivi/ is heard as /i:/ (xx2.b), but the underlying form is revealed by inflected forms such as the Imperative (îrîyà 'get up!'). In cases like téwsî-(xx2.c), the form is phonetically bisyllabic but given the unusual consonant cluster one suspects Syncope from /téwisi/. A more convincing argument could be made if the tone contour were rising, since /CvCiCv/ with LHH tone should appear as CČCC after Syncope, in contrast to CČCC from an unsyncopated true bisyllabic. This is clearly the case in the derived transitive verb daw-ri-'hide (sth)' from /dawi-ri-/, cf. dawi-yi- 'hide (oneself)'. In principle there should be underived verbs of the same type, but I can cite none from my lexical data.

(xx2) stem gloss

```
a. initial vowel other than {e o}, final /i/
     súmúr<sup>n</sup>i- ~ súmir<sup>n</sup>i-
                                        'rest'
                                       'scrub'
    púgúsî-
    îngîrî-
                                       'accompany'
    kémîr<sup>n</sup>î-
                                       'have fun'
     gònir<sup>n</sup>i-
                                       'go around'
    bógisi-
                                        'punch' (Fr boxer)
    dàgiri-
                                        'get ready'
b. final ...iyi heard as [i:]
                                        'get up'
    irîyî-
                                       'be tangled'
    nîndîyî-
c. syncope to CvCCi-
    téwsi-
                                       'tamp down'
```

For the trisyllables, if we combine the data in (xx1) and (xx2), we can say that initial-syllable $\{u\ i\}$ allow either a final i or a final $\{o\ e\}$ (agreeing in rounding), while non-high initial-syllable vowels allow us to predict the final-syllable vowel. The allowable sequences (disregarding occasional assimilatory rounding of the medial vowel) are therefore u...i., u...u., o, i...i., i...i., e...e., e...i., o...u., o, e...i., o...i., i, and a...i., Factoring out i as the basic medial vowel, these sequences have a partial correspondence to those allowed in bisyllabic verbs. However, one notes the absence of i and i o...i. and i orresponding to the (not very common) bisyllabic e...i and o...i, and of i a...i...a corresponding to bisyllabic a...a.

The preceding discussion of trisyllabic vowel sequences is somewhat idealized. However, the generalizations are valid for trisyllabic stems that are under no whiff of suspicion of including a derivational suffix. When we include stems that do seem to have such a suffix (even if frozen and no longer clearly segmentable), additional vowel sequences emerge. Consider the data in (xx3).

```
(xx3) stem gloss

a. jòrîyé- 'fight'

b. làrágî- 'scrub lightly' 'hold on chest'

c. bèndé-mî- 'hit hard'
```

pînjé-mî- 'wring'

In (xx3.a), an initial /o/ or /u/ co-occurs with a final /e/ (instead of expected /o/). Since the cognate nominal is jòriyè '(a) fight', one suspects that the verb is just following the vocalism of the noun. An additional factor is that -yé can be a derivational suffix (Mediopassive), and there are some examples of this suffix after a stem containing o or u, e.g. nóngî-yé- 'be caught (stuck)', tónjî-yé- 'be curved', and púndî-yé- 'be clumped'.

In (xx3.b), the first two syllables of the stem have the correct vocalism for CvCv- stems. The following -gî- syllable resembles a minor Causative allomorph -gî- which likewise does not alter the vocalism of a preceding CvCv-stem (§9.2.2). However, I know of no unsuffixed counterpart to làrágî, and ámbî- is recorded only in the sense 'apply (a compress)'.

The productive Causative suffix -mi- likewise does not alter the vocalism of a preceding CvCv-. The forms in (xx3.c) are phonologically compatible with causative morphophonology, though they do not function as causatives; compare pinjé '(water) splash', bèndi 'swim'.

10.2 Positive indicative AN categories

10.2.1 Perfective positive system (including perfect)

10.2.1.1 Unsuffixed Perfective with all-low toned stem

The unsuffixed Perfective consists morphologically of the stem without audible AN suffix, and with tones dropped. It is typical of clauses with a focalized constituent (other than the verb itself). In Nanga, the notion of "focalized" is quite broad, and the unsuffixed Perfective is very common when any preverbal constituent is present.

Sample paradigms are in (xx1). The verbs are dògó- 'leave', túwé- 'die', gùró- 'steal, rob', and yě:- 'come'. The latter shows that a Cv:- stem shortens its vowel before the consonant of a 1st/2nd person suffix.

(xx1)	category	'leave'	'die'	'rob'	'come'
	1Sg 1Pl 2Sg 2Pl	dògò-ỳ dògò-ỳ∴ dògò-ẁ dògò-ẁ∴	tùwè-y tùwè-ỳ∴ tùwè-w tùwè-ẁ∴	gùrò-y gùrò-ỳ∴ gùrò-w gùrò-ẁ∴	yè-y yè-ỳ∴ yè-w yè-ẁ∴
	3Sg/InanSg	dògè	tùwè	gùrè	yè:

3Pl/InanPl dòg-ò tùw-à gùr-ò y-ò:

The **3Sg form** ends in $\{e \ \epsilon\}$ in the examples in (xx1), but if the verb is nonmonosyllabic and ends in /i/, the 3Sg form also ends in /i/. This makes it difficult to segment the final vowel as a 3Sg suffix, and I transcribe the 3Sg form with the usual 3Sg $-\emptyset$ suffix, taking the final vowel as part of the stem. Nevertheless, it should be understood that this is a special form of the verb, which I call the E/I-stem, since it must end in $\{e \ \epsilon \ i\}$, whereas the basic form of the verb may end in a wider range of vowels.

The 3Pl, by contrast, ends in $\{o \circ a\}$. In other AN categories, the 3Pl is marked by a suffix, contrasting with 3Sg $-\emptyset$. I therefore hyphenate the final vowel of the 3Pl in the unsuffixed Perfective. Analytically, I take the 3Pl as an underlying back or low vowel (underspecified). When it follows a vowel that is already back or low $\{o \circ a\}$, the 3Pl suffix does not change the vowel quality. It also combines with preceding /i/ to give long /i:/ ('see'). The 3Pl suffix appears as /-à/ after /ɛ/, and combines with preceding /e/ to give /o:/.

The 3Sg and 3Pl forms of monosyllabic Cv:- stems are illustrated in (xx2). Of particular interest are diphthongal /àè/, /òè/, and /òè/ in the 3Sg, and /èà/ in the 3Pl.

(xx2)	gloss	bare stem	3Sg Perf	3Pl Perf
	a. regular Cv:- verbs			
	'reply'	sá:-	sàè-∅	$s-\grave{a}$: (= $s\grave{a}-\grave{a}$) etc.
	'go out'	gŏ:-	gòè-Ø	g-ò:
	'eat'	kó:-	k∂è-Ø	k-ð:
	'come'	yě:-	yè:-∅	y-ò:
	'lay out (mat)'	té:-	tè:-Ø	tè-à
	'avoid taboo'	tá: ⁿ -	tà ⁿ è ⁿ -∅	t-à ⁿ :
	b. irregular verb 'see'			
	'see'	yĭ:-	yì:-∅	y-à:

The verbs of type Cvy-, which were interpreted as /Cvyi/ in §10.xxx, above, have the unsuffixed Perfectives in (xx3). túyî- ~ tîyî- 'put down' can be treated as in (xx3.a) or as in (xx3.b). In all of the forms shown, a final /i/ syncopates or apocopates after /y/, hence in the 3Sg forms (heard as nuy^n - \varnothing etc.) as well as in the bare stem.

(xx3)	gloss	bare stem	3Sg Perf	3Pl Perf
	a. Cuyi 'hear' 'go in'	nùy ⁿ î- núy ⁿ î-	nùy ⁿ i-∅ nùy ⁿ i-∅	nùy ⁿ -ò nùy ⁿ -ò
	b. Ciyi 'send' 'put earth in'	tîyî- bìy ⁿ î-	tĭyĭ-Ø bĭy ⁿ i-Ø	tiy-à biy ⁿ -à

The 3Sg and 3Pl unsuffixed Perfective forms of nCv- and other bisyllabic stems are illustrated in (xx4). Here there are no final diphthongs. The 3Sg ends in $\{e \in o\}$, with /e/ replacing stem-final /o/. The 3Pl ends in $\{o \circ a\}$, with /a/ replacing /ɛ/. In (xx4.d), we see that when the stem ends in /i/, the 3Sg Perfective retains this /i/, while the 3Pl replaces it by copying the quality of the vowel of the first syllable (if this is a non-high vowel). If the first syllable has a high vowel this does not work; we get -à after first-syllable /i/ and -ò after firstsyllable /u/. The data in (xx4.e) should be taken with salt; the vowel sequences e...i and o...i are uncommon and may fluctuate with ε ...i and o...i, respectively. My assistant pronounced the first syllable of the bare stem and of the 3Sg Perfective of these verbs with $\{e \circ\}$, but switched to $\{\epsilon \circ\}$ in the 3Pl Perfective form, with consequences for the suffixal allomorphy. The three nCvstems in (xx4.f) have messy Perfective forms. 'Go' switches from /e/ in the basic stem to /ɛ/ in the 3Sg Perfective, but 3Pl nn-ò is consistent with underlying /ńné-/. For 'give', the 3Pl form nd-à 'they gave' (homophonous with 'they went up') suggests that the variant onset pronunciation îndî- for ńdî- is psychologically basic; compare the suffixal vocalism with that of tini- 'speak' (xx4.d).

(xx4)	gloss	bare stem	3Sg Perf	3Pl Perf
	a. bisyllabics w	ith repeated nor	n-high vowel	
	'touch'	táwá-	tàwè-∅	tàw-à
	'get'	bèré-	bèrè-∅	bèr-à
	'do well'	ké:ndé-	kè:ndè-∅	kè:nd-ò
	'weep'	kóyó-	kòyè-∅	kòy-ò

```
'leave'
                    dògó-
                                    dògò-Ø
                                                     dòg-ò
b. high vowel followed by non-high vowel
                    gùró-
    'steal'
                                    gùrè-Ø
                                                     gùr-ò
    'insult'
                    dùyó-
                                    dùyè-Ø
                                                     dùy-ò
    'return'
                    bindé-
                                    bindè-Ø
                                                     bind-ò
    'swing'
                    wisé-
                                    wisè-Ø
                                                     wis-à
                                    pisè-Ø
    'spray'
                   pîsé-
                                                     pis-ò
c. final /ɛ/ after a different vowel (loanwords)
    'win'
                    gáné-
                                    gànè-Ø
                                                     gàn-à
d. \{i \in a \circ u\} plus final /i
    'put'
                                    gàr"i-Ø
                                                     gàr<sup>n</sup>-à
                    gàr<sup>n</sup>i-
    'cover'
                    dèwi- [děw]
                                    dèwi-∅
                                                     dèw-à
    'put down'
                    dùŋi-
                                    dùŋi-Ø
                                                     dùŋ-ò
    'speak'
                    tiŋi-
                                    tiŋi-Ø
                                                     tiŋ-à
                                    bògi-Ø
    'perpetrate'
                   bògi-
                                                     bàg-à
e. \{e \ o\} plus final /i/ (uncommon, \{e \ o\} may vary with \{\epsilon \ o\})
    'cure'
                   jòŋî-
                                    jòŋì-Ø
                                                     jàŋ-à
    'tamp'
                   dèŋi-
                                    dèŋi-Ø
                                                     dèŋ-à
f. nCv- verbs
                    ńné-
                                    'nnὲ-Ø
                                                     nn-ò
    'go'
                                                     nd-à
    'go up'
                    ìdέ-
                                    ndε-Ø
    'give'
                   ńdî-
                                    'nďi-Ø
                                                     nd-à
```

Some **trisyllabic** examples are in (xx5). In (xx5.a), we see that both stem-final /e/ or /o/ in the bare stem split into a 3Sg with /e/ and a 3Pl with /o/. In (xx5.b-c), the final /i/ is replaced in the 3Pl Perfective by a copy of a first-syllable $\{o \circ a\}$, by /a/ after first-syllable /i/ or /ɛ/, and by /o/ after first-syllable /u/.

(xx5)	gloss	bare stem	3Sg Perf	3Pl Perf
	a. final {e o} 'cut up' 'dream' 'look up at' 'ransack'	yègîsé- mònjúró- dògîyé- lúgúró-	yègisè-Ø mònjùrè-Ø dògiyè-Ø lùgùrè-Ø	yègĭs-ò mònjùr-ò dògĭy-ò lùgùr-ò
	b. final {i}			
	'accompany'	îŋgîrî-	ĭŋgĭrĭ-Ø	iŋgir-à
	'have fun'	kémîr ⁿ î-	kèmìr ⁿ i-∅	kèmir ⁿ -à

```
'get ready'
                    dàgiri-
                                     dàgiri-Ø
                                                     dàgir-à
    'punch'
                                     bàgisi-Ø
                    bógisi-
                                                     bàgis-à
    'scrub'
                    púgúsî-
                                    pùgùsì-∅
                                                     pùgùs-ò
c. final {i} with derivational suffix
    'demolish'
                    wòró-gî-
                                    wòrò-gì-Ø
                                                     wòrò-g-ò
    'make go in' núy<sup>n</sup>5-mi-
                                    nùy<sup>n</sup>ò-mì-∅
                                                     nùy<sup>n</sup>ò-m-ò
    'make jump' péré-mi-
                                    pèrè-mì-∅
                                                     pèrè-m-ò
    'recover'
                    màllî-rî-
                                    màlli-rì-∅
                                                     màlli-ri-à
```

Sentence examples containing an unsuffixed Perfective as in (xx6).

```
yè-y
(xx6) a. yéŋir<sup>n</sup>i:
           yesterday
                           come.Perf.L-1SgS
            'It was yesterday [focus] that I came.'
       b. ăŋ
                      [yi:
                                  nέ-η]
                                                    sùyè-Ø
                                                                        mà
           who?
                      [child
                                  DefAnSg-Acc]
                                                    hit.Perf.L-3SgS
                                                                        Q
            'Who hit the child?'
           àrnáŋá
                        nà:-w<sup>n</sup>
                                                       mà
            where?
                        spend.night.Perf.L-2SgS
                                                       Q
            'Where did you-Sg sleep?'
```

10.2.1.2 Perfective-1a -èrè-, Perfective-1b -tì-

There are two suffixally marked Perfective forms. The suffixes are -èrè-, which I label Perfective-1a, and -ti-, which I label Perfective I-b. The stem has its lexical form (not the special form used in the unsuffixed Perfective) before both of these suffixes.

The two Perfective-1 suffixes compete most directly with the Perfective-2 with suffix -só-. (The unsuffixed Perfective is used when a non-verb constituent has at least some degree of focalization.)

Perfective-la -ere- is used with motion and stance verbs ('go', 'sit down'), with deadjectival inchoatives and other non-active intransitives, with a few low-impact transitives like 'forget', and optionally with transitive verbs of holding and wearing (which often contain Mediopassive -y1-/-yé-)

```
(xxx) Perfective-1a -èrè-
gloss stem Perfective
```

```
a. monosyllabic Cv:-
    'go out'
                      gŏ:-
                                          gŏ-èrè-
    'arrive'
                      dš:-
                                          dŏ-èrè-
    'spend night'
                      ná:-
                                          ná-èrè-
    'come'
                                          yě-èrè-
                      yě:-
b. nCv-
    'go'
                      ńné-
                                          ńnέ-èrè-
                      'ndέ-
                                          ndέ-èrè-
    'go up'
c. bisyllabic with final non-high vowel
    'die'
                      túwέ-
                                          túwé-èrè-
    'jump'
                      tómbó-
                                          tómbó-èrè-
    'run'
                      -ègéy
                                          yògó-èrè-
    'sit down'
                      éw-yé-
                                          éw-yé-èrè-
    'be finished'
                      đimέ-
                                          dimé-èrè-
d. Cvyî- and ...iyî-
    'enter'
                      núv<sup>n</sup>i-
                                          núv<sup>n</sup>-èrè-
    'stand'
                                          î:-y-èrè-
                      î:-yî-
e. bisyllabic with final /i/
    'become'
                      táŋî-
                                          táŋi-èrè- ~ táŋy-èrè-
    'lie down'
                      bivé-
                                          bivé-èrè-
    'be covered'
                      dèwi-
                                          dèwi-èrè-
```

-èrè- is not subject to Nasalization-Spreading even when the preceding stem ends in a nasalized syllabe. The main phonological problem is how VV sequences are treated (núy¹-èrè- 'went in' is the easy case, since this stem is C-final). With a Cv:- stem, the stem vowel is shortened and we get a surface bimoraic /vɛ/ sequence with both vowels clearly articulated, without harmonization (gó-èrè, etc.). With longer stems, there is fluctuation between bimoraic and monomoraic (contracted) articulations. In casual speech, the tendency is toward contraction in the direction of a single mora, preserving the suffixal ε clearly and at least a trace of the stem-final vowel. For example, in yɔgó-èrè- 'ran', the middle vowel is pronounced [ɔɛ], with the [ɔ] somewhat faint. The duration of this complex vowel is variable, but it often approaches the duration of a single non-complex short vowel like /ɛ/. When the stem-final vowel is /o/ (but not /ɔ/), the articulation may approach that of labialized C^w-ɛ, as in tómbó-èrè 'jumped', which can approach [tómbw-ère]. However, I think that tómbó-èrè is a better transcription, even phonetically, since the H-tone on

the second /o/ remains audible in the falling pitch of the medial complex vowel, and since consonantal labialization is not consistently heard. When the stem-final is /ɛ/, of course there is no shift in vowel quality at the boundary, and I often hear the duration as short, but the falling pitch is still there, e.g. dimé-èrè-'was finished'. Stem-final /e/ is clearly heard in the onset of the medial complex vowel, as in éwyé-èrè- 'sat'.

A form <code>èré</code> is attested as a kind of equivalent of the Perfective-1a suffix preceding the clause-final different-subject subordinator na 'then'. See ...gŏ: <code>ńné èré na</code> '(When hare) had gone out (completely), ...' in (xxx) in the sample text. In this example, <code>èré</code> is separated from gŏ: 'go out' by an intervening 3Sg subject pronoun.

Perfective-1b -ti- is used with most transitives, and with active intransitives denoting speech or thought.

(xxx) Perfective-1b -ti-

gloss	stem	Perfective
a. monosyllabic Cv	:-	
'shave'	ká:-	ká:-tì-
'catch'	wŏ:-	wŏ:-tì-
'see'	yĭ:-	yĭ:-tì-
b. nCv-		
'give'	ńdî-	ńdî-tì-
c. bisyllabic with fi	nal non-high vowe	el
'hit'	súyó-	súyó-tì-
'shout'	kóyó-	kóyó-tì-
'bathe'	dîyé-	dîyé-tì-
'cut (slice)'	késé-	késé-ti-
d. bisyllabic with fi	nal /i/	
'send'	tîyî- [tî:]	tîy-tì-
'cover'	dèwî- [děw]	děw-tì-
'speak'	tîŋî-	tîŋ-tì- ~ tîŋî-tì-
'think'	mă:ndî-	mă:ndî-tì-
'laugh'	màndi-	màndi-tì-

Many non-monosyllabic verbs that take -ti- (but not those that take -ere-) end in a lexical /i/, and this vowel is usually syncopated before -ti- where syllabically possible, i.e. after a single C: lá:r-ti- 'chased' from lá:ri-, pág-ti-

'tied' from págí-. Pronunciations with the /i/ are also possible, especially in careful speech.

The pronominal paradigms are given in (xxx). The second person forms of -ti- show optional assimilation of the /i/ to the following /w/. Homorganic sequences of vowel and semivowel (i.e. /iy/, /uw/) are pronounced as long vowels ([i:], [u:]). The 1Pl and 2Pl forms have their usual prolongation and [LHL] pitch of the final syllable.

(xxx)	category	form of -ti-	form of -èrè-
	1Sg	-tĭ-ỳ	-èrè-y
	1Pl	-tì̈-y∴ [tìîi]	-èrè-y∴ [èrèéèj]
	2Sg	-ti-w ~ -tù-w	-èrè-w
	2P1	-tì-w∴ ~ -tù-w∴	-èrè-w∴
	3Sg/Inan	-tĭ-∅	-èrè-∅
	3Pl	-tì-yà	-èr-à

The distinction between Perfective-1a -èrè- and Perfective-1b -ti- is neutralized under negation (§10.xxx, below). It is also generally neutralized in relative clauses, where Perfective Participial -sè replaces them (and Perfective-2-só-). With some effort it was possible to elicit relatives with -ti-, see §14.1.7.1, below, but these did not occur spontaneously.

10.2.1.3 Perfective-2 (-só-)

Another positive perfective-system form is expressed by a suffix H-toned suffix -só-. This form is common with verbs of perception ('see', 'hear'), but can be used with any verb and competes with the suffixal perfectives just described. It is the most common perfective form obtained in elicitation, with French past-tense cue sentences.

```
(xx1) a. ŋgî: yĭ:-só-Ø
1SgO see-Perf2-3SgS
'He/She saw me.'
b. bògòrô nŭy<sup>n</sup>-só-y
noise hear-Perf2-1SgS
'I heard a noise.'
```

The stem has its lexical final vowel, as before -ti- and several other suffixes. Examples are in (xx2).

```
Perfective-2
(xx2)
              gloss
                                 stem
         a. Cv:- monosyllabics
              'catch'
                                 wŏ:-
                                                     wŏ:-só-
              'go out'
                                 gŏ:-
                                                     gŏ:-só-
              'spend night'
                                 ná:-
                                                     ná:-só-
              'see'
                                 yĭ:-
                                                     yĭ:-só-
         b. nCv-
              'go'
                                 ńné-
                                                     ńné-só-
                                 ìdέ-
                                                     ndέ-só-
              'go up'
              'give'
                                 ńdî-
                                                     ńdi-só-
         c. bisyllabics ending in non-high vowel
              'hit'
                                 súyó-
                                                     súyó-só-
              'bathe'
                                 ďiyé-
                                                     ďiyé-só-
              'run'
                                 yògó-
                                                     yògó-só-
              'sit down'
                                 éw-yé-
                                                     éw-yé-só-
         d. bisyllabics ending in /i/
              'do'
                                                     kár<sup>n</sup>i-só-
                                 kár<sup>n</sup>i-
              'enter'
                                                     núy<sup>n</sup>-só-
                                 núy<sup>n</sup>i- [núj<sup>n</sup>]
              'think'
                                                     mă:ndî-só-
                                 mă:ndî-
```

The pronominal paradigm is (xx3). The 1Pl and 2Pl have their usual bell-shaped [LHL] pitch contour in the final syllable.

Representative stem combinations are in (xx4), with the Imperative and Progressive for comparison.

(xx4) gloss Imprt Perfective-2 Progressive

'see'	yĭ:	yĭ:-só-	yì-yî:-sò-
'understand'	pá:mò	pá:mî-só-	pà-pá:mè(:)-sò-
'hit'	súyô	súyó-só-	sù-súyò(:)-sò-
'hear'	nùy ⁿ i [nŭj ⁿ]	nŭy ⁿ -só-	nù-núy ⁿ -sò-
'tie'	págî	págî-só-	pà-págì(:)-sò-
'have fun'	kémîr ⁿ à	kémîr ⁿ î-só-	kè-kémìr ⁿ è(:)-sò-

For Progressive -sò- see 10.xxx. The Perfective-2 and the Progressive both have -so- suffix but differ phonologically as indicated in xx5. The 1Pl and 2Pl forms are disregarded here.

(xx5)	feature	Perfective-2	Progressive
	suffixal tone stem tone	H lexical	L H, HL, HHL
	stem-final vowel length	lexical	lengthened (in some cases optionally)
	final /i/ in long stems reduplication	no change none	shifts to non-high yes, if clause-initial

The opposition is illustrated in (xx6.a) versus (xx6.b), and (xx6.c) versus (xx6.d-e).

- (xx6) a. nàná yǐ:-só-y cow see-Perf2-1SgS 'I saw the cow.'
 - c. nàŋá yî:-sò-y
 cow see-Prog-1SgS
 'I see (=am seeing) the cow.'
 - c. sémbî-só-y sweep-Perf2-1SgS 'I swept.'
 - d. sè-sémbi(:)-sò-y Rdp-sweep-Prog-1SgS 'I am sweeping.'
 - e. dámbí sémbǐ(:)-sò-y courtyard sweep-Prog-1SgS

'I am sweeping the courtyard.'

10.2.1.4 Experiential Perfect 'have ever' (tá:-só-)

The suffix tá:- occurs before Perfective-2 -só- in the experiential perfect sense 'have (once, ever) done' and its negation 'have never done'. The positive form is common in questions ('have you ever ...?') like (xx1.a), but can also be used in statements ('I once ...' or 'I have ...'). The tá:- with its long vowel sounds prosodically like a chained verb rather than like the other perfective-system AN suffixes.

The corresponding negative is based on the Perfective Negative form tà:-rî-. The tone of the chained verb also drops in this combination, as in nnè tà:-rî-'has never gone' (xx1.b) from nné 'go'. This tone-dropping pattern could be used as an argument for suffixal status for tá:-, so that the chained verb is really part of the same word. In any event, it shows that the chaining is compound-like in its tightness.

```
(xxx) a. pà:ŋgŏ: yĭ: tá:-só-ẃ mà elephant see ExpPf-Perf2-2SgS Q 'Have you-Sg (ever) seen an elephant?'
```

```
b. [dúwá<sup>n</sup>sán y<sup>n</sup>ɛ̃] nnè tà:-rí-∅

[Douentza to] go ExpPf-Perf.Neg-3SgS

'He/She has never gone to Douentza.'
```

10.2.1.5 Recent Perfect ($j\hat{\epsilon}$ -)

The inflectable stem $j\hat{\epsilon}$ - indicates recency of the (positive) event denoted. It can often be translated with 'already' or 'just'. It often means 'have finished VP-ing'. It combines with the bare-stem form of the primary verb. It is written as a separate (chained) verb rather than as a suffix, because a preverbal subject pronoun (e.g. in nonsubject relatives) intervenes between the primary verb and $j\hat{\epsilon}$ (xx1.d).

```
(xxx) a. nàmá kúwó jè-y
meat eat.meat RecPf-1SgS
'I have already eaten (or: I have finished eating) the meat.'
```

```
b. nàr<sup>n</sup>á jè-∅
give.birth RecPf-3SgS
```

'She has (just) given birth.'

```
c. gir<sup>n</sup>£ jÈ-y∴
harvest RecPf-1PlS
'We have (already) harvested.'
```

```
d. nă: wàgàdi kó: i:n jè-sè gà
meal time.L eat 1SgS RecPf-Ppl.Perf in
'at the time when I finished eating'
```

The pronominal paradigm is (xx2). The dying-quail effect in the 1Pl and 2Pl results is expressed as [LHL] pitch on the final syllable.

```
(xx2) category Recent Perfect

\begin{array}{cccc} 1Sg & j\grave{\epsilon}\text{-y} \\ 1Pl & j\grave{\epsilon}\text{-y} \therefore & [d3\grave{\epsilon}\grave{\epsilon}\grave{\epsilon}] \\ 2Sg & j\grave{\epsilon}\text{-w} \\ 2Pl & j\grave{\epsilon}\text{-w} \therefore & [d3\grave{\epsilon}\grave{\epsilon}\grave{\epsilon}\grave{w}] \\ 3Sg/Inan & j\grave{\epsilon}\text{-}\emptyset \\ 3Pl & j-\grave{a} \end{array}
```

10.2.1.6 Reduplicated Perfective (Cý-)

An initial reduplication, with no further affixation, expresses another perfective category. It is not very common in my texts, but it is used in conversation in somewhat emphatic contexts, as in 'So-and-so has already eaten (plenty)' (and therefore doesn't need any more food).

The reduplicative syllable has high tone, while the following base is low-toned. The stem has the same vocalism as in the unsuffixed Perfective; it therefore ends in $\{e\ \epsilon\ i\}$ depending on the stem, and has the usual Perfective diphthongs with Cv:- stems. The pronominal-subject suffixes are the same as for the unsuffixed Perfective.

```
'sleep'
b. niy<sup>n</sup>ε-
                                           nî-nìy<sup>n</sup>ὲ-
    kósó-
                   'cough'
                                           kó-kòsè-
d. k5:-
                    'eat'
                                           kó-kòè-
                    'drink'
    nó:-
                                           nó-nòè-
    gŏ:-
                    'go out'
                                           gó-gòè-
    ká:-
                    'shave'
                                           ká-kàè-
```

Some examples are in (xx2).

```
(xx2) a. gèr<sup>n</sup>é ké-kè:ndè-∅ rainy.season Rdp-be.good.Perf.L-3SgS ná-nàmè-∅ céw

Rdp-be.ruined.Perf.L-3SgS all 'whether the rainy season has been good or has been bad'
```

```
b. nú-nùy<sup>n</sup>-∅ yè-rí-∅
Rdp-hear-3SgS see-Perf.Neg-3SgS
'He has (definitely) heard of (it), (but) he hasn't seen (it).'
```

A phonologically similar reduplicated stem, also without a final AN suffix but with telltale stem-final {a o} rather than {e ɛ i}, is the reduplicated Stative; see §10.3. For the reduplicated Imperfective, which has a low-toned reduplicative syllable and a high-toned stem, see §10.xxx, below.

10.2.2 Imperfective positive system

10.2.2.1 Imperfective (positive) $(-\dot{m}-\sim -\dot{\eta}\sim -\dot{\eta}\sim -\dot{\eta}$

A suffix $-\dot{m}$ - occurs before a nonzero pronominal subject suffix (all but 3Sg). For the zero 3Sg, we get $-\dot{\eta}$, and this is optionally realized as just $\dot{-}^n$, i.e. as vocalic nasalization and lengthening, with a final low-tone element that combines with the preceding high-tone to produce a final falling tone. The falling tone is also heard in the most common pronunciations of the 1Sg $(-\dot{m}-\dot{\varnothing})$ and 2Sg $(-\dot{m}-\dot{\omega})$.

Preceding the low-toned suffix, the stem has high tones, except that it preserves the initial low-tone element of $\{LH\}$ stems of three or more moras $(\mathring{C}\mathring{V}\mathring{C}\mathring{V}-, \mathring{C}\mathring{V}:\mathring{C}\mathring{V}-, \mathring{C}\mathring{V}\mathring{C}\mathring{V}-, etc.)$. In the Imperfective, the tone break in these $\{LH\}$ stems is after the first mora, not after the first syllable. The effect is that $\mathring{C}\mathring{V}\mathring{C}\mathring{V}-$ verbs have Imperfective $\mathring{C}\mathring{V}\mathring{C}\mathring{V}-$. Bimoraic $\{LH\}$ stems, i.e. $\mathring{C}\mathring{V}-$ and

CỳCý-, become all-high toned in the Imperfective, and are therefore not distinguished tonally from their high-toned counterparts.

Examples showing the Imperfective of stems **ending in a non-high vowel** are in (xx1). The stem vocalism, including the stem-final vowel, is identical in the bare stem and in the Imperfective, except for the irregular shift of $\acute{n}\acute{n}\acute{e}$ - 'go' to $\acute{n}\acute{n}\acute{e}$ - (xx1.b). Note the merger of all-high and {LH} tone contours in (xx1.b-c), versus the preservation of this tone-contour distinction in the trimoraic and longer stems in (xx1.d-f).

(xx1) Imperfective (stem ends in non-high vowel)

bare stem	Imperfective	gloss	
a. monosyllabic Cv	/:-		
ká:-	ká:-m̀-	'shave'	
gŏ:-	gó:-m̀-	'go out'	
té:-	té:-m-	'lay out'	
b. nCv-			
ńné-	ńnî-m-	ʻgo'	
'ndέ-	ńdé-m-	'go up'	
c. CvCv- (bimoraio	e)		
dògó-	dógó-m-	'leave'	
gùró-	gúró-m-	'steal'	
pîsé-	pîsé-m-	'spray'	
gáné-	gán ⁿ έ-m̀-	'win'	
túwé-	túwé-m-	'die'	
d. CvCCv-			
dòŋgó-	dóŋgó-m-	'rub on'	
tîmbé-	tîmbé-m-	'lean on'	
e. Cv:C(C)v-			
ké:ndé-	ké:ndé-m-	'do well'	
yŏ:ró-	yó:ró-m-	'stalk'	
f. trisyllabic			
yègîsé-	yègîsé-m-	'cut up'	
mònjúró-	mònjúró-m-	'dream'	
kónďiyé-	kóndîyé-m-	'droop, sag	

When the verb ends in /i/, the Imperfective has the same tone contours as above, except that the lexical distinction between {LH} and all-high is

respected not only in stems that have three or more moras, but also in the shorter stems. Therefore 'see' (xx2.a) along with 'hear', 'cover', and 'put down' (xx2.c), begin with a low-tone element in the Imperfective. Segmentally, the final /i/ is preserved in stems of two moras (xx2-c) and in CvCCi- (xx2.d), but in stems with three or more vocalic moras, including Cv:Cv-, it is replaced by a non-high vowel (xx2.e-f). The non-high vowel in this event is copied from the initial syllable vowel (if non-high). If the stem has no non-high vowel, we get /ɔ/ after /u/, and /ɛ/ after /i/. The Imperfective of Causative -mi- is -mɛ́-mi regardless of the vocalism or syllable count of the input stem (xx2.g)

(xx2) Imperfective (stem ends in high vowel)

bare stem	Imperfective	gloss		
a. monosyllabic Ci:-				
yĭ:-	yî:-m-	'see'		
b. nCi-				
ńdî-	ńdi-m-	'give'		
c. CvCi-				
nùy ⁿ î- [nǔj ⁿ]	nŭy ⁿ -m̀-	'hear'		
núy ⁿ î- [núj ⁿ]	núy ⁿ -m̀-	'go in'		
dèwî-	dèwî-m-	'cover'		
dùŋî-	dùŋî-m-	'put down'		
kár ⁿ î-	kár ⁿ i-m-	'do'		
d. CvCCi-				
tîmbî-	tîmbî-m-	'cover, close'		
témbî-	témbî-m-	'find'		
dàmbi-	dămbî-m-	'push'		
e. Cv:C(C)i-				
bă:rî-	bă:rá-m-	'help'		
tî:rî-	tî:ré-m-	'pour over'		
pέ:ndî-	pé:ndé-m-	'stimulate'		
f. trisyllabic with final /i/				
kémîr ⁿ î-	kémîr ⁿ é-m-	'have fun'		
púgúsî-	púgúsó-m-	'scrub'		
bègiri-	bègiré-m-	'winnow'		
g. Causative -mi-				
kó:-mî-	k5:-mέ-m̀-	'feed'		

```
péré-mî- péré-mé-m- 'make jump'
núy<sup>n</sup>5-mî- núy<sup>n</sup>5-mé-m- 'make go in'
```

The pronominal paradigm is given in (xx3).

(xx3) Imperfective

```
category
                 form
1Sg
                 -m̀-∅ ~ -m-i
1P1
                                            (or: -m-iy)
                 -mi-y
                 -\dot{m}-^{\rm w} ~ -m-\dot{u}
2Sg
2P1
                 -mù-w
                                            (or: -m-ùw)
                 -\dot{\eta} \sim :-\dot{\eta} (i.e. nasalized long vowel, falling tone)
3Sg/Inan
3P1
                 -m-è
```

In the 1Sg and 2Sg forms, the assistant wavered between syllabic variants (1Sg -mi, 2Sg -mw) and apocopated variants (1Sg -m, 2Sg -mw). In the shortened forms, 2Sg -mw was heard with slight rounding at the transition from the stem-final vowel to the nasal; in effect, the nasal is "prelabialized." The corresponding plurals have long vowels, but the pitch of the final syllable is level low.

In the 3Sg, I hear variably a final velar nasal (L-toned), or a weakened version of the same involving final falling-toned stem-vowel with terminal nasalization.

Analytically, one could argue that a **floating low tone** (L) occurs at the end of this imperfective stem, hence /STEM-L-PRONOMINAL/. If so, the L is realized phonetically on the pronominal suffix if the latter is audible (all of the nonzero suffixes consist of a Cv syllable or of a nasal, so these suffixes all have at least one mora). In the 3Sg, if the -i) suffix weakens to just vocalic nasalization, the low-tone component is expressed as the final part of a falling tone.

Examples of the Imperfective are in (xx3).

- (xxx) a. àsú⇒ wórî wárá-m̂-Ø every.day farming farm-Impf-1SgS 'I farm (=work in the field) every day.'
 - b. tê: dógó-m-∅ tea leave-Impf-1SgS

'I will leave (the) tea.'

```
c. èsî bîré-ŋ (or: bîrê:-n)
good work-Impf.3SgS
'He/She works well.'
```

```
d. [nà: mòsî] dàŋî-ŋ (or: dàŋî:-n) [meal.L bad] cook-Impf.3SgS 'He/She cooks badly.'
```

```
e. [ndo ngá] sigé-m-Ø [house.L Dem.InanSg.Loc] go.down-Impf-1SgS 'I go down (= lodge) in this house.'
```

More examples of the suffixes are given in the following section on the reduplicated Imperfective, which differs from the simple Imperfective only in adding an initial reduplication.

10.2.2.2 Unsuffixed Imperfective absent

There is no Imperfective form of the Jamsay type with no (segmental) Imperfective suffix, i.e. with just a pronominal-subject suffix added directly to the verb (with perhaps a tone change on the verb).

However, in disjunctions of clauses with future-time reference (as well as past-time reference), the unsuffixed Perfective (or a homophonous form) is in use. See §7.2.2, above.

10.2.2.3 Reduplicated Imperfective (Cv-)

The reduplicated Imperfective has an initial (C)v-reduplication. The consonant is present unless the verb begins with a vowel. The vowel is a copy of the first-syllable vowel of the stem. The base (i.e. the stem minus the reduplication) is identical (segmentally and tonally) to the unreduplicated Imperfective just described above.

```
b. {H}-toned bisyllabic

'jump' tómbó- tò-tómbó-mè-

'hit' súyó- sù-súyó-mè-

'sit down' éw-yé- è-éw-yé-mè-
```

c. {LH}-toned bisyllabic shifting to all-high in Imperfective 'leave' dɔgɔ́- dɔ-dɔ́gɔ́-m̀-

```
d. longer {LH}-toned stem preserving initial low tone 'laugh' màndî- mà-măndî-m- 'dream' mònjúró- mò-mònjúró-mò-
```

A sample paradigm ('sit down') is (xx2).

(xx2) 'sit down' (reduplicated Imperfective)

```
      1Sg
      è-éw-yé-mì

      1Pl
      è-éw-yé-mì-y

      2Sg
      è-éw-yé-mì ~

      2Pl
      è-éw-yé-mù-w

      3Sg
      è-éw-yé-n) ~
      ~ è-éw-yê:-n

      3Pl
      è-éw-yé-m-è
```

The reduplicated Imperfective can be distinguished from the reduplicated Perfective by the presence of high tones on the stem, and by the low tone on the reduplicant.

10.2.2.4 *Progressive* (-sò-)

A verb form used in progressive contexts involves L-toned AN suffix -sò-, following a stem-form ending in a low- or falling-toned vowel. A short stem-final vowel is also lengthened. This is a historical reflex of an original Imperfective morpheme (*-m- or *-ŋ-) added to the stem before *sò 'have'. I have heard variant pronunciations with -ŋ- in the Progressive: gô:-ŋ-sò- 'be going out', variant of gô:-sò-. The fuller form suggests a two-word construction similar to that with Imperfective -ŋ (-m) plus bù- 'be', see §15.2.2.2. However, the fuller form is no longer in common use, at least in my assistant's dialect. See also the discussion of -m-sè in §15.2.3, below.

When the verb is clause-initial (i.e. in the absence of an object or other preverbal constituent), reduplication ($\mathbb{C}\tilde{v}$ -) is present.

Progressive forms of stems **ending in a non-high vowel** are in (xx1). In monosyllabics and short bisyllabics, the lexical distinction between all-high and {LH} tones is neutralized, and we get a {HL} contour on the (simple) Progressive stem. In Cv:Cv- bisyllabics and in all trisyllabics, the tone of the first syllable of the (simple) Progressive stem is lexically determined; the final two syllables have {HL}. 'Go' shifts from /e/ to /i/, as it does in the Imperfective (xx1.b). I hear consistent lengthening of the stem-final vowel in the nCv- stems (xx1.b), but not in the other stems.

(xx1) Progressive (stem ends in non-high vowel)

gloss	bare stem	Progressive	
-		simple	reduplicated
a. monosyllabic	Cv:-		
'eat'	kó:-	kô:-sò-	kò-kô:-sò-
'lay out'	tέ:-	tê:-sò-	tè-tê:-sò-
'drink'	nă:-	nô:-sò-	nò-nô:-sò-
b. nCv-			
ʻgo'	ńné	ńnì:-sò-	ì-ńnì:-sò-
'go up'	'nnέ	ńdè:-sò-	ì-ńdè:-sò-
c. bisyllabic			
'hit'	súyó-	súyð(:)-sò-	sù-súyò(:)-sò-
'leave'	dògó-	dógò(:)-sò-	dò-dógò(:)-sò-
ʻjump'	tómbó-	tómbò(:)-sò-	tò-tómbò(:)-sò-
'rub on'	dòŋgó-	dóŋgò(:)-sò-	dò-dóŋgò(:)-sò-
'stalk'	yŏ:ró-	yŏ:rò(:)-sò-	yò-yó:rò(:)-sò-
d. trisyllabic			
'cut up'	yègîsé-	yègîsè(:)-sò-	yè-yègisè(:)-sò-
'dream'	mònjúró-	mònjúrò(:)-sò-	mò-mònjúrò(:)-sò-
'droop, sag'	kóndîyé-	kóndíyè(:)-sò-	kò-kóndíyè(:)-sò-

Stems ending in /i/ are illustrated in (xx2). I hear consistent lengthening of the final vowel in the one nCi- verb (xx2.b), and also in the lexically {LH}-toned CvCCi- verbs (xx2.d), which express the final <HL> characteristic of the Progressive on the stem-final syllable (the first syllable carrying the lexical initial low tone). In the other verbs, I heard sporadic but inconsistent lengthening of the stem-final vowel, except that the /Cvyi/ (Cvy) verbs were always heard as just Cvy (xx2.c). Segmentally, the stem-final /i/ is retained in (xx2.a-c), but stems that have three or more vocalic moras, including Cv:Cv-, replace the final /i/ by a non-high vowel. If the stem has an initial non-high

vowel, its vowel quality is copied on the stem-final, otherwise we get stem-final /ɔ/ after /u/ and /ɛ/ after /i/.

(xx2) Progressive (stem ends in /i/)

gloss	stem	Progressive		
C		simple	reduplicated	
a. monosyllabic C		^ \	× ^ ×	
'see'	yĭ:-	yî:-sò-	yi-yi:-sò-	
b. nCi-				
'give'	ńnî	ńďi:-sò-	ì-ńdî:-sò-	
8-11				
c. CvCi-				
'do'	kár ⁿ í-	kár ⁿ i(:)-sò-	kà-kár"i(:)-sò-	
'cover'	dèwî-	déwi(:)-sò-	dè-déwì(:)-sò-	
'go in'	núy ⁿ i- [núj ⁿ]	nûy ⁿ -sò-	nù-nûy ⁿ -sò-	
'hear'	nùy ⁿ i- [nǔj ⁿ]	nûy ⁿ -sò-	nù-nûy ⁿ -sò-	
d. CvCCi-				
'grill'	sîmbî-	sîmbì(:)-sò-	sì-sîmbì(:)-sò-	
'laugh'	màndi-	màndì:-sò-	mà-mànđi:-sò-	
'push'	dàmbi-	dàmbî:-sò-	dà-dàmbî:-sò-	
pusii	damor-	damoiso-	da-damoiso-	
e. Cv:C(C)i-				
'pour over'	tî:rî-	tî:rè(:)-sò-	tì-tî:rè(:)-sò-	
'stimulate'	pέ:ndi-	pέ:ndε(:)-sò-	pè-pé:ndè(:)-sò-	
'help'	bă:rî-	bă:rà(:)-sò-	bà-bă:rà(:)-sò-	
f. trisyllabic				
'have fun'	kémîr ⁿ î	kémîr ⁿ è(:)-sò-	kè-kémîr ⁿ è(:)-sò-	
'winnow'	bègîrî-	bègîrè(:)-sò-	bè-bègirè(:)-sò-	
'scrub'	púgúsi-	púgúsð(:)-sò-	pù-púgúsò(:)-sò-	

The pronominal paradigm is (xxx). The 1Pl and 2Pl have F-tone, while all other forms have L-tone.

```
Progressive
(xx3) category
       1Sg
                     -sò-ỳ
       1P1
                     -sò-ỳ∴
                                     [sòóòj]
       2Sg
                     -sò-ẁ
       2P1
                     -sò-ẁ∴
                                     [sòóòw]
       3Sg/Inan
                     -sò-∅
       3P1
                     -s-è
```

Examples are in (xx4).

- (xxx) a. bîrá bîrè(:)-sò-ỳ work(noun) work-Prog-1SgS 'I am working.'
 - b. móndi màndi:-sò-ỳ laughter laugh-Prog-1SgS'I am laughing.'
 - c. kèmirⁿé kémirⁿè(:)-s-è fun have.fun-Prog-3PlS 'They are having fun.'
 - d. tómbì tómbò(:)-s-è jump(noun) jump-Prog-3PlS 'They are jumping.'

10.2.3 Negation of indicative verbs

Except for statives (including the progressive), which have a special Stative Negative -ndó-, the positive indicative inflections correspond to a Perfective Negative -rı̂- or to an Imperfective Negative -ŋɔ̂:-.

10.2.3.1 Perfective Negative -ri-

For most purposes, there is a single negative perfective-system form, that with suffix -ri- added directly to the stem. This corresponds to all of the major positive perfective-system forms: unsuffixed Perfective, Perfective-1a -èrè-, Perfective-1b -ti-, Perfective-2 -só-, and Recent Perfective jè-.

```
(xx1) a. nă: kò:-rî-∅ meal eat-Perf.Neg-3SgS 'He/She hasn't eaten (yet).'
b. ĭ:n gò:-rî-y 1SgS go.out-Perf.Neg-1SgS 'I didn't go (= haven't gone) out.'
c. kò-kámâ yè-rî-y anything see-Perf.Neg-1SgS 'I didn't see anything.'
```

The plural-subject (1Pl, 2Pl, 3Pl) forms are somewhat distinct from the singular-subject forms. For 1Pl and 2Pl, the differences are limited to tone contours. For 3Pl, the differences are more substantial. The plural forms are covered at the end of this section, after the description of the singular forms.

In the singular-subject forms, the stem drops tones to all-low before the high-toned $-\hat{r_1}$ -.

If the verb ends in a **non-high vowel** $\{e \in a \circ o\}$, the only change to the stem is the tone-dropping. Some examples are in (xx2).

```
Perfective Negative gloss
(xx2)
             stem
      a. monosyllabic Cv:
                                                   'remain'
             bě:-
                            bè:-rî-
                                                   'go out'
             gŏ:-
                            gò:-rî-
                                                   'bring'
                            jè:-rî-
             iě:-
                                                   'shave'
                            kà:-rî-
             ká:-
                                                   'come'
             yě:-
                            yè:-rî-
      b. nCv
             ìdέ-
                            ndε-rî-
                                                   'go up'
      c. bisyllabic
             yògó-
                            yògò-rî-
                                                   'run'
             tómbó-
                            tòmbò-rî-
                                                   'iump'
             ké:ndé-
                            kè:ndè-ri-
                                                   'make well'
      d. longer stems
             mònjúró-
                            mònjùrò-rî-
                                                   'dream'
             pîgîré-
                                                   'screw in'
                            pigirè-ri-
```

Verbs ending in /i/ in the bare stem form must replace it by a non-high vowel before -rî-.

The nonlexical vowel that appears before -ri- in these cases is in most cases **copied** from the non-high stem vowel if there is one. This is the case with **bisyllabic** stems, where all nonhigh vowel qualities are attested (xx3.a). If the verb has only high vowels {u i}, the "copied" vowel is /ɔ/ or /ɛ/, respectively, agreeing with the lexical high vowel in backness and rounding (xx3.b). In addition, the few verbs of the shapes Cuy- and Ciy- (Ci:) add a non-high vowel before -ri- (xx3.c). For the nCi- verbs, the shape is nCè- before -ri-, harmonizing with the displaced final vowel (xx3.d). The verb 'see' is slightly irregular in having a short vowel before -ri- (xx3.e).

```
(xx3)
                stem
                                     PerfNeg
                                                               gloss
        a. final /i/, bisyllabic, lexical non-high vowel present
                                                               'urinate'
                ónjî-
                                     -în-cinc
                págî-
                                     pàgà-ri-
                                                               'tie'
                dèwî ([děw])
                                    dèwè-ri-
                                                               'cover'
                dèni-
                                     dènè-r<sup>n</sup>i-
                                                               'tamp'
                                     jònò-r<sup>n</sup>1-
                                                               'cure'
                jòŋî-
        b. final /i/, bisyllabic, no lexical non-high vowel present
                                                                'sing'
                nùŋi-
                                     nùŋɔ̀-rʰi-
                                                               'put down'
                dùnî-
                                     dùnò-r<sup>n</sup>i-
                                                               'speak'
                tîŋî-
                                     tinè-r<sup>n</sup>i-
                                     timbè-ri-
                                                                'put lid on'
                tîmbî-
        c. Cuy-
                nŭy<sup>n</sup>-
                                     nùy<sup>n</sup>ò-r<sup>n</sup>i-
                                                               'hear'
                núv<sup>n</sup>-
                                     nùy<sup>n</sup>ò-r<sup>n</sup>i-
                                                                'go in'
                túy-
                                     tùyò-rî-
                                                               'put down'
                tîy-
                                     tiyè-ri-
                                                               'send'
        d. nCi-
                                                               'give'
                ńdi-
                                     ndε-ri-
                ńnî-
                                     ìnὲ-r<sup>n</sup>i-
                                                                'go'
        e. irregular
                yĭ:-
                                     yὲ-rî-
                                                                'see'
```

In **trisyllabic and longer** stems, there are further examples involving $\{\varepsilon \circ a\}$ and a small number with $\{e \circ\}$. In most such verbs there is also an intervening medial high vowel that is disregarded in determining the stem-final vowel before -ri- (xx4.a). There are also some additional stems that have nothing but high vowels, where /ɔ/ appears in connection with /u/ and /ɛ/ appears in connection with /i/ (xx4.b). Some Fulfulde loanwords show final /ɛ/ where one would expect /ɔ/, reflecting the fact that the usual form in which Fulfulde verbs are borrowed into Dogon (and Songhay) languages is with final /ɛ/ (xx4.c).

```
(xx4)
                                   PerfNeg
                stem
                                                           gloss
        a. final /i/, trisyllabic, lexical non-high vowel
                táná-ndí-
                                   tànàndà-ri-
                                                           'ignite (fire)'
                bègîrî-
                                   bègirè-ri-
                                                           'winnow'
                wàgiri-
                                   wògùrò-rî-
                                                           'take grain out of mortar'
                kónúr<sup>n</sup>i-
                                   kònùr<sup>n</sup>ò-r<sup>n</sup>i-
                                                           'gnaw (bones)'
                dèŋi-r<sup>n</sup>i-
                                   dèŋi-r<sup>n</sup>è-r<sup>n</sup>i-
                                                           'un-stick flour'
        b. final /i/, no lexical non-high vowel present
                kúgîrî-
                                   kùgùrò-rí-
                                                           'skim'
                îriyî- [îri:]
                                   ìriyè-ri-
                                                           'get up'
        d. Fulfulde loanwords
                yúrúmî-
                                   vùrùmè-r<sup>n</sup>i-
                                                           'pity'
                sógînî-
                                   sòginè-r<sup>n</sup>i-
                                                           'take to pasture at night'
```

Causative -mî- becomes -mè- before -rî- (the result is -mè-rnî-), regardless of the vocalism of the preceding stem (xx5).

```
(xx5)
               stem
                                       PerfNeg
                                                                   gloss
               bàr<sup>n</sup>á-ndíyé-mí-
                                       bàrnà-nđiyè-mè-rni-
                                                                    'make red'
               jèmé-mî-
                                       jèmè-mè-r<sup>n</sup>i-
                                                                    'make black'
               tómbó-mi-
                                       tòmbò-mè-r<sup>n</sup>i-
                                                                    'cause to jump'
                                                                    'seat, have sit'
               éw-yé-mî-
                                       èw-yè-mè-r<sup>n</sup>i-
```

Stems ending in minor Causative allomorphs, the Reversive suffix, or the Mediopassive suffix, are treated for inflectional purposes just like underived stems of the same (often trisyllabic) shape.

The suffixal /r/ of the Perfective Negative suffix is subject to **Nasalization-Spreading** (§3.xxx), hence $-r^n$ i- after a nasal syllable (one containing a

nasalized vowel, or beginning with a prevocalic nasal consonant or nasalized sonorant) (xx6.a). The Cuy- verbs are expanded to Cuyv- before -rı̂- (see above), and since two of them begin with /n/ they expand as nuy^n in this inflection and therefore induce Nasalization-Spreading on the suffix (xx6.b).

(xx6)		stem	gloss	Perfective Negative
	a.	nàr ⁿ á- tá: ⁿ - níy ⁿ é- nó:-	'bear (child)' 'spread fingers' 'sleep' 'drink'	nàr ⁿ à-r ⁿ i- tà: ⁿ -r ⁿ i- nìy ⁿ è-r ⁿ i nò:-r ⁿ i-
	b.	núy- nǔy-	'go in' 'hear'	nùy ⁿ ð-r ⁿ 1- nùy ⁿ ð-r ⁿ 1-

We now consider the pronominal-subject paradigm of -rı́-. In the forms with 3Sg- \emptyset , 1Sg-y, and 2Sg-w, the stem is tone-dropped as illustrated above, and the suffix is -rı́-. In the 1Sg and 2Sg, the suffixal semivowel extends the high-tone of the AN suffix. 2Sg /-rı́-w⁄/ assimilates to -rú-w⁄ (phonetic [rú:]). This gives us the singular-subject paradigm in (xx7).

(xx7)	category	PerfNeg form
	1Sg	-r i -ý
	2Sg	-rú-ẃ
	3Sg	-rî-∅

To complete the exposition of the singular-subject forms, sample paradigms are given in (xx8). The verbs in question are identified in the key at the end of the list.

(xx8)	1Sg	2Sg	3Sg	gloss
	yè-rî-ý	yè-rú-ẃ	yè-rî-∅	'see'
	yè:-rî-ý	yè:-rú-ẃ	yè:-rî-∅	'come'
	gò:-r i -ý	gò:-rú-ẃ	gò:-rî-∅	'go out'
	'nnὲ-r ⁿ i-ý ⁿ	nnè-r ⁿ ú-w ⁿ	'nnὲ-r ⁿ î-∅	ʻgo'
	ndè-ri-ý	ndè-rú-w	ìdὲ-rî-∅	'go up'
	ndè-rî-ý	ndè-rú-w	ìdὲ-rî-∅	'give'
	bàyà-rî-ý	bàyà-rú-ẃ	bàyà-rî-∅	'be cured'
	gò-ndò-rî-ý	gò-ndò-rú-ẃ	gò-ndò-rî-∅	'take out'
	kùwò-mè-r ⁿ i-ý ⁿ	kùwò-mè-r ⁿ ú-ẃ ⁿ	kùwò-mè-r ⁿ i-∅	'burn'

```
tànà-ndà-rí-ý tànà-ndà-rú-ẃ tànà-ndà-rí-\varnothing 'ignite' gisè-rí-\checkmark gisè-rú-\checkmark gisè-rí-\varnothing 'throw'
```

The **1Pl and 2Pl** show vowel-length and tonal changes, but are otherwise segmentally identical to the corresponding singulars (with 1Sg -y and 2Sg -w). If not already long, the **stem-final vowel is lengthened** before the Perfective Negative suffix. The tone contour of the stem is {LH} realized as <LH> (rising) tone on a monosyllabic stem, LH on a bisyllabic, LLH on a trisyllabic, etc. The final syllable consisting of the suffix -rî- plus the pronominal-subject suffix is heard with **low tone**, in contrast to the high tone of the singular-subject forms For example, 1Sg yè-rî-ý 'I did not see' corresponds to 1Pl yě:-rì-y 'we did not see'. I do not hear 1Pl -rì-y or 2Pl -rù-w suffixes themselves as longer than their singular counterparts.

The special prosody of the 1Pl and 2Pl forms is roughly consistent with the analysis of other 1Pl and 2Pl forms in terms of **dying-quail intonation** ($\S xxx$). In the Perfective Negative, both the suffixal syllable and the stem-final syllable in the 1Pl and 2Pl forms are modified vis-à-vis the corresponding singulars. My usual transcription is with \dot{x} after the transcription of the singular form, which brings out the morphological structure. It is understood that the vowel-lengths and pitch contours are adjusted by phonetic implementation rules. Examples below of the 1Pl (xx9) and 2Pl (xx10) show the phonetic realizations in brackets.

(xx9)	1Pl (1Sg plus ∴)	1Pl phonetic	gloss
	a. monosyllabic		
	yè-rî-y∴	[jě:rì:]	'see'
	yè:-rî-y∴	[jě:rì:]	'come'
	gò:-rî-y∴	[gŏ:rì:]	'go out'
	tà:-rî-y∴	[tǎ:rì:]	'shoot'
	b. bisyllabic		
	ṅ̀nὲ-r ⁿ i-y∴	[nné:r ⁿ i:]	ʻgoʻ
	ǹdὲ-rī-y∴	[ndé:ri:]	'go up'
	ǹdὲ-rı̂-y∴	[ndé:ri:]	'give'
	bàyà-rî-y∴	[bàjá:rì:]	'be cured'
	gi̇̀sè-rî-y∴	[gisé:ri:]	'throw'
	gò-ndò-rî-y∴	[gòndó:rì:]	'take out'
	c. trisyllabic		
	kùwò-mè-r ⁿ i-y∴	[kùwòmé:r ⁿ i:]	'burn'
	tàŋà-ndà-rî-y∴	[tàŋàndá:rì:]	'ignite'

```
(xx10)
            2Pl (2Sg plus :.)
                                     2Pl phonetic
                                                              gloss
        a. monosyllabic
                                                              'see'
            yè-rú-w∴
                                     [jě:rù:]
                                                              'come'
            yè:-rú-w∴
                                     [jě:rù:]
                                                              'go out'
            gò:-rú-w∴
                                     [gŏ:rù:]
            tà:-rú-w∴
                                                              'shoot'
                                     [tă:rù:]
        b. bisyllabic
            'nnὲ-r<sup>n</sup>ú-w<sup>n</sup>∴
                                     nné:rnù:
                                                              'go'
            ndè-rú-w∴
                                     ndέ:rù:
                                                              'go up'
            ndè-rú-w∴
                                     [ndé:rù:]
                                                              'give'
            bàyà-rú-w∴
                                     [bàjá:rù:]
                                                              'be cured'
                                                              'throw'
            gisè-rú-w∴
                                     gisé:rù:
                                     [gòndó:rù:]
                                                              'take out'
            gò-ndò-rú-w∴
        c. trisyllabic
                                                              'burn'
            kùwò-mè-r<sup>n</sup>ú-w<sup>n</sup>∴
                                     [kùwòmé:rnù:]
                                                              'ignite'
                                     [tàŋàndá:rù:]
            tànà-ndà-rú-w∴
```

In these 1Pl and 2Pl forms, note the following. The final syllable (the suffixal syllable) is long and low-pitched. Its length is arguably due to the monophthongization of /iy/ and /uw/ to [i:] and [u:], rather than to the phonetics of the dying-quail intonation. However, lengthening also occurs in other 1Pl and 2Pl forms with comparable pitch contours in other AN inflections (§3.xxx). The low pitch of the ultimate contrasts sharply with its final high tone in the singulars. The penult has its vowel lengthened. In addition, a rising pitch contour is realized on the stem. Specifically, monosyllabic stems have rising pitch, bisyllabics are LH, and trisyllabics are LLH. That is, the high pitch is limited to the penult, but occupies the entire penult except in monosyllabic stems (which preserves the {L}-tone contour of the stem in its initial mora). The dying-quail intonation elsewhere is characterized by (at least) [LHL] pitch, usually just on the final syllable. Here in the Perfective Negative, the [LHL] pitch is distributed over the entire word.

The **3Pl form** shows a radical divergence from the rest of the paradigm, since the Perfective Negative marker (elsewhere -ri-) combines with 3Pl to give the portmanteau -ndú. The preceding stem is **low-toned** as in the singular forms.

There are also some segmental changes in the final stem vowel in the 3Pl, suggestive of an underlying suffixal form /-àndú/ with an initial /a/, but only in certain combinations. Cross-cutting this is a process by which final {e ɛ} are rounded to {o o}, presumably under the influence of the suffixal /u/ (though this cannot be attributed to a productive phonological rule). Moreover, in some cases it appears that /-àndú/ or /-ndú/ is added to a form of the stem whose

vocalism is consistent with that of the singular-subject Perfective Negative forms (i.e. before -ri-) rather than that of the bare stem, to the extent the two can be distinguished.

In (xx11), note especially the following. Regarding the possible allomorph /-àndú/, note 3Pl Cèà-ndú- for the Cé:- stems 'shine' and 'sprout' in (xx11.a), and the shift of final /ɛ/ to /a/ in most bisyllabic and longer stems, e.g. 'obtain' and 'take handful'. The /ɛ/ that shifts to /a/ is specifically the /ɛ/ of the Perfective Negative stem, not that of the bare stem form, in the cases of 'give' (xx11.b), the irregular verb 'see' (xx11.c), and 'fear' and 'take handful' in (xx11.e), all of which have final /i/ in the bare stem form. The alternative rounding and backing of {e ɛ} to {o o} is relatively systematic in the case of /e/ > /o/, being obligatory in monosyllabics, see 'come' and 'bring' in (xx11.a), and common in spite of considerable phonetic variation (from the same assistant) in the stem-final /e/ of longer stems like those at the end of (xx11.d). The verb 'go' (xx11.b) shows the shift of /ɛ/ to /ɔ/ in 3Pl nnò-ndú (based on Perfective Negative nnè-rⁿi-, already shifted from lexical /e/ to /ɛ/)

(xx11)	stem	PerfNeg (Sg)	PerfNeg 3Pl	gloss
a. n	nonosyllabic (Cv:-		
	gŏ:-	gò:-rî-	gò:-ndú	'go out'
	jě:-	jè:-rî-	jò:-ndú	'bring'
	yě:-	yè:-rî-	yò:-ndú	'come'
	ká:-	kà:-rî-	kà:-ndú	'shave'
	té:-	tè:-rî-	tèà-ndú-	'sprout'
	kέ:-	kè:-rî-	kèà-ndú	'shine'
	kó:-	kò:-rî-	kð:-ndú	'eat (meal)'
b. r	ıCv-			
	ńné-	nnè-r ⁿ i-	nnò-ndú	ʻgo'
	'ndέ-	ndε-ri-	ǹdà-ndú	'go up'
	ńdi-	ndè-ri-	ǹdà-ndú	'give'
c. '	see' (irregula	r)		
	yĭ:-	yè-rî-	yà:-ndú	'see'
d. t	oisyllabic or lo	onger, ending in i	non-high vowel	
	bàyá-	bàyà-ri-	bàyà-ndú	'be cured'
	yògó-	yògò-rî-	yògò-ndú	'tun'
	gò-ndó-	gò-ndò-rî-		'take out'
	bèré-	bèrè-ri-	bèrà-ndú	'obtain'
	gisé-	gisè-ri-	gisè-ndú ~ gisò-ndú	'throw'

```
sîgé-
                                                                    'go down'
                     sìgè-ri-
                                        sìgò-ndú ~ sùgò-ndú
                     pèrè-ri-
                                        pèrè-ndú ~ pèrò-ndú
                                                                    'jump (off)'
     péré-
     mèngiré-
                     mèngirè-ri-
                                        mèngirò-ndú
                                                                    'make into balls'
e. bisyllabic or longer, ending in high vowel
     ú:-yî-
                     ù:-yè-rî-
                                        ù:-yà-ndú
                                                                    'fear'
     nŭy<sup>n</sup>-
                     nùy<sup>n</sup>ò-r<sup>n</sup>i-
                                        nùy<sup>n</sup>ò-ndú
                                                                   'hear'
     pέnjî-
                     pènjè-ri-
                                        pènjà-ndú
                                                                    'take handful'
                     kùwò-mè-r<sup>n</sup>i-
                                        kùwò-mò-ndú
                                                                    'burn'
     kúwó-mî-
     táná-ndi-
                                                                    'ignite'
                     tànà-ndà-rî-
                                        tànà-ndà-ndú
```

10.2.3.2 Experiential Negative -tà:-rî-

For the sense 'have never VERB-ed', Perfective Negative -rî- is added to the verb tá:-, which is chained to a substantive VP (§10.1.2.xxx). A peculiarity of this combination is that the preceding verb is also tone-dropped.

```
(xx1) a. [bàmàkó nnè] tà:-rî-y
[Bamako go.L] ExpPerf.L-PerfNeg-1SgS
'I have never gone to Bamako.'
b. [yî: sùyò] tă:-rǐ-y
[child hit.L] ExpPerf-PerfNeg-1PlS
'We have never struck a child.'
```

10.2.3.3 Recent Perfect Negative

The Recent Perfect jè- (§10.2.1.xxx), which cannot easily be distinguished from the 'finish VP-ing' construction, is usually negated by the simple Perfective Negative -rî-. In other words, in the negative it is not usually distinguished from other subtypes of the perfective (positive). Therefore the usual negative answer to the question nă: kố: jè-w mà 'have you-Sg already eaten a meal?' is simply kò:-rî-y 'I have not eaten'.

However, it is possible to elicit examples with Recent Perfect jè-followed by the Perfective Negative suffix to produce jè-rî-. The Perfective Negative form jè-rî- is usually understood to mean 'have not finished (VP-ing)', as in nă: kố: jè-rî-y 'I have not finished eating the meal'. The 3Pl subject form is nă: kố: jà:-ndú 'they have not finished eating'.

The tone-dropping usually enforced by Perfective Negative -rî- on a preceding stem does not apply to the stem preceding jè-, another indication that jè- patterns phonologically like a chained verb.

10.2.3.4 Imperfective Negative -ŋɔ:-

The all-purpose negation of imperfectives ('doesn't VP', 'isn't VP-ing', 'will not VP') is formed by adding the Imperfective Negative suffix. It has a basic form -ŋɔ̂:- with long vowel for third person categories (3Sg -ŋɔ̂:-Ø, 3Pl -ŋ-ɛ̂:), but the first and second person forms are based on -ŋɔ̂-, which is followed by the suffixal semivowel (1Sg -ŋɔ̂-yʰ, 2Sg -ŋɔ̂-wʰ). The phonetic long vowel in 1Pl -ŋɔ̂-yʰ.: and 2Pl -ŋɔ̂-wʰ.: is attributable to the dying-quail intonational effect associated with the pronominal suffix, i.e. -ŋɔ̂- plus -y:. (1Pl) or -w:. (2Pl).

The pronominal-subject **paradigm** of $-\eta\delta$:- is given in (xx1), with 'run' as the example in the right-hand column. The \therefore symbol indicates dying-quail intonation in the 1Pl and 2Pl, which is expressed by prolongation and [LHL] pitch on the suffixal syllable.

(xx1)	category	ImpfNeg		'run'
	1Sg	-ŋɔ̞-yʰ		yágá-ŋà-y ⁿ
	1Pl	-ŋò-y ⁿ ∴	[ŋὸόὸϳ]	yágá-ŋà-y ⁿ ∴
	2Sg	-ŋð-w ⁿ		yágá-ŋà-w ⁿ
	2PL	-ŋò-w ⁿ ∴	[ŋɔ̇ɔ́òw]	yágá-ŋà-w ⁿ ∴
	3Sg	-ŋò:-∅		yágá-ŋà:-∅
	3Pl	-ŋ-ὲ:		yágà-ŋ-è:

The stem preceding the Imperfective Negative suffix has a full or partial tone-contour overlay controlled by the suffix. Verbs ending in /i/ differ from those ending in non-high vowels.

If the bare stem of the verb ends in a non-high vowel, and has no more than two vocalic moras (Cv:, CvCv, CvCCv), the lexical tone is erased and the stem has H-tone before the suffix (xx2).

(xx2) Imperfective Negative (light stem ending in nonhigh vowel)

```
stem ImpfNeg gloss

a. monosyllabic {LH}-toned
```

```
gŏ:-
                gó:-ŋò:-
                                 'go out'
                nố:-ŋồ:-
                                 'drink'
    nó:-
    bě:-
                bé:-ŋò:-
                                 'remain'
                jé:-ŋò:-
                                 'bring'
    jě:-
                                 'catch'
    wŏ:-
                wó:-ŋà:-
                                 'come'
    yě:-
                yé:-ŋò:-
  {H}-toned
                                 'shoot'
    tá:-
                tá:-ŋò:-
                                 'sow'
    tó:-
                tó:-ηò:-
b. CvCv
  {LH}-toned
                yágá-ŋà:-
                                 'run'
    yògó-
    dògó-
                dógó-ŋò:-
                                 'leave'
                bárá-ŋò:-
                                 'gather'
    bàrá-
  {H}-toned
    péré-
                péré-ŋò:-
                                 'jump (off)'
    tóró-
                                 'pound'
                tóró-ŋò:-
c. nCv
  {LH}-toned
    ìdέ-
                ńdέ-ŋò:-
                                 'go up'
d. CvCCv, lexical {LH} tone
  {LH}-toned
                   gómbó-ŋò:-
                                          'pull in (stomach)'
    gòmbó-
    gùnjó-
                   gúnjó-ŋò:-
                                          'harvest (peanuts)'
  {H}-toned
                                          'bend, flex'
    tónjó-
                  tónjó-ηὸ:-
    éw-yé-
                   éw-yé-ηὸ:-
                                          'sit down'
```

Stems ending in a nonhigh vowel with more than two vocalic moras, i.e. Cv:Cv, CvCvCv, and longer, distinguish lexical {H} from {LH} tones. As in the bare stem, {LH}-toned verbs have an audible L-toned onset (first mora of Cv:Cv, first syllable of trisyllabic and longer stems) (xx3).

(xx3) Imperfective Negative (heavy stem ending in nonhigh vowel)

```
stem ImpfNeg gloss

a. Cv:Cv
{LH}-tone
yŏ:ró-
yŏ:ró-
th}-tone
sé:ré-
sé:ré-
sé:ré-
yŏ:ró-
yŏ:ró-
yŏ:ró-
yŏ:ró-
yŏ:ró-
yŏ:ró-
yŏ:ró-yò:-
'encounter'
```

```
b. trisyllabic
{LH}-tone
mònjúró- mònjúró-nò:- 'dream'
{H}-tone
pîgîré- pîgîré-ŋò:- 'screw in'
```

Stems ending in a high vowel /i/ respect the lexical $\{H\}$ versus $\{LH\}$ distinction more strongly than do the stems ending in a non-high vowel. In particular, CvCv stems make the tone-class distinction. If the stem is prosodically light, with no more than two vocalic moras, the final /i/ is also preserved, so the stem used in the Imperfective Negative is indistinguishable from the bare stem (xx4).

(xx4) Imperfective Negative (light stem ending in /i/ or /y/)

```
ImpfNeg
                                      gloss
     stem
a. Cvy
  {H}-toned
                   núy<sup>n</sup>-nà:-
    núy<sup>n</sup>-
                                       'go in'
                   túy-ŋɔ:-
                                       'put down'
     túy-
b. nCi
  {H}-toned
     ńnî-
                   ńnî-ŋò:-
                                       'go'
     ńdi-
                   ńdi-ŋò:-
                                       'give'
c. CvCi
  {LH}-toned
     gàr<sup>n</sup>i-
                   gàr<sup>n</sup>i-ŋɔ:-
                                       'put (in)'
     dèwi-
                   dèwi-ŋò:-
                                       'cover'
  {H}-toned
     kár<sup>n</sup>i-
                   kár<sup>n</sup>i-ŋɔ:-
                                       'do, make'
d. CvCCi
  {LH}-toned
                      màndî-ŋò:-
     màndi-
                                                 'laugh'
                                                 'dig'
     gànji-
                      gànjî-ŋò:-
  {H}-toned
                      sándî-ŋɔ:-
     sándi-
                                                 'pray'
     túngi-
                      túŋgi-ŋò:-
                                                 'rest (one's head)'
```

Quite irregular are two perception verbs (yǐ:- 'see' and nuyn- 'hear'), which have a low-toned stem and a high-toned form of the Imperfective Negative suffix. In addition, the stem is shortened to Cv- and the suffix has a short vowel even in the third person (xx5). For 'see', the initial /y/ is nasalized to /p/ by a unique case of **Backward Nasalization**, and the vowel is irregularly backed and rounded to /u/, perhaps by analogy to the corresponding form of 'hear'.

```
(xx5) stem ImpfNeg gloss

a. Cvy
{LH}-toned
nuyn- nu-ŋó- 'hear'

b. Cv:
{LH}-toned
yi:- nu-ŋó- 'see'
(alongside regular variant yi:-ŋò:-)
```

In heavy stems ending in /i/, the {LH} versus {H} lexical tone distinction is again audible. However, in contrast to the prosodically light stems ending in the same vowel, these heavier stems shift the final /i/ to a non-high vowel before the Imperfective Negative suffix -ŋɔ̂:-. The quality of this vowel is copied from the (other) non-high vowel, if any, in the stem (xx6).

(xx6) Imperfective Negative (heavy stem with non-high vowel and final /i/)

```
ImpfNeg
    stem
                                   gloss
c. Cv:Ci, lexical {LH} tone
  {LH}-toned
                                            'dare'
                    dă:rá-ŋò:-
    dă:rî-
                                            'take, deliver'
    gě:rni-
                    gě:r<sup>n</sup>é-ŋà:-
  {H}-toned
                   ká:rá-ŋò:-
                                            'rip'
    ká:rî-
                                            'authorize'
    tó:ri-
                   tó:ró-ŋò:-
a. trisyllabic, lexical {LH} tone, final /i/, non-high vowel in stem
  {LH}-toned
                                            'sift'
    bègiri-
                    bègiré-nò:-
                   jèmbîré-ηò:-
                                            'hit off-center'
    jèmbîrî-
                                            'awaken'
    wùró-gi-
                    wùró-gó-ŋò:-
    gàgiri-
                    gàgirá-ŋò:-
                                            'rub into the ground'
  {H}-toned
    tigi-ri-
                   tigiré-ŋò:-
                                            '(griot) call out names'
```

```
kégîrî- kégîré-ŋò:- 'trim off'
```

If the stem has only high vowels $\{i\ u\}$, a final non-high vowel (harmonizing with the lexical vocalism in backness and rounding) is conjured up before the suffix (xx7). In my limited data, the vowel appears as $\{\epsilon\ o\}$ rather than $\{e\ o\}$, suggesting that the suffixal /o/ favors a stem-final vowel of the same harmonic type, other things being equal.

(xx7) Imperfective Negative (heavy stem with high vowels and final /i/)

stem	ImpfNeg	gloss
a. with ε î:-rî- yúrúmî-	î:-ré-ŋò:- yúrúmé-ŋò:-	'cause to stop' 'pity' (cf. noun yúrúmè)
b. with o pú:rî- súmúr ⁿ î-	pú:ró-ŋò:- súmúr ⁿ ó-ŋò:-	'frisk' 'rest'

10.2.3.5 Progressive Negative -sò≡ndò-

Progressive -sò- does have a special negative form -sò=ndò-, ending with a L-toned version of the conjugatable Stative Negative clitic =ndó-.

```
(xx1) kò-kámâ yĩ:-sò=ndò-ỳ
anything see-Progr=StatNeg-1SgS
'I don't see anything.'
```

The stem has the same form as in the positive Progressive. The -sò- suffix is invariant in this construction, except that the 3Pl subject is marked both with -sò- and with the following clitic, resulting in -s- $\dot{\epsilon}$ =nd- $\dot{\epsilon}$. In the 1Pl and 2Pl forms, -sò=ndò- plus dying-quail intonation is realized as [-sòóńdòò-], as both the penult and the final are lengthened, and as the [LHL] pitch of the dying-quail effect is split into [LH] on the penult and L on the ultimate. The full paradigm is (xx2).

(xx2)	category	form	with 'see'	with 'hit'
	1Sg 1Pl	-sò≡ndò-ỳ -sò≡ndò-ỳ∴	yî:-sò≡ndò-ỳ yî:-sò≡ndò-ỳ∴	súyò:-sò≡ndò-ỳ súyò:-sò≡ndò-ỳ∴
	2Sg	-sò≡ndò-ẁ	yî:-sò=ndò-ẁ vî:-sò≡ndò-ẁ	súyɔ:-sò=ndò-ẁ

```
      2Pl
      -sò≡ndò-ŵ..
      yî:-sò≡ndò-ŵ..
      súyò:-sò≡ndò-ŵ..

      3Sg
      -sò≡ndò-Ø
      yî:-sò≡ndò-Ø
      súyò:-sò≡ndò-Ø

      3Pl
      -s-è≡nd-è
      yî:-s-è≡nd-è
      súyò:-s-è≡nd-è
```

10.3 Pronominal paradigms for non-imperative verbs

10.3.1 Subject pronominal suffixes

The forms used to index subject pronominal category on verbs and some other predicators are given in (xxx). The third person forms are used for inanimate as well as animate referents.

(xxx)	category	suffix
	1Sg 1Pl	-y -y∴
	2Sg 2Pl	-W -W∴
	3Sg 3Pl	-∅ [see below]

The third person forms are more difficult to analyse. The 3Sg can generally be taken as $-\emptyset$, but in the positive imperfective we get $-\eta$ (or just $-\eta$, i.e. vowel nasalization) rather than the expected $\#-m-\emptyset$.

The 3Pl category is expressed by adding $-\hat{\epsilon}$ to Imperfective -m-, by zero in the unsuffixed perfective (where 3Pl is therefore identical to 3Sg), by a shift of a preceding back vowel {o o} to ϵ (preserving the vowel length), by shift of a preceding ϵ to a, and by apparent suppletion in the case of Perfective Negative -ri- becoming -ndó-. Using 1Sg and 1Pl for comparison (2Sg -w and 2Pl -w: are exactly parallel), note the 3Sg versus 3Pl alternations across various AN categories in (xxx).

(xxx)	category	1Sg	1Pl	3Sg	3P1	suffix
	unsuffixed Perf	-ỳ	-ỳ∴	(è\è\ì)-∅	-à\ò\à	_
	Perfective-1a	-èrè-ỳ	-èrè-ỳ∴	-èrè-∅	-èr-à	-èrè-
	Perfective-1b	-tì-ỳ	-tì-ỳ∴	-tìi-∅	-tì-yà	-tì-

Perfective-2	-só-ý	-só-ý∴	-só-∅	-s-έ	-só-
Recent Perfect	-jὲ-ỳ	-jὲ-ỳ∴	-jὲ-Ø	-j-à	-jὲ-
Perf.Neg	-rî-ý	-rî-ý∴	-rî-∅	-ndó	-rî-
Imperfective	-m̀-(i)	-m-i ∴	-ỳ	-m-ὲ	-m̀-
Progressive	-sò-ỳ	-sò-ỳ∴	-sò-∅	-s-È	-sò-
Impf.Neg	-ŋò-y	-ŋà-y ⁿ ∴	-ŋò:-∅	-ŋ-è:	-ŋბ:-
Past	-bὲ-v	-bὲ-v∴	-bὲ-∅	-b-à	-bè-

10.3.2 Nonhuman versus 3Sg subject

There is no distinction between animate 3Sg and inanimate third person subjects in verbal inflection.

10.3.3 Vowel-semivowel interactions of AN and pronominal suffixes

Perfective-1b -ti- combines with 2Sg -w as -tù-w more other than as -tù-w. Likewise, Perfective Negative -ri- has a 2Sg form -rú-w. The 2Pl forms have the same vowel quality.

Quasi-verb bù- 'be' combines with 1Sg -y as bì-ỳ or bù-ỳ. The 1Pl form has the same vowel quality.

What I write as uw and iy are, in syllable-final position, pronounced [u:] and [i:].

These alternations are of course assimilations of a short high vowel to a following nonhomorganic semivowel, followed by Monophthongization. See §3.xxx.

10.3.4 Tones of pronominal-subject suffixes

Pronominal-subject suffixes used in indicative categories lack intrinsic tones. The nonzero suffixes (except 3Pl) consist of a sonorant, whose tone is acquired by spreading from the preceding vowel.

This does not apply to the modal inflectional categories, most of which have syllabic and tone-bearing suffixes (e.g. Prohibitive -rá).

10.4 Stative form of verbs (reduplicated and unreduplicated)

10.4.1 Stative positive

Verbs of stance (sitting, standing, etc.), holding and carrying, and certain others like 'be closed', are used in both active ('sit down') and stative ('be sitting, be seated') contexts. The regular indicative conjugations (perfective and imperfective) are used in the active sense. In the stative sense, denoting a static position, the (reduplicated or unreduplicated) Stative inflection is used. It makes no aspectual distinctions, falling outside of the perfective and imperfective systems that apply to active verbs.

The reduplicated Stative has an initial Cv- reduplication (high-toned), followed by a low-toned stem ending in /o/ or /a/. For those stems that transparently contain a -yv- suffix in the active forms, this suffix is omitted in the reduplicated Stative. The criterion for transparency is syllabic in nature; the stem minus the -yv- formative must be CvCv- or CvC- (xx1.b-c). Original *Cv-yv- with just a *Cv- stem have arguably become unsegmentable, and in any event require the second syllable to meet the bisyllabic shape requirement, so the /yv/ segment remains in the Stative stem (xx1.d). This might also be argued to be the case with some of the *Cv:-yv- stems in (xx1.e), but 'fear' in particular suggests that in these verbs the initial Ci:- or Cu:- splits into Ciya-and Cuwa-, in which case there is no need to assume that the *-yv- formative is present in the Stative.

```
(xx1)
                              Imperative
                                             Perfective-1a reduplicated Stative
             gloss
        a. bisyllabic stem (unsegmentable)
                                             tîsé-èrè-
             'lean (on)'
                              tîsô
                                                             tí-tisò-
        b. bisyllabic stem plus Mediopassive -vv-
             'lean back'
                                             disî-èrè-
                                                             dî-disò-
                              ďisî-yô
             'kneel'
                              túŋi-ynà
                                             túŋi-y-èrè-
                                                             tú-tùŋò-
             'hold'
                              ágî-yà
                                             ágí-y-èrè-
                                                             á-àgà-
             'hold onto'
                              kúmbî-yò
                                             kúmbi-èrè-
                                                             kú-kùmbò-
             'carry on back' bàmbi-yâ
                                             bàmbi-èrè-
                                                             bá-bàmbà-
        c. CvC- stem plus Mediopassive -vv-
             'sit'
                              éw-yò
                                             éw-yé-èrè-
                                                             é-èwò-
             'squat'
                              sów-yò
                                             sów-yé-èrè-
                                                             só-sòwò-
                                             téw-yé-èrè-
             'perch'
                              téw-yò
                                                             té-tèwò-
        d. Cvyv- stem (with frozen Mediopassive *-yv-)
                                                             nî-niy<sup>n</sup>à-
             'sleep'
                              nîy<sup>n</sup>â
                                             niy<sup>n</sup>-èrè-
```

```
'lie down'
                         bîyô
                                          bîyé-èrè-
                                                             bî-biyò-
e. Cv:yv- stem (with frozen Mediopassive *-yv-)
     'be closed'
                         pî:-y<sup>n</sup>à
                                          pî:-y<sup>n</sup>-èrè-
                                                             pî-piy<sup>n</sup>à-
     'stand'
                         î:-yà
                                          î:-y-èrè-
                                                             î-iyà-
     'fear'
                         ú:-yà
                                          ú:-y-èrè-
                                                             ú-ùwà-
```

The pronominal-suffix **paradigm** is (xx2), with 'be sitting' as the example. In the 3Pl, final /o/ shifts to /e/. This could be taken as a slightly irregular assimilation to the following -yè. It could also be a case of double 3Pl conjugation (as in the negative paradigm, below). However, a stem-final /a/ does not shift: uwà-yè 'they fear'.

(xx2)	category	Stative	'be sitting'
	1Sg	-y	èwò-ỳ
	1Pl	-y∴	èwò-y∴
	2Sg	-w	èwò-ẁ
	2Pl	-w∴	èwò-w∴
	3Sg/Inan	-∅	èwò-∅
	3Pl	-yὲ	èwè-yὲ (èw-è-yὲ)

The full reduplicated form of the stem given above is used when no location is overtly specified. When the verb is preceded by a locational adverb, the reduplicated segment is optionally (but usually) omitted (xx3).

- (xx3) a. ŋgà-gá èwò-∅ there sit.Stat-3SgS 'He/She is sitting over there.'
 - b. ŋgà-gá ¡yà-yɛ̂ there stand.Stat-3PlS 'They are standing there.'
 - c. [nídó gó] bìyò-Ø
 [house in] lie.down.Stat-3SgS
 'He/She is lying down over there.'

10.4.2 Stative Negative (≡ndó-)

The negative forms are based on Stative Negative clitic =ndó-. The paradigm, and the forms for 'not be sitting', are in (xx1). The final syllable of the verb stem becomes high-toned before the clitic-suffix complex (in addition to the dying-quail intonation). With 'sit', the stem-final /o/ shifts to /e/ before the clitics, regularly in the 3Pl and sporadically in the other cases.

(xx1)	category	Stative Negative	'not be sitting'	
	1Sg 1Pl 2Sg 2Pl	=ndó-ý =ndo-y∴ =ndó-ẃ =ndo-w∴	èwò≡ndó-ý èwó≡ndo-y∴ èwò≡ndó-ẃ èwó≡ndo-w∴	[èwòóndòóòj] [èwóndòóòw]
	3Sg/Inan 3Pl	≡ndó-∅ ≡nd-έ	èwò≡ndó-∅ èwè≡nd-έ	

10.5 Post-verbal temporal particles

10.5.1 Past clitic ($\equiv b\epsilon$ -)

The Past suffix $=b\varepsilon$ - carries over the preceding tone (subject to further tone rules due to a following pronominal suffix). We therefore get e.g. $-\dot{m}=b\dot{\varepsilon}$ - with low tone but e.g. $-\dot{r}i=b\dot{\varepsilon}$ - with high tone. The Past clitic combines with certain AN forms of the verb, and is itself conjugated for pronominal subject. The combinations with preceding AN suffixes are given in (xxx).

(xx1)	AN category	AN suffix	AN + Past
I	positive		
	Imperfective	-m̀-	-m̀≡bὲ-
	Progressive	-sò-	-sò≡bὲ-
	unsuffixed Perfective	(zero)	=bὲ - (Past Perfect)
	Perfective-1b	-tĭ-	-tî≡bέ-
	Perfective-1a	-èrè-	-èrè≡bè-
	Perfective-2	-só-	-só≡bέ-
	Recent Perfect	-jè-	-jὲ≡bὲ-
1	negative		
	perfective	-rî-	-rî≡bέ-

imperfective -ŋɔ̂:- [pronominal]≡bε̂-

=bε- is followed by the usual subject pronominals. The 3Pl form is **=**b-a. The paradigm is given in two forms, based on the preceding tone. The two tonal forms are not distinguished in the 1Pl and 2Pl, which have their bell-shaped pitch of LHL type.

(xx2)	category	form with = bε-		
		after high tone	after low tone	
	1Sg	≡bέ-y	≡bὲ-y	
	1Pl	≡bὲ-y∴	≡bὲ-y∴	
	2Sg	≡bέ-w	≡bὲ-w	
	2P1	≡bὲ-w∴	≡bὲ-w∴	
	3Sg	≡ bέ-∅	≡bὲ-∅	
	3Pl	≡b-á∴	≡b-à	

There are some cases of double conjugation, whereby the pronominal subject is marked both on the Past clitic and on the preceding verb. This is especially true for the 3Pl.

10.5.1.1 Past Imperfective (positive and negative)

A **Past Imperfective** (positive) is formed with the complex $-\dot{m}=b\dot{\epsilon}$. Here $-\dot{m}$ is equatable with the Imperfective suffix -m seen above. This leaves $=b\dot{\epsilon}$ as the specifically Past morpheme. The 3Sg form is $-\dot{m}=b\dot{\epsilon}-\emptyset$, and the 3Pl form is $-\dot{m}=b-\dot{a}$.

- (xxx) a. àmây¹-àmây¹ ká kár¹i-m̀≡bɛ̀-ẁ∴ how?-how? there do-Impf≡Past-2PlS 'What did you-Pl use to do there (= about it)?'
 - b. kiyaw [ndo ngá] sígé-m=bè-y first [house.L Dem.InanSg.Loc] go.down-Impf=Past-1SgS 'I used to go down (= lodge) in this house.'
 - c. bàmàkó ńnî-m̀≡bè-∅ Bamako go-Impf≡Past-3SgS 'He/She used to go to Bamako'

d. bàmàkó ńni-m=b-à
Bamako go-Impf=Past-3PIS
'They used to go to Bamako'

The **Past Imperfective Negative** combines the Imperfective Negative suffix $-\eta\delta$:- with the inflected form of $=b\dot{\epsilon}$ -. In the 3Pl, both $-\eta\delta$:- and $=b\dot{\epsilon}$ - are conjugated, so the 3Pl category is doubly marked. For other nonzero categories, i.e. first and second persons, $=b\dot{\epsilon}$ - is conjugated, and my assistant occasionally also conjugated $-\eta\delta$:-, but it seems that the singly-conjugated variants in e.g. (xx2.a) are preferred.

```
(xx2) a. kiyaw
                      [ndò
                                       ŋgá]
           first
                      [house.L
                                      Dem.InanSg.Loc]
            sîgé-η∂:≡bὲ-y
       or: sigé-ηò-y≡bè-y
           go.down-Impf.Neg(-1SgS)=Past-1SgS
           'I did not use to go down (= lodge) in this house.'
       b. sigé-ŋò:-∅≡bè-∅
           go.down-Impf.Neg-3SgS≡Past-3SgS
           'He/She did not use to go down.'
            sîgé-ŋɔ:≡bè-w∴
       or: sîgé-ŋò-w≡bὲ-w∴
           go.down-Impf.Neg(-2PIS)≡Past-2PIS
           'You-Pl did not use to go down.'
       d. sîgé-ŋ-è:-∅≡b-à
           go.down-Impf.Neg-3PlS≡Past-3PlS
           'They did not use to go down.' (double conjugation required)
```

10.5.1.2 Past forms of stative quasi-verbs ('be', 'have')

The stative **quasi-verbs** bù- 'be' (existential-locational) and sò- 'have', which do not make the perfective/imperfective aspecutal distinction, take -m=bè- for past time reference (bù-m=bè-, sò-m=bè-). The -m- is unusual for these quasi-verbs, but suggests a crypto-connection (made overt in the Past only) between these statives and the Imperfective category of active verbs.

The special negative forms of 'be' and 'have' can directly take $=b\dot{\epsilon}$ -.

(xxx) gloss regular form Past

For 3Pl, 'not have' takes doubly conjugated Past forms sè-ndé≡b-á 'did not have', ngé≡b-á 'were not', and sè-m≡b-à 'had'. (Positive) '(they) were' is singly conjugated: bù-m≡b-à.

For (positive) 'have', the variant without -m- is used in $3Sg s = b = \emptyset$. All other forms (e.g. 1Sg s = b = y and 3Pl s = b = b) have the -m-. For (positive) 'be', the -m- is required in all forms, including $3Sg b = b = \emptyset$.

10.5.1.3 Past Perfect (positive and negative)

The Past clitic =b\(\varepsilon\)- can also be used after the bare stem (with lexical tone contour as well as vocalism), and after suffixal forms of the perfective system, in **Past Perfect** (positive sense) ('X had VERB-ed'). This requires establishment of a separate temporal reference point, before which the eventuality in question occurred.

In the bare-stem construction, the bare stem functions as a substitute for the unsuffixed Perfective, which does not combine with $=b\dot{\epsilon}$.

Examples follow with the bare stem (xx1.a), Perfective-1a - $\epsilon r \epsilon$ - (xx1.b), Perfective-1b - ϵr - (xx1.c), Perfective-2 -só- (xxx.d), and Recent Perfect ϵr - (xxx.e). Perfective-1b - ϵr - occurs here in high-toned form.

- (xxx) a. bàyá≡bέ-y be.cured≡Past-1SgS 'I had been cured.'
 - b. gó-èrè=b-à go.out-Perf1a=Past-3PIS 'They had gone out.'
 - c. njî-ŋ súyó-tî=bε-∅
 1Sg-Acc hit-Perf1b=Past-3SgS
 'He/She had hit me.'

```
    d. njî-ŋ yĭ:-só≡bέ-∅
    1Sg-Acc see-Perf2≡Past-3SgS
    'He/She had seen me.'
```

```
e. . wàgàti ńné yé:-m sè gà,
time.L 3SgS come-Ppl.Impf Ppl.Perf in,
nă: kó: jè≡bè-y∴
meal eat RecPf-Past-1PlS
'When he was coming, we had already eaten.'
```

The **Past Perfect Negative** is formed by adding $=b\hat{\epsilon}$ - (with its inflection) to Perfective Negative suffix $-r\hat{\imath}$ -. The latter takes its usual 3Pl form $-n\hat{\iota}d\hat{u}$ -, but is otherwise invariant before $=b\hat{\epsilon}$ -. The paradigm is therefore (xxx). Examples are in (xxx).

```
(xxx) category
                        negative past perfect
        1Sg
                        -rî≡bέ-y
       1P1
                        -rî≡bὲ-y∴
       2Sg
                        -rî≡bέ-w
       2P1
                        -rî≡bὲ-w∴
       3Sg
                        -rî≡bέ-∅
       3P1
                        -ǹdú≡b-á
       a. wàgàtì
                       ńnέ
                                vé:-m
(xxx)
                                                sègà,
           time.L
                       3SgS
                                come-Ppl.Impf while,
           ŋă:
                     kà:-rî≡bè-y∴
                     eat-Perf.Neg≡Past-1PlS
           meal
           'When he came, we had not (yet) eaten.'
       b. nă:
                     kà:-rî≡bέ-∅
                     eat-Perf.Neg=Past-3SgS
           meal
           'He had not (yet) eaten.'
       c. nă:
                     kà:-ndú≡b-á
                     eat-Perf.Neg.3PlS≡Past-3PlS
           'They had not (yet) eaten.'
```

Past Perfect verb forms with =be- are also used in both the antecedent and consequent clauses of **counterfactual conditionals** (§16.xxx). Thus the past imperfectgive negative in (xxx.a) has two free translations, the second being appropriate to a counterfactual (xxx.b) exemplifies the first reading in context,

and (xxx.c) does likewise for the second reading. The main verb 'know' in the antecedent clause in (xxx.c) has another past imperfective, this time positive (there is also an example of bù-m=bè- 'was'). (xxx.d) is another counterfactual example, this time with -be- added to perfective-system stems in both clauses.

```
(xxx) a. bàmàkó ńnî-ŋò:≡bɛ-y
Bamako go-Impf.Neg≡Past-1SgS
'I did not use to go to Bamako.'
'I would not have gone to Bamako.'
```

- b. bàmàkó ńnî-ŋò:=bè-y
 Bamako go-Impf.Neg=Past-1SgS
 [isè dùgî] jóró-ŋò-y dèrⁿ1
 [village big] want-Impf.Neg-1SgS for
 'I didn't use to go to Bamako, because I didn't like big cities.'
- c. [kòmbó yá bù-m≡bè-Ø] júgó-m̀≡bè-y ndè, [war Exist be-Impf≡Past-3SgS] know-Impf≡Past-1SgS if, bàmàkó ńnî-ŋò:=bè-y
 Bamako go-Impf.Neg≡Past-1SgS
 'If I had known that there was a war, I would not have gone to Bamako.'
- d. 6: ǎy≡bé-y ndè, bàyá-èrè≡bè-y medication take.Perf≡Past-1SgS if, be.cured-Perf≡Past-1SgS 'If I had taken the medicine, I would have been cured.'

10.5.1.4 Past Passive (positive and negative)

The Past Passive is the passive plus conjugatable **Past** clitic $=b\epsilon$ - (§10.xxx). In this case, we get H-toned Passive $-y\dot{\epsilon}=$ for all person subject categories except 3Pl, but the Past clitic is low toned. The complete pronominal paradigm is (xx1).

```
(xx1) 1Sg -y\hat{\epsilon}=\hat{m}=b\hat{\epsilon}-y
1Pl -y\hat{\epsilon}=\hat{m}=b\hat{\epsilon}-y.
2Sg -y\hat{\epsilon}=\hat{m}=b\hat{\epsilon}-w
2Pl -y\hat{\epsilon}=\hat{m}=b\hat{\epsilon}-w.
3Sg -y\hat{\epsilon}=\hat{m}=b\hat{\epsilon}-\emptyset (</=\hat{\eta}=b\hat{\epsilon}-/)
3Pl -y\hat{\epsilon}=\emptyset=b-\hat{a}
```

InanSg $-y \hat{\epsilon} = \hat{w} = b - \hat{\epsilon}$ InanPl $-y \hat{\epsilon} = \hat{w} = b - \hat{a}$

Pronominal inflection occurs on Past $=b\hat{\epsilon}$, and for inanimates and for 3Pl also on Passive $-y\hat{\epsilon}=$. The expected $\#=\eta=b\hat{\epsilon}$ - sequence (3SgS plus Past) is pronounced $=m=b\hat{\epsilon}$ - with Nasal-Assimilation, and therefore falls together with $=m=b\hat{\epsilon}$ - in the first and second person forms.

Examples of the Past Passive are in (xx2).

- (xx2) a. ndo kémé-yé≡w=bè-∅ house.L build-Pass=it.is.InanSgS=Past-3SgS 'A/The house had been built.'
 - b. tóŋ-yέ≡∅≡b-à
 write-Pass=it.is.3PlS=Past-3PlS
 'They (= books) had been written.'
 - c. pèrgè sémé-yé≡m≡bè-∅ sheep.L slaughter-Pass≡it.is.3SgS=Past-3SgS 'A/The sheep-Sg had been slaughtered.' (from /...=η≡bè-/)
 - d. [pèrgè bû:] sémé-yé≡∅=b-à
 [sheep.L Def.AnPl] slaughter-Pass=it.is.3PlS=Past-3PlS

 'The sheep-Pl had been slaughtered.'

The Past Passive Negative is based on combining either of the two Passive Negative constructions mentioned in (\$xxx) with Past $=b*\ensuremath{\varepsilon}$ - (xx3.a-b).

- (xx3) a. tóŋó-yé≡ẃ=ndŏ:-∅≡bè-∅ write-Pass=it.is.InanSgS=it.is.not-3SgS=Past-3Sgs 'It had not been written.'
 - b. tòŋò-rºi-yé=ẃ=bè-∅ write-Perf.Neg-Pass=it.is.InanS 'It has not been written.'

10.5.2 'Still', 'up to now', (not) yet'

'Still' and 'up to now, for the present' can be expressed by [nîŋày yŋà], instrumental PP ($\S 8.1.2$) from nîŋèyⁿ 'now'. If the predicate is negative, the translation is 'not (yet)'.

- (xx1) a. [ninèyⁿ ynà] [ŏ: gó] bù-Ø [now with] [field.L in] be-3SgS 'He/She is still in the fields.'
 - b. [níŋèyⁿ yŋà] yè:-rî-Ø [now with] come-PerfNeg-3SgS 'Up to now he/she hasn't come.' 'He/She hasn't come yet.'

10.6 Imperatives and Hortatives

10.6.1 Imperatives and Prohibitives

10.6.1.1 Positive imperatives (Imperative stem, Plural -ndi)

Positive imperatives have distinctive forms for 2Sg and 2Pl subject. The **2Sg imperative** has no affix (and so is equivalent to the **Imperative stem**), but does show vocalic and tonal changes vis-à-vis the bare stem (see discussion below). The Imperative stem favors stem-final {a o o} vowels. Tonally, it is characterized chiefly by a final L-tone.

The **2Pl imperative** is expressed by adding suffix allomorph -ndi to the singular imperative; it has also been heard as -ni after longer stems (xx1). I gloss the suffix as "2Pl.Imprt" in interlinears.

(xx1)		number	Sg imperative	Pl imperative
	a.	'run' 'go out' 'buy' 'go'	yógô gô: éwâ ńnô	yógô-nđi [jógónđi] gô:-nđi éwâ-nđi ńnô-nđi
	b.	'make weep' 'return' 'open'	kóyó-mô bindò pí: ⁿ -r ⁿ à	kóyó-mô-nđi bîndò-nđi pî: ⁿ -r ⁿ à-nđi

Examples with objects are in (xx2). In the free translation, -2Sg or -2Pl is added to the verb to indicate subject number.

'Hit-2Pl it!'

The final vowel of the stem is subject to modifications in the imperative. It suffices here to discuss the singular, since the plural is based on it.

If the regular stem ends in /e/, this vowel shifts to /o/ in the imperative. The shift applies to monosyllabic as well as longer stems (xx3).

(xx3)		gloss	bare stem	Sg imperative
	a.	'bring' 'come'	jě: yě:	jô: yô:
	b.	ʻgoʻ	ńné	ńnô
	c.	'do well' 'fight' 'jump off'	ké:ndé jòrîyé péré	ké:ndò jòriyô pérô

If the stem ends in /ɛ/, this vowel shifts to /a/ in the imperative in stems of two or more syllables. ndé 'go up' is counted as bisyllabic for this purpose (xx4.a). Monosyllabic Cɛ: has an imperative of the diphthongal form Cɛa, except for té: 'lay (mat)', which also has a bisyllabic variant form Cɛya (xx4.b). Cognates of té: have a semivowel in some neighboring languages, e.g. Ben Tey téy(1).

(xx4)		gloss	bare stem	Sg imperative
	a.	'go up' 'build' 'fall' 'look'	ndé kémé yègé ηĭr ⁿ é	ńdâ kémâ yégâ ŋîr ⁿ â
	b.	'lay (mat)'	té:	téyâ, téâ

'shine'	kέ:	kéâ
'knock off'	pέ:	pέâ
'get old'	pέ:	pέâ

Monosyllabic Ci: is segmentally stable in the imperative in the one relevant verb, 'see'. There is no audible segmental distinction between the syllabic nucleus of yı̃: 'see' and that of fiy 'send' (xx5.a). Stems of the bisyllabic shapes (C)vCi and (C)vCCi, with "v" a short vowel, have a stable final /i/ that is retained in the imperative (as well as in the Imperfective) (xx5.b). hdı̃ 'give' is treated as bisyllabic for this purpose (xx5.c). Longer stems, including bisyllabics with a long vowel in the first syllable (xx5.d), and trisyllabics (xx5.e), have a final /i/ in the bare form (and in the suffixal Perfective) that is replaced, in the imperative (and Imperfective) by a vowel from the set $\{a \ni o\}$. In the trisyllabics, if the final vowel is rounded $\{ \ni o\}$, this vowel quality also spreads to the medial syllable; see 'rest' and 'go around' in (xx5.e).

(xx5) Imperatives (verbs with final /i/)

gloss	bare stem	Sg imperative
a. Ci: and Ciyi		
'see'	yĭ:	yî:
'send'	tiyî [tî:]	tîy
b. CvCi and CvCCi		
'speak'	tiŋi	tiŋî
'put'	kúr ⁿ i	kúr ^m i
'tie'	págî	págî
'do'	kár ⁿ î	kár ⁿ i
'put up'	náŋî	náŋî
'cure'	jòŋî	jòŋî
'sing'	nùŋî	nùŋî
'cover'	děw (<td>dèwî</td>	dèwî
'find'	témbî	témbi
'put lid'	tîmbî	tîmbì
'urinate'	ónjî	ó njì
'laugh'	màndî	mànđi
'step in'	nàmbî	nàmbî
c. nCi		
'give'	ńdî	ńdî

```
d. Cv:Ci
  underived
        'call'
                               nă:r<sup>n</sup>î
                                                        nă:r<sup>n</sup>à
        'help'
                               bă:rî
                                                         bă:rà
        'take away'
                               gě:r<sup>n</sup>i
                                                         gě:r<sup>n</sup>à
        'gather'
                               m<sub>5</sub>:ndî
                                                         mð:ndò
        'think'
                               mă:ndî
                                                         mă:ndà
  mediopassive
        'carry on head'
                               dŭ:-yî
                                                         dŭ:-yâ
        'rest'
                               súmîr<sup>n</sup>î
                                                         súmór<sup>n</sup>ò
e.
        'demolish'
                               wòró-gî
                                                         wòró-gô
        'go around'
                               gòŋîr<sup>n</sup>î
                                                         gàŋár<sup>n</sup>â
        'get up'
                               îrî-yî [îrî:]
                                                         îrîyà
                               kémîr<sup>n</sup>î
        'have fun'
                                                         kémîr<sup>n</sup>à ~ kémîr<sup>n</sup>ò
        'winnow'
                               bègîrî
                                                         bègirâ ~ bègirô
        'get ready'
                               dàgiri
                                                         dàgirâ
        'get rid of'
                                                         màrá-gâ
                               màrá-gî
        'remember'
                               ìlî-rî
                                                        ìlî-râ
```

Several examples in (xx5.d-e) show that the stem-final vowel in the imperative is a **copy of an earlier stem vowel** {a ɔ o}, as in wòró-gô, màrá-gâ, gòŋórⁿô. An earlier ε is copied as ɔ or a, often in free variation, but my assistant accepted only ɔ in the high-frequency imperative gě:rⁿò 'bring!'. If the nonfinal syllables of the stem have such a vowel followed by a high vowel {i u}, the high vowel is disregarded in determining the imperative stem-final vowel. For example, in gòŋîrⁿî 'go around' the first two syllables have an ɔ...i sequence, but only the ɔ is relevant in determining the imperative vocalism.

If the nonfinal syllables of the stem have **only high vowels** $\{i\ u\}$, the imperative ends in a vowel from the set $\{a\ o\}$. In the available examples, I find a after i-vowels (corresponding to ϵ in the Imperfective), and a lexical choice of a or a after u-vowels a (xx6).

(xx6) Stems with all high vowels in nonfinal syllables

gloss	bare stem	Sg imperative
a. Imperative ends in a	ı	
'remember'	ìlî-rî	ìlî-râ
'get up'	îrî: (îrî-yî/)</th <th>îrîyà</th>	îrîyà
'accompany'	îŋgîrî	îŋgîrà

b. Imperative ends in o

'rest' súmîrⁿî súmórⁿò

c. Imperative ends in o

'skim' kúgîrî kúgórò

The remaining stem-final vowels, namely $\{o \circ a\}$, are segmentally stable as we go from the bare stem to the Imperative. This remark applies to monosyllabic stems (xx7.a) as well as to longer stems (xx7.b).

(xx7)		gloss	bare stem	Sg imperative
	a.	ʻdrink' ʻreply' ʻgo out'	nó: sá: gŏ:	nô: sâ: gô:
	b.	'run' 'bite' 'touch'	yògó kúwó táwá	yógô kúwô táwâ

Tonally, the Imperative stem is characterized by a final L-tone, which combines with the lexical {H} or {LH} to result in word-level {HL} or {LHL}. However, the expected {LHL} is reduced to {HL} under some conditions, which distinguish stems ending in a high vowel (or y) from those ending in non-high vowels.. The following exposition on tones re-uses the forms given above to demonstrate imperative vocalism.

rewrite analysis

In (xx8), we see that the Imperative stem (i.e. the singular imperative) is $\{HL\}$ -toned for Cv:, CvCv, and most CvCCv stems (i.e. stems with two vocalic moras) that end in a non-high vowel, regardless of lexical tone contour. Therefore even the $\{LH\}$ -toned stems have imperatives beginning with a H-tone.

(xx8) Tone of prosodically light stems with final non-high vowel

gloss	bare stem	Imperative
a. Cv: {H}-toned		
'reply'	sá:	sâ:

```
{LH}-toned
      'go out'
                                          gô: (contrast gồ: 'fire')
                       gŏ:
      'drink'
                       nă:
                                          nô:
b. CvCv (including nCv)
  {H}-toned
      'go'
                                          ńnô (variant înô)
                       ńné
      'get bogged'
                       pîdé
                                          pîdô
  {LH}-toned
      'go up'
                       'ndέ
                                          ńdâ (variant indâ)
      'steal'
                       gùró
                                          gúrô
      'run'
                       yògó
                                          yágâ
c. CvCCv with simple CC cluster (see text below)
  {H}-toned
      'tamp down'
                       túmbó
                                          túmbò
  {LH}-toned
      'dig'
                       gùnjó
                                          gúnjò
      'churn'
                       jùmbó
                                          júmbò
```

There are certain CvCv and CvCCv stems that are treated differently. Most of these include a derivational suffix (xx9.a-b), and we will see below that the productive causative derivative (which always has more than two vocalic moras) also has special tonal treatment. There is a type with complex medial CC cluster that also has special treatment (xx9.b).

(xx9) Irregular CvCv and CvCCv

```
Imperative
      gloss
                      bare stem
a. initial L-tone of lexical {LH} verb preserved
   Mediopassive Cv-yv
      'lie down'
                      bì-yé
                                         bì-yê
   Cỳ-ndý (frozen causative)
      'take out'
                      gò-ndó
                                         gò-ndô:
b. H<HL> instead of HL sequence for CvCCv
   Mediopassive CvC-yv
      'sit'
                      éw-yé
                                         éw-yô
   Cýmjý
                                         kúmjô
      'crumple'
                      kúmjó
```

In (xx9.a), the L-tone of lexical {LH} remains on the entire first syllable in the imperative. In effect, these stems are treated as prosodically heavy (see below). In (xx9.b), The irregularity in (xx9.b), by contrast, is that the H-tone does not end at the syllabic boundary, but instead spreads into the onset of the final vowel, as with CvCv stems. In the case of 'crumple', the heavy, nonhomorganic cluster mj seems to somehow be involved; this cluster is heard as [mnd3] in careful speech, i.e. as even heavier than my regular transcription suggests. Perhaps with such a heavy cluster, the final syllable is treated as prosodically autonomous.

Prosodically heavy stems (those with at least three vocalic moras) ending in a non-high vowel are shown in (xx10). Here the lexical distinction between $\{H\}$ - and $\{LH\}$ -toned stems is audible in the imperative, since the $\{LH\}$ -toned stems preserve the L-tone at the stem onset. The final syllable is L-toned. There are no available non-causative quadrisyllabic stems to test.

(xx10) Tones of prosodically heavy stems with final non-high vowel

gloss	bare stem	Imperative
a. Cv:Cv and Cv:CCv {H}-toned		
'weigh'	pé:sé	pé:sò
'do well'	ké:ndé	ké:ndò
[LH}-toned		
'stalk'	yŏ:ró	yŏ:rò
'file'	ďi:sé	ďi:sò
b. CvCvCv		
{H}-toned		
'screw in'	pîgîré	pîgîrò
'rub'	ligîsé	ligisò
[LH}-toned		
'fight'	jòr î yé	jòriyô
'lean'	disîyé	disîyô

This completes the tonal analysis of imperatives for stems ending in non-high vowel. Stems ending in i (often varying with u) or in y (arguably /yi/) are illustrated in (xx11). Causative -mî is not included (see below for separate discussion).

(xx11) Tones of stems with final non-high vowel

glos	S	bare stem	Imperative
a. Cvy(i)			
{ <i>H</i> }-tone	ed		
ʻgo i		núy ⁿ	núy ⁿ [tonally irregular]
{LH}-to.		·	
ok 'hear		nŭy ⁿ	$n\check{u}y^n$
b. CvCi			
<i>{H}-tone</i>	ed		
'cros	ss'	táŋî	táŋì
ʻaffi	х'	tárî	tári
'put { <i>LH</i> }- <i>to</i> .	up on'	náŋî	náŋĭ
cure 'cure		jòŋî	jòŋî
'cov		děw (< /dèwi/)	dèwî
c. CvCCi			
<i>{H}-tone</i>	ed		
'pind'		kémbî	kémbì
{ <i>LH</i> }-to.			
ʻlaug		màndi	mànđi
d. Cv:Ci ar	nd Cv:CCi		
<i>{H}-tone</i>	ed		
'cha		lá:rî	lá:rà
'scra	itch'	kó:sî	kó:sò
{LH}-to	ned		
'mix		gă:r ⁿ î	gă:r ⁿ à
	away'	gě:r ⁿ î	gě:r ⁿ ò
'call		nă:r ⁿ î	ŋǎ:r ⁿ à
'gath	ner'	mɔ̃:ndĩ	mð:ndð
e. trisyllab			
<i>{H}-tone</i>			
	e fun'	kémîr ⁿ î	kémîr ⁿ à
$\{LH\}$ -to			
'win	now'	bègîrî	bègîrâ

The i-final verbs in (xx11.b-e) clearly distinguish lexical $\{H\}$ and $\{LH\}$ tones, with $\{LH\}$ stems showing L-toned initial syllable. This is true even in CvCi stems, which contrast in this respect from CvCv stems with final non-high

vowel. In (xx11.a), we see an irregular {H}-toned imperative núyⁿ 'go in!', audibly distinct from the regular imperative of nǔyⁿ 'hear' (xx11.a).

Causative suffix -mî is treated as a chained verb stem for this purpose. For example, in Imperative kóyó-mɔ̂ 'make-2Sg weep!' and its plural-subject form kóyó-mɔ̂-ndī, the Causative suffix has its own {HL} tone contour, while the preceding stem has the same form it would have as a bare stem. If kóyó-mī 'make weep' were treated as an ordinary trisyllabic, the H-tone would extend from the left edge only to the second syllable, giving the incorrect #kóyó-mɔ̂, #kóyó-mɔ̂-ndī.

10.6.1.2 Prohibitives (-rá, -ndá, -ndà:)

The Prohibitive is the negative imperative (xx1).

```
(xx1) tê: n5:-r<sup>n</sup>á-nđi
tea drink-Prohib-Imprt.Pl
'Don't-2Pl drink the tea!'
```

The Prohibitive is formed from mono- and bisyllabic stems with a suffix -rá that has a variant -ndá. The form -rá is usual in Anda, but a Wakara informant generally used -ndá. This suffix is distinct in form from other negative suffixes on verbs (Perfective Negative -rî-, Imperfective Negative -ŋɔ:-). The suffix -rá undergoes **nasalization** to -rná under the influence of a preceding nasal or nasalized segment, as does Perfective Negative -rî-. For allomorph -ndà: with longer stems, see below.

The plural Prohibituve adds -ndi (sporadically reduced to -ni) to the singular Prohibitive.

Examples with monosyllabic stems are in (xx2). Observe that the lexical tone contour, {H} versus {LH}, is respected in the Prohibitive.

(xx2)	gloss	bare stem	Sg Prohib	Pl Prohib
	'go out'	gŏ:	gŏ:-rá	gŏ:-rá-nďi
	'drink'	nš:	nă:-r ⁿ á	nă:-r ⁿ á-nđi
	'bring'	jě:	jě:-rá	jě:-rá-nďi
	'see'	yĭ:	yĭ:-rá	yĭ:-rá-nďi
	'reply'	sá:	sá:-rá	sá:-rá-nďi
	'lay (mat)'	tέ:	té:-rá	té:-rá-nďi

Since the plural form is always easily predictable from the singular, I will omit the plurals in the remaining tables.

For **bisyllabics** with just two vocalic moras (no long vowel), the **stem-final vowel is replaced by** /i/ (xx3.a-b). The /i/ is usually syncopated when the syllabic and segmental conditions permit, i.e., after certain unclustered sonorants, especially semivowels and /r/ (xx3.c). The suffix is usually heard as -ndá after a rhotic (xx3.c). A syllable-final /iy/ resulting from syncope contracts phonetically to a long [i:] (xx3.d) by Monophthongization. Likewise, a syllable-final /uw/ resulting from syncope contracts to a long [u:], though in one case ('bite') my assistant preferred a variant with /uy/ (xx3.e). Throughout (xx3), the lexical tone is respected in the Prohibitive.

(xx3) Prohibitive (bimoraic bisyllabic stem ending in non-high vowel)

	gloss	bare stem	Sg Prohib
a.	'tie'	págî-	pág i -rá
	'cut'	késé-	késî-rá
	'build'	kémé-	kémî-r ⁿ á
	'throw'	gisé-	gisî-rá
	'run'	yògó-	yògî-rá
	'fall'	yègé-	yègî-rá
	'go back'	bindé-	bìndî-rá
	'nibble'	jòmbó-	jòmbî-rá
	'dig'	gùnjó	gùnjî-rá
b.	ʻgo'	ńné-	ńn i -r ⁿ á
	'go up'	ìdέ-	ndi-rá
c.	'work'	bĭré-	bĭr-ndá
	'begin'	tóró-	tór-ndá
	'sell'	túró-	túr-ndá
	'skin'	úró-	úr-ndá
	'hit'	súyó-	súy-rá
	'buy'	έwέ-	éw-rá
	'look'	ŋìr ⁿ é-	ŋìr ⁿ î-ndá
	'give birth'	nàr ⁿ á-	nàr ⁿ i-ndá ~ năn-dá)
d.	'kill'	gìyé-	gĭy-rá [gĭ:rá]
	'lie down'	bìyé-	bĭy-rá [bĭ:rá]
e.	'fan'	jùwó-	jŭw-rá [dʒŭ:rá]
	'brush'	bùwó-	bŭw-rá [bŭ:rá]
	'bite'	kúwó-	kúy-rá

(kúw-rá OK but dispreferred)

Stems with **three or more vocalic moras** (Cv:Cv-, CvCvCv-) ending in a **non-high vowel** are in (xx4). The Prohibitive suffix is now -ndà:, which is heard in L-toned form word-finally. It combines with the Plural suffix as -ndà:-ndì, bringing out a latent rising tone that is also heard when a clause-final particle is added. The final vowel of the stem is **not shifted** to /i/.

(xx4) Prohibitive (trisyllabic stem ending in non-high vowel)

	gloss	bare stem	Sg Prohib
a.	'do well'	ké:ndé-	ké:ndé-ndà:
b.	'dream 'hiccup'	mònjúró- bègîré-	mònjúró-ndà: bègîré-ndà:
c.	'screw in' 'poke'	pîgîré- dùsúró-	pîgîré-ndà: dùsúró-ndà:

We now turn to stems ending in /i/ (xx5). Of course any morphophonological switch to final /i/ for these verbs would be inaudible. The tonology and suffixal allomorphy are consistent with those seen above for verbs ending in non-high vowel. We get -rá- (which is nasalized -rná- after a nasal syllable) with stems that have just two vocalic moras, including CvCCv-, but -ndà: after stems with three vocalic moras, including Cv:Cv-. Causative -mî-tends to be reduced to segmental zero before -ndà:, though a fuller pronunciation as -m-dà: is also possible (xx5.g).

(xx5) Prohibitive (bisyllabic stem ending in /i/)

	gloss	bare stem	Sg Prohib
a.	'give'	ńdi-	ńdi-rá
b.	'speak' 'perpetrate'	tiŋi- bògi-	tiŋi-r ⁿ á bògi-rá
c.	'cover' 'go in' 'hear'	dèwî- [děw] núy ⁿ î- [núj ⁿ] nùy ⁿ î- [nǔj ⁿ]	děw-rá núy ⁿ -r ⁿ á nŭy ⁿ -r ⁿ á

```
d.
      'open wide'
                        gòmbi-
                                             gòmbi-rá
      'find'
                       témbi-
                                             témbi-rá
      'help'
                       bă:rî-
                                             bă:r-ndà:
e.
      'stop'
                       î:-yî- [i:]
                                             î:-y-ndà:
      'gather'
                       mɔs:ndi-
                                             mɔ̃:ndi-ndà:
      'open'
                       pî:-r<sup>n</sup>î
                                             pî:"-r"i-ndà:
f.
      'uncover'
                       tîmbî-rî-
                                             tîmbî-r-ndà:
      'accompany'
                       îŋgîrî-
                                             îŋgîr-ndà:
      'scrub'
                       púgúsî-
                                             púgúsî-ndà:
                                             dàgir-ndà:
      'get ready'
                       dàgiri-
      'make go out'
                       gŏ:-mî-
                                             gŏ:-m-dà: ~ gŏ:-Ø-ndà:
g.
```

10.6.2 Positive hortatives (-má, plural -màyⁿ)

For dual number (speaker and one addressee), the hortative ('Let's ...!') is expressed by adding a suffix -má. For larger numbers (speaker and two or more others, including the addressee), the form is -màyⁿ.

(xx1)	gloss dual hortative		3+ hortative	
	'run'	yògó-má	yògó-mày ⁿ	
	'go out'	gŏ:-má	gŏ:-mày ⁿ	
	'buy'	έwέ-má	έwέ-mày ⁿ	
	ʻgo'	ńné-má	ńné-mày ⁿ	

Further examples showing the hortative of verbs with **final non-high vowel** are in (xx2). As the dual form is predictable from the (more common) 3+ hortative, only the latter is shown. In every example in (xx2), the stem used before the Hortative suffix is identical (segmentally and tonally) to the bare stem.

(xx2) Hortative (verb with final non-high vowel)

	gloss	bare stem	3+ hortative
a.	'shave'	ká:-	ká:-mày ⁿ
	'drink'	nó:-	nó:-mày ⁿ

b.	ʻgo' ʻgo up'	ńné- ndé-	ńné-mày ⁿ ǹdέ-mày ⁿ
c.	'steal' 'run' 'spray'	gùró- yògó- písé-	gùró-mày ⁿ yògó-mày ⁿ pîsé-mày ⁿ
	'hiccup' 'dream' 'screw in'	bègiré- mònjúró- pigiré-	bègîré-mày ⁿ mònjúró-mày ⁿ pîgîré-mày ⁿ

When the stem ends in /i/, on the other hand, this final vowel must be changed to a non-high vowel, regardless of syllable or mora structure. The final /i/ is replaced by a vowel with the same quality as the initial-syllable vowel, if non-high. Stems with only /u/ have final /ɔ/, and those with only /i/ have final /ɛ/ (xx3). The lexical tone contour is retained in the hortative.

(xx3) Hortative (verb with final /i/)

	gloss	bare stem	3+ hortative
a.	'see'	yĭ:-	yἕ:-mày ⁿ
b.	'give'	ńdi-	ńdέ-mày ⁿ
c.	'go in' 'hear' 'send' 'cover' 'perpetrate' 'put down'	núy ⁿ î- [núj ⁿ] nùy ⁿ î- [nǔj ⁿ] tîyî- [tî:] dèwî- bògî- dùŋî-	núy ⁿ ó-mày ⁿ nùy ⁿ ó-mày ⁿ tíyé-mày ⁿ dèwé-mày ⁿ bògó-mày ⁿ dùŋó-mày ⁿ
d.	'find' 'open wide' 'put lid on'	témbi- gòmbi- timbi-	témbé-mày ⁿ gòmbó-mày ⁿ tímbé-mày ⁿ
e.	'help' 'take away' 'stop' 'gather'	bă:rî- gĕ:r ⁿ î- î:-yî- mŏ:ndî-	bă:rá-mày ⁿ gĕ:r ⁿ é-mày ⁿ î:-yé-mày ⁿ mŏ:ndî-mày ⁿ
f.	'scrub' 'get ready' 'have fun'	púgúsî- dàgirî- kémîr ⁿ i-	púgúsó-mày ⁿ dàgirá-mày ⁿ kémir ⁿ é-mày ⁿ

10.6.3 Hortative Negative (-rá and variants, plural -rá-màyⁿ)

The Imperative Negative form of the verb, with suffix -rá-, -ndá-, or -ndá:- (word-final ndà:), is followed by the (positive) Hortative suffix -má (dual subject) or -màyⁿ (three or more) to form a the Hortative Negative. Some examples showing the morphological connection to the (singular-subject) Prohibitive are in (xx1).

(xx1)		gloss	bare stem	Sg Prohibitive	Hortative Neg
	a.	'go out' 'drink' 'tie' 'go'	gŏ: nó: págî- ńné-	gŏ:-rá nɔ:-r ⁿ á pági-rá ńni-r ⁿ á	gŏ:-rá-mày ⁿ nɔ:-r ⁿ á-mày ⁿ pági-rá-mày ⁿ ńni-r ⁿ á-mày ⁿ
	b.	'begin'	tóró-	tór-ndá	tór-ndá-mày ⁿ
	c.	'screw in'	pîgîré-	pîgîré-ndà:	pîgîré-ndă:-mày ⁿ

Examples are (xx2).

```
(xx2) a. ńnî-r<sup>n</sup>á-má
go-Hort.Neg-Hort.Du
'Let's-2 (= the two of us) not go!'
```

```
b. gŏ:-rá-mày<sup>n</sup>
go.out-Hort.Neg-Hort.3+
'Let's-3+ (= all of us) not go out!'
```

10.6.4 Third-person Hortative

A third-person subject form occurs primarily in wishes (and curses) of the form 'may God VERB you!'. It can also be used as a kind of indirect command, whether or not the speaker intends the command to be conveyed to the third person by an intermediary. The third person agent can be singular or plural.

We begin with the positive third-person Hortative. When the verb ends in a **non-high vowel**, we get the forms in (xx1). The Cv:- verbs take a suffix - \acute{y} , and shorten their vowel. They also divide into two tonal sets, one with a high-toned form (xx1.a) and another with a rising-toned form (xx1.b), reflecting

etymological stem-tone contours. Bisyllabic verbs with just two vocalic moras shift their final vowel to /i/, which however remains high-toned (xx1.b-d). An exception is gondó- 'take out', an original causative that patterns like a longer stem. Verbs with three or more vocalic moras, including Cv:Cv-, have a final low-tone element that is joined to the right edge, producing {HL} or {LHL} depending on the lexical tone. The added final low tone has its own syllable if preceded by two high-toned syllables, otherwise it merges with the final-syllable high tone to produce a falling tone (xx1.e-g). Longer stems also differ in final vocalism; the trisyllabics the Cv:Cv- verbs kept a final /e/ rather than switching it to /i/. Trisyllabics and Cv:Cv- verbs with final /o/ were more difficult for my assistant, who either switched the /o/ to its harmonic stablemate /e/ ('dream') or switched it to /i/ ('stalk').

(xx1) Third-Person Hortative positive (verb with final non-high vowel)

	gloss	bare stem	3rd person hortative
a.	'shave' 'spend night' 'reply' 'eat'	ká:- ná:- sá:- kó:-	ká-y ná-y ⁿ sá-y kó-y
b.	'arrive' 'bring' 'go out'	dŏ:- jě:- gŏ:-	dŏ-y jĕ-y gŏ-y
c.	ʻgoʻ ʻgo up'	ńné- ndé-	ńnî ndî
d.	'steal' 'run' 'spray' 'pound' 'jump off'	gùró- yògó- pîsé- tóró- péré-	gùri yògi pisi tóri péri
e.	'go back' 'lift up'	bindé- indé-	bindî îndî
f.	'stalk' 'turn over' 'bathe [tr]' 'do well' 'take out'	yŏ:ró- jŭw-ró- dĭ:-ré- ké:ndé- gòndó-	yŏ:rî jŭw-rî dĭ:-rê ké:ndè gòndê

```
g. 'hiccup' bègîré- bègîrê
 'dream' mònjúró- mònjúrê ~ mònjîrî
 'screw in' pîgîré- pîgîrè
```

When the stem already ends in /i/, there is no change in vocalism. The tonology is different from what we just saw with the stems ending in a non-high vowel, in that the final low-tone element is obligatory regardless of syllable count. Therefore 'see' and (at least phonetically) 'hear' end up as <LHL> toned, and the other verbs in (xx2.b-c), and the lexically {LH} toned verbs in (xx2.d-f), end up with a final falling-toned syllable. The lexically high-toned verbs in (xx2.d-f), like 'find', 'stop', and 'have fun', have final low-toned syllables.

(xx2) Third-Person Hortative positive (verb with final /i/)

	gloss	bare stem	3rd person hortative
a.	'see'	yĭ:-	yì:
b.	'give'	ńdi-	ńdî
c.	'go in' 'hear' 'send' 'cover' 'perpetrate' 'put down'	núy ⁿ î- [núj ⁿ] nùy ⁿ î- [nŭj ⁿ] tîyî- [tî:] dèwî- bògî- dùŋî-	nûy ⁿ nǔy ⁿ tíyî [tî:] dèwî bògî dùŋî
d.	'find' 'open wide' 'put lid on'	témbi- gòmbi- tímbi-	témbi gòmbi timbi
e.	'help' 'take away' 'stop' 'gather'	bă:rî- gĕ:r ⁿ î- î:-yî- mŏ:ndî-	bă:rî gĕ:r ^m î î:-yĭ mŏ:ndî
f.	'scrub' 'get ready' 'have fun'	púgúsî- dàgîrî- kémîr ⁿ î-	púgúsi dàgiri kémîr ⁿ i

Causatives are fairly common in the third-person hortative construction, since wishes like 'let him jump off!' can be expressed as 'may God make him jump off!' (děnjê ńné-ń péré-m-ĭ).

The elicited Negative counterparts end in -rà, -ndà, or -ndà: for singular subject. The plural-subject forms are, respectively, -rá-ndǐ, -ndá-ndǐ, and -ndǎ:-ndǐ for plural subject. These forms are related to those of the Prohibitive (=imperative negative), but are low-toned in the singular-subject form, and the stem vocalism differs significantly in the two morphological categories. There is probably dialectal and even idiolect-internal variation in the allomorphy, here as with the Prohibitive. For my assistant, third-person Hortative -ndà: occurs with far more types of verbs than does the similar Prohibitive allomorph -ndà:. Therefore many short stems have third-person Hortative -ndà: but Prohibitive -rá. A generous set of forms is given in (xxx). Note -ndà: after all verbs whose bare stem ends in a nonhigh vowel (xxx.a), as well as with many stems with final high vowel (xxx.b). -rà was recorded with CvCi and nCi stems (xxx.c), and -ndà (note the short vowel) after CvCvCi stems (xxx.d).

(xxx) Third-Person Hortative Negative (verb ending in nonhigh vowel)

	gloss	bare stem	3rd person hort	tative negative Pl
a.	'shave' 'eat' 'arrive' 'bring' 'go' 'go up' 'steal' 'pound' 'dream'	ká:- kó:- dŏ:- jě:- ńné- ňdé- gùró- tóró- mònjúró-	ká:-ndà: kó:-ndà: dŏ:-ndà: jĕ:-ndà: ńné-ndà: 'ndé-ndà: gùró-ndà: tóró-ndà: mònjúró-ndà:	ká:-ndă:-ndī kó:-ndă:-ndī dŏ:-ndă:-ndī dŏ:-ndă:-ndī ńné-ndă:-ndī ńdé-ndă:-ndī gùró-ndă:-ndī tóró-ndă:-ndī
b.	'see' 'go in' 'find' 'open wide' 'gather' 'take away' 'help'	yǐ:- núy ⁿ î- [núj ⁿ] témbî- gòmbî- mŏ:ndî- gě:r ⁿ i- bă:rî-	yě:-ndà: núy ⁿ ó-ndà: témbé-ndà: gòmbó-ndà: mŏ:ndi-ndà: gě:r ⁿ i-ndà: bă:ri-ndà:	yě:-ndă:-ndī núy ⁿ ó-ndă:-ndī témbé-ndă:-ndī gòmbó-ndă:-ndī mŏ:ndī-ndă:-ndī gě:r ⁿ i-ndă:-ndī bă:ri-ndă:-ndī
c.	'give' 'cover'	ńdi- dèwi-	ńdî-rà dĕw-rà	ńdi-rá-ndi děw-rá-ndi
d.	'scrub' 'have fun'	púgúsî- kémîr ⁿ î-	púgúsî-ndà kémîr ⁿ î-ndà	púgúsî-ndá-ndi kémîr ⁿ î-ndá-ndi

10.6.5 Third-Person Hortative with implied first person singular subject

The third-person hortative can also be used with a first-person subject, in a special (but high-frequency) context where the speakers needs clarification as to whether he/she is commanded or authorized to perform an action. For example, if the boss asks you to 'bring some fruits!', and you want to know what kind of fruits the boss wants, you can say (in English) *should I bring mangoes?* In Nanga, the relevant construction is hortative in form, with a final interrogative particle: măŋgórò jĕ-y má? There is no overt subject pronominal here, but in context it is understood that the speaker is, in effect, quoting a command directed at himself.

11 VP and predicate structure

11.1 Regular verbs and VP structure

11.1.1 Verb types (valency)

Since Nanga clearly distinguishes subjects (pronominal-subject agreement on verbs) from direct objects (Accusative case suffix on animate nouns and pronouns), the prototypical transitive verbs are clearly identifiable from their syntactic behavior. These include the usual impact verbs ('hit', 'cut'), but also perception verbs ('see', 'hear')

```
(xx1) a. ńné-ń yĭ:-só-ý
3Sg-Acc see-Perf2-1SgS
'I saw him/her.'
```

b. ńnś-ń súyó-só-ý 3Sg-Acc hit-Perf2-1SgS 'I hit him/her.'

Basic **motion verbs** ('go', 'come') are intransitive. Sentences like [[isé gó] ńné-èrè-ŷ] 'I went [to the village]' have adverbial (e.g. PP) rather than simple NP complements, usually with Locative /ga/ (/go/, etc.). Toponyms omit the Locative postposition, so sentences like [mó:ti ńné-èrè-ŷ] 'I went [to Mopti]' mimic transitives, but the adverbial nature of these place names is suggested by the fact that they correspond to 'where?' (àr áŋá) rather than to 'what?' in questions, and by the fact that the place name cannot be replaced by an Accusative pronominal.

ńdi- 'give' takes two direct objects, either or both of which may show Accusative -ŋ. In the usual case where the recipient is animate and the thing given inanimate, Accusative marking is much more common on the recipient than on the noun denoting the thing given (xx2.a), following the usual pattern with the Accusative morpheme. When both NPs are animate, double Accusative marking is common (xx2.b). kɛ́:ri- 'show X to Y' has similar syntax.

```
(xx2) a. [bă: yɛ̃:-ŋ] kĕ:rɛ̃ ńdi-só-ý

[father 1SgP.Poss.AnSg-Acc] money give-Perf2-1SgS

'I gave some money to my father.'
```

```
b. pèrgé-ý ú-ý ńdî-só-ý sheep.Sg-Acc 2Sg-Acc give-Perf2-1SgS 'I gave you-Sg a sheep.'
```

kîyε- 'say' can take a direct object referring to a quotation ('I didn't say that') and/or a dative NP representing the person addressed.

```
(xx3) [[bă: yɛ̃:] bày] ŋgú-ŋ́ kiyɛ̃-ri-ý

[[father 1SgP.Poss.AnSg] Dat] Dem.InanSg-Acc say-PerfNeg-1SgS

'I didn't say that to my father.'
```

Verbs of **putting** like **kúr**ⁿó- 'put (object) in (a container)', **gàr**ⁿi- 'put (e.g. liquid, grain) in (container)', and náni- 'put (object) up on (something)' normally take a direct object and a locational expression, though the latter is sometimes obvious and can be omitted ('I put the tea kettle on [the burner]').

The boundary between intransitives and transitives is blurred by the existence of many verbs that take a conventional complement, usually a **cognate nominal**. Syntactically, the cognate nominal can be taken as a direct object; see §11.1.6.2. However, it does not normally allow Accusative -ŋ.

There are also some fixed subject-verb combinations whose subject NP does not appear to have full subject properties; see §11.1.4, below.

11.1.2 Valency of causatives

The subject (agent) of an intransitive clause becomes a direct object when the clause is causativized, and may therefore take Accusative -ŋ.

```
(xx1) [dèré yè:-ŋ] yě:-m-só-ý

[elder.sibling 1SgP.Poss.AnSg-Acc] come-Caus-Perf2-1SgS

'I made/had my older sibling come.'
```

When an already transitive clause is causativized, we end up with two direct objects, either or both of which may take Accusative $-\eta$ (xx2).

What I call the Mediopassive (MP) verb form (suffix -yî- etc.) cuts across transitivity lines, as it can apply not only to classic mediopassives like 'be hung up', but also to transitives denoting actions that create a state for the agent (verbs of carrying, wearing clothes, etc.). A transitive Mediopassive verb (xx2.a) often corresponds to a related verb with Transitive suffix -rî- that functions much like a causative and takes two objects (xx2.b).

```
(xx2) a. yî:-ŋ bàmbî-ý-só-ý
child-Acc carry.on.back-MP-Perf2-1SgS
'I carried a child (on my back).'

b. yî:-ŋ ńnέ-ń bàmbî-rî-só-ý
child-Acc 3Sg-Acc put.on.back-MP-Perf2-1SgS
'I put a child on him/her (on his/her back).'
```

11.1.3 Verb Phrase

The notion of VP is most useful in connection with VP-chains, which are essentially chains of clauses with a shared subject; see §15.1.

11.1.4 Fixed subject-verb combinations

Fixed combinations of a subject NP and a verb are mainly found with meteorological and seasonal expressions (xx1.a-b) and some emotional and medical expressons (xx1.c).

```
(xx1) a. involving ùsî 'sun' or variant (cf. òsù-dér<sup>n</sup>î 'daytime')
              ùsî
                            sîvé-
                                          '(day) break'
                            dèr<sup>n</sup>έ-
                                          'night fall' (cf. dèR<sup>n</sup>é- 'spend mid-day')
              ùsî
                            bàrná-
                                          'be summer' (cf. bár<sup>n</sup>í 'red')
              ùsîyé
         b. rain and rainy season
                                          'rainy season be about to start' ("sky
              yàrî
                            dă:-
                                          go.out")
                                          'be just after the harvest' ("sky go.out")
              yàrî
                            gŏ:-
              bòndi
                            wă:-
                                          'rain fall'
         c. emotions and medical
                            bàrná-
                                          'get angry' (kéndè 'heart/liver', bàrná-
              kéndè
                                          'become red')
```

```
kǐr<sup>n</sup>è-dèr<sup>n</sup>í gŏ:- 'have a nosebleed' (kír<sup>n</sup>ê 'nose', gŏ:- 'go out')
kĭr<sup>n</sup>è-dèr<sup>n</sup>í dèr<sup>n</sup>é- 'have a nosebleed' (cognate verb)
```

The nouns in these constructions tend not to have full subject properties. The noun usually occurs close to the verb, following spatiotemporal adverbs, whereas fully referential subject NPs often precede such adverbs. However, these are tendencies rather than strict rules.

The seasonal and meteorological expressions can occur in same-subject constructions. This is possible since some cyclical sequences can be expressed by pairing combinations sharing a conventionalized subject (xx2).

```
(xx2) [yàrî dŏ: ý] bìndé-èrè-∅

[sky arrived and.SS] go.back-Perf1a-3SgS

'The sky (=cloudy weather of rainy season) arrived and left.'
```

With the emotional and medical expressions in (xx1.c), the absence of subject properties is due to the (frequent) presence of a separate human subject, in addition to the bodily term shown. The construction suggests "possessor raising," as the choice of verb makes most sense if selected by the bodily term. kéndè bàrⁿá- by itself would mean 'heart/liver (seat of emotions) became red (=burned)', and kĭrⁿè-dèrⁿi gŏ:- would mean 'nosebleed (nose blood) go out'. These would make good sense with possessors: 'my heart became red', 'my nose-blood came out'. However, the actual constructions have human subject NPs, not possessors. This is shown by subject agreement on the verb and by the absence of possessed-noun tone contour on the bodily term.

```
bàr<sup>n</sup>á-só-y
(xx3) a. kéndé
              heart/liver
                                     get.red-Perf2-1SgS
              'I got angry.'
                                                   bàr<sup>n</sup>á-só-∅
         b. á:mádù
                                 kéndè
                                 heart/liver
                                                    get.red-Perf2-3SgS
              'Amadou got angry.'
         c. kir<sup>n</sup>è-dèr<sup>n</sup>i
                                     gŏ:-só-y
                                     go.out-Perf2-1SgS
              nose-bleed
              'I had a nosebleed.'
```

Therefore at least in the emotional and medical expressions, the apparent subject noun functions somewhat like an adverb.

11.1.5 Idiomatic and cognate objects

Many verbs are commonly paired with a default nominal, usually functioning as direct object (and usually omitted if there is a more concrete object NP). In some cases, the default object nominal and the verb are non-cognate (xx1).

```
(xx1) noun verb gloss of combination

nă: k5:- 'eat (a meal)'
nî: dĩyé- 'bathe' (nî: 'water')
nî: n5:- 'drink (water)'
```

11.1.5.1 Formal relationships between cognate nominal and verb

Many verbs have a **lexicalized cognate nominal** from the same word family. The present focus is on the relationship among the nominal and verbal forms. For the grammatical functions of the cognate object, see §11.1.5.xxx, below.

Any verb that does not have a lexicalized cognate nominal can simply use its regular verbal noun in -ndé. Example: játi-ndé jàti- 'do a calculation'. Such cases are not at issue in the present section since verbal nouns are predictable in form.

A generous set of examples of cognate noun-verb pairs is in (xx1). Since the tone contour of a verb is closely associated with its initial consonant, it seems most useful to organize the data around the tone contour (and syllable count) of the noun. Of interest is the distinction between dùrî dùró- 'let out a groan' (xx1.e) and dúrî dùró- '(lion etc.) roar' (xx1.d), distinguished only by the tone of the noun.

```
(xx1)
                               verb
                                               gloss of combination
              noun
         a. monosyllabic
                               tá:n-
                                                'build a shed (shelter)'
              tă:n
              tă:n
                               tá:n-
                                                'avoid, respect (a taboo)'
                                                'give out a whistle'
              pô:
                               pó:-
                                                'tie a knot'
              mš:
                               mó:-
         b. bisyllabic, noun {H}
                               biré-
                                                'work, do a job'
              bîrá
              gîr<sup>n</sup>á
                               gir<sup>n</sup>έ-
                                                'harvest millet, do the millet harvest'
              dér<sup>n</sup>î
                               dèr<sup>n</sup>έ-
                                                'spend the mid-day'
                                                'roll turban (on head)'
              dómbó
                               dòmbi-yé-
              má:ndî
                               mă:ndî-
                                                'think a thought'
```

```
c. bisyllabic, noun {LH}
    tùwá
                                   '(a) death occur'
                   túwέ-
    òró
                    óró-
                                   'make a heap'
    èré
                    éré-
                                   'be rivals, have a rivalry'
                                   'dance'
    giyé
                    giyé-
    giynέ
                    giy<sup>n</sup>έ-
                                   'fart, let out a fart'
                                   'defecate, take a shit'
    sùgś
                    súgó-
                                   'speak'
    dàmá
                    dàmá-
    pěw
                   ρέψέ-
                                   'give a reprimand'
    tir<sup>n</sup>î
                   tîr<sup>n</sup>î-
                                   'go search for firewood'
                                   'write, do some writing'
    tàŋś
                   tóŋi-
                                   'treat (medically), provide care to'
    jòŋî
                   jònî-
                                   'sing, perform a song'
    nùŋá
                   nùni-
                                   'compete, be in a race'
    pòmbó
                    pómbi-
    sàmbá
                    sámbi-
                                   'do the second round of weeding'
    ĵimbî
                   ĵimbî-
                                   'double up, have two'
d. bisyllabic, noun {HL}
                                   'give a description'
    sîbâ
                   sîbέ-
    bîgâ
                    bìgέ-
                                   'chew cud'
    bógî
                    bògó-
                                   '(dog) bark'
                                   '(lion, hyena, elephant) roar'
    dúrî
                    dùró-
    jîŋâ
                   jiŋi-
                                   '(plant stem) split into two'
    gʻʻsr<sup>n</sup>ĵ
                   gòr<sup>n</sup>ó-
                                   'be stronger (than)'
    gósô
                   gòsó-
                                   'divide into halves'
    îsê
                   ìsé-
                                   'sneeze'
    tîŋâ
                    tîŋî-
                                   'speak'
                                   'belch, emit a belch'
    jîgâ
                   jigî-
                                   'vomit'
    úrô
                   úró-
    máŋî
                    mànî-
                                   'cook a dish including cottonseed'
    bέrê
                                   'gain, make a profit'
                    bèré-
                                   'foam, be frothy'
    pútô
                    pútó-
                                   'foam up'
    púdε̂
                    púdέ-
                                   'poke fun at'
    jáyrè
                   jǎyrέ-
    bémbè
                   bèmbé-
                                    'stutter'
    jángè
                   jàngî-
                                   'study, go to school'
                                   'pray, perform the Muslim prayer'
    sándi
                   sándi-
    tómbi
                   tómbó-
                                   'jump, take a jump'
                                   'make a profit'
    tî:nà
                   tî:nî-
                                   'preach a sermon'
    wá:jè
                    wá:jî-
    wé:tè
                    wέ:tî-
                                    'spend a half-day (morning)'
    wá:tè
                    wá:tέ-
                                   'swear an oath' (<Fulfulde)
    té:njè
                   tέ:njî-
                                   'tell a story'
```

```
e. bisyllabic, noun {LHL}
                                    'make a payment'
    tàsâ
                    tósî-
    kòyô
                    kóyó-
                                    'weep'
    lùgô
                    lúgó-
                                    'count (recite numbers)'
                                    'be deceptive, trick'
    bàgâ
                    bàgî-
                                    'make an insult'
    dùyâ
                    dùyó-
    dùwâ
                    dùwó-
                                    'forge (tools)'
                                    'stand/ stop in a position'
    ĭyâ
                    î:-yî-
                                    'clear one's throat'
    kàgâ
                    kágá-
    ùsâ
                    úsî-
                                    'ask a question'
                                    'cook para (dish with cow-peas, or
    pàrâ
                    párá-
                                    millet mixed with roselle leaves)'
    à:njâ
                    á:njî-
                                    'yawn, make a yawn'
    dùrî
                    dùró-
                                    'let out a groan'
                                    'urinate'
    îjnć
                    ónjî-
    tànji
                    tónjî-
                                    'spit, emit a spit'
    jinjâ
                    jinjî-
                                    'make noise'
    tà:rî
                    tá:rî-
                                    'lay egg'
    sè:r<sup>m</sup>i
                    sέ:r<sup>n</sup>i-
                                    '(woman) emit cry of joy'
f. trisyllabic, noun {LH}
    kèmìr<sup>n</sup>é
                    kémir<sup>n</sup>i-
                                    'have fun, stage festivities'
    èmir<sup>n</sup>é
                    émîr<sup>n</sup>î-
                                    'converse, chat'
                                    'utter a formal greeting'
    sàlàmi
                    sálmí-
    bèrèmbi
                    bèrémi-
                                    'take animals to pasture'
             (often pronounced bèrém bèrémi-)
g. trisyllabic, noun {HL}
                                    'do wage labor (by the day)'
    sámár<sup>n</sup>i
                    sámár<sup>n</sup>i-
    sógînè
                                    'take cows out at night'
                    sógînî-
    vîmbérè
                    yimbiri-
                                    '(beggar) sing koranic verses'
    jóriyè
                    jòriyé-
                                    'fight, engage in a fight'
                                    'dream a dream'
    mónjórò
                    mònjúró-
h. trisyllabic, noun {LHL}
    gòròndô
                    gòróndi-
                                    'snore; (lion) roar'
    tàsàrâ
                    tósîrî-
                                    'have a discussion'
                                    'request, beg'
    nèmbirê
                    némbîré-
    bàgàrâ
                    bàgiri-
                                    'make loud noises'
                                    '(animal) bellow'
    bògòrô
                    bògóró-
                                    'formally counsel (a young person)'
    tingiri
                    tîngîrî-
                                    'perfume with incense'
    ùgirî
                    ùgúró-
                                    'have a rest'
    sùmùr<sup>n</sup>i
                    súmúr<sup>n</sup>i-
```

Some of the trisyllabic examples above show distinctions between the noun and the verb in the treatment of the medial stem syllable (raised /i/ versus a repeated non-high vowel); see '(beggar) sing koranic verses' and 'dream a dream' in (xx1.g), and 'have a discussion', 'make loud noises', and '(animal) bellow' in (xx1.h).

In a few cases, there is an **irregular vocalic change** affecting the initial vowel. Some examples involve switches between $\{e \ o\}$ in the noun and $\{\epsilon \ o\}$ in the verb (xx2.a). There are also several cases of $\{o \ o\}$ in the noun versus /a/ in the verb (xx2.b). The example in (xx2.c) is similar but likely involves syncope of *g in addition to the vocalic change.

(xx2)		noun	verb	gloss
	a.	tŏ: yógî jéw yégî ségî yémbî	tó:- yògó- jèwé- yègé- ségi- yèmbi-yi-	'sow (seeds); sow the seedstock' 'run' 'curse, utter a curse' 'fall down, take a fall' 'pay dues, make a contribution' 'cover oneself with blanket'
	b.	mánďi bó:ri yóri ò:ri bómbi wóri	màndi- bă:ri- yàri-yi- á:ri- bàmbi-yi- wàrá-	'laugh, let out a laugh' 'make an addition (top-off)' 'take a walk' 'crawl, drag oneself' 'hold on one's back' 'do (manual) farm work (in field)'
	c.	sò:rî	sógîrî-	'(sth unseen) make a noise'

In (xx3), there is a **partial cognate** relationship. The default object nominal is a compound, whose initial or final is related to the verb.

```
(xx3)
              noun
                                  verb
                                                gloss
         a. verb related formally to the final element of a nominal compound
                                  kúwó-
                                                'perform black magic'
              yε-kŭ:
              girè-niy<sup>n</sup>ê
                                  niyέ-
                                                'sleep' (giré 'eye')
              àr<sup>n</sup>à-bó:rì
                                  bă:rî-
                                                'provide assistance to'
              àr<sup>n</sup>à-tš:
                                  tó:-
                                                'scold'
              àr<sup>n</sup>à-pă:<sup>n</sup>
                                                'take a step'
                                  pá:n-
                                                'emit some slobber, drool'
              nò:-já:rà
                                  jă:rî-
              gò:-kòndùgó
                                  kóndúgó-
                                                'build a conical roof'
              nà:-pérê
                                                'clap, applaud'
                                  péré
                                                'draw a line (with the hand)'
              nà:-tinjî
                                  tînjé-
```

```
b. [noun adjective] combination, verb based on adjective
   nà: pîrî
                                 'cook nà: pîrî (lit. "white meal," a
                     pîré-
                                 millet dish)
c. noun arguably with frozen *an- not included in verb (§4.xxx)
    àntá:ri (~ àtá:ri) tá:ri-
                                 'hunt, go on a hunt' (árna 'man'?)
d. verb based on final ... CvCv of trisyllabic noun
    ná:pîlà
                     pîlέ-
                                 'perform an individual prayer'
e. verb based on initial of nominal compound
                                 'hiccup'
    bègirè-bê:
                     bègé-
    kòyò-kè:sî
                     kóyó-
                                 'give out a shout'
```

A few morphologically messier cases are in (xx4). In (xx4.a), the verb is a somewhat frozen combination of a monosyllabic stem plus Mediopassive -y \acute{v} -($\S9.xxx$). In (xx4.b), the verb has a somewhat frozen transitive suffix *-r \acute{v} -. In (xx4.c), the noun has a reduplicative segment that is disregarded in the verb.

(xx4)		noun	verb	gloss
	a.	ùwá ĭyâ	ú:-yî- î:-yî-	'be afraid' 'stand, be in a position'
	b.	nèŋî	néŋgîré-	'cook sauce'
	c.	kà-kàri	kárá-	'lie, tell a lie'

11.1.5.2 Grammatical status of cognate nominal

Often the cognate nominal is rather pro forma, as in 'dance (a dance)' or 'stutter (a stuttering)'. However, the cognate nominal may be **quantified over or modified** adjectivally where this makes sense semantically. If the activity is divisable into bounded units, these can be quantified over (xx1.a). Evaluative or other adverbial modification normally takes the form of adding a modifying adjective to the nominal (xx1.b), cf. \$8.xxx.

```
    (xx1) a. [nùŋá tà:nđi:] nùŋî-só-∅ [song three] sing-Perf2-3SgS 'He/She sang three songs.'
    b. [nùŋà èsî] nùŋî-só-∅
```

[song.L good] sing-Perf2-3SgS 'He/She sang well ("sang a good song")."

11.2 'Be', 'become', 'have', and other statives

11.2.1 'It is' clitics

11.2.1.1 Positive 'it is' (
$$\equiv m$$
-, $\equiv \eta$ -, $\equiv y \hat{\epsilon}$ -, $\equiv W$ -)

The clitic =m- 'it is' is added to NPs. It can be conjugated, and has the **paradigm** (xx1). (The postconsonantal forms are rare, see below, so readers should focus on the postvocalic paradigm.) Except for 3Pl =y ε and the special Inanimate form =w, both of which suggest adjectival morphology, the paradigm is closely related to that of the Imperfective suffix -m- (and 3Sg portmanteau -ŋ) with regular verbs ($\S10.xxx$). However, the pronominal endings with the Imperfective suffix are shorter, e.g. 1Sg Imperfective -m- \varnothing ~ -m-ı versus 1Sg 'it is' clitic =mi-y, phonetic [mi:].

(xx1)	category	postvocalic	postconsonantal (uncommon)
	1Sg 1Pl 2Sg 2Pl	=mi-y [mi:] =mi-y∴ [miîi] =mu-w [mu:] =mu-w∴ [mùúù]	[same as postvocalic] [same as postvocalic] [same as postvocalic]
	3Sg 3Pl Inan	$= \mathfrak{y} \sim :=^{n} \sim := \emptyset$ $= \mathfrak{y} \hat{\varepsilon} \sim = \mathfrak{y} \varepsilon$ $= \mathfrak{W}$	$=y\grave{\varepsilon} \sim =y\varepsilon$ $=y\grave{\varepsilon} \sim =y\varepsilon$ $=y\grave{\varepsilon} \sim =y\varepsilon$

I generally normalize the transcription of the 3Sg form to =ŋ. However, it may be realized phonetically as nasalization (and brief lengthening) of the final vowel, and (especially in an already nasal context) may reduce segmentally to zero.

3Sg =ŋ and Inanimate =w consist only of a consonant, and do not occur in postconsonantal position. However, postconsonantal position is rare for the 'it is' clitic since noun and adjective stems are essentially all vowel-final. Even a noun (from Fulfulde) like álâl 'Sunday' that are usually heard with final sonorant seem to be treated as vowel-final (with apocopated final high vowel), so we get 'it is' forms like álâlù=w 'it's Sunday'. True postconsonantal position is therefore effectively restricted to NPs ending in a numeral wŏy 'two', in a consonant-final determiner (wŏŋ 'this-Animate', yĕy 'these-Inanimate',

Definite Plural \circ), or in particle săy 'only'. After these true consonant-final elements, = η and Inanimate = ψ are replaced by the syllabic morpheme = ψ = ψ , which is otherwise (i.e. postvocalically) a 3Pl form.

The fact that nonmonosyllabic nouns and adjectives do not end in $/u/(\S3.xxx)$ is very helpful, since =w is easily audible after other vowels. Stemfinal /i/, which is quite common in nouns, combines with =w as [u:], as in $b \`o n d \'u = \'w \'i is rain' (b \'o n d \'i)$.

The 1Pl and 2Pl forms have their usual dying-quail intonation with [LHL] pitch on the clitic syllable, at least in careful pronunciation. This is most easily heard when the 'it is' form is followed by a particle, such as interrogative /ma/.

The 1Sg, 2Sg, 3Sg, and Inanimate 'it is' forms are **atonal**, meaning that they acquire a surface tone by spreading from the preceding morpheme. Thus $\frac{\partial g}{\partial m}$ 'I am a white person' with final high tone, but $\frac{\partial g}{\partial m}$ 'I am a Dogon' with final low tone; likewise $\frac{\partial g}{\partial m}$ 'he/she is a white person', $\frac{\partial g}{\partial m}$ 'he/she is a Dogon', $\frac{\partial g}{\partial m}$ 'it is a meal', and $\frac{\partial g}{\partial m}$ 'it is an egg' ($\frac{\partial g}{\partial m}$).

3Sg =ŋ and Inanimate =w are atonal in most combinations as just stated. However, when added to third person pronouns, demonstrative pronouns, and interrogative pronouns, they are **low-toned** even when the preceding form ends in a high tone: kò-pé= \hat{w}^n 'what is it?' (kò-pé), ńné= \hat{y} 'it's him/her' (ńné), \hat{y} gú= \hat{w} 'it's this/that' (\hat{y} gú). I am inclined to interpret this as a special grammatical low-tone element grafted onto the clitic, but the effect is that the 3Sg and Inanimate clitics have a marked low tone in these combinations.

The tonal behavior of $3Pl = y\hat{\epsilon} \sim y\hat{\epsilon}$ is equivocal in my data. After a final-high-toned noun, I have recorded both $y\hat{\epsilon}$ and $y\hat{\epsilon}$ (the latter includes surface $y\hat{\epsilon}$ after a rising-toned syllable, whose final high-tone element spills over), with $y\hat{\epsilon}$ predominating. For 'it's women' (stem $y\hat{\epsilon}$), $y\hat{\epsilon}$: $y\hat{\epsilon}$ is more common than $y\hat{\epsilon}$: $y\hat{\epsilon}$, though my assistant accepts both. For 'it's white people' (stem $y\hat{\epsilon}$), only $y\hat{\epsilon}$: $y\hat{\epsilon}$ was accepted. For 'it's sheep-Pl' ($y\hat{\epsilon}$), both $y\hat{\epsilon}$: $y\hat{\epsilon}$ and $y\hat{\epsilon}$: $y\hat{\epsilon}$ were recorded.

=yε (like Inanimate =w) also occurs in the inflection of stative verbs (§10.4.1). Here =yε (like =w) is atonal, and therefore appears with high tone when following a final-high-toned stem, as in $mb\acute{a}=y\acute{e}$ 'they love' (§11.2.6.3). Incidentally, in these stative paradigms, the 3Sg form is zero (not =η).

For occasional extensions of $=y\hat{\epsilon}$ to 1Pl and 2Pl subjects ($=y\hat{\epsilon}-m\hat{i}-\hat{y}...$, $=y\hat{\epsilon}-m\hat{u}-\hat{w}...$), see §11.xxx, below.

Examples with **animate subjects** are in (xx2). An initial independent pronoun (topicalized) is possible but not required, though it is common in the 3Pl (which has no suffixal expression).

```
'I am (a) Dogon.'
```

- b. (i:) dógô≡m-ĭy∴ (1Pl) Dogon≡it.is-1PlS 'We are Dogon.'
- c. (ńné) dógô=ŋ)
 (3Sg) Dogon=it.is.3SgS
 'He/She is Dogon.'
- d. (bû:) dógô≡yè (3Pl) Dogon≡it.is.3PlS 'They are Dogon.'
- e. pèrgé≡ý sheep=it.is.3SgS 'It is a sheep'
- f. äŋ≡∅ who?≡it.is.3SgS 'Who is it?' (from /ἄη≡η)/)
- g. [á:mádù sǎy]=mî-y
 [Amadou only]=it.is-1SgS
 'I am (= it's) just Amadou.'
- h. ńné [fatùmátâ săy]=yê
 3Sg [Fatoumata only]=it.is.3S
 'She is (= it's just) Fatoumata.'
- i. bû: [yù:-wàrî sǎy]=yê
 3Pl [millet.L-farm.Agent only]=it.is.3S
 'They are only millet farmers.'

In **identificational predicates** with an animate pronoun (even a first or second person pronoun) as predicate, as in 'it's me' (in answer to e.g. 'who ate the meat?' or 'who is that knocking at the door?'), the relevant pronominal category appears as the subject (expressed by pronominal-subject suffix) as well as the predicate (expressed as independent pronoun), hence literally 'I am me', etc. Recall the comment above that third person pronouns (along with demonstratives) have a final low tone in the 'it is' form; this accounts for 3Sg $\text{nn}\hat{\epsilon}=\hat{\eta}$ (xx3.d). The 3Pl is $\hat{\text{bu}}:=\emptyset$ with no overt clitic (xx3.e).

```
a. ĭ:n=mî-y
1Sg=it.is-1SgS
'It is me.'

b. ú=mú-w
2Sg=it.is-2SgS
'It is you-Sg.'

c. î:=mĭ-y...
1Pl=it.is-1PlS
'It is us.'

d. ńnέ=ŋ
3Sg=it.is.3SgS
'It is he/she.'

e. bû:=∅
3Pl-be.3PlS
'It's them.
```

Examples with **inanimate subject** are in (xx4). $\equiv w$ is nasalized to $\equiv w^n$ by regular Nasalization-Spreading when preceded by a nasal syllable.

```
(xx4) a. nàmâ≡w<sup>n</sup>
               meat≡it.is.InanS
               'It's meat.' (nàmâ)
          b. pòrú≡ẃ
               knife≡it.is.InanS
               'It's a knife.' (pòri)
          c. kúr<sup>n</sup>ô≡ẁ<sup>n</sup>
               stone≡it.is.InanS
               'It's a stone.' (kúr<sup>n</sup>ô)
          d. tùmá≡ẃ<sup>n</sup>
               tree≡it.is.InanS
               'It's a tree.' (tùmá)
          e. kờ-μέ≡ẁ<sup>n</sup>
                                          mà
               what?≡it.is.InanS
                                          Q
```

```
'What is it?
```

```
f. ŋgú [ně:m sǎy]=yê
Dem.InanSg [salt only]=it.is.3SgS
'That is just salt.' (nè:mî)
```

Identificational predicates with inanimate pronoun or demonstrative as predicate have $=\hat{\eta}$, i.e. 3Sg as subject, rather than the specifically inanimate =w. The same is true of interrogative pronouns. Perhaps the substitution was originally a device to avoid adding =w to any of the high-frequency grammatical morphemes already ending in /u. As noted above, some of these combinations are also irregular in having low-toned $=\hat{\eta}$ after a high tone.

```
(xxx) a. kú=ἢ
    InanSg=it.is.3SgS
    'It is (= that is) it.'
b. kû:=yè
    InanPl=it.is.3PlS
    'It is (=that is) them-Inan.'
c. àr<sup>n</sup>áŋá=ἢ
    where?=it.is.3SgS
    'where is it?' (compare àr<sup>n</sup>áŋá bù-∅)
d. ἢgú=ἢ
    Dem.InanSg=it.is.3SgS
    'it's this'
```

11.2.1.2 'It is not' (**=ndŏ:-**)

L-toned when clause-final

The negative counterpart of \equiv m- 'it is' is \equiv ndŏ:- 'it is not'. It has this form after consonants as well as vowels. It does not co-occur with -yɛ (cf. the following section). It is phonologically slightly distinct from Stative Negative -ndó-, which is used with other statives, but the two are probably related.

■ndŏ:- 'it is not' can be conjugated; the paradigm is (xx1). 3Sg (including Inanimate) is the zero category. There is a suffixed 3Pl form. The nasal in ■ndŏ:- gets its tone (not marked in the transcription) from the preceding stem, which is not tone-dropped.

```
form
(xx1). category
       1Sg
                      =ndŏ-y
       1Pl
                                    [ndòóòj]
                      ≡ndŏ-y∴
       2Sg
                      =ndŏ-w
       2P1
                      ≡ndŏ-w∴
                                    [ndòóòw]
       3Sg, Inan
                      ≡ndŏ:-Ø
       3P1
                      ≡nd-ě:
```

Examples are in (xx2).

(xx2) a. dógô≡ndŏ-y Dogon-**not.be**-1SgS 'I am not a Dogon.' (dógô) phonetic [dógôndôĵ]

- b. wŏŋ nàŋá≡ndŏ:-∅
 Dem cow-not.be-3SgS
 'That is not a cow.'
 phonetic [nàŋáńdòó]
- c. ŋgú [nĕ:m săy]≡ndŏ:-∅

 Dem.InanSg [salt only]-not.be-3SgS

 'That is not just salt.' (nɛ:mi)

 phonetic [sàjńdòó]
- d. [ànsá:rá sǎy]≡nd-ĕ: [white.person only]≡not.be-3PlS 'They are not only white people.' phonetic [sàjídèé]

11.2.1.3 Extensions of yέ-m- to 1Pl and 2Pl subject

The form $-y\epsilon$ is mentioned above as a postconsonantal allomorph of the 'be' clitic for third person subject, used for example after săy 'only'. A further example is (xxx).

```
(xx1) [bû: gày] [ànsá:rá sǎy]≡y€
[3Pl Topic] [white.person only]=it.is.3SgS
```

'Them, they're only white people.'

However, $\equiv y \varepsilon$ is morphologically rather obscure, as it fails to make the usual distinction between 3Sg, 3Pl, and inanimate categories. Not surprising, it shows signs of partial reanalysis as a nonpronominal element. There are (therefore) occasional attestations of $\equiv y \varepsilon$ - with following $\equiv m$ - 'it is' and a 1Pl or 2Pl pronominal-subject suffix. Thus (xx2.b) occurs as an optional variant of (xx2.a). My assistant rejected $\equiv y \varepsilon$ - with 1Sg and 2Sg suffixes.

```
(xx2) a. [î: ànsá:rá sǎy]≡m-ì∴
[1PlS white.person only]≡it.is-1PlS
'We are only white people.'
```

```
b. î: ànsá:rá sǎy≡y€≡m-ì∴

1Pl white.person only≡it.is=it.is-1PlS

[= (a)]
```

11.2.2 Existential and locative quasi-verbs and particles

11.2.2.1 Existential (yá)

This particle is used in conjunction with a following locational 'be' quasi-verb (bù-), or with a following 'have' quasi-verb (sò-). It is not used in negative clauses, or in (positive) relative clauses. See §11.xxx and §11.xxx for examples.

11.2.2.2 Locational quasi-verb (bù-, negative ngó-)

The 'be' quasi-verb used in positive contexts with an overt or implied locational expression ('be in the village', 'be here', 'be present', etc.), or in general existential sense ('exist, be somewhere'), is bù-.

(xx1).	category	form	
	1Sg 1Pl 2Sg 2Pl	 bù-ỳ ~ bǐ-ỳ bù-ỳ∴ ~ bǐ-ỳ∴ bù-w bù-w∴ 	[bùúùj]
	3Sg 3Pl	bù-∅ b-è	

Examples with specific locationals are in (xx2).

When there is no other locational complement, Existential yá precedes it.

The **negative** counterpart of bù- is $\eta g \circ f$ - 'not be (somewhere)'. Its paradigm is (xx4). The initial nasal is normally low-toned, but in the 1Pl and 2Pl the nasal and is $f \circ f$ are high-pitched.

(xx4)	category	form	
	1Sg 1Pl 2Sg 2Pl	ŋgó-ý ŋgó-ý∴ [ŋ́gòòj] ŋgó-ẃ ŋgó-ώ∴ [ŋ́gòòw	
	3Sg 3Pl	ὴgó-∅ ὴg-έ: ~ ὴgέ-yέ	

Existential yá is not used in negative contexts. Examples of ngó- with and without overt locational complements are in (xxx).

b. [ndó gó] ngó-ý [house in] not.be-1SgS 'I am not in the house.'

11.2.3 'Be put in/on' (kùrⁿò-, gàrⁿà-, nànà-)

To say that a person, animal, or a large or whole object (e.g. a grain spike) is 'in' a container or, metaphorically, 'in' a situation (such as being in post-partum quarantine), but not e.g. being in a village or in a house, the Stative verb k u n o 'be in' (also 'be put in') is used instead of b u. It is the Stative form of active transitive verb k u n o 'put (something) inside (a container)'. It occurs with an explicit locational in (xx1.a), with optional Existential yá. The Existential yá is required in the absence of an explicit locational (xx1.b).

- (xx1) a. nàmâ [sà:g gá] (yá) kùrⁿò-∅ meat [sack Def.InanSg.Loc] (Exist) be.in.Stat-3SgS 'The meat is in the sack.' (sá:gǐ)
 - b. nàmâ yá kùrⁿò-∅ meat Exist be.in.Stat-3SgS 'The meat is in (it).'

To say that a liquid or granulated substance (water, millet grain, flour, granulated sugar or salt) is 'in' a container (waterjar, grain sack), a different Stative verb garna- is put to use. The corresponding transitive is garna- 'put, place'. The syntax is as for kurno-.

- (xx2) a. nî: [jînjà gá] (yá) gàrⁿà-∅ sugar [waterjar.L Def.InanSg.Loc] (Exist) be.in.Stat-3SgS 'The water is in the waterjar.' (jînjá)
 - b. yû: [sà:g gá] (yá) gàrⁿà-Ø millet [waterjar.L Def.InanSg.Loc] (Exist) be.in.Stat-3SgS 'The millet (grain) is in the (grain) sack.' (sá:gǐ)
 - c. sigórò yá gàrⁿà-∅ sugar Exist be.in.Stat-3SgS 'The sugar is in (it).'

To say that an object is '(up) on' another object or raised surface, the Stative verb used is nànà- 'be (put) up on'. Contexts include putting a tea-kettle

on a burner, a cooking pot on a raised (three-stone) hearth with fire underneath, and putting anything (mattress, peanuts) on a roof. The syntax is as before.

- (xx3) a. màtĭlâ: [dèw gá] (yá) nàŋà-∅ sugar [roof.L Def.InanSg.Loc] (Exist) be.up.on.Stat-3SgS 'The mattress is up on the roof.' (dèwî)
 - b. tê: yá nàŋà-∅
 tea Exist be.up.on.Stat-3SgS
 'The tea (kettle) is up (on the burner).'

The three Stative locational verbs described here are the most important alternatives to bù- 'be'. However, the Stative form of verbs is productive and many others occur to describe more specific positions ('be hanging up', 'be lying down', etc.)

11.2.4 Morphologically regular verbs

11.2.4.1 'Remain, happen' (bě:-)

This morphologically regular verb has a full set of AN stems including Perfective-1a bě-èrè-, Perfective Negative bè:-rî-, and Imperfective bè-bé:-m̀-. For present time (stative), the perfective is used.

- (xxx) a. ké-kéw bě-èr-à
 Rdp-same remain-Perf1a-3PlS
 'They remain (= are) the same.'
 - b. ké-kéw bè-bé:-m-è Rdp-same Rdp-be-Impf-3PIS 'They will remain (= be) the same.'

11.2.4.2 'Become, happen' (táŋî-)

This morphologically regular verb has Perfective-1a táŋî-ere- (or táŋy-ere-), Perfective Negative tàŋà-rⁿî-, Imperfective tà-táŋî-m-, etc. In the sense 'become X', the X is most often a noun or NP, but can also be an adjective or a descriptive adverbial. (For 'become A' with adjective A, the inchoative derived verb is common; see §9.xxx).

```
(xxx) a. yă-ŋ táŋî-èrè-∅
woman-Sg become-Perf1a-3SgS
'He/She has become a woman.'
```

```
b. [[kìyǎw kô gù] mày<sup>n</sup>]
[[before Poss.InanSg Def.InanSg.L] like]
táŋî-èrè-w
become-Perf1a-2SgS
'You-Sg have become like before.' (lit. "like before's thing")
```

11.2.4.3 'Want, like' (jòró-, nàmà-, m^bùrá-)

'X want Y' denoting a momentary wish may be expressed by the morphologically regular verb jɔrɔ́-. This verb may also (in other contexts) be translated 'like, love', denoting an enduring attitude. An alternative is a verb that usually occurs in stative form as nama-, though Imperfective stem nama-mis also attested.

```
b. [kò nɛ] jóró-m-w mà
[what?] want-Impf-2SgS Q
'What do you-Sg want?' (or: 'What do you-Sg like/love?')
```

In positive utterances, stative nama- may be preceded by Existential ya. In the negative, it takes Stative Negative -ndó-, without ya.

```
    b. nî: nàmà-ndó-ý
water want-StatNeg-1SgS
'I don't want water.'
```

In the strong sense 'love, be very fond of (someone)', an irregular stative verb $\mathring{\mathbf{m}} b \mathring{\mathbf{a}} \sim \mathring{\mathbf{m}} b \mathring{\mathbf{a}} \sim \mathring{\mathbf{m}} b \mathring{\mathbf{a}}$ is also recorded: $\mathring{\mathbf{u}}^{-n} \mathring{\mathbf{m}} b \mathring{\mathbf{a}} = \mathbf{m} - \mathring{\mathbf{1}}$ 'I love you-Sg', $\mathring{\mathbf{n}} j \mathring{\mathbf{1}} \mathring{\mathbf{m}} b \mathring{\mathbf{a}} = y \mathring{\mathbf{c}}$ 'they love me'. The paradigm, which has some unusual tonal features, is (xx3).

(xx3) mbá- 'love'

category	form	
1Sg 1Pl 2Sg 2Pl	mbá≡mí-y ⁿ mbá≡mí-y ⁿ ∴ mbá≡mú-w ⁿ mbá≡mú-w ⁿ ∴	[m̀bámi:] [m̀bámi:] [m̀bámú:] [m̀bámù:]
3Sg/Inan	m̀bá-∅	
3Pl	ṁbá≡yέ	

The regular Stative Negative of this is mba=ndó- 'not love'.

A distinct, irregular negative stem mbură- is more common, often being used as the negation of jɔró-, but tending toward a lexically separate sense 'dislike, hate' rather than the mere absence of liking. A segmentation as mbură- is possible but not transparent.

(xx3) mbură- 'not want; dislike'

category	form	
1Sg 1Pl 2Sg 2Pl	mbùrà-ý mbùrà-ý∴ mbùrà-ẃ mbùrà-ẃ∴	[m̀búráàj] [m̀búráàw]
3Sg/Inan	m̀bùrá-∅	
3Pl	ṁbùrá≡yέ	

11.2.4.4 'Fear' (ú:-yí, ùwà-)

'X fear Y', 'X be afraid of Y' may be expressed by the morphologically regular verb $\acute{\mathbf{u}}$:- $\acute{\mathbf{y}}$ i-, which contains Mediopassive - $\acute{\mathbf{y}}$ i-.

```
(xx1) a. ńnś-ń ù:-yè-rî-ý
3Sg-Acc fear-MP-PerfNeg-1SgS
'I was not afraid of him/her.'
```

b. njî ú:-y-èrè-∅ 1SgO fear-MP-Perf1a-3SgS 'He/She was afraid of me.'

The Stative form is <u>uwa-</u>, or reduplicated <u>u-uwa-</u> (§10.4). It optionally takes Existential ya in the positive. The negative has the regular Stative Negative suffix, and does not allow the Existential particle. These Stative forms are used for states including the present time.

- (xx2) a. njî yá ùwà-yè 1SgO Exist fear.Stat-3PlS 'They are afraid of me.'
 - b. njî ùwà≡nd-€
 1SgO fear≡StatNeg-3PlS
 'They are not afraid of me.'

11.3 Quotative verb

11.3.1 'Say' (**k**iyέ-)

The fully inflectable verb 'say' is kiyé-.

- (xx1) a. [yèbùmbà wŏŋ], kð-kðsî kîyé-m-è [snake.L Dem.AnSg], viper say-Impf-3PlS 'This snake, they call it "viper".'
 - b. [kò kámâ] kìyè-rî-y [thing.L any] say-Perf.Neg-1SgS 'I didn't say anything.'
 - c. [ńné kĭyè-sè-∅ gú↓ pú⇒] kà-kàrú≡ẃ [3SgS say-Ppl.Perf-3SgS Def.InanSg.L all] Rdp-lie≡it.is.Inan 'Everything he/she said, it is false.'

For quotative complements, see §17.xxx. For jussives (embedded imperatives), see §17.xxx.

For uninflectable quotative particle wà, see §17.xxx.

11.4 Adjectival predicates

The predicates discussed here are stative in nature 'X is heavy', etc. Inchoatives ('become heavy') and factitives ('make it heavy') are expressed by fully inflectable derived verbs; see §9.xxx.

11.4.1 Positive adjectival predicates with 'it is' clitic

Adjectives may function as predicates when followed by the 'it is' clitic = \hat{m} -($\S xxx$). The 3Sg subject form is, however, zero instead of the usual portmanteau = \hat{n} .

The form of the adjective, specifically its final vowel, is often distinct in modifying as opposed to predicative function. Specifically, stem-final /i/ in modifying function corresponds to stem-final /u/ in predicative function: kùrnò dúsî 'a heavy stone', [kùrnò ŋgú] dúsû 'this stone is heavy'. In the case of 'difficult' (and no other adjective known to me), there is also a tonal change, the final syllable appearing as high-rather than falling-toned: kò: nòmî 'a difficult thing', [kò: ŋgú] nòmú 'this thing is difficult'. For a list of adjectives in modifying and predicative function, see §4.5.1.

Examples with **first and second person** subjects are in (xx1). Optional independent pronouns at the beginning are topicalized.

```
(xx1) a. (î:n) dúsû≡m-ĭ
(1Sg) heavy≡it.is-1SgS
'I am heavy.'

b. (û:) èwré≡m-ùw∴
(2Pl) small≡it.is-2PlS
'You-Pl are small.'
```

Examples with **3Sg subject** are in (xx2). Here the expected 3Sg 'it is' clitic = n is absent. Again, an independent pronoun is optional.

```
(xxx) a. (ńné) dúsû (3Sg) heavy 'He/She is heavy.'

b. (ńné) èwré (3Sg) small 'He/She is small.'
```

```
c. [kà: gú] èwré

[thing.L Def.InanSg] small

'The thing is small.'
```

For 3Pl, the suffix $-y\hat{\epsilon}$ is added (without 'it is' clitic =m-). After stem-final $/\epsilon$ /, the semivowel tends to elide, resulting in a final $/\hat{\epsilon}$:/.

```
dúsû-yè
(xxx) a. (bû:)
                        heavy-be.3PIS
            (3Pl)
            'They are heavy.'
        b. (bû:)
                        èwrέ-yè
            (3Pl)
                        small
            'They are small.' [variant èwrê:-∅]
incorporate
        [kàrnâ ú
                                 'njî-η
                                         έrû
                         gŝ]
        'Your action pleases me.'
        [kàr<sup>n</sup>â ú
                         gŝ]
                                 njî-ŋ
                                         èr≡ndó
        'Your action doesn't please me.'
```

11.4.2 Negative adjectival and stative predicates (≡ndó-)

Negative counterparts of the positive adjectival predicates illustrated above are formed by adding an inflected form of Stative Negative clitic =ndó- to the adjective stem, which drops its tones.

```
(xxx)
        a. (\acute{n}n\acute{\epsilon})
                         dùsù≡ndó-Ø
            (3Sg)
                         heavy.L≡StatNeg-3SgS
             'He/She is heavy.'
        b. (bû:)
                         èwrè≡nd-έ
                         small.L≡StatNeg-3PlS
            (3Pl)
             'They are not small.'
        c. (i:^n)
                         dùsù≡ndó-ý
                         heavy.L≡StatNeg-1SgS
            (1Sg)
             'I am not heavy.'
```

11.5 Possessive predicates

11.5.1 'Have' (sò-)

Positive '(X) have Y' is expressed as [Y yá sò-], with Existential particle yá and defective quasi-verb sò-, which takes pronominal subject suffixes but allows no overt marking of aspectual categories. Like other statives, it does combine with Past $\equiv b\tilde{e}$ -, which in this combination oddly requires what appears to be Imperfective -m- (xx1.b) except in the 3Sg, see §10.5.1.2, above.

- (xx1) a. lègèsô: yá sò-ỳ bicycle Exist have-1SgS 'I have a bicycle.'
 - b. lègèsô: yá sò-m≡bè-ỳ
 bicycle Exist have-Impf=Past-1SgS
 'I had (= used to have) a bicycle.'
 - c. yă: [ndò dùgi] yá s-è woman [house.L big] Exist have-3PIS 'The women have a big house.'

sò- can be relativized on (xx2). This is the only instance where it appears in H-toned form (só-). Existential yá is omitted.

- ĭ:n só-m-sè (xx2) a. [wàgàti lègèsô: gàl bicycle have-Impf-Ppl.Perf [time.L 1SgS Loc [àsú⇒ á:ndε̂≡∅ ńnî-m̀≡bε-y] go-Impf≡Past-1SgS] always Anda 'When I had a bicycle, I used to go to Anda all the time.'
 - b. [àrnà [ídô wǒy] só-mǐ] ðgð-yî:≡ŋ̂
 [man.L [house two] have-Ppl.Impf] chief.L-child≡it.is.3SgS
 'A man who has two houses is a rich person.'

The **negative** of sò- is sò-ndó-, cf. Stative Negative clitic =ndó- Existential yá is not allowed in negatives (xx3). (xx3.b) is one of the most common utterances in Nanga speech.

(xx3) a. lègèsô: sò-ndó-y bicycle have-StatNeg-1SgS 'I don't have a bicycle.' b. dăyⁿ sò-ndó-∅ limit have-StatNeg-3SgS 'It has no limit (= is abundant).'

The positive and negative **paradigms** are in (xx4).

```
'have'
                                          'do not have'
(xx4)
       category
                                          sò-ndó-ý
       1Sg
                  (yá) sò-ỳ
       1P1
                  (yá) sò-ỳ∴ [sòóòj]
                                          sò-ndó-ý∴ [sòóńdòòj]
       2Sg
                                          sò-ndó-ẃ
                  (yá) sò-w
       2P1
                  (yá) sò-ẁ∴ [sòóòw]
                                          sò-ndó-w∴ [sòóńdòòw]
       3Sg/Inan
                 (yá) sò-∅
                                          sò-ndó-Ø
       3P1
                  (yá) s-è
                                          sè-nd-έ
```

11.5.2 'Belong to' predicates

'X belongs to Y' is expressed as 'X, it is Y's thing (possession)'. The X NP is optional, and normally preposed (like a topic) when overt. Y appears in possessor form, hence with $k\hat{o}$ (or $g\hat{o}$) for an inanimate possessed entity and $y\hat{e}$ for an animate one ($\S6.xxx$).

- (xxx) a. ŋgú [ặŋ kô-]≡ŋ
 Dem.InanSg [who? Poss.InanSg]≡it.is.3SgS
 'That is whose (= belongs to whom)?'
 - b. [nàkòmbò gú] [á:mádù kò]≡ŋ [shoulder.bag.L Def] [Amadou Poss.InanSg]≡it.is.3SgS 'The shoulder bag is Amadou's.'
 - c. $[n \epsilon r^m i \quad n \epsilon]$ $[an \quad y \epsilon] = i \gamma$ $[dog.L \quad Dem.AnSg]$ $[who? \quad Poss.AnSg] = it.is.3SgS$ 'This dog belongs to who(m)?'
 - d. ŋgú kɔ̃:≡ŋ
 Dem.InaSg 1SgPoss.Inan≡it.is.3SgS
 'That is mine.'

11.6 Verb iteration

11.6.1 Uninflected iteration of type $[\hat{v}_1-\hat{v}_1(-\hat{v}_1...)]$ in narratives

In narrative, a common device for backgrounding a durative activity (often motion), setting up a following foregrounded event predication, is to iterate the activity verb one or more times, without inflectional suffixation. The first occurrence has superimposed {HL} tone, and the following iterations are all-low-toned.

In the sample text, note kémè-kèmè-kèmè '(they) kept building and building' from verb kémé- (§xxx), jórò-jòrò-jòrò-jòrò '(he) was looking and looking' from verb jòró- (§xxx), and yê:-yè: '(as they) were coming' from verb yě:- (§xxx).

With a trisyllabic or longer verb, the high tone of {HL} is limited to the first syllable: bɛ́giri-bẽgiri-bẽgiri '(they) kept winnowing' from verb bẽgiri-.

12 Comparatives

12.1 Asymmetrical comparatives

12.1.1 Simple adjective with dé:rè≡w 'than' and comparandum

In one construction, the predicate is an adjective. The comparandum precedes the predicate, with dé:rè=w.

For first and second person subject,, the predicative adjective is followed by the appropriate conjugated form of $\equiv m$ - 'it is' clitic.

```
(xx1) a. [ńné dé:rè≡w] gàwá≡m-îy

[3Sg more] long≡it.is-1SgS

'I am taller than he/she (is).'
```

- b. ĭ:ⁿ [ú dé:rè=ẁ] dúsû=m-ĭy 1SgS [2Sg more] heavy≡it.is-1SgS 'I am heavier than you-Sg.'
- c. [î: dé:rè≡ẁ] dùgú≡m-ùw∴ [1Pl more] fat≡it.is-2PlS 'You-Pl are fatter than we (are).'

For 3Pl subject, the verb lacks =m- but is inflected with clitic =y ϵ (xx2). This is the regular (non-comparative) 3Pl predicative form ('they are fat').

```
\begin{array}{cccc} (xx2) & a. & \  \  \, [i:^n & d\acute{e}:r\grave{e}=\grave{w}] & d\grave{u}g\acute{u}=y\acute{e} \\ & [1Sg & more] & fat=3PlS \\ & `They are fatter than I (am). ` \end{array}
```

```
b. [i:n dé:rè=w] dúsû=yè
[1Sg more] heavy=3PlS
'They are heavier than I (am).'
```

For 3Sg subject, the **bare adjective** form is used (one might alternatively transcribe with a zero 3Sg suffix). This is again the regular (non-comparative) predicative form.

- (xx3) a. pà:ŋgŏ: [tà-tã: dé:rè≡w] dùgú elephant [Rdp-hyena more] big 'An elephant is bigger than a hyena (is).'
- b. [ĭ:n dé:rè≡ẁ] gùró
 [1Sg more] long
 'He/She is taller than I (am).'

For **past time**, the Past clitic $=b\epsilon$ - is added. For first or second person subject, the result is either $=b\epsilon$ - or $=m=b\epsilon$ - plus the correct pronominal-subject inflection. $=b\epsilon$ - (as well as the optional =m-) acquires its tone from the final tone of the preceding morpheme. For 3Sg, the form is $=b\epsilon$ - \varnothing without =m-. For 3Pl, the form is $(=y\epsilon)=b$ -à where $=y\epsilon$ - gets its tone from the left but where =b-à is always low-toned. A variant $=y\epsilon=m=b$ à was recorded early in the fieldwork but was later rejected by the same assistant. Excluding this doubtful variant, the paradigm is (xx4).

(xx4) Past of comparative adjectival predicate

category	suffix-clitic complex (after adjective stem)		
	after H-tone	after L-tone	
1Sg	(≡ ḿ) ≡ bέ-ý	(= m̀) = bɛ̀-ỳ	
1Pl	(≡ḿ)≡bέ-ý∴	(= m̀) = bὲ-ỳ∴	
2Sg	(≡ḿ)≡bέ-ẃ	(≡m̀)≡bὲ-ẁ	
2Pl	(≡ḿ)≡bέ-w∴	(≡m̀)≡bὲ-w∴	
3Sg	= bέ-∅	≡bὲ-Ø	
C			
3Pl	(≡ yέ) ≡ b-à	(= yɛ̀) = b-à	

Examples of the past-time positive comparative are in (xx5).

- (xx5) a. kĩyǎw [ńnɛ́ dé:rè=ẁ] dùgú(=m)=bɛ́-y before [3Sg more] fat(=it.is)=Past-1SgS 'I used to be fatter than he/she (was).'
 - b. [ĭ:n dé:rè≡w] gùró≡bέ-∅
 [1Sg more] long≡Past-3SgS
 'He/She was taller than I (was).'
 - c. [î: $d\acute{e}:r\grave{e}=\grave{w}$] $j\acute{e}m\i(=y\acute{e})=b-\grave{a}$ [1Pl more] black(=it.is.3Pl)=Past-3PlS

'They were blacker (=darker) than we (were).'

- d. [ńnś dé:rè=w] dúsû(=m)=bè-y
 [3Sg more] heavy(=it.is)=Past-1SgS
 'I used to be heavier than he/she (was).'
- e. [ĭ:n dé:rè=w] dúsû=bè-∅ [1Sg more] heavy=Past-3SgS 'He/She used to be heavier than I (was).'
- f. [i:n dé:rè=w] dúsû(=yè(=m))=b-à
 [1Sg more] heavy(=it.is.3Pl(=it.is))=Past-3PlS
 'They used to be heavier than I (was).'

The adjective may be directly **negated** with Stative Negative = ndo- 'not be', which requires L-toned stem. For the paradigm of = ndo-, see §10.4.2. The stem-final high tone in the 1Pl and 2Pl forms before = ndo- is applicable here.

- (xx6) a. [ĭ:ⁿ dé:rè≡w] gàwà=ndó-∅ [1Sg more] tall.L=StatNeg-3Sg 'He/She is not taller than I (am).'
 - b. [ú dé:rè=w] dùgù=ndó-ý
 [2Sg more] fat.L=StatNeg-1SgS
 'I am not fatter than you-Sg (are).'
 - c. [û: dé:rè=w] dùgú=ndó-ý∴ [2Pl more] fat.L=StatNeg-1PlS 'We am not fatter than you-Pl (are).'
 - d. [i:n dé:rè=w] dùsù=nd-£
 [1Sg more] heavy.L=StatNeg-3PlS
 'They are not heavier than I (am).'

The **past negative** is =ndó=b ε - with the usual morphology of Past =b ε -. The paradigm is (xx7).

(xx7) Past Negative of comparative adjectival predicate

category suffix-clitic complex (after adjective stem)

1Sg ≡ndó≡bέ-ý

```
1Pl =ndò=b-y∴

2Sg =ndó=bε-ẃ

2Pl =ndò=bε-w∴

3Sg =ndó=bε-Ø

3Pl =nd-ε=b-á
```

Examples of the past negative are (xx8).

```
(xx8) a. [ĭ:<sup>n</sup> dé:rè=ẁ] gàwà=ndó=bé-∅

[1Pl more] tall.L=StatNeg=Past-3SgS

'He/She was not taller than I (was).'
```

```
b. [î: dé:rè=ẁ] dùsù=nd-é=b-á
[1Pl more] heavy.L≡StatNeg-3PlS=Past-3PlS
'They were not heavier than I (was).'
```

12.1.2 Verbal predicate plus dé:rè≡w 'than'

In (xx1), the predicate is a verb, imperfective (xx1.a) or perfective (xx1.b). There is no explicit 'more' adverb. $d\acute{e}:r\grave{e}=\grave{w}$ 'than' follows the comparandum. The unmarked interpretation is 'X VP's more (than) Y'.

```
(xx1) a. [ú dé:rè≡ŵ] kô:-<sup>n</sup>
[2Sg more] eat-Impf.3SgS
'He/She eats more than you (eat).'
```

```
b. [ú dé:rè≡ẁ] njî-ŋ ndî-∅

[2Sg more] 1SgO-Acc give.Perf.L-3SgS

'He/She gave me more than you-Sg (gave me).'

or: 'He/She gave me more than (he/she gave) to you.'
```

The comparandum may take the form of a PP, such as the dative in (xx2.a), which is followed by $d\acute{e}:r\grave{e}=\grave{w}$. Accusative $-\grave{\eta}$ is optionally used before $d\acute{e}:r\grave{e}=\grave{w}$ when the form functions as direct object (xx2.b). This theoretically allows the speaker to distinguish (xx2.b) from (xx2.c), but Accusative $-\eta$ is optional in (xx1.b) as elsewhere, so this cue is far from reliable.

```
(xx1) a. [[ú báy] dé:rè≡ẁ] bàr<sup>n</sup>i kîyé-só-∅
[[2Sg Dat] more] eat-Impf.3SgS say-Perf2-3SgS
'He/She said more to me than to you.'
```

- b. [ú-ŋ dé:rè≡ẁ] njî-ý súyó-só-∅ [2Sg-Acc more] 1Sg-Acc hit-Perf2-3SgS 'He/She hit me more than (he/she hit) you-Sg.'
- c. [ú dé:rè≡ẁ] njî-ŋ súy5-só-∅ [2Sg more] 1Sg-Acc hit-Perf2-3SgS 'He/She hit me more than (he/she hit) you-Sg.'

12.1.3 'Surpass' (láwá-)

láwá- 'pass (by)' can be used in the sense 'surpass, exceed'. It specifically denotes the transition from equality or inferiority to superiority in the relevant dimension. The latter is specified by a nonfinal chained verb or VP.

- (xx1) a. njî-j gawa lawa-êrê-∅
 1Sg-Acc be.tall pass-Perf1a-3SgS
 'He/She has surpassed me in tallness (= has become taller than me).'
 - b. li:gi nji-n giyé láwá-só-∅ bird 1Sg-Acc kill pass-Perf2-3SgS 'He/She has surpassed me in killing birds.'

12.1.4 'Be better, more' (dé:rè-)

In this construction, dé:rè- itself is conjugated by adding the conjugated 'it is' clitic =m- with first or second person inflection, or 3Sg =w or 3Pl =y\varepsilon.

- (xx2) a. ú dé:rè=m-ì: 2SgO better=it.is-1SgS 'I am better than you-Sg (are).'
 - b. măŋgórò kùró dé:rè≡ẁ
 mango wild.grape better-be.3SgS
 'Mangoes are better than wild grapes (are).'
 - c. ĭ:ⁿ dé:rè=yè 1SgO better-be.3PIS 'They are better than I (am).'

For past time reference, the forms take Past clitic $=b\hat{\epsilon}$ -, but there is some variation in the morphological construction in my data. My assistant preferred the paradigm in (xx2). However, for the first and second person forms, a variant with $=\hat{m}=$ instead of $=\hat{m}=$ was also recorded.

(xx2) 'Was better than'

```
1Sg dé:rè≡ẁ≡bὲ-ỳ
1Pl dé:rè≡ẁ≡bè-ỳ∴ [...bèéèj]
2Sg dé:rè≡ẁ≡bè-ẁ
2Pl dé:rè≡ẁ≡bè-ẁ∴ [...bèéèw]
3Sg dé:rè≡ẁ≡bè-∅
3Pl dé:rè≡ẁ≡b-à
```

An example is (xx3).

```
(xx3) ú dé:rè≡ẁ≡bè-ỳ
2SgO better≡it.is≡Past-1SgS
'I used to be better than you-Sg.'
```

12.1.5 'Best' (kǎy)

The noun kǎy is used to specify that the referent in question is the best (of a set). The conjugation is the same as for dé:rè- except that 3Sg = w is not pronounceable after the stem-final semivowel. The paradigm therefore consists of conjugated forms of =m- 'it is' for first or second person, zero for 3Sg, and $=y\varepsilon$ for 3Pl. Thus kǎy=m-iy 'I am the best', kǎy= \emptyset 'he/she is the best', kǎy= $y\varepsilon$ 'they are the best'.

Past forms: $k\check{a}y=\check{m}=b\acute{\epsilon}-\acute{y}$ 'I am the best', $k\check{a}y=b\acute{\epsilon}-\varnothing$ 'he/she is the best', $k\check{a}y=y\acute{\epsilon}=b-\grave{a}$ 'they are the best'.

When the reference set is specified, it functions as possessor of kay, which therefore takes possessed-noun tone contour, {HL} or all-low depending on whether the possessor ends in a high or low tone.

```
(xx1) ĭ:<sup>n</sup> wòrĭ-wàrí pú⇒ kây≡m-ĭy
1SgS farming.L-do.farming.Agent all best.HL≡it.is-1SgS
'I am the best of all the farmers.'
```

12.2 Symmetrical comparatives

12.2.1 'Equal; be as good as' (bă:-)

The verb bă: 'equal, be as much as' most often occurs in negative sentences ('not be as much as' = 'be less than'). When the domain of comparison is specified in a following expression, negation is expressed on the latter rather than on bă: itself, the effect being to reduce bă: to a particle-like status.

- (xxx) a. [á bâ:] bá:-ŋɔ̂:-∅ [ReflP father] equal-Impf.Neg-3SgS 'He/She isn't as good as his father.'
 - b. [á bâ:] bă: gáwá-ŋɔ:-∅ [ReflP father] equal be.tall-Impf.Neg-3SgS 'He/She is not as tall as his/her father.'

12.2.2 'Same (equal)' (kêw)

Another way to indicate equality along a measure is to use kêw 'each, all' as a predicate, reminiscent of the adverb ké-kéw 'same, equal'. The form is always kêw-yè, which I interpret as having adjectival Plural -yè.

- (xx1) a. [î: wŏy] îgîrî kêw-yè [1Pl two] height same-Pl 'We two are of the same height.'
 - b. [sèŋǐ ý] [â: wŏy] gùró kêw-yè [rope.L Def.InanPl] [ReflPl two] long same-Pl 'The two ropes are of the same length.'

12.2.3 'Attain, equal' (dš:-)

dŏ:- 'arrive at, reach (a place)' can be used in the abstract sense 'attain the level of (someone, in some respect).' It denotes the transition from inferiority to equality. The domain of comparison may be specified by a chained VP or verb.

(xx1) njî-ý dùsú-ndîyé dŏ-èrè-∅ 1Sg-Acc heavy-Inch arrive-Perf1a-3SgS 'He/She has become equally heavy as me.'

12.3 'A fortiori' (sákô, yê:)

The 'a fortiori' expression (i.e. 'much less ...', 'not to mention ...', or 'never mind ...' as part of a comparison) is either sákô (from Fulfulde) at the end of the second phrase, or yê: (shared with Jamsay) at the beginning of the second phrase. This phrase may also contain the possessed noun dàmá 'talk (of ...)' or some similar expression (cf. local French *ne parlons pas de* ... as an 'a fortiori' expression). With yê: the second phrase specifies a far greater or more difficult task (xx1.a). With sákô the second phrase may be of this type, or it may simply describe a less likely or less appropriate task (xx1.b).

```
(xx1) a
            [[kè:rè
                         bérî
                                  éwè-mì]
                                                 sò-ndó-ý]
                                                have-Neg-1Sgs]
                                  buy-Ppl]]
            money
                         goat
                                        dámà]≡w<sup>n</sup>]
            [yê:
                           [nàηá
                           cow
                                        talk.HL]≡it.is.Inan
            [a.fortiori
            'I don't have the money to buy a goat, much less (talk of) a cow.'
```

```
b. [kö: éwé bèrè-rî-ý]
[1SgP.Poss.InanSg buy get-PerfNeg-1SgS]
[[ú gô] sákô]
[[2Sg Poss.InanSg] a.fortiori]
'I couldn't buy one for myself, never mind (me buying) one for you.'
```

13 Focalization and interrogation

13.1 Focalization

When a focalized constituent is present, it is preverbal but not necessarily clause-initial

The remainder of a clause beginning with a focalized constituent undergoes modifications of a simplificatory nature (xx1), but the verb has regular subject-pronominal inflection.

- (xx1) a. Imperfective and Progressive are both possible, but the optional reduplication in the Imperfective is omitted
 - b. unsuffixed Perfective (bare stem with all-L tones) replaces the suffixally marked Perfective-1a/b or Perfective-2
 - c. Perfective Negative -ri- is usually low-toned
 - d. Imperfective Negative verbs (preceding suffix -ŋɔ̀:-) are often all-low toned
 - e. pro forma cognate nominal-objects are usually omitted

In imperfective examples like é:ŋî yé:-mò-Ø 'I will come tomorrow', we cannot be sure whether é:ŋî 'tomorrow' is focalized ('It is tomorrow [focus] that I will come') or unfocalized ('I will come tomorrow'). In practice, focalization is most easily detected in positive perfective clauses, since the unsuffixed Perfective is specifically associated with focalization. With other inflectional categories,

13.1.1 Subject focalization

The verb has its usual pronominal subject suffix. In (xx1.a), the focalized constituent is a pre-clausal topic NP, and is resumed within the clause proper by a 3Pl pronoun.

```
(xx1) a. yǐ-tègê, bû: bîré-m-è child-Pl 3Pl work-Impf-3PlS 'The children, it's they [focus] who work.'
```

```
    b. ĭ:<sup>n</sup> sémbî-m-Ø
    1Sg sweep-Impf-1SgS
    'It's <u>I</u> [focus] who will sweep.'
```

- c. î: sémbî-m-ì:

 1Pl sweep-Impf-1PlS

 'It's we [focus] who will sweep.'
- d. ú yègè-w
 2SgS fall.Perf-2SgS
 'It's you-Sg [focus] who fell.'
- e. ĭ:ⁿ ú-íj sùyò-ỳ
 1SgS 2Sg-Acc hit.Perf.L-1SgS
 'It was <u>I</u> [focus] who hit you-Sg.'

13.1.2 Object focalization

The object NP or pronoun is preposed. It may have Accusative marking, and 1Sg object is obligatorily expressed by the special 1Sg Accusative nji-n.

- (xxx) a. ŋgú jórð-sô-y
 Dem.InanSg look.for-Perf-1SgS
 'That [focus] is what I'm looking for.'
 - b. ńné-ŋ sùyò-y
 3Sg-Acc hit.Perf-1SgS
 'It is him/her [focus] that I hit.'
 - c. ú-ⁿ nîrⁿé-m-ĭ:
 2Sg-Acc look.at-Impf-1PlS
 'It's you-Sg [focus] that we will look at.'
 - d. njî-ý sùyè-∅ 1Sg-Acc hit.Perf.L-3SgS 'It was me [focus] that he/she hit.'

13.1.3 Focalization of PP or other adverbial

A spatial or temporal adverbial, such as a PP, is focalized in (xx1).

```
\begin{array}{cccc} (xx1) & a. & \begin{tabular}{ll} $\hat{o}$: & $g\acute{o}$ & $\hat{m}\hat{n}$-$\hat{m}$-$\emptyset \\ & & [field.L & in] & go-Impf-1SgS \\ & & `It's $\underline{to}$ the $\underline{fields}$ [focus] that $I$ am going.' \\ \end{array}
```

b. é:ŋî [ò: gó] ńnî-m̀-∅ tomorrow [field.L in] go-Impf-1SgS 'It's tomorrow [focus] that I will go to the field(s).'

A dative example is (xx2.a), and an instrumental example is (xx2.b).

b. [ŋgú yàŋà] bìrè-y
[Dem.InanSg with] work.Perf.L-1SgS
'It was with this [focus] that I worked.'

13.1.4 Focalization of postpositional complement

The full postposition is focalized as a unit; there is no way to distinguish focalization of a PP from focalization of just its NP complement.

13.2 Interrogatives

13.2.1 Polar (yes/no) interrogatives (ma)

The interrogative particle /ma/ is added at the end of an otherwise indicate sentence. It is optionally prolonged (symbol \Rightarrow). It is of variable pitch because of intonational effects (see below on parallelistic constructions).

It is heard as mà with **low tone** after verbs, adjectives, and anything else that ends in a low tone. Interestingly, it also has low tone after verb forms containing Perfective-2 -só- that end in a high tone, like 2Sg -só-w in (xx1.c). The Perfective-2 is the only positive indicative inflection with a final high tone on the suffix, so one can generalize that all positive indicative verbs are followed by low-toned mà.

- b. £:ŋî yé:-m-ùw mà tomorrow come-Impf-2PIS Q 'Are you-Pl coming tomorrow?'
- c. yě:-só-ẃ mà come-Perf2-2SgS Q 'Did you come?' (likewise 1Sg yě:-só-ý mà, 3Sg yě:-só-∅ mà)
- d. dúsû mà be.heavy Q 'Is he/she heavy?'
- e. yé:-ŋɔ̂-ẁ¹ mà come-ImpfNeg-2SgS Q 'You-Sg will not come?'

High-toned má occurs after all other words ending in a high tone. For example, it occurs after final-high-toned words containing (or ending in) any of the main negative forms except the Imperfective Negative -ŋò:-. High tone is heard after forms ending in a high tone based on a negative predicate: Perfective Negative -ri-, ŋgó- 'not be', Stative Negative -ndó- (with adjectives or Stative verbs), or -ndŏ:- 'not be' (with nouns).

- (xx2) a. yè:-rî-Ø má
 come-Perf.Neg-3SgS Q
 'Did he/she not come?'
 - b. nă: kò:-rú-w má
 meal eat-Perf.Neg-2Sg Q
 'Have you-Sg not eaten?'
 - c. ŋgá ŋg-ɛ: má
 here not.be-3PIS Q
 'Are they not here?'
 - d. dùsù≡ndó-∅ má heavy.L≡StatNeg-3SgS Q 'Is he/she not heavy?'
 - e. dógð≡ndŏ:-ẃ má Dogon=not.be-2SgS Q

'You-Sg are not a Dogon?'

The high-toned form is also used after predicative adjectives that end in a high tone: dùgú má 'is he/she big?'

When both polar alternatives are presented, /ma/ appears after the first segment with pitch raised (\uparrow) and often with the vowel prolonged (\Rightarrow). The second segment may be complete, exactly parallel to the first (but ending in /ma/ without much intonational modification). Alternatively, the second segment may omit /ma/ (this is rather common), and may also show other simplifications.

```
(xx3) a. nimî jóró-m-w mà⇒↑, nàmâ jóró-m-w cow-peas want-Impf-2SgS Q, meat want-Impf-2SgS 'Do you like cow-peas, (or) meat?'
```

```
b. ńni-m-w mà⇒, bé:-m-w mà
go-Impf-2SgS Q, stay-Impf-2SgS Q
'Will you-Sg go, (or) will you-Sg stay?'
```

As also noted in (§7.2), there is no clear distinction between interrogative mà and the 'or' disjunctive particle mà in such parallel polar interrogatives.

13.2.2 'Who?' (ăŋ)

In WH-interrogatives, the interrogative particle mà occurs optionally (but redundantly) at the end of the clause.

The 'who?' word is /āŋ/. It is (less often) expanded as nù ǎŋ, with the preadjectival form of nü: 'person'. The /ŋ/ is subject to assimilation to the position of a following consonant, hence [ǎm], [ǎn], and [a˙jⁿ] before various following consonants. In isolation it can also be pronounced [a˙n] or [a˙sⁿ]. I normalize transcription as a˙ŋ.

```
(xx1) a. [kè:rè gú] [ǎŋ báy] ǹdi-ẁ [money.L Def.InanSg] [who? Dat] give.Perf.L-2SgS 'To who(m) did you-Sg give the money?
```

c. ăŋ nàŋá jè:-
$$\varnothing$$
 mà

```
who? cow bring.Perf.L-3sgS Q 'Who brought the cow?'
```

- d. ăŋ ŋgà-gá yì:-w mà
 who? there see.Perf.L-2SgS Q
 'Who(m) did you-Sg see there?'
- e. [[nù ăŋ] yàŋà] y-ò:
 [[person.L who? with] come.Perf.L-3PlS
 'Who did they come with?'

Predicative function ('who is ...') is expressed by ǎŋ=ŋ (pronounced ǎŋ]). This presumably includes some form of the 3Sg form of the 'it is' clitic (§11.xxx). However, one cannot reliably hear the second velar nasal, especially since the velar nasal(s) assimilate to a following consonant. The only reliably audible indicator of predicative status is the <LHL> tone of the syllable, and there is evidence (see below) that a final low-tone element is the real morphological feature here.

```
(xx2) a. ăŋ≡ŋ who?≡it.is.3SgS 'Who is he/she?'
```

```
b. ăŋ≡ỳ mà
who?≡it.is.3SgS Q
'Who is it?' (pronounced [àḿmà])
```

If an overt "subject" NP is present, it is topicalized (xx3).

```
(xx3) wŏŋ [ăŋ≡ŋ mà]
Dem.AnSg [who?≡it.is.3SgS Q]
'This/that (person), who is it?' (= 'Who is that?')
```

There is a **plural predicative** 'who-Pl is/are ...?' form <u>ă</u>:=yè (xx4). Again, note the final low tone.

```
(xx4) ă:≡yè mà
who?≡it.is.3Pl Q
'Who-Pl is it?
```

With first or second person "subject," we get forms like those in (xx5), where conjugated clitic =m- 'it is' is added to $\check{a}\eta$. The $n\grave{u}$ in (xx5.a) is optional.

Note especially the final falling tone in the 2Sg form (xx5.a) and the 1Sg form (xx5.b), which are therefore followed by low-toned mà. So there are clear indications that a final low-tone element is characteristic of predicative forms of 'who?'.

```
(xx5) a. ú [nù ăŋ]=m-ûw mà
2Sg [person.L who?]=it.is-2SgS Q
'Who are you-Sg?'
```

13.2.3 'What?' (kò-μέ), 'with what?', 'why?'

'What?' is kò-nɛ́ (xx1). The first syllable is presumably related to kó ~ kón 'thing', leaving -nɛ́ as the real interrogative element (historically). The combination is rather frozen synchronically.

```
(xx1) a. kð-né bîrè:-sò-w
what? work-Prog-2SgS
'What are you-Sg doing?'
```

```
b. kò-né kó:-m-iy
what? eat-Impf-1PlS
'What are we going to eat?'
```

'With what?' (instrumental) is kò-nế ỳŋà (xx2), contracted from *kò-nế yàŋà.

```
(xx2) [kònê:-ŋà] wárá-m̀-w

[what?-with] farm-Impf-2SgS

'With what do you-Sg farm?'
```

'For what?' = 'why?' is kò-né dèrⁿí, with Purposive postposition (xx3).

```
(xx3) [kò-né dèr<sup>n</sup>í] yè-w

[what? for] come.Perf.L-2SgS

'Why did you-Sg come?'
```

/nné/ by itself can be used as the **plural** 'what (things)?', but it is fairly uncommon (xx4). The initial /nn/ cluster is of historical phonological interest; see \S

The **predicative** form is seen in (xx5)

The ending in $k\eth -n\acute{\epsilon} = \grave{w}^n$ is the Inanimate conjugated form of the 'it is' clitic, namely /=w/, which elsewhere gets its tone from the preceding morpheme (§11.2.1.1). The fact that it appears as low-toned $=\grave{w}^n$ in $k\eth -n\acute{\epsilon} = \grave{w}^n$ is more evidence for a final low-tone feature in predicative interrogatives; compare the predicative forms of 'who?' discussed in the preceding section.

```
13.2.4 'Where?' (àr<sup>n</sup>áŋá)
```

'Where?' is àrⁿáná. Adverbial examples are in (xx1).

```
(xx1) a. àr<sup>n</sup>áŋá ńnî-m̀-<sup>w</sup>
where? go-Impf-2SgS
'Where are you-Sg going?'
```

```
    b. yì-tègè bû: àr<sup>n</sup>áŋá yèg-à child-Pl Def.AnPlwhere? fall-3PlS
    'The children [topic], where did they fall?'
```

Predicative '(someone/someething) be where?' is the same àrⁿáŋá plus the regular form of locational 'be' (§11.xxx) (xx2.a-b). 'Where is it?' with

unspecified topic is $ar^n an a = n$, with the by-now familiar final low tone element (xx2.c).

- (xx2) a. bòndî àrnáná bù-Ø rain where? be-3SgS 'The rain [topic], where is it?' (='Where is the rain?)
 - b. àrⁿáŋá bù-w where? be-2SgS 'Where are you-Sg?'
 - c. $ar^n an a=n$ where=it.is.3SgS 'Where is it?'

13.2.5 'When?' ([àngú tû:] gò, wàgàti àr náná)

'When?' is expressed as [àngú tû:] gò, which ends in Locative gò. It begins with àngú, which can be connected to àngú 'which?' (but the latter is an adjective that follows its modified noun). This suggests that tû: (perhaps a possessed noun?) should mean 'time', but no such noun with this meaning occurs elsewhere in Nanga. (Compare, however, Toro Tegu tùwó 'time'.)

- (xx1) a. á:mádù [[àŋgú tû:] gò] yé:-ŋ̀
 A [[which? time.L] in] come-Impf.3SgS
 'When is Amadou coming?'
 - b. [[àŋgú tû:] gò] tò-ò
 [[which? time.L] in] sow.Perf.L-3PlS
 'When did they sow (= plant the seeds)?'

An alternative 'when?' interrogative is wàgàti àrnáná, which consists of a low-toned form of wàgáti 'time, moment', an Arabic loan that is widespread in languages of the zone (Fulfulde, Songhay, etc.), plus àrnáná 'where?'.

- (xx2) a. [wàgàtĩ àrⁿáŋá] bòndî wòε-∅ [time.L which?] rain rain.fall.Perf.L-3Sgs 'When did the rain fall?'
 - b. [wàgàti àrnáná] á:ndé ńni-m-w [time.L which?] Anda go-Impf-2SgS

'When are you-Sg going to Anda?'

```
13.2.6 'How?' (amây<sup>n</sup>)
```

'How?' can be expressed by the interrogative adverb $\frac{\partial m}{\partial y}$, or its reduplication $\frac{\partial m}{\partial y}$. These forms are subject to phonetic attrition in sentential context; the y^n is often elided, and the reduplication may be reduced to $\frac{\partial m}{\partial y}$.

- (xx1) a. amay nni-m-w how? go-Impf-2SgS 'How will you-Sg go?'
 - b. tóndî àmâyⁿ-àmâyⁿ ké:ndé-m̀-^w
 basket how?-how? fix-Impf-2SgS
 'How are you-Sg going to fix the basket?'

àmâyⁿ is often combined with kárⁿá 'do' in the sense 'do what?', and this phrase as a whole can be used in 'how?' interrogatives.

(xx2) [àmâyⁿ kàrⁿi gáy] úmî lá:rá-m-iy [how? do- and.then.SS] mosquito chase.away-Impf-1PlS 'How (= doing what?) will we chase away the mosquitoes?'

13.2.7 'How much/many?' (à:ŋgǎy)

'How much?' or 'how many?' is $\grave{a}:ng\check{a}y$. It can be used like a numeral, following a noun (without plural marking) that keeps its normal tones (xx1.a). It can also be used absolutely (xx1.b).

- (xx1) a. [pèrgé à:ŋgǎy] sò-w [sheep how.many?] have-2SgS 'How many sheep do you-Sg have?'
 - b. à:ŋgǎy jóró-m̀-w how.much? want-Impf-2SgS 'How much (or: how many) do you-Sg want?'

A complement may also be expressed by a topicalized NP (xx2).

(xx2) nàná yề:, à:ŋgǎy màrà-∅

cow 1SgP.Poss.AnPl, how.many? be.lost.Perf.L-3SgS 'My cows, how many (of them) were lost?'

The distributive reduplication, used in markets to indicate price per unit of sale, is à:ngăy-à:ngăy.

(xx3) [éwé gá] măŋgórò à:ŋgǎy-à:ŋgǎy túró-m-è [market in] mango how.much?-how.much? sell-Impf-3PlS 'They sell mangoes for how much apiece in the market?'

13.2.8 'Which?' (angú)

As interrogative modifying adjective, 'which?' is expressed as angú (xx1.a). It may also be used absolutely (xx1.b). Before a Cv onset it optionally reduces to an, and the velar nasal may then assimilate (xx1.c).

- (xx1) a. [màngòrò àngú] jɔ́rɔ́-m̀-ẅ [mango.L which?] want-Impf-2SgS 'Which mango do you-Sg want?'
 - b. angú jóró-m-w which? want-Impf-2SgS 'Which (one) do you-Sg want?'
 - c. [nàŋá ú yê] ăŋ túró-m̀-w [cow 2SgP Poss.AnPl] which? sell-Impf-2SgS 'Which of your cows are you-Sg selling?' (can also mean 'You will sell your cows to whom?')

àngú gets some competition from other interrogatives. In (xx2.a), àrnáná 'where?' is used as a modifier (or perhaps compound final), with preceding L-toned noun, since a fixed location is involved. In (xx2.b), a possessive construction is used with kò-né 'what?' as the possessor, since the question concerns the substance (i.e. fruit) from which the possessed noun is constituted.

- (xx2) a. [ú tíyà] [ìdò àr¹áŋá] sîgé-ŋ̀
 [2SgP friend] [house.L where?] go.down-Impf.3SgS
 'In which house does your-Sg friend go down (=lodge)?'
 - b. [kò-nɛ́ nî:] kárⁿi-m-iy [what? water] make-Impf-1PIS

'Which juice (=juice of what) will we make?'

13.2.9 'So-and-so' (àmá:nà)

The word for so-and-so, substituting for any of a range of personal names, is àmá:nà.

(xxx) [nàŋà wŏŋ], [àmá:nà yè]≡ŋ [cow.L Dem], [so.and.so Poss.AnSg]≡it.is.3SgS 'This cow belongs to so-and-so.'

13.2.10 Embedded interrogatives

Embedded WH-interrogatives can take the same form as unembedded ones, with the original WH-interrogative word and with the interrogative particle, which is typically pronounced with some prolongation and with falling pitch in this construction (xxx).

- (xx1) a. [ăŋ yè:-Ø mà⇒] yè-rî-y [who? come.Perf.L Q] see-Perf.Neg-1SgS 'I didn't see who came.'
 - b. [à:ŋgǎy èwè-∅ mà⇒] júgó-ŋò-y how.much? pay.Perf.L-3SgS Q know-Impf.Neg-1SgS 'I don't know how much he/she paid.'
 - c. [àrnáŋá nnè-Ø mà⇒] júgó-ŋð-y
 [where? go.Perf.L-3SgS Q] know-Impf.Neg-1PIS
 'I don't know where he/she went.'

The construction gets some competition from relative clauses, such as the object relative clause in (xx2). For $-\dot{m}$ -sè, see the temporal adverbial clause type with $-\dot{m}$ -sè gà (§15.2.3, below).

(xx2) [kò ĭn jóró-m-sè gú] ìré bèsì-y [NonhO 1SgS look.for-Impf-PplPerf Def] forget bury.Perf-1SgS 'I have forgotten that which (=what) I was looking for.'

14 Relativization

14.1 Basics of relative clauses

Relative clauses have the features in (xxx).

- (xxx) a. the head NP remains in its clause-internal position (internal head), but (except for a possessor within the NP) it drops tones;
 - b. the verb ends in a special participle, which is marked for a reduced set of aspectual and polarity categories; except in negatives with 3Pl subject, the participle contains no marking in the verb of the pronominal subject category, and no agreement with the head NP in nominal features
 - c. subject pronominals are expressed by preverbal independent pronouns;
 - d. in the predominant case where the relative clause is semantically definite, the clause ends in a postverbal determiner, agreeing in animacy and number with the head noun.

14.1.1 Coordinated relatives with a shared head

In (xx1), the head noun (nŭ: 'person') appears once, but is understood to be the unexpressed head of the second relative clause ('who go') as well. The first relative clause is structurally complete, including a final Definite morpheme. The second clause contains a resumptive 3Pl pronoun.

```
(xx1)
                                                            bû:]
       nù
                      ηgá
                                  nà:-mì
       [person.L
                      here
                                  spend.night-Ppl.Impf
                                                            Def.AnPl]
       [kòrnóŋòyn
                         bû:
                                      ńnî-mì]
       [morning
                         AnPl
                                      go-Ppl.Impf]
       jóró-m-∅
       want-Impf-1SgS
        'I like people who spend the night here and go (away) in the morning.'
```

Three people who spend the hight here and go (away) in the morning.

In (xx2), even the main clause ('I dislike') is adjacent to the first relative clause, suggesting a kind of extraposition of the second relative clause.

```
[kirná
(xx2)
                     nù
                                  gîsé-mì]
                                                       mbùrà-ý
                                  throw-Ppl.Impf]
                                                       dislike-1SgS
        [bone
                     person.L
        [[tùndú
                   gó]
                            kirná
                                      nù
                                                    sémî-ŋò:]
                                                    sweep-Ppl.ImpfNeg]
        [[rear
                   in]
                            bone
                                      person.L
        'I don't like people who throw bones and don't sweep up afterwards.'
```

Of course a simpler construction occurs when the two co-events are merged into a chain, since here only the final verb is participalized (xx3).

```
(xx3) nù tómbó sígé-mì
person.L jump go.down-Ppl.Impf
'a person (or: people) who jump down'
```

14.1.2 Tone-dropping on final word(s) of NP in relative clause

The regular NPs in the left column in (xx1) are converted into the relativeclause head NPs in the right column.

(xx1)		regular NP	relative head NP	gloss
	a.	nŭ: ńdô	nù ndò	'person' 'house'
	b.	nù mòsi ndò èsi	nù mòsi ndò èsi	'bad person' 'good house'
	c.	nǔ: wŏy ńdô kúrê	nù wòy ndò kùrè	'two people' 'six houses'
	d.	[nù mòsî] wŏy [ǹdò ɛ̀sî] kúrê	[nù mòsi] wòy [ǹdò èsi] kùrè	'two bad people' 'six good houses'

In (xx1.a-b), whether the NP is just a noun or is a noun plus adjective, the final word has its regular tones as a regular NP. This word drops its tones when the NP functions as relative head. (xx1.c-d) add numerals to (xx1.a-b), respectively. In regular NPs, the numeral has no tonal interation with the core NP consisting of the noun plus any modifying adjectives. In a relative head NP, both the final word of the core NP, and the numeral, drop tones simultaneously.

These tone-dropping facts are parallel to those seen in Jamsay and some other Dogon languages. However, it is also necessary to consider how possessed NPs behave when they function as head NPs. When the possessor is postnominal (i.e. a pronominal possessor plus possessive classifier, §6.xxx), the

possessed noun is not tone-dropped. Rather, the possessive classifier (originally a noun meaning 'thing' or 'animate being') is tone-dropped, if not already low-toned. The tone-dropping is easiest to hear in the 1Sg forms, where the 1Sg possessor fuses with the classifier (adding only an initial low tone to the basic {HL} possessed-noun contour of the classifier).

- (xx2) a. ndô ú gô house 2SgP Poss.InanSg 'your-Sg house'
 - b. [ndô ú gồ] yègè-sè gú [house 2SgP **Poss.InanSg.L**] fall-Ppl.Perf Def.InanSg 'your house that fell'
 - c. pèrgé yề: sheep 1SgP.Poss.AnSg 'my sheep-Sg'
 - d. [pèrgé yè:] yègè-sè né [sheep 1SgP.Poss.AnSg.L] fall-Ppl.Perf Def.AnSg 'my sheep-Sg that fell'

In other words, the possessed noun in (xx2.b,d) is a **tonosyntactic island**, protected from the tone-dropping power of the relative-clause construction. So is the possessor pronoun, as we see from the high tone of \mathbf{u} in (xx2.b).

A **prenominal possessor** may be any nonpronominal NP, and may be a pronoun in the case of inalienable possession (kin terms). The data in (xx3) show that the entire possessor-possessed NP is again a tonosyntactic island with respect to relativization. That is, neither the possessor NP nor the possessed noun drops its tones. In (xx3.a), 'house' is already tone-dropped within the possessed NP since the possessor ends in a low tone (§xxx), so its low tone is not attributable to the relative construction. In the other examples in (xx3), the possessed NP has {HL} possessed-noun tone since the possessor ends in a high tone, and there is no change when the entire NP functions as relative-clause head.

- (xx3) a. [á:dámà ndò] yègè-sè gú
 [A house.L] fall-Ppl.Perf Def.InanSg
 'Adama's house that fell.'
 - b. [yǎ-ŋ pérgè] yègè-sè né [woman-Sg sheep.HL] fall-Ppl.Perf Def.AnSg 'a/the woman's sheep-Sg that fell'

- c. [[àrⁿà wǒŋ] pérgè] yègè-sè né [[man.L Dem.AnSg] sheep.HL] fall-Ppl.Perf Def.AnSg 'this man's sheep-Sg that fell'
- d. [ú dérè] bàmàkô:≡Ø bù-mì né
 [2SgP elder.sibling.HL] B≡in be-Ppl.Stat Def.AnSg
 'your-Sg elder sibling who is in Bamako'

The possessive NP data need to be completed by adding modifying adjectives and numerals. If the possessor is **postnominal**, a **modifying adjective** is adjacent to the noun, so e.g. 'your big house' is expressed by the linear sequence [[house.L big] 2SgPossessor Classifier]. Since 'big' is tightly bound to the noun, it is of course included in the tonological island and is not tone-dropped in relatives (xx4.a). A **numeral** may precede or follow the postnominal possessor, and which choice it makes determines whether it is in the tonological island and is protected from relative-clause tone-dropping (xx4.b), or is vulnerable to this tone-dropping (xx4.c).

- - b. [ndô kurê u yè] yègè-sè ý
 [house six 2SgP Poss.InanPl.L] fall-Ppl.Perf Def.InanPl
 'your-Sg six houses that fell'
 - c. [[ńdô ú yè] kùrè] yègè-sè ý [house 2SgP Poss.InanPl.L] six.L] fall-Ppl.Perf Def.InanPl [=(b)]

When an adjective is part of a possessed NP with **preposed possessor** (whether nonpronominal or pronominal), the **adjective is already tone-dropped** as an extension of the {HL} or all-low possessed noun tone imposed by the possessor (xx5.a,b), so there is no way to determine whether the relative-clause construction would otherwise have induced tone-dropping on the adjective (xx5.b,d).

- (xx5) a. wǒn [ńdò òwò]
 Dem.AnSg [house.HL big.L]
 'a big house of this (person)' (òwó)
 - b. [wŏŋ [ńdò ɔwɔ]] yègè-sè gú [Dem.AnSg [house.HL big.L]] fall-Ppl.Perf Def.InanSg

'a big house of this (person) that fell'

- c. ú [dérè mòsì]
 2SgP [elder.sibling.HL bad.L]
 'your bad (older) brother'
- d. ú [dérè mòsi] yègè-sè né
 2SgP [elder.sibling.HL bad.L] fall-Ppl.Perf Def.AnSg
 'your bad (older) brother who fell'

When a numeral follows a noun that has a prenominal possessor, whether the numeral is tone-dropped under the influence of the possessor depends on whether the possessor is nonpronominal (tone-dropping occurs) or pronominal (no tone-dropping) (§6.2.2.2). In the former case, whether any further tone-dropping in relatives would occur is moot. In the latter case, we get audible tone-dropping on the numeral when the NP is head of a relative (xx6).

```
(xx6) [ú dérè kùrè] yègè-sè bû:

[2SgP elder.sibling.HL six.L] fall-Ppl.Perf Def.AnPl

'Your-Sg six older brothers who fell' (kúrê)
```

Determiners and non-numeral quantifiers are not relevant to this section, sicne they are relocated after the verbal participle; see §14.xxx, below.

14.1.3 Restrictions on the head noun in a relative clause

A pronoun may not directly head a relative clause. Instead, nǔ: 'person' (for humans) occurs as head NP (low-toned nù), and the pronoun is preposed, in apposition.

14.1.4 Relative clause with conjoined NP as head

A conjoined NP behaves as a tonological island with respect to relative-clause tone-dropping, so there is no audible effect in cases like (xx1). However, the conjunction yò is already low-toned, so it is conceivable that it might have been subject to tone-dropping if it had been high toned.

(xxx) [[yă: yò árⁿà yò] jòriyè-sè bû:] àrⁿáŋá b-è [[woman and man and] fight-Ppl.Perf Def.AnPl] where be-3PlS 'Where are the men and the women who squabbled?'

14.1.5 Headless relative clause

In idiomatic Nanga, a relative head NP that is understood to refer to an animate or inanimate object is normally expressed by at least a semantically light noun, such as nǔ: 'person', kó: 'thing', or yɛ´ 'things; critter' (in low-toned form as nù, kò, yɛ̀). However, relative clauses with no overt head are acceptable. In a case like jòriyè-sɛ̆ bû: 'the ones who squabbled', the Definite Animate Plural determiner at the end gives some clues to the identity of the omitted head.

14.1.6 Preverbal subject pronominal in relative clause

In non-subject relatives, if the subject is a pronoun, it is expressed by an independent pronoun **immediately preceding the verb**. In (xx1.c), the subject pronoun follows a direct-object pronoun, and in (xx1.d) it intervenes between two directly chained verbs ('fall' and 'go down'). The forms used include 1Sg in and 3Sg ńné.

- (xx1) a. kò ú júgó-ŋò: thing.L 2SgS know-Ppl.ImpfNeg 'something that you-Sg do not know'
 - b. ùsì i:n yè:-sè gú
 day.L **1SgS** come-Ppl.Perf Def.InanSg
 'the day (when) I came'
 - c. ùsì njî-n ńné sùyò-sè gú day.L 1Sg-Acc **3SgS** hit-Ppl.Perf Def.InanSg 'the day (when) he/she hit me'
 - d. ùsì yègé ú sìgè-sè gú day.L fall **2SgS** go.down-Ppl.Perf Def.InanSg 'the day you-Sg fell down'

14.1.7 Verbal participle

Participles replacing inflected verb forms are a defining feature of relative clauses. Excluding some issues involving 3Pl head NP combined with a negative participle, the four basic participial suffixes reflect the intersection of aspect (Perfective/Imperfective) and polarity (positive/negative), i.e., a simplified version of the fuller aspect-negation (AN) system seen with inflected forms. Each of the four Participial suffixes in (xx1) is in fact closely related in phonological form to an inflectable AN suffix, and requires the same stem-tone pattern and any applicable stem-final vocalic shifts as that AN suffix. The negative participles in particular are identical in form to the corresponding negative inflectable stems.

(xx1) Participles (four basic suffixes)

category suffix	related AN suffix
Perfective -sè	Perfective-2 -só-
PerfNeg -rî	Perfective Negative -ri-
Imperfective -mi ~ -mi	Imperfective -m-
[for -m-se, see discussion b	pelow]
ImpfNeg -ŋà:	Imperfective Negative -ŋò:-

The (positive) Imperfective Participial suffix has a broader range than its inflectable counterpart. Specifically, it is used with Stative verbs and quasiverbs, as well as with active verbs in imperfective aspect. It may also be added to a verb already containing Progressive suffix -sò- to produce the suffix complex -só-mì (note the change in tone).

Likewise, the (positive) Perfective Participial suffix has a broader range than any specific inflectable counterpart. It corresponds to the amalgamation of Perfective-1a -èrè-, Perfective-1b -tì-, and Perfective-2 -só-. It may also be added to Recent Perfect jè- to produce the sequence jè-sè, and to Experiential Perfect tá:- (really a chained verb) to produce tá:-sè.

Including these combinations, and some similar combinations with negative Participial suffixes, (xx1) can be expanded as (xx2).

(xx2) Participles (all categories of active verbs)

category	suffix(es)	similar AN morpheme
Perfective	-sè	Perfective-2 -só-

Recent Perfect	jè-sè	Recent Perfect jε-
Experiential Perfect	tá:-sè	ExpPerf tá:-
Perfective Negative	-rî	Perfective Negative -rî-
Recent Perfect Negative	jè-rî	Recent Perfect Neg jè-rî-
Experiential Perfect Neg	tà:-rî	ExpPerfNeg tà:-rî-
Imperfective	-mì ~ -m̀	Imperfective -m-
Progressive	-só-mì	Progressive -so-
Imperfective Negative	-ŋɔ̀:	Imperfective Negative -ŋɔ̂:-
Progressive Negative	-sò-ndó	Progressive Negative -sò-ndó

14.1.7.1 Participles of positive perfective-system verbs

Before getting into examples, it is necessary to point out that the participle is very frequently tone-dropped by a following determiner. Therefore the tones described here as basic are often overridden.

Examples of the basic **Perfective Participial** suffix -sè are in (xx3). The suffix is added to the unmodified bare stem (with lexical vocalism and tone contour). When followed by a Definite morpheme or other determiner, all tones in the participal (audibly, those of the verb stem) drop, hence nù yè:-sè né 'the person who came', etc.

Perfective Participial -sè is the relative-clause counterpart of the Perfective-1a, Perfective-1b, and Perfective-2 forms in main clauses. In practice it is also the usual counterpart of Recent Perfect jè- as well.

(xx3) Perfective participles

bare stem	Perf participle	gloss
yě:-	yě:-sè	'come'
gŏ:-	gŏ:-sè	'go out'
yĭ:-	yĭ:-sè	'see'
núy ⁿ -	núy ⁿ -sè	'go in'
ńné-	ńné-sè	ʻgo'
yègé-	yègé-sè	'fall'
kár ⁿ î-	kár ⁿ i-sè	'do'
bă:rî-	bă:rî-sè	'help'
bègîrî-	bègîrî-sè	'winnow'

Examples are in (xx4). (xx4.b) shows the tone-dropping on the participle before a determiner.

- (xx4) a. nànà bû: émé-sè cow.L 3AnPlS milk-Ppl.Perf 'a cow (or: cows) that they milked'
 - b. nàŋà ĭ:n èmè-sè né
 cow.L 1SgS milk-Ppl.Perf.L Def.AnSg
 'the cow that I milked'

Recent Perfect $j\hat{\epsilon}$ - has participial form $j\hat{\epsilon}$ -s $\hat{\epsilon}$. It is used sparingly, since Perfective Participial -s $\hat{\epsilon}$ is often used as a relative-clause counterpart of $j\hat{\epsilon}$ - in main clauses. Like inflected $j\hat{\epsilon}$ -, the participial complex $j\hat{\epsilon}$ -s $\hat{\epsilon}$ follows the bare stem with its lexical tones. When a determiner is added, there is no change to the tones. In other words, the tone-dropping effect of the determiner does not reach beyond the $j\hat{\epsilon}$ -, which is already low-toned. This is evidence that $j\hat{\epsilon}$ - is phonologically similar to a chained verb, rather than a simple suffix. This view is reinforced by the fact that a subject pronominal intervenes between $j\hat{\epsilon}$ - and the preceding verb (xx5.c).

- (xx5) a. àrⁿà nă: kố: jè-sè man.L meal eat RecPf-Ppl.Perf 'a man who has already eaten (or: who has finished eating)'
 - b. àr"à nă: kó: jè-sè né
 man.L meal eat RecPf-Ppl.Perf Def.AnSg
 'the man who has already eaten (or: who has finished eating)'
 - c. àrⁿà súyó ĭ:ⁿ jè-sè né man.L hit 1SgS RecPf-Ppl.Perf Def.AnSg 'the man who(m) I have already hit (or: whom I have finished hitting).'

Experiential Perfect tá:- with its long vowel is prosodically even more clearly a chained verb phonologically. A simple example of a participle is (xx6.a). If a determiner is added, tá:- is tone-dropped but the preceding verb is not (xx6.b). A pronominal subject intervenes between the two verbs (xx6.c).

(xx6) a. yà: [isè gó] ńné tá:-sè woman.L [village.L in] go ExpPerf-Ppl.Perf 'a woman who has (ever) gone to the village'

- b. yà: [isè gó] ńné tà:-sè né woman.L [village.L in] go ExpPerf-Ppl.Perf.LDef.AnSg 'the woman who has (ever) gone to the village'
- c. yà: yǐ: ǐ:ⁿ tà:-sè né woman.L see 1SgS ExpPerf-Ppl.Perf.L Def.AnSg 'the woman who has (ever) gone to the village'

Although **Perfective-1b** -ti- was not observed in spontaneously produced relative clauses, it was possible with some effort to elicit -ti- in relatives. In nonsubject relatives, my assistant tended to allow a preverbal subject pronominal to intervene between the main verb stem and ti-, suggesting that the latter is a separate word (i.e. a chained verb) rather than a suffix (xx7). However, he sometimes repeated such elicited sentences with the subject pronominal preceding the main verb. The construction does not seem to be fully productive and it is not surprising that its syntax is inconsistent.

14.1.7.2 Participles of positive imperfective-system and stative verbs

Here the basic Participial suffix is $-mi \sim -m$. The syllabic allomorph is usual in simple participles not followed by a determiner. The nonsyllabic allomorph is usual before a determiner.

Examples showing the stem-shapes of Imperfective participles from verbs that end in a non-high vowel are in (xx1). Here the stem used in the participle is identical (segmentally and tonally) to the stem used for the same verb before inflected Imperfective - \dot{m} -. However, the participle is **never reduplicated**.

(xx1) Imperfective participles (stem ends in non-high vowel)

bare stem	Impf participle	gloss
yě:-	yé:-mì	'come'
gŏ:-	gŏ:-mì	'go out'
yègé-	yégé-mì	'fall'

When a determiner is added, aside from the tone-dropping effect there is no change in the participial forms for the verbs illustrated in (xx1) that end in a

non-high vowel. However, for those verb stems that end in /i/ in this participial form, the /i/ shifts to a non-high vowel when a determiner is added (in addition to the tone-dropping). In (xx2), therefore, an extra column is added. Note that the longer verbs (those of more than two vocalic moras) whose stem ends in /i/ have already shifted the /i/ to a non-high vowel in the simple form of the Imperfective participle, so for these verbs the forms in the two columns have identical vocalism.

(xx1) Imperfective participles (stem ends in /i/)

```
bare stem
                        Impf participle
                                                                gloss
                                   before determiner
                   simple
a. two vocalic moras
                                   yὲ:-m
                                                                 'see'
     yĭ:-
                   yî:-mì
     núy<sup>n</sup>-
                   núy<sup>n</sup>-mì
                                   nùy<sup>n</sup>ò-m
                                                                 'go in'
                                                                 'go'
     ńné-
                   ńnî-mì
                                   ìnε-m
     kár<sup>n</sup>i-
                   kár<sup>n</sup>i-mì
                                   kàr<sup>n</sup>à-m
                                                                 'do'
b. more than two vocalic moras
     bă:rî-
                   bă:rá-mì
                                   bà:rà-m
                                                                'help'
                                   bègirè-m
                                                                 'winnow'
     bègiri-
                   bègîré-mì
```

Examples are in (xx3). (xx3.b,d) end in determiners.

- (xx3) a. nàŋà ĭ:n émé-mǐ cow.L 1SgS milk-**Ppl.Impf** 'a cow (or: cows) that I will milk'
 - b. nàŋà bû: èmè-m bû:
 cow.L 1SgS milk-**Ppl.Impf.L** Def.AnPl
 'the cows that they will milk'
 - c. arnanuyn-mi man.L go.in-Ppl.Impf 'a man who goes in'
 - d. àrⁿà nùyⁿ5-m nế man.L go.in-Ppl.Impf.L Def.AnSg 'the man who goes in'

The same Imperfective Participial suffix is used with stative verbs, which are (in other respects) aspect-neutral but generally include present-time

reference (in the absence of the Past clitic). Representative forms are in (xx1), with examples in (xx4). A determiner forces tone-dropping on the whole participle as for the imperfectives (xx4.b). éwé-mǐ 'who is sitting' has medial /e/ from /o/ by assimilation.

(xx3) Stative participles

	ba	re stem	Stative	Stative participle	gloss
	î:-: só:	r-yé- yî- w-yé- yî-	é-èwò- 1-ĭyà- só-sòwò- ú-ùwà-	éwé-mi îyá-mi sówó-mi úwá-mi	'be sitting' 'be standing' 'be squatting' 'be afraid'
(xx4)	a.	àr ⁿ à man.L 'a man v	úwá-mì fear.Stat-P vho is afraid'	pl.Impf	
	b.	àr ⁿ à	ùwà-m	nέ	

fear.Stat-Ppl.Impf.L

Representative **Progressive** participle forms are in (xx5). The stem has the same form as in the inflected Progressive, but the reduplication is gone. The suffix -sò- shifts to high-toned -só- in the participle. Examples are in (xx6). Tone-dropping forced by a determiner applies to -só- (which reverts to -sò-) but not to the preceding verb stem (xx6.b).

Def.AnSg

(xx5) Progressive participles

man.L

'the man who is afraid.'

bare stem	Progressive	Progr participle	gloss
¥.	>> ->		' ²
yě:-	yè-yê:-sò-	yê:-só-mì	'come'
gŏ:-	gò-gô:-sò-	gô:-só-mì	'go out'
yĭ:-	yì-yî:-sò-	yî:-só-mì	'see'
núy ⁿ -	nù-nûy ⁿ -sò-	nûy ⁿ -só-mì	'go in'
ńné-	ì-ńnî:-sò-	ńnî:-só-mì	ʻgoʻ
yègé-	yè-yégè(:)-sò-	yégè(:)-só-mì	'fall'
kár ⁿ í-	kà-kár"i(:)-sò-	kár ^m i(:)-só-mì	'do'
bă:rî-	bà-bă:rà(:)-sò-	bă:rà(:)-só-mì	'help'
bègîrî-	bè-bègîrè(:)-sò-	bègîrè(:)-só-mì	'winnow'

- (xx4) a. yà: gô:-só-mì woman.L go.out-Progr-Ppl.Impf 'a woman who is going out'
 - b. yà: gô:-sò-m nế woman.L go.out-Progr-Ppl.Impf.LDef.AnSg 'the woman who is going out.'

14.1.7.3 Participles of negative perfective-system verbs

The basic **Perfective Negative** suffix $-r\hat{i}$ - occurs in participles, but without the following pronominal-subject suffix. As with the inflectable suffix (in the 1Sg, 2Sg, and 3Sg), the corresponding participles have a low-toned stem, and shift a stem-final /i/ to a non-high vowel. As usual, the /r/ is nasalized to /rⁿ/ after a nasal syllable. When a determiner follows, the suffix $-r\hat{i}$ drops its tones (the stem is already low-toned so we cannot tell whether the determiner would have also dropped any high tones on the stem). Some examples are in (xx1).

- (xx1) a. nànà ĭ:n èmè-rⁿi
 cow.L 1SgS milk-**Ppl.PerfNeg**'a cow that I didn't milk'
 - b. nàŋà î: èmè-rⁿi bû:
 cow.L 1Pl milk-**Perf.Neg.L** Def.AnPl
 'the cows that we didn't milk'
 - c. yà: nùyⁿò-rⁿi
 woman.L go.in-**Ppl.PerfNeg**'a woman who didn't go in'
 - d. àr"àsóy ndè-ri man.L all give-**Ppl.PerfNeg** 'a man who didn't give anything' (ndi- 'give')

In the inflectable Perfective Negative, the 3Pl subject form is rather irregular, replacing -ri- by a 3Pl Perfective Negative portmanteau -àndú, as in èm-à:ndú 'they didn't milk (cow)', cf. èmè-ri-Ø 'he/she didn't milk (cow)'. This peculiarity extends to the corresponding participles, but here the unusual form is associated with an **Animate Plural head NP**. Compare the participial forms in (xx2.a) and (xx2.b).

```
(xx2) a. nànà bû: èmè-r<sup>n</sup>i
cow.L 3PlS milk-Ppl.PerfNeg
'a cow that they didn't milk'
```

```
c. nànà bû: èm-à:ndú
cow.L 3PlS milk-Ppl.PerfNeg.AnPl
'cows that they didn't milk'
```

There is no similar suffixal switch with **inanimate** head NPs. Therefore (xx3.a) is ambiguous as to number, in the absence of a final determiner. The explicitly plural (xx3.b) has the same -ri Participial suffix as (xx3.a), though in low-toned form before the determiner.

```
(xx3) a. kùr<sup>n</sup>ò ĭ:<sup>n</sup> gĭsè-r<sup>n</sup>1
stone.L 1SgS throw-Ppl.PerfNeg
'a stone (or: stones) that I didn't throw'
```

```
b. kùr<sup>n</sup>ò i:<sup>n</sup> gisè-r<sup>n</sup>i ý
stone.L 1SgS throw-Ppl.PerfNeg.L Def.InanPl
'the stones that I didn't throw'
```

The **Recent Perfect Negative** jè-rî-, which often means 'has not finished VP-ing', can be participialized. The form of the stem is the same as in the inflected paradigm; in particular, -rî- (which elsewhere forces tone-dropping on a preceding verb stem) does not affect the tones of the stem preceding jè-. Much less does a following determiner force tone-dropping on this stem, though as usual it does drop the tones of -rî-.

```
(xx4) a. nù nă: k5: jè-rî
person meal eat RecPf-Ppl.PerfNeg
'a person who has not finished eating'
```

b. nù nă: kó: jà:-ndù bû:
 person meal eat RecPf-Ppl.PerfNeg.L Def.AnPl
 'the people who have not finished eating'

The **Experiential Perfect Negative** based on chained verb tá:- has a similar pattern. The negative inflectable form is tà:-rî-, including Perfective Negative -rî-, and this is copied in Participial tà:-rî. The participles also copy the unusual tone-dropping pattern seen in the inflected forms, by which -rî- induces tone-dropping not only on tá:- but also on the preceding verb stem.

- (xx5) a. yà: [isè gó] nnè tà:-rî
 woman.L [village in] go.L ExpNeg-**Ppl.PerfNeg**'a woman who has never gone to the village'
 - b. yà: [isè gó] nnè tà:-ri né
 woman.L [village in] go.L ExpNeg-**Ppl.PerfNeg** Def,AnSg
 'a woman who has never gone to the village'

14.1.7.4 Participles of negative imperfective-system and stative verbs

The Imperfective Negative suffix -ŋɔ: is also used in participles, with the same stem-shapes as with the inflected stems. A few examples are in (xx1). The same irregularities with 'hear', 'see', and 'go' as in the inflected forms are observed in the participles. The 3Pl agreement forms are discussed below.

(xx1) Imperfective Negative participles

bare stem	ImpfNeg	ImpfNeg participle	gloss
gŏ:- ńné- nŭy ⁿ - yĭ:- gɛ̃:r ⁿ î-	gó:-ŋò:- ńné-ŋò:- nù-ŋó- ɲù-ŋó- gĕ:r ⁿ é-ŋò:-	gó:-ŋò: ńné-ŋò: nù-ŋó- ɲù-ŋó gĕ:r ⁿ é-ŋò:	'go out' 'go' 'hear' 'see' 'take, deliver'
gc.i i-	gc.1 c-1j5	gc.1 c-1j5.	take, deliver

Examples are in (xx2). When a determiner follows, the verb stem is tone-dropped (xx2.b).

- (xx2) a. nànà ú émé-nò: cow.L 2SgS milk-**Ppl.Impf.Neg** 'a cow that you-Sg do (will) not milk'
 - b. nàŋà û: èmè-ŋò: bû:
 cow.L 2PlS milk-**Ppl.Impf.Neg.L** Def.AnPl
 'the cow that you-Sg do (will) not milk'

As with the Perfective Negative Participle, there is a special form used for Animate Plural head-NP agreement with the Imperfective Negative Participle. In the inflected paradigm, the 3Pl form of $-\eta$: is $-\eta$ - ϵ : 'they don't/won't come'. The form $-\eta$ - ϵ : is used in the participles for Animate Plural head NPs (xx3).

- (xx3) a. nàŋà bû: émé-ŋ-è:
 cow.L 3PlS milk-**Ppl.Impf.Neg-3PlS**'cows that they will not milk'
 - b. nàŋà bû: èmè-ŋ-è: bû:
 cow.L 3PlS milk-**Ppl.Impf.Neg-3PlS.L** Def.AnPl
 'the cows that they will not milk'

Stative Negative =ndó- also has a participial counterpart with the identical ending, though without pronominal-subject agreement (=ndó). This allows participles to be created from negative forms of Stative verbs and of predicate adjectives. Similarly, a participle can be directly constructed from 'not be (noun)' clitic =ndő:-.

- (xx4) a. àrⁿà èwò≡ndó man.L sit.Stat≡Ppl.StatNeg 'a man who is not sitting'
 - b. $ar^n agawa = ndo$ man.L tall = Ppl.StatNeg'a man who is not tall'
 - c. $ar^n ad 6g = nd$ iman.L Dogon=Ppl.not.be

 'a man who is not a Dogon (person)'

The counterparts of (xx4.a-c) with Animate Plural head NP are in (xx5). They show the usual imitation of the 3Pl subject form of the inflected clitics.

- (xx5) a. àrⁿà èwè=nd-é man.L sit.Stat=Ppl.StatNeg-AnPl 'men who are not sitting'
 - b. àrnàgàwà≡nd-€ man.L tall.L≡Ppl.StatNeg-AnPl 'men who are not tall'
 - c. àrⁿàdógô≡nd-ĕ: man.L Dogon≡Ppl.not.be-AnPl 'men who are not Dogon (people)'

The Progressive Negative inflected form ends in Stative Negative =ndó and is participialized accordingly (xx6.a). sò-ndó- 'not have' and other negative forms of stative verbs and quasi-verbs are also participializable in the same way.

```
(xx6) àr<sup>n</sup>à sémí-ndé sémbi-sò≡ndó
man.L sweep-VblN sweep-Progr≡Ppl.StatNeg
'a man who is not sweeping'
```

14.1.7.5 Participle of Past $=b\varepsilon$

Since everything else can be participialized, it is no surprise that verbs and other predicates with Past clitic $=b\varepsilon$ can be participialized.

For positive categories, the relationship between the inflected stem and the participle is summarized in (xx1). These are rather uncommon in actual use.

AN category	inflected	participle
positive		
Imperfective	-m̀≡bὲ-	-m̀≡bὲ-s
	(but	-m̀=sè before determiner)
Progressive	-sò≡bὲ-	-só-ḿ≡bè ~ -só-≡bè
Past Perfect	≡bέ-	-yέ-ḿ≡bὲ ~ -yέ-≡bε
Perfective-1b	-tî≡bέ-	-tî≡bέ
Perfective-1a	-èrè≡bè-	-èrè≡bè
Perfective-2	-só≡bέ-	-só≡bε
Recent Perfect	jὲ≡bὲ-	jὲ≡bὲ
	positive Imperfective Progressive Past Perfect Perfective-1b Perfective-1a Perfective-2	positive Imperfective -m=bè- (but Progressive -sò=bè- Past Perfect Perfective-1b -ti=bé- Perfective-1a -èrè=bè- Perfective-2 -só=bé-

For negative forms of $=b\epsilon$ -, the participial forms in (xx2) were recorded. The special Animate Plural head NP forms are in the far right column.

(xx1)	AN category	inflected	participle	
			regular	AnPl
	negative			
	C	4.17	4.17	N 12 1 2
	Past Perfect	-rî≡bέ-	-rî≡bέ	-àndú≡b-á
	Imperfective	-ηὸ:≡bὲ-	-ŋò:≡bὲ	-ηὲ:≡b-à

14.1.8 Relative clause involving verb- or VP-chain

When verbs (and VPs) are chained, only the final verb is participialized, the nonfinal ones remaining in their usual form. In the case of direct chains (without a chaining morpheme), the nonfinal verbs have their usual bare stem form with lexical vocalism and tones. In a non-subject relative, if a subject pronominal is present, it intervenes between the penultimate verb in the chain and the final verbal participle. In (xx1), śwś 'buy' is the nonfinal chained verb.

A nonpronominal NP does not intervene between the verbs of a direct chain. However, in non-subject relatives involving directly chained verbs (the final verb, therefore, being a participle), if the subject is a nonpronominal NP, it is optionally (but often) resumed by a 3Sg or 3Pl subject pronoun that directly preceded the final verb. Note the parenthesized, optional pronouns in (xx2), each of which is coindexed with the subject NP of the relative clause ('the man', 'some women').

- [àrnà (xx2)a. nànà nέ] έwέ (ńné) Def.AnSg] cow.L [man.L buy (3SgS)jè:-sὲ nέ bring.Ppl.Perf.L Def.AnSg 'the cow that the man bought and brought'
 - b. nàŋà [yà: bû:] śwś (bû:)
 cow.L [woman Def.AnPl] buy (3PlS)
 jè:-sè nś
 bring.Ppl.Perf.L Def.AnSg
 'the cow that some women bought and brought'
 - c. [àrnà né] ùsù nàŋá éwé (ńné)
 [man.L Def.AnSg] day cow buy (3SgS)
 jè:-sè gú
 bring.Ppl.Perf.L Def.InanSg
 'the day the man bought and brought a cow'

14.1.9 Determiners following the participle

If the head NP (and therefore the entire NP including the relative) is definite, the relevant Definite morpheme is added after the participle. Definite morphemes have tone-dropping effects on preceding words within simple NPs. In relatives, Definite morphemes have some tone-dropping effects on the participles themselves. The limitations of these tone-dropping effects are described in detail in the preceding sections on the form of participles for various inflectional categories. Abundant examples with and without Definite morphemes are given there.

In careful pronunciation, or before the 'it is' clitic or other particle, the Definite morphemes at the end of relatives have their regular pronunciation including high or falling tones. In rapid speech they tend to be articulated casually and the high or falling tone is not always audible.

Less often, a demonstrative pronoun is present, again at the end of the relative clause following the participle. A demonstrative has the same tone-dropping effects as a Definite morpheme,

```
(xx1) àr<sup>n</sup>à ú yè:-m wǒŋ
man.L 2SgS see-Ppl.Impf Dem.AnSg
'this man whom you-Sg see'
```

14.1.10 Non-numeral quantifiers following the participle

Universal and distributive quantifiers ('all', 'each') are also shifted from NP-final position (on the head NP) to post-participial position. kéréw 'all' may also follow a determiner (xx1.a). kámâ 'each' directly follows the participle, and forces tone-dropping on it (xx1.b-c).

```
(xx1) a. [nàŋà ú èmè-r<sup>n</sup>i bû: kéréw] jô: [cow.L 2SgS milk-Ppl.PerfNeg.L Def.AnPl all] bring.Imprt 'Bring-2Sg all the cows that you-Sg have not milked.'
```

```
b. [ndò yègè-sè kámâ],
[house.L fall-Ppl.Perf.L each]
[mîlyó wòy] jóró-m-i:y
[million two] look.for-Impf-1PIS
'(For) each house that fell, we are seeking two million (francs).'
```

```
c. nù nă: kó: jè-rì kámâ
person meal eat RecPf-Ppl.PerfNeg.L each
```

'any person who has not finished eating'

14.1.11 Indefinite relatives

These relatives omit the Definite morpheme, and therefore end with a participial.

- - b. [[nàŋà wŏy] ú èmè-rⁿi] jóró-sò-y [[cow.L two.L] 2SgS milk-Perf.Neg] look.for-Impf-1SgS 'I'm looking for two cows that you-Sg haven't milked.'

14.2 Subject relative clause

14.2.1 Ordinary subject relative clause

To resume the comments scattered across several preceding sections, a subject relative is characterized by the following: a) head NP is clause-internal but is marked by tone-dropping; b) no preparticipial subject pronominal is present; c) verb is replaced by a participle; d) determiners and non-numeral quantifiers that would normally be part of the head NP are repositioned after the participle. The examples in (xx1) involve positive participles.

- (xx1) a. [éwé gá] àrⁿà njî-ń yì:-sè né [market in] man.L 1Sg-Acc see-Ppl.Perf.L Def.AnSg 'the man who saw me in the market'
 - b. [éwé gá] àrⁿà njî-ń yĭ:-sè [market in] man.L 1Sg-Acc see-Ppl.Perf 'some men who saw me in the market'
 - c. [kùrⁿò njî-ń bàrⁿimĭ-sè gú]
 [stone.L 1Sg-Acc injure-Ppl.Perf.L Def.InanSg]
 àrⁿáŋá bù-Ø
 where? be-3SgS
 'Where is the stone that injured me?'

- d. [kùrnò njî-ń bàrnimi-sè ý]
 [stone.L 1Sg-Acc injure-Ppl.Perf.L Def.InanPl]
 àrnáŋá b-è
 where? be-3PlS
 'Where are the stones that injured me?'
- e. nù njî-ý súyó-sè person.L 1Sg-Acc hit-Ppl.Perf 'a person who hit me'
- f. nù njî-ŋ sùyò-sè bû:

 person.L 1SgO-Acc hit-Ppl.Perf.L Def.AnPl

 'the people who hit-Past me'
- g. nù njî-ý sùyð-m nế
 person.L 1SgO hit-Ppl.Impf.L Def.AnSg
 'the person who hits me'
- h. nù njî-ij súyó-mi person.L 1SgO hit-Ppl.Impf.L 'a person (or: people) who hit-Present me'

In the negative categories, the participle has a distinctive form for Animate Plural head NP, as in (xx2.b) and (xx2.d).

- (xx2) a. nù njî-ý sùyò-ri nế
 person.L 1Sg-Acc hit-Ppl.Perf.Neg.L Def.AnSg
 'the person who didn't hit me'
 - b. nù njî-ŋ sùyô-ndù bû:

 person.L 1Sg-Acc hit-Ppl.Perf.Neg.AnPl

 'the people who didn't hit me'
 - c. nù njî-ý sùyò-ŋò: nế
 person.L 1Sg-Acc hit-Ppl.ImpfNeg Def.AnSg
 'the person who doesn't hit me'
 - d. nù njî-ý sùyò-ŋ-è: bû:
 person.L 1Sg-Acc hit-Ppl.ImpfNeg-AnPl
 'the people who don't hit me'

14.3 Object relative clause

14.3.1 Ordinary object relative clause

Again the verb is replaced by a participle. The head NP is tone-dropped, and any determiners or non-numeral quantifiers belong to the head NP are placed following the participle. There is **no Accusative marking**. If the subject is pronominal, it is expressed by an independent pronoun immediately before the participle. Positive examples are in (xx1).

- - b. [éwé gá] àrⁿà ĭ:ⁿ yǐ:-sè bû: [market in] man.L 1SgS see-Ppl.Perf.L Def.AnSg 'the men who(m) I saw in the market'
 - c. [éwé gá] màngòrò ĭ:n yǐ:-sè gú [market in] mango.L 1SgS see-Ppl.Perf.L Def.InanSg 'the mango that I saw in the market'
 - d. [éwé gá] màngòrò ĭ:n yǐ:-sè ỳ
 [market in] mango.L 1SgS see-Ppl.Perf.L Def.InanPl
 'the mangoes that I saw in the market'
 - e. [lègè-sò ú èwè-sè gú]
 [bicycle.L 2SgS buy-Ppl.Perf.L Def.InanSg]
 nàmá-èrè-Ø
 be.ruined-Perf-3SgS
 'The bike that you-Sg bought has malfunctioned.'
 - f. [lègè-sò ĭ:" èwè-m kù] ŋgú kû" [bicycle.L 1SgS buy-? Def] Dem be 'This is the bike that I will buy.'
 - g. [î: kò:-sè gú] nàmâ≡ẁ=ndŏ:
 [1PIS eat-Ppl.Perf.L Def] meat≡it.is=not.be
 nìmî=ẁ nà
 cow.pea-be rather
 'What we ate was not meat, rather cow-peas'.

Negative examples are (xx2).

- (xx2) a. kò û: kó:-ŋò: thing.L 2PIS eat-Ppl.ImpfNeg 'what you-Pl do not eat'
 - b. nàŋà ĭ:n tùrò:-ndù bû:
 cow 1SgS sell-Ppl.PerfNeg.AnPl Def.AnPl
 'the cows that I did not sell'

14.4 Possessor relative clause

The possessor (always nonpronominal and always preceding the possessed noun) is tone-dropped, while the possessed noun reverts to its regular (non-possessed) tone contour. Especially with kin and other inalienable relationship terms, a pronominal postnominal possessor may also appear.

- (xx1) a. [àrⁿà ńdô[yègè-sè né [man.L house fall-Ppl.Perf.L Def.AnSg 'the man whose house fell'

 - c. àr"à [bă: nò] nnè-sè né man.L [father 3SgP] go-Ppl.Perf.L Def.AnSg 'the man whose father has gone'

14.5 Relativization on the complement of a postposition

In (xx1.a), the head NP is logically the dative indirect object, but there is no sign of the Dative postposition /bay/. Likewise, in (xx1.b), 'daba' (native hoe) is logically instrumental, but the usual Instrumental postpostion yana is absent. And in (xx1.c), 'honey' is purposive, cf. (xx1.a) in §8.3, above, but Purposive postposition der^n is nowhere to be seen. Clearly the regular way to form a

relative clause with the complement of a basic postposition as head NP is to delete the postposition entirely and then treat the head NP in the usual way.

- (xx1) a. yà: ŋgú ĭ:n kǐyè-sèné
 woman.L Dem.InanSg 1SgS say-Ppl.Perf.L Def.AnSg
 'the woman to whom I said that'
 - b. wàrà yû: ĭ:ⁿ wàrà-m gú daba.L millet 1SgS do.farm.work-Ppl.Impf.L Def.InanSg 'the daba (hoe) with which I do farming' (wárâ)
 - c. ò:ndò bû: yè:-sè gú honey.L 3PIS come-Ppl.Perf.L Def.InanSg 'the honey for which they came'

15 Verb (VP) chaining and adverbial clauses

I use the term **chain** to denote a sequence of two or more verbs, or VPs, where the nonfinal verbs are not inflected for pronominal subject. **Direct chains** have nonfinal verbs in their bare form (with lexical tone and vocalism), usually directly adjecent to the following (perhaps final) verb in the chain, except that (in non-subject relative clauses) a subject pronominal may intervene. Looser chains, often involving more complete VPs or clauses, make use of VP-final chaining morphemes, which to some extent distinguish same-subject from different-subject clause sequences (switch-reference).

15.1 Direct chains (without chaining morpheme)

An example of a direct chain is tómbó sígé- 'fall go.down', i.e., 'fall down'. In such cases, we can speak of two co-events that fuse into a single complex cognitive scenario. Since direct chaining of verbs is fairly productive in Nanga, a direct chain in Nanga may correspond to a single clause with an adverbial phrase in English.

15.1.1 Verbal Noun of directly chained verbs

When a direct verb chain is converted into a verbal noun, the Verbal Noun suffix -ndé is added to the final verb only. The nonfinal verbs have their regular form (there is no tone-dropping). Thus tómbó sigé-ndé 'fall(ing) and going down' = 'falling down'.

15.1.2 Presence of AN suffix in nonfinal verb in direct chains

In my data, the basic AN suffixes that do not pattern phonologically as chained verbs (i.e. Perfective-1a -èrè-, Perfective-1b -ti-, Perfective-2 -só-, Perfective Negative -ri-, Imperfective Negative -ŋò:-) do not occur in nonfinal verbs in chains, without an overt chaining morpheme. Nanga uses other devices to specify temporal relations among fuller, more loosely chained VPs and clauses, and generally does not allow separate negation of nonfinal verbs or VPs in chains.

However, Imperfective -m- (3Sg form - $\dot{\eta}$) does correspond to an Imperfective subordinator - $\dot{\eta}$ used in durative clauses (§15.xxx, below).

15.1.3 Arguments of directly chained verbs

Consider a chain consisting of two transitive verbs, with a direct object, as in (xx1).

```
(xx1) pèrgé-ý súyó jĩyè-∅
sheep-Acc hit kill.Perf.L-3SgS
'He/She hit and killed the sheep.'
```

In effect, 'hit' and 'kill' fuse into a single scenario, and it may not be meaningful to ask whether 'sheep' should be bracketed with 'hit' or with 'kill'. My assistant rejected a version of (xx1) with 'sheep' intervening between the verbs (which would favor bracketing specifically with 'kill'). Replacing 'sheep' by a pronoun results in no change in order (ú-ý súyó jîyé-ỳ 'he/she will hit and kill you-Sg').

15.1.4 Negation of direct verb chains

Morphologically, a direct verb chain is negated as a whole, with the negative morpheme appearing on the final verb.

```
(xx1) tómbó sùgò-ndú
junp go.down-PerfNeg.3PlS
'They didn't jump down.'
```

15.1.5 Direct chains including dògó- 'leave'

As usual in Dogon languages, the transitive verb 'leave, abandon' is often added after another verb that states (or implies) fixing the position of the object NP. In a free English translation, 'leave' could easily be omitted.

```
(xx1) a. [jînjà gú] ŋgá dǔŋ dɔg-ɔ [water.jar.L Def.InanSg] here put.down leave.Perf.L-3PlS 'They put down (and left) the water jar.' (dùŋî)
```

```
b. nàná pági dògó-só-ý
```

cow tie leave-Perf2-1SgS 'I tied up and left the cow.'

15.1.6 Direct chains including a motion verb or 'pick up, take'

'Come/go and VP' can be expressed by a nonfinal motion verb plus a chained VP. The motion verb directly precedes the final verb, and if the latter is transitive the complements occur to the left of the two-verb chain. This construction is especially common in imperatives and hortatives.

yě:- 'come' takes the low-toned, short-voweled form yè in this type of chain (xx1.a). ńné- 'go' has its regular form (xx1.b).

- (xx1) a. nă: yè kô: meal come eat.Imprt 'Come eat (a meal)!'
 - b. pèrgé-ŋ ńné sémâ sheep-Acc go slaughter.Imprt 'Go slaughter (the) sheep-Sg!'
 - c. pèrgé-ń ńné sémé-màyⁿ sheep-Acc go slaughter-Hortative.1Pl 'Let's-1Pl go slaughter (a/the) sheep-Sg!'
 - d. é:ŋî pèrgé ńné sémé-m-Ø tomorrow sheep go slaughter-Impf-1SgS 'Tomorrow I will go and slaughter (a/the) sheep-Sg.'

15.1.7 Durative verb-iterations chained to a motion verb

One type of durative adverbial clause (or its functional equivalent) is constructed by iterating the uninflected verb stem, with {HL} tone on the first occurrence and all-low tone on the second and (if present) later iterations. In (xx1), the initial give is a cognate nominal and is not part of the iteration, which begins with the following verb stem.

(xx1) [gǐyé gîyè-gǐyè] yè:-Ø [dance(noun) dance.HL-dance.L] come.Perf.L-3SgS 'He/She came (while)dancing.'

My assistant regularly produced examples of this type when combining a final motion verb with a VP denoting a co-occurring activity. For more examples of this tonally defined iterative pattern and a discussion of tonal contours, see §xxx.

15.1.8 Chains including mɔs:ndi:- 'be/do together'

By itself, the verb mɔ̃:ndi-yi- [mɔ̃:ndi:] is an intransitive verb meaning 'gather together, assemble'. It occurs chained with another VP to translate adverbial 'together'. Thus 'work together' is expressed as 'get together and work'.

- (xx1) a. m<u>o:ndî-yî bîrê-m-ê</u> gather-MP work-Impf-3PIS 'They work together.'
 - b. mɔ̃:ndi-yi ńni-m-ɛ̃
 gather-MP go-Impf-3PlS
 'They will go together.'

15.1.9 Chaining with jéjè⇒ 'go with'

The stem $j \in j = is$ a specialized element (cf. Jamsay $j \in j = is$) that functions syntactically like a transitive verb with a sense like 'have/take (something) with oneself', but occurs only in nonfinal position in chains, chiefly with motion verbs. It may take a direct object (denoting anything from an inanimate object to a human) that is not otherwise part of the argument structure of the following verb. The object may take Accusative marking.

- (xx1) a. sùmǎylâ pèrgé jéjè⇒ ǹnè-∅
 S sheep have.with go.Perf.L-3SgS
 'Soumaila went with (a/the) sheep.'
 or: 'Sumaila took (a/the) sheep along with him.'
 - b. ńnέ-ń jέjὲ⇒ nn-ò
 3SgS-Acc have.with go.Perf.L-3PlS
 'They took him/her along.'

15.2 Adverbial clauses with overt chaining or subordinating morpheme

This section begins with a number of alternative durative or imperfective VP or clause types. See also the direct chain type with iterated uninflected verb (§15.1.xxx, above). After covering these constructions, we look at others involving a temporal separation between the chained eventualities.

15.2.1 Backgrounded durative clauses (m)

Particle mò is used in backgrounded durative clauses. That is, the eventuality denoted by the /mò/ clause persists through a temporal extent T that leads up to (and may overlap with) a following foregrounded event.

/mò/ is added directly to the same form of the stem used before Imperfective -mì (-n). It would be possible to analyse /mò/ as containing, or as a variant form of, Imperfective -mì, but the vowel is obscure. Some examples showing the stem form are in (xx1).

bare stem	with mò	gloss	
yě:-	yé: mò	'come'	
yĭ:-	yî: mò	'see'	
ńné-	ńnî mò	'go'	
yègé-	yégé mò	'fall'	
gùnjó-	gŭnjó mò	'dig'	
bă:rî-	bă:rá mò	'help'	
bègîrî-	bègiré mò	'winnow'	
	yě:- yĭ:- ńné- yègé- gùnjó- bă:rî-	yě:- yé: mò yǐ:- yî: mò ńné- ńnî mò yègé- yégé mò gùnjó- gǔnjó mò bǎ:ri- bǎ:rá mò	

In elicited examples, the subjects of the $/m\delta$ / clause and the following clause may or may not be coindexed. If they are coindexed, the subject is omitted in the $/m\delta$ / clause (xx2.b). If the two subjects are not coindexed, the subject is marked in the $/m\delta$ / clause, and if this subject is pronominal it is expressed by an independent pronoun (xx2.a).

```
(xx2) a. [î: yé: mò] [bòndi wòè-Ø]
[1PlS come while] [rain(n) rain.fall.Perf.L-3SgS]

'As we were on our way here, it rained.'
```

```
b. [yé: mò] yèbúmbà yǐ:-só-ý

[come while] snake see-Perf2-1SgS

'On my way (= while coming) here, I saw a snake.'
```

In the sample text presented at the end of this work, the occurrences of mò are used in specific narrative contexts. An activity verb (such as a motion verb) is introduced in one clause (with its subject NP or pronominal), then from two to four identical mò clauses occur with background-clause intonation to indicate prolongation of this activity (usually with no repetition of the subject pronominal), then a new foregrounded event is introduced:

'The two of them were coming; **come** $m\partial$, **come** $m\partial$, **come** $m\partial$ [= they kept coming and coming]. (Then) a storm arose.' (xxx).

'The two of them **come** $m\hat{o}$, **come** $m\hat{o}$ [= were coming and coming], (and) when they had gone a little way, he (= hyena) said to hare: ...' (xxx)

mò may also be used after quasi-verb bú 'be' when the latter functions as an auxiliary verb, following another imperfective verb. A pronominal subject is again expressed by an independent pronoun. the In the sample text we find básâ-ⁿ ńné bú mò 'while he (= hyena) was pulling' (followed by: 'the goat lay down motionless') (xxx).

See also /mɔ/ in durative complements of 'see' and 'find' (§17.2.2.1).

15.2.2 Imperfective - n as subordinator

15.2.2.1 Imperfective -n on activity verb plus time-of-day verb

A time-of-day verb ('spend the night', 'spend the mid-day', etc.) may be chained with a preceding activity VP. The verb of the activity VP takes Imperfective subordinating suffix -ŋ, which is not inflected for pronominal subject. The verb stem has the same form as in the inflected Imperfective.

- (xx1) a. [gǐyé gîyé-ŋ̀] nàè-∅ [dance(noun) dance-**Impf**] **spend.night**.Perf.L-3SgS 'He/She spent the night dancing.' (= 'danced all night')
 - b. [gǐyé gîyé-ŋ̀] nà:-y
 [dance(noun) dance-**Impf**] **spend.night**.Perf.L-1SgS
 'I spent the night dancing.' (= 'danced all night')
 - c. [wórî wárá-ŋ] dĭrnε-Ø
 [farming do.farm.work-Impf] spend.midday.Perf.L-3SgS
 'He/She spent the (mid-)day farming.'

15.2.2.2 Imperfective -i) (-ii) plus bù- 'be'

This combination does not seem to be common, since there is a more productive Progressive verb form with sò- (§10.xxx). However, it was elicitable. From my assistant's comments, the bù- here has its literal meaning 'be present, be (in a place)', so the -m clause can be taken as subordinated.

An example is (xx1.a). When directly preceding $b\tilde{u}$ -, the Imperfective subordinator is always pronounced [m], which I take to reflect assimilation to the following labial. The negative counterpart has $[\eta]$ before $\tilde{\eta}g\acute{o}$ - 'not be', and the two velar nasals contract. This too could be a point-of-articulation assimilation, but in view of the clear $-\tilde{\eta}$ in the time-of-day construction (see just above) I normalize transcription as $-\tilde{\eta}$.

```
(xx1) a. [[tiyá yɛ:] nă: k5:-m] bù-∅ [[friend 1SgP.Poss.AnSg] meal eat-Impf] be-3SgS 'My friend is (present) eating a meal.'
```

```
b. nă: kó:-ŋ] ŋgó-ý
[meal eat-Impf] not.be-1SgS
'I am not (present) eating a meal.'
```

15.2.3 Imperfective adverbial clause with -m-sè gà

In this construction, the verb ends in Imperfective -m followed by -sè gà. The clause is otherwise in relative-clause form, with wágádî (or wágátî) 'time' in low-toned form as the head NP. A pronominal subject is expressed as an preverbal pronoun.

The clause can be translated as 'while' or 'when', with a progressive, stative, or other durative VP. The -sè gà clause and the main clause may have coindexed or disjoint subjects. Morphologically, gà can be taken as the Locative postposition, here 'at (the time when...)'. This leaves -sè, which is identical in form to the Perfective Participial suffix. I will gloss it accordingly. It is difficult to make sense of the combination of imperfective and perfective, but perhaps -sè here functions as a past-time marker here, rather than as a conventional perfective. If so, the construction might be parsed as 'at [the time when X was VP-ing]'. However, another option would be to take sègà as a fused morpheme.

```
(xx1) a. wàgàđi yû: ĭ:n wárá-m̀-sè gà, time.L millet 1SgS do.farm.work-Impf-Ppl.Perf at, bòndî wòè-∅ rain rain.fall.Perf.L-3SgS 'While I was (in the fields) farming millet, it rained.'
```

b. [tègé gá] ú bú-m-sè gà, [childhood in] 2SgS be-Impf-Ppl.Perf while, ŋgá yé:-m=bè-w here come-Impf=Past-2SgS 'When you were a child, you used to come here (often).'

-m̀-sɛ̀ is attested without gà. See (xx2) in §13.2.10, above. Historically, -m̀-sɛ̀ may have originated as a participial (i.e. relative-clause) form of the Progressive. The latter is now expressed by a suffix -sò- following a form of the verb stem with lengthened final stem vowel with a final L-tone element, with a less common variant -ŋ-sò- that retains an audible variant of Imperfective -m ~ -ŋ (§10.2.2.4). However, the Progressive now has a distinct participial form in -só-mǐ (§14.1.7.2), so there is no clear synchronic connection between it and -m̀-sɛ̀.

15.2.4 Simultaneous same-subject jè: before motion verb

A construction with -jè: added to a VP with its verb in bare-stem form, followed by a motion verb, denotes simultaneous co-events with the same subject. jè: has some resemblance to Recent Perfect jè- ($\S10.xxx$), which also follows the bare stem, but it cannot be identified with any specific form of jè-. There is no obvious semantic connection since the co-events are simultaneous. Another possible connection is with $j\acute{\epsilon}j\grave{\epsilon}\Rightarrow$ 'have/take (something) with oneself' ($\S15.1.9$).

- (xx2) a. [hálî mó:tî=yè] [yògó jè:] nnè-Ø [all.the.way.to M=Loc] [run while.SS] go.Perf.L-3SgS 'He/She ran all the way to Mopti.' (lit. "He/She went all the way to Mopti while running.")
 - b. [barkô dàmbî jè:] yè-y
 [gas.drum push while.SS] come.Perf.L-1SgS
 'I came (here) pushing a gas drum (large barrel).'

My assistant rejected combinations of jè: with following non-motion verbs.

15.2.5 Durative ... \(\xi\): on complement of \(/\d\xi\):-/ 'be tired/do long time'

The usual verb 'become tired (weary)' (or more generally 'suffer physically') is áyá-. Another verb, dě:-, is used in contexts like 'I worked until I got tired', or more freely 'I worked a very long time'.

In (xx1), from the sample text, the complement of $d\check{\epsilon}$:- is a verb with suffix - $\acute{\epsilon}$:. As we will see, the suffix is actually based (segmentally) on the unsuffixed Perfective (ending in $\{\epsilon \in i\}$ depending on the verb), but with lexical onset tone, and with the final vowel high-toned and (for mono- and bisyllabic verbs) long. I will gloss the ending in interlinears as "Dur[ative]," but will not hyphenate it.

```
(xx1)
       [tà-tã:
                   [bêr
                                           yê]
                              ReflSgP
       [hyena
                   [goat
                                           Poss.AnSg]
       bàsέ:
                    ńnέ
                                                       ńnî-ηὸ:-Ø
                              dě:
                                          nà]
                                                       go-ImpfNeg-3SgS
        pull.Dur
                    3SgS
                              be.tired
                                          then.DS]
        'Hyena tugged on his goat until he (= hyena) was exhausted (= for a
        long time), (but) it wouldn't go.'
```

I have found no syntactic context other than the construction with de:- that calls for this form of the complement. Similar examples with other verbs were readily elicited. They show that a non-3Sg pronominal subject is expressed by an independent pronoun in complement, not as a pronominal-subject suffix on the de:- verb (xx2). This suggests that even in (unsubordinated) main clauses, de:- in this construction is impersonal, rather than constituting a typical verb-chain (with shared subjects).

```
(xx2) [in birá biré:] dè:-∅

[1Sg work(noun) work(verb).Dur] be.tired.Perf.L-3SgS

'I worked until getting tired (= for a very long time).'
```

Forms of the complement verb from various stems are in (xx3). Observe that the lexical tone of the stem is always preserved in the onset. Also note the absence of lengthening of the final vowel in trisyllabic and longer stems.

```
(xx3)
             stem
                            Perfective
                                            before dě:-
                                                           gloss
         a. bisyllabic, final £:
             bàsá-
                            bàsè-
                                            bàsέ:
                                                            'pull'
                                                            'work'
             birέ-
                            birè-
                                            birέ:
                                                            'hit'
             súyś
                            sùyè-
                                            súyέ:
         b. bisyllabic, final é:
```

```
kóyó-
                   kòyè-
                                 kóyé:
                                                 'weep'
                                                 'jump off'
    péré-
                   pèrè-
                                 péré:
    gùnjó-
                   gùnjè-
                                 gùnjé:
                                                 'dig'
c. bisyllabic, final 1:
                                                 'tie'
    pági-
                   pàgi-
                                  págî:
    kó:sî-
                   kà:sì-
                                  kɔ́:sî:
                                                 'scratch'
    dèwî- [děw]
                  dèwi-
                                 dèwi:
                                                 'cover'
d. monosyllabic
                   kàè-
                                 kόέ:
                                                 'eat (meal)'
    k5:-
    ká:-
                   kàὲ-
                                 káέ:
                                                 'shave'
                                                 'go out'
                                 nòέ:
    nă:-
                   nàè-
                                 gòé:
                                                 'go out'
    gŏ:-
                   gòè-
    tέ:-
                                 tέ:
                                                 'lay out (mat)'
                   tὲ:-
                                                 'see'
    yĭ:-
                   yi:-
                                 yĭ:
e. Cvy
                                                 'hear'
                   nùy-
                                 nŭy
    nŭy-
                                                 'go in'
    núy-
                   nùy-
                                  núy
f. nCv
                                                 'go'
    ńné-
                   'nnὲ-
                                  ńnέ:
    ńdi-
                   ndi-
                                  ńdi:
                                                 'go'
                                                 ʻgoʻ
    ìdέ-
                   ndε-
                                 ìdέ:
g. trisyllabic with final /i/
                                                 'winnow by shaking'
    bègîrî-
                   bègiri-
                                 bègiri
                                                 'shake (grain)'
    yàgibi-
                   yàgibi-
                                 yàgibi
h. trisyllabic with final /e/
    bògóró-
                   bògòrè-
                                 bògóré
                                                 'bellow'
                                                 'dream'
    mònjúró-
                   mònjùrè-
                                 mònjúré
```

A **negative** counterpart can be formed by adding the same 3Sg Perfective dè:- 'be tired' to a pronominally inflected Perfective Negative verb (xx4).

```
(xx4) [nàmâ kùwò-rî-ý] dè:-∅

[meat eat-Perf.Neg-1SgS] be.tired.Perf.L-3SgS

'I didn't eat meat for a long time.'

(= 'I went a long time without eating meat.')
```

15.2.6 Clauses with /nà/ 'and then' (different subject, anterior)

Clauses with /nà/ denote eventualities that precede in time the reference time (in the main clause). The subject of the /nà/ clause is referentially disjoint from that of the main clause, and is therefore overtly expressed (for example, by an independent pronoun). This different-subject ("DS") subordinator follows a bare verb stem with lexical vocalism and tone, hence yǐ: nà 'see and', ńné nà 'go and', bă:rî nà 'help and ...', etc. A pronominal subject, if present, is expressed by an **independent pronoun** (as in a non-subject relative clause.) There are some examples in the sample text. Other examples are in (xx1).

```
(xx1) a.
          [á:mádù
                       [sèn
                                               ńnέ
                                                        bàsá
                                                                  nà]
                                 gú]
           [A
                       [rope.L
                                Def.InanSg] 3SgS
                                                        pull
                                                                  and.DS]
           pàrè-Ø
           snap.Perf.L-3SgS
           'Amadou pulled the rope and (then) it snapped.' (seni)
           (lit. "Amadou having pulled the rope, it snapped.")
       b. [[ńné
                      báy]
                               ú
                                         ńné
                                                   nà]
           [[3Sg
                      Dat]
                               2SgS
                                                   and.DS]
                                         go
           ſú-n
                             ndε-ri-∅]
           [2Sg-Acc
                             give-PerfNeg-3SgS]
           'You-Sg went to him, but (then) he didn't give (it) to you.'
```

15.2.7 Clauses with $\frac{\dot{\eta}}{(same-subject, anterior)}$

/ŋ/ appears at the end of a clause whose time frame precedes that of the main clause. The subjects are coindexed, and the interlinear gloss is 'and.SS' (for "same subject"). The subject is normally expressed only once in the two-clause sequence. This form is extremely common and is clearly the basic same-subject subordinator involving chronologically sequenced events. The verb always has its **bare-stem form**, with no tone changes or vocalic irregularities: wàrá ŋ 'do farm work ...', bă:rî ŋ 'help and ...', yǐ: ŋ 'see and ...', nné ŋ 'go and ...'. Examples, in addition to several in the sample text, are in (xx1).

```
(xx1) a. [yě: ý] èwy-à
[come and.SS] sit.Perf.L-3PIS
'They came and sat.'

b. [[ńnɛ báy] ńne ý] [ńnɛ-ŋ
```

Dat]

[[3Sg

tèmbè-rú-w]

and.SS] [3Sg-Acc find-PerfNeg-2SgS]

'You-Sg went to him (= to his place), and (= but) you didn't find him (there).'

15.2.8 /gáy/ or /-wndé/ '(and) then' (same-subject, anterior, future time)

/gáy is another same-subject subordinator. It can be glossed 'and then', and specifies that the events of the two chained clauses are temporally sequenced (but not separated by a lengthy interval). In elicitation, my assistant allowed it only in future contexts, above all when the subsequent clause is an imperative or hortative, but he also accepted indicative clauses like that in (xx1.b). This type of context is also seen in the occurrence of gáy in the sample text (xxx). See also (xx1) in $\S 8.4.6.2$ ('we'll work first, then we'll eat') and (xx2) in $\S 13.2.6$.

The verb in the gáy clause is in bare-stem form, and drops its tones.

- (xx1) a. [nă: kò: gáy] ńné-má [meal eat.L then.SS] go-Hort.1Dual 'Let's eat and then go!' (kó:)]
 - b. [wàrà gáy] ńnî-ŋ
 [do.farm.work then.SS] go-Impf.3SgS
 'He/She did farm will work (= worked in the field) and then go.'
 (wàrá)
 - c. [ŋgá sèmbi gáy] nho [here sweep.L then.SS] go-Impf.3SgS 'Sweep up here and then go!' (sémbî)

Combinations with verbs that show irregularities elsewhere have regular form: yi: gáy 'see and then ...', nnè gáy 'go and then ...'.

An alternative to gáy with similar restrictions is -wndé, which also controls tone-dropping on the preceding verb. It is attested in nonfinal verbs in chains ending with an imperative.

- (xx2) a. yè:-wìdé bíndò come.L-then.SS go.back.Imprt 'Come-2Sg (here) and then go back!'
 - b. nnè-wndé yô:
 go.L-then.SS come.Imprt
 'Go-2Sg and then come (back)!'

-wndé most likely contains -nde 'if' (Chapter 16), but it is semantically specialized and has no conditional nuances.

15.3 Other temporal adverbial clauses

```
15.3.1 'Since ...' clauses (sê dày<sup>n</sup>)
```

The bare verb stem of a 'since' clause is followed by /sê/ and then optionally by dàyⁿ (cf. noun dăyⁿ 'limit, bounds'). The clause is normally positive. However, a negative 'since' clause can be constructed using a Perfective Negative

- (xxx) a. [bû: yě: sê dàyⁿ] [nî: dǐyò-ndú] [3PlS come since since] [water bathe-PerfNeg.3PlS] 'Since they came, they have not bathed.'
 - b. [mùr¹á ńnέ-ń wŏ: sê dày¹] gò:-rí-∅
 [sickness 3Sg-Acc catch since since] go.out-PerfNeg-3SgS
 'Since she got sick ("sickness caught her"), she has not gone out.'
 - c. [làwà-rî â: kárⁿî sê dàyⁿ] gò:-ndú
 [pass-PerfNeg 3PlLogo do since since] go.out-PerfNeg.3PlS
 'Since (the day when) they did not pass (= they failed exams), they haven't gone out.'

'Since X' with a temporal NP as X is expressed as [X bă:] or (less often) [X dàyⁿ]. Thus yéŋĭrⁿĭ: bă: 'since yesterday', less often yéŋĭrⁿĭ: dàyⁿ.

15.4 Noun-headed temporal clause ('the time when ...')

wágádi 'time, moment in time' (a regional word ultimately from Arabic) can be used as the (low-toned) head of a relative clause that functions as a temporal adverbial clause. The relative clause as a whole takes the Locative postposition, which fuses with a Definite morpheme (e.g. $/g\acute{u}$ gá $/>g\acute{a}$). The tone-dropping on the participle is attributable to the Definite morpheme.

(xx1) wàgàdi ńné yègè-sè gá time.L 3SgS fall-Ppl.Perf.L Def.InanSg.Loc 'at the time when he fell' (= 'when he fell') Of course any temporal noun such as 'day', 'morning', 'month', 'year', etc. can also be the head a temporal adverbial relative clause.

15.4.1 Reverse anteriority clause with final múgò 'before ...'

'Before' clauses are expressed by clause-final múgò, with tone-dropping on the preceding bare verb stem. Although I struggle to segment or parse it, my assistant's intuition is that múgò is the combination of a suffix -mî on the verb (which drops tones), followed by /gò/. If forced to gloss each morpheme, one might connect -mú (-mî) with the Imperfective suffixes (e.g. -mì in relative-clause participles), and /gò/ with the Locative postposition (gó and allomorphs). However, the tone pattern makes no sense in this interpretation, and I consider it to be unanalysable.

The subject may be expressed by an independent pronoun, like $b\hat{u}$: 'they' in (xx1.a). The subject is usually omitted when the subject of the 'before' clause and that of a juxtaposed main clause are coindexed (xx1.b).

- (xx1) a. [bǐndé bû: yè: múgò] dàwî-yî-èrè
 [go.back 3PlS come.L before] hide-MP-Perf1a-1SgS
 'Before they came back, I hid (myself).'
 - b. [nă: kò: múgò] bîrέ-m-iy
 [meal eat.L before] work-Impf-1PIS
 'We will work before eating.'

15.4.2 'No sooner ..., than ...' ($b\acute{u} \Rightarrow$, $p\acute{u} \Rightarrow$)

In (xx1), the particle $b\acute{u}\Rightarrow$ (variant $p\acute{u}\Rightarrow$ or $f\acute{u}\Rightarrow$) with H-tone is added to a headless definite adverbial relative clause ending in Definite Inanimate Singular $/g\acute{u}/.p\acute{u}\Rightarrow$ varying with $f\acute{u}\Rightarrow$ (less often $b\acute{u}\Rightarrow$) is elsewhere one of the 'all' quantifiers. In this construction, Definite $/g\acute{u}/$ itself is pronounced with low pitch, which I interpret to be an intonational effect on an element followed by the emphatically pronounced $/b\acute{u}\Rightarrow/$. The pronominal subject of the 'as soon as' clause is expressed as an independent pronoun. The subjects of the two clauses may the same or different.

(xx1) a. [î: yè:-sè
$$gú \Downarrow bú \Rightarrow$$
]
[1Pl come-Ppl.Perf.L Def.InanSg all]
bî-bĭyè-y.:

Rdp-lie.down.Perf.L-1PIS 'As soon as we came (= arrived), we went to bed.'

- b. [ňă: ĭ:ⁿ kò:-sè gù bú⇒] nnè-yⁿ
 [meal 1Sg eat-Ppl.Perf.L Def.InanSg all] go.Perf.L-1SgS
 'As soon as I had eaten the meal, I went away.'
- c. [î: yè:-sè gú↓ bú⇒]
 [1Pl come-Ppl.Perf.L Def.InanSg all]
 [bòndî wòè-∅]
 [rain rain.fall.Perf.L-3SgS]

 'As soon as we came, the rain fell.'

In (xx2), $b\acute{u}\Rightarrow (p\acute{u}\Rightarrow)$ occurs at the end of temporal clause with bare verb stem and Different-Subject subordinator $/n\grave{a}/$. The pronominal subject of the 'as soon as' clause is expressed by an independent pronoun. This construction is only used when the **subjects are disjoint** and the time reference is past.

(xx2) [búrâ: ĭ:ⁿ táwá nà pú⇒] kòyè-Ø [B 1Sg touch and.DS all] weep.Perf.L-3SgS 'As soon as I touched Boura, he wept (=began weeping).'

In (xx3), $b\acute{u}\Rightarrow$ with L-tone follows a **regular inflected verb** (Perfective aspect). The subjects of the two clauses may be coindexed (xx3.a) or disjoint (xx3.b). This construction is used when the time frame is **in the future** or is **gnomic** (denoting a recurrent event).

- (xx3) a. [séwá:ré dŏ-èrè-y∴ bú⇒] bǐyé-m-ĭy [S arrive-Perf-1PlS **as.soon**] lie.down-Impf-1PlS 'As soon as we arrive in Sevare, we will go to bed.'
 - b. [búrâ: táwá-tù-w bú⇒] kóyô-ⁿ
 [B touch-Perf-2SgS as.soon] weep-Impf.3SgS
 'As soon as you-Sg touch Boura, he will weep.' (warning)
 - c. [fántà éwyé-èrè-Ø bú⇒] nîyⁿê-ⁿ
 [F sit-Perf-3SgS as.soon] sleep-Impf.3SgS
 'As soon as Fanta sits down, she falls asleep.' (general statement)

Other 'all' quantifiers are more common in postnominal position (kéréw, bé:ndè). Variants of pú=> or fú=> occur in other languages of the zone in both 'as soon as' and 'all' functions (Jamsay, Fulfulde, etc.).

A less common 'as soon as' construction is with final tán (§16.2.2).

15.5 Spatial and manner adverbials

15.5.1 Spatial adverbial clause ('where ...')

The noun /3:/ 'place' occurs in L-toned form /3:/ as the relative head.

```
(xx1) a. [[bèrèmbî ɔ: bèrè-mbè-m] gá]
[[herder place.L graze-Caus-Ppl.Impf.L] Def.InanSg.Loc]
nnô
go.Imprt
'Go to the place where the herders are having (the livestock) graze!'
```

b. [[5: ńné yègè-sè gú] wàgá [[place.L 3Sg fall-Ppl.Perf.L Def.InanSg] be.far 'The place where he/she fell is far away.'

15.5.2 Manner adverbial clause ('how ...')

A relative clause with $d\check{a}y^n$ 'manner' may function as a NP (xx1.a). With may 'like' this can become a manner adverbial clause (xx1.b).

```
(xx1) a. [dày<sup>n</sup> ńnέ bîré-m̀] èsù≡ndó-∅

[manner.L 3SgS work-Ppl.Impf] be.good≡not.be-3SgS

'The way he/she works isn't good.'
```

b. [[dàyⁿ ńnέ bîrέ-m] màyⁿ] bîrέ-m-∅
 [[manner.L 3SgS work-Ppl.Impf] like] work-Impf-1SgS
 'I work the (same) way he/she works.'

15.5.3 Headless adverbial clause as spatiotemporal or manner clause

The 'time', 'place', or 'manner' head noun may be omitted. The result is a headless adverbial relative clause whose exact interpretation requires contextual decipherment. In some cases there may be no determinable specific head noun, and something like 'situation' may be useful in translation. In cases like (xx1.a), my assistant preferred a temporal reading even though 'be far' might have

hinted at a spatial reading. In (xx1.b), the 'like' particle forces a manner adverbial reading.

```
(xx1) a. [[ńné yègè-sè gú] wàgá

[[3Sg fall-Ppl.Perf.L Def.InanSg] be.far

'(The time) when he/she fell is far away (= was long ago).'
```

```
b. [[ńné biré-m] mày"] biré-m-Ø [[3SgS work-Ppl.Impf] like] work-Impf-1SgS 'I work like (the way) he/she works.'
```

15.5.4 'From X, until (or: all the way to) Y'

Parallel adverbial relatives ending with dayⁿ (possessed, low-toned form of dayⁿ 'limit') can be used to specify initial and terminal points of an extended duration. hale 'all the way to' is optional at the beginning of the second clause.

```
nàr<sup>n</sup>á-sè]
                                                               dày<sup>n</sup>]
(xx1)
         [[bû:
                     bû:-n
                                   bear.child-Ppl.Perf]
         [[3P1S
                     3Pl-Acc
                                                              limit.L]
                                        túwé-m]
                                                             dày<sup>n</sup>]
         [[hálè)
                         â:
                         ReflP1
                                        die-Ppl.Impf]
         [[(until)
                                                             limit.L]
         mòsî-yέ
         be.bad-3PIS
          'From when they (= their mothers) bear them<sub>x</sub>, until when they<sub>x</sub> die,
```

they are wicked.'

15.5.5 'As though ...' clause

A clause may be followed by /may/ 'like' (§xxx) in an 'as though' manner adverbial. Since there is no subjunctive in Nanga, there is no distinction between 'as/like ...' clauses and 'as though ...' clauses (with the special pragmatic force of the latter). If the manner adverbial has a pronominal subject, it appears as an independent pronoun, and there is no pronominal-subject suffix on the verb, except for 3Pl subject (xx1.b).

```
(xx1) a. [[ú nă: kò:-rî] máy] kóyò-sò-ẁ [[2SgS meal eat-PerfNeg] like] weep-Progr-2SgS 'You-Sg are crying as though you hadn't eaten.'
```

```
b. [[nă: kò:-ndú] máy] kóyò-s-è
```

[[meal eat-PerfNeg.3PIS] like] weep-Progr-3PIS 'They are crying as though they hadn't eaten.'

16 Conditional constructions

16.1 Hypothetical conditional with -nde 'if'

In a hypothetical conditional, the antedecent clause ends in -nde 'if', suffixed to the verb or other predicate. When it follows any **negative predicate** ending in a high vowel (forms of Perfective Negative -ri-, Stative Negative clitic =ndó-, ngó- 'not be (present)', and clitic =ndó:- 'not be'), -nde **assimilates the high tone**, as in yè:-rú-w-ndé 'if you don't come' and yǎ-n=ndó:-Ø-ndé 'if he/she is not a woman'. However, when -nde is added to a **positive predicate**, it has **low tone even when the predicate ends in a high tone**: kiyé-só-w-ndè 'if you-Sg say' (Perfective-2), yǎ-ŋ=ndè 'if he/she is a woman'. An exception is the unsuffixed Perfective, which is low-toned but is followed by a high-toned -ndé, thus kùwò-ndé 'if he ate (meat)', (xxx) in the sample text.

Aside from -nde (and its phonological interactions with the verb), both the antecedent and consequent clauses have main-clause-like form. When the antecedent denotes a time-bounded event, and the consequent denotes an eventuality that follows the antecedent event, the unmarked pattern is for a **perfective-system predicate** in the antecedent and either an imperfective-system predicate, an imperative, or a question in the consequent. Either clause may be negated, and there is no requirement of shared subjects.

Perhaps related to -nde but formally and semantically specialized is -wndé in nonfinal position in verb chains with future time reference, see §15.2.7.

16.1.1 Regular antecedent clause with pronominal subject suffix

In simple hypothetical considitionals, the usual consequent clause is an imperative (or hortative) as in (xx1.a), or an imperfective clause as in (xx1.b).

```
á:mádù
(xxx) a. [éwé
                     gá]
                                       yĭ:
                                              jὲ-ẁ-ndè,
           [market in]
                           Amadou
                                              RecPf-2SgS-if,
                                       see
           yágâ
           flee.Imprt
           'If you-Sg see Amadou in the market, flee!'
       b. [éwé
                     gál
                             á:mádù
                                       yĭ:
                                              įὲ-ỳ-ndè,
```

Amadou

[market in]

see

RecPf-1SgS-if

yò-yógó-m̀-∅

Rdp-run-Impf-1SgS

'If I see Amadou in the market, I will flee.'

c. háwâ nă: kó:-ŋò:-Ø-ndè, túwé-ŋ̀ H meal eat-**Impf.Neg-**3SgS-i**f**, die-**Impf**.3SgS 'If Hawa doesn't eat, she will die.'

The antecedent may have almost any form from the perfective-system arsenal, including the unsuffixed perfective (xx2.a), Perfective-1a -èrè-, Perfective-1b -tì- (xx2.b), and Recent Perfect jè-. The verb may also be stative, for example with quasi-verb 'be' or 'have' (xx2.c-d).

Perfective-2 -só- is avoided in this construction with verbs like yǐ:- 'see' that, in main clauses, regularly take -só- in preference to other perfectives; these verbs switch to Recent Perfject jɛ-. However, -só- is recorded in kiyɛ-só- \emptyset -nde 'if he/she says', so there is no strict grammatical constraint against this combination.

- (xx2) a. á:mádù yì:-w-ndè
 Amadou see.Perf.L-2SgS-if
 'if you-Sg see Amadou'
 - b. yî: súyó-tì-w-ndè child hit-Perf-2SgS-if 'if you-Sg strike the child'
 - c. nàná yá sò-w-ndè cow Exist have-2SgS-if 'if you-Sg have a cow'
 - d. yá b-è-ndè
 Exist be-3PlS-if
 'if they are (there)'
 - e. njî-ý yĭ: jè-w-ndè 1Sg-Acc see RecPf-2SgS-if 'if you-Sg see me'
 - f. [kò kámâ] sò-ndó-ẃ-ndé [thing.L any] have-Neg-2SgS-if 'if you-Sg have nothing'

g. dógð≡ndő:-Ø-ndé Dogon=not.be-3SgS-if 'if he/she is not a Dogon (person)'

As usual, pronominal-subject suffixes consisting of a semivowel (1Sg -y, 1Pl -y., 2Sg -w, 2Pl -w.) monophthongize with a preceding homorganic vowel (i, u). In addition, before -ndè 'if', 1Sg -y is sometimes monophthongized with a preceding $\{e\ \epsilon\}$, resulting in a long vowel with no clearly articulated upglide. The fact that ynd is a triple cluster is undoubtedly behind this. In (xx3.a), 1Sg ńnè-rè-y-ndè varies with monophthongized ńnè-rè:- \emptyset -ndè. The latter is still audibly distinct both from 3Sg ńnè-rè- \emptyset -ndè 'if he/she goes' and from 1Pl ńnè-rè:-y-ndè 'if we go'.

```
(xx3) á:ndê≡∅ ńné-èrè:-∅-ndè,

Anda≡in go-Perf-1SgS-if,

[[ámbéri ndò] gò] sigé-mò-∅

[[chief house.L] in] go.down-Impf-1SgS

'If I go to Anda, I will go down (= lodge) at the chief's home.'
```

16.1.2 'Unless' antecedent

This is simply a regular hypothetical conditional in which the antecedent clause is negative.

- - b. [ámbéri ng-é-ndé] [pèrgé sémé béré-ŋô-y∴] [chief not.be-3PlS-if] [sheep slaughter can-ImpfNeg-1PlS] 'Unless the chiefs are here ("if the chiefs are not present"), we cannot slaughter a sheep.'
 - c. [wará-rú-w-ndé]
 [do.farm.work-PerfNeg-2SgS-if]
 [ú amâyⁿ nă: kɔ́:-m-w]
 [2Sg how meal eat-Impf-2SgS]

'Unless you-2Sg do ("f you-2Sg don't do) the farm work, how will you eat?'

16.2 Alternative 'if' particles

```
16.2.1 'Even if ...' (dèr''i, mò)
```

Elicited 'even if ...' examples have (what looks like) Purposive postposition derni at the end of the 'if' clause, replacing the usual 'if' morpheme. The verb of the 'even if' clause has regular pronominal-subject inflections (xx1.a-c).

- (xx1) a. [yĕ:-só-Ø dèrⁿi] [ŋgá nă: kó:-ŋɔ̄:-Ø] [come-Perf2-3SgS even] [here meal eat-Impf.Neg-3SgS] 'Even if he/she comes, he/she won't eat here.'
 - b. [yě:-s-é dèrⁿi] [ŋgá nă: kó:-ŋ-è:]
 [come-Perf2-3PlS **even**] [here meal eat-Impf.Neg-3PlS]
 'Even if they come, they won't eat here.'
 - c. [yě:-só-w dèr"i] [ŋgá nă: kớ:-ŋɔ-w¹] [come-Perf2-2SgS even] [here meal eat-Impf.Neg-2SgS] 'Even if you-Sg come, you won't eat here.'
 - d. [bòndi wǒ:-só-∅ dèri] ńni-m-iy [rain rain.fall-Perf2-3SgS even] go-Impf-1PlS 'Even if it rains, we'll go.'

16.2.2 'As soon as ...' (tán)

tán (variant táŋ) is a Fulfulde particle meaning 'only'. It is widely used in other nearby Dogon and Songhay languages, but only as an alternative clause-final 'if' particle. It does not appear to be very common in Nanga but examples were elicited, with the meaning 'as soon as'. In this function it combines with preceding same-subject subordinator $\hat{\eta}$ or different-subject subordinator nà, depending on whether the two clauses have the same subject.

'As soon as you-Sg came, the rain began.'

The more common 'as soon as' construction is with clause-final $b\hat{u} \Rightarrow$ (§15.4.2).

16.3 Willy-nilly and disjunctive antecedents ('whether X or Y ...')

In this construction, the two mutually incompatible conditions (both irrelevant to the consequent) are spelled out, sometimes with a final $k\hat{\epsilon}w$ 'each/all'. (xx1.a) is a simple example where the second condition is the negation of the first. In (xx1.b), the two conditions are understood to be thought quotations from the reported agent's perspective, and each of the conditions is framed by 'he looks (= considers)'.

```
(xx1) a. [bòndi wɔ́:-ŋ̀ wɔ́:-ŋò:-Ø kɛ̂w]
[rain rain.fall-Impf.3SgS rain.fall-Impf.Neg-3SgS all]
nni-m-iy
go-Impf-1PIS
'Whether it rains or not, we are going.'
```

```
b. [[nǔ:
                       gŝ]
                                              \mathfrak{pir}^{n} \hat{\epsilon} - \mathfrak{p} \hat{\sigma} : -\emptyset
                                              look-Impf.Neg-3SgS
                       Poss.InanSg]
     [[person
     [[á
                       gŝ]
                                              \mathfrak{g}^n \hat{\epsilon} - \mathfrak{g} \hat{\delta} : -\emptyset
     [[LogoSgP
                      Poss.InanSg]
                                              look-Impf.Neg-3SgS
     ſńnέ
                     gàl
                                    kéréw
                                                       kò-kó:-n
     [3Sg
                     Topic]
                                    all
                                                       Rdp-eat-Impf.3SgS
     'He doesn't look at (= care whether) "it somebody's", he doesn't
     look at "it is mine," as for him, he eats everything.'
```

16.4 Counterfactual conditional

In this construction, the antecedent clause has -ndè 'if' as with hypothetical conditionals. However, now both the antecedent and the consequent have Past clitic = $b\varepsilon$ -. Examples are in (xx1).

```
(xx1) a. kà-kã: yò:-ndú=b-á-ndè,
Rdp-grasshopper come-Perf.Neg.3PIS=Past-3PIS-if,
[tàrá gá] yû: èsî⇒ béré-m̀=bè-y∴
[granary in] millet very.much get-Past-1PIS
```

'If the locusts hadn't come, we would have gotten (= were going to get) a lot of millet in the granary.'

b. dògòtórô yá bù-m≡bè-⊘-ndè, doctor Exist be-3SgS≡Past-3Pl-if, bàyá-è-rè≡bè-ỳ

be.cured-Perf≡Past-1SgS

'If the doctor had been there, I would have been cured.'

17 Complement and purposive clauses

17.1 Quotative complement

Quotations are marked by up to three distinct features:

- (xx1) a. inflectable 'say' verb (kîyέ-), preceding or following the quotation, §17.1.2;
 - b. invariable **quotative particle** /wa/ (i.e. wà or wá) following the quotation (or multiple segments of the quotation), §17.1.3;
 - c. **logophoric** pronouns substituting for (original) first person pronouns, §18.xxx.

17.1.1 Direct versus indirect in quotative complements

write

17.1.2 'Say that ...' with inflectable 'say' verb (kîyε-)

kîyé- 'say' may precede or follow the quotation. When it precedes, it is set off prosodically from the quotation. When it follows, there is no obligatory prosodic break.

In normal perfective positive use ('X said that ...'), a common form of the verb is invariant kîyêsêwndê, with variant kîyêsêgúndê. Morphologically, this looks like a sequence beginning in kîyê- 'say' and ending with -ndê 'if'. The middle portion is not easily parsable, though we might see Perfective Participial -sê, Inanimate -w, and Inanimate Singular Definite gú, as components of one or both variants. Since it has no conditional-antecedent modal quality, the morphemic identification of the final ...nde as the 'if suffix raises more questions than it answers. I will therefore transcribe without morphemic hypthes.

This 'say' form is preceded by an independent pronoun, even when this pronoun merely resumes an immediately preceding nonpronominal NP. This suggests that the nonpronominal NP is topicalized, and provides further evidence that kîyésêwndè at least originated as a non-subject relative (cf. 'what I said [was]: ...).

- [mò:dibò kiyésêwndè (xx1) a. bû:] bû: Def.AnPl] [holy.man.L **AnPlS** say kà-kã: yé:-η-è: wà come-Impf.Neg-3PIS Rdp-grasshopper say 'The holy men (= clerics) said that the locusts won't come (back).'
 - b. ńnś kiyésêwndè, á ś:ŋî ńnî-ŋ̀ wà 3Sg **say**, LogoSgS tomorrow go-Impf.3SgS say 'He_x said that he_x is going tomorrow.'
 - c. ĭ:ⁿ kiyésêwndè, ńnî-ŋɔ-yⁿ 1SgS **say**, go-Impf.Neg-1SgS 'I said that I am not going.'

In imperfective positive (e.g. present or future) contexts, the morphologically regular imperfective of $kiy\dot{\epsilon}$ - is used (xx2). If a subject NP for 'say' is present, there is no resumptive subject pronoun (xx2.b).

- (xx2) a. ńné ki-kiyé-ŋ, á mùrⁿá≡ŋ wà 3SgS Rdp-say-Impf.3SgS, LogoSgS sick.one≡it.is.3SgS say 'She_x will say that she_x is sick.'
 - b. nǔ: kǐ-kíyé-m-è, â: bàyá-èrè wà person Rdp-say-Impf-3PlS, LogoPlS be.cured-Perf1a say '(The) people will say that they are cured.'

The 'say' verb may also **follow the quoted clause**. This is usual when the 'say' verb itself is negated or hypothetical.

- (xx3) a. [kà-kà: 1:-yέ-m bέrέ-m-∅] kǐyè-rî-ý [Rdp-grasshopper stand-Caus can-Impf-1SgS] say-Perf.Neg-1SgS 'I didn't say that I can stop the locusts.'
 - b. bû: [ĭ:" wá ú dé:rè-w] kîyé-ŋ-è:
 3PlS [1SgS say 2Sg more] say-Impf.Neg-3PlS
 'They don't say that I am better than you-Sg.'
 - c. [iyé ŋgá kó:-ŋɔ-yʰ] [nínɛ báy] kiyâ [today here eat-Impf.Neg-1SgS] [3Sg Dat] say.Imprt 'Tell him/her that I will not eat here today.'
 - d. \acute{n} n \acute{n} [\acute{a} \acute{s} ir \acute{n} \acute{e} - \acute{s} o- \acute{o}] \acute{k} iy \acute{e} - \acute{s} o- \acute{o} - \acute{n} d \grave{e} ,

```
3SgS [LogoSgS be.sated-Perf2-3SgS] say-Perf2-3SgS-if kà-kǎr kárá-ŋ̀ falsehood lie-Impf.3SgS 'If he says that he is full (of food), he is lying.' (kà-kàrú)
```

e. yé:-m-è kìyè-rî-∅ come-Impf-3PIS say-Perf.Neg-3SgS 'He_x didn't say that they_v will come.'

The complement of kúyé- 'say' may also be an NP such as interrogative 'what?' or a demonstrative. In this case, the 'say' verb follows the complement, behaving like a normal transitive.

```
(xx4) a. kò-né kiy-à what? say.Perf.L-3PlS 'What did they say?'

b. ŋgú kiyè-rú-ẃ
```

Dem.InanSg say-Perf.Neg-2SgS 'You-Sg didn't say that.'

The 'say' verb kiyé- is often omitted when the quotative particle /wa/ (see below) is present. When entire back-and-forth conversations are quoted, as happens often in tales, kiyé- is largely absent, while /wa/ occurs constantly.

17.1.3 Quotative clitic /wa/

This particle is very common when the quoted speaker is a third person. When the embedded quotation is propositional in nature, the particle can therefore have 'hearsay' evidential quality, suggesting that the present speaker does not vouch for the truth of the quoted proposition. However, the particle is also used with jussive complements (embedded imperatives), where truth is not at issue.

The particle occurs **at the end** of the quotation, with no prosodic break. When the preceding word ends in low tone (including falling tone), /wa/ has L-tone. When the preceding word ends in high tone (including rising tone), which occurs in negative predicates and some 'it is' predicates, we hear /wa/ with high tone. In other words, /wa/ is intrinsically atonal, but allows spreading of the preceding tone segment. Even after the phonological tone is determined in a given context, the pitch is subject to intonational modification.

(xx1) a. ńné kiyésêwndè,

```
3SgS say,

á sikórò sò-ndó-⊘ wá

LogoSgS sugar have-Neg-3SgS say

'He/she<sub>x</sub> said that he/she<sub>x</sub> has no sugar.'
```

- b. á sîkórò yá sò-∅ wà LogoSgS sugar Exist have-3SgS say '(He/she_x) said that he/she_x has some sugar.'
- c. á:mádù ńné kîyésêwndè,
 Amadou 3SgS say,
 [nù bû:] tŏ: tó:-s-é wá
 [person.L Def.AnPl] sowing sow-Perf2-3PlS say
 'Amadou said that the people have sown (the millet).'

If the quotation is itself multi-clausal, /wa/ is repeated after each clause (xx2.a). In addition, /wa/ is optionally repeated after an overt subject NP, if any, in a quoted clause. This is analogous to the subject-quotative particles in languages like Ben Tey. In addition, many quotations begin with an initial quoted vocative of the '(hey) you!' type, converted to third person pronoun (unless the original addressee is also a participant in the present speech act). This quoted vocative is obligatorily followed by /wa/. The effect is to indicate who the original quotation was directed to, and also to index a switch in speaker in narrative reporting a back-and-forth conversation.

In (xx2.a), we see /wa/ at the end of both quoted clauses, and optionally following the overt subject NP in the first clause. In (xx2.b), since 'you-Pl have' is conjugated, the clause-initial 2Pl û: is interpreted as a quoted vocative. The conversion to third person does not take place here, indicating that the original addressees are also the addressees in the current speech event. Therefore such a vocative makes it unnecessary to specify the original addressee in a dative complement to the 'say' verb ('He said, hey you, ...' = 'He said to you, ...'). In (xx2.c), the original addressee is not a participant in the current speech event, so the original 'hey you!' is expressed with a 3Sg pronoun (there are many examples like this in the sample text). The alternative is to quote the actual appellation of the original addressee as the quoted vocative (xx2.d).

```
(xx2)
                        kiyésêwndè,
      a. ńné
           3SgS
                        say,
           [nù
                       bû:
                                            yé:-m-è
                                   (wà)]
                                                                 wà
                      Def.AnPl]
                                  (say)]
                                            come-Impf-3PIS
           [person.L
                                                                 say
           ŋgá
                       nă:
                                 k5:-η-è:
                                                        wà
                                 eat-Impf.Neg-3PIS
           here
                       meal
                                                        say
```

'He/she said that the people will come, (but that) they won't eat here.'

```
b. ńné kiyésêwndè,

3SgS say,

[û: (wà)] sîkórò sò-ndó-ẃ∴ wà

[2PIS (say)] sugar have-Neg-2PIS say

'He/she<sub>x</sub> said (to you-Pl) that you-Pl [topic] have no sugar.'
```

```
kîyésêwndè,
c. tà-tã:
              cmći
                       báy]
                               ńnέ
              [hare
   hyena
                       Dat]
                               3SgS
                                        say,
    ſńnέ
              wá]
                        ńnî
                                          wà
   [3Sg
              say]
                        go.Hort.3Sg
                                          say
    'Hyena said to hare: hey you, go!'
```

```
d. tà-tà: ńné kîyésêwndè,
hyena 3SgS say,
[jòmó wá] ńnî wà
[hare say] go.Hort.3Sg say
'Hyena said: hey hare, go!'
```

The particle /wa/ is omitted (because redundant) when it would otherwise be **adjacent to the 'say' verb**, i.e., when the latter immediately follows the quotation (xx3.a). Even an intervening subject pronominal is enough to allow both the particle and the 'say' verb to appear (xx3.b).

```
(xx3) a. á:mádù [[nù bû: (wà)] tŏ: tó:-s-é
Amadou [[person.L Def.AnPl (say)] sowing sow-Perf2-3PlS
(#wà) kîyé-só-Ø
(#say) say-Perf2-3SgS
'Amadou said that the people have sown (the millet).'
```

```
b. [nù bû:] tŏ: tɔ́:-s-ɛ́ wà]
[person.L Def.AnPl] sowing sow-Perf2-3PlS say]
nnɛ́ kiyɛ́-só-Ø
3SgS say-Perf2-3SgS
'He said that the people have sown (the millet).'
```

The particle is also omitted when the 'say' verb is negated, even when the 'say' verb is (atypically) preposed, so that adjacency of 'say' verb and quotative particle is not possible (xx4.a). The particle is also omitted when the 'say' clause is a polar interrogative (xx4.b), or a conditional antecedent (xx4.c). In

other words, when the fact of the quoted speech event is itself **modally problematic**, the quotative particle is not added to the quotation.

```
(xx4) a. á:mádù ńné kǐyè-rí-Ø,
Amadou 3SgS say-Perf.Neg-3SgS,
[nù bû:] tŏ: tɔ́:-s-ɛ́ (#wá)
[person.L Def.AnPl] sowing sow-Perf2-3PlS (#say)
'Amadou did not say that the people have sown (the millet).'
```

```
    b. ńnέ kiyé-só-Ø má
    3SgS say-Perf2-3SgS Q
    [nǔ: tŏ: tó:-s-έ (#wá)]
    [person sowing sow-Perf2-3PlS (#say)]
    'Did he say that the people have sown (the millet)?'
```

When the quoted material takes the form of a factive complement ($\S17.xxx$), with final Definite morpheme /gú/, the quotative particle is not used.

```
(xx5)
                   kiyésêwndè,
        ńnέ
                                             [[nù
                                                             bû:]
        3SgS
                   say,
                                             [[person.L
                                                             Def.AnPl]
                                                               kà-kàrú:≡Ø
        tŏ:
                   tà:-sè
                                                   (\#w\grave{a})],
                                     gú
                   sow-Ppl.Perf.L Def.InanSg (#say)],
                                                               Rdp-lie-it.is.3SgS
        'If he says (= claims) that the people have sown (the millet), it's false.'
```

Quotative /wa/ is possible, but less systematic, when the quotation is attributed to the current addressee (xx6.a). In ordinary contexts (with no special modal attributes), it is absent when the addressee is also the subject of the quoted clause (xx6.b). The particle can be used in repetitions of what the addressee has just said if the speaker needs confirmation, but in this case there is usually no overt 2Sg pronominal.

```
(xx6) a. [ńnέ báy] kîyâ, á:mádù ńné-èrè-∅ (wà)

[3Sg Dat] say.Imprt, Amadou go-Perf1a-3SgS (say)

'Tell her that Amadou has gone.'
```

```
b. [bàr<sup>n</sup>î ú kîyésêwndè]
[1SgDat 2SgS say]
ú yègè-ẁ (?#wà)
2SgS fall.Perf.L-2SgS (say)
'You-Sg told me that it was <u>you</u> [focus] who fell down.'
```

c. é:nî wá

```
tomorrow say 
'(Did you say) "tomorrow"?'
```

The quotative particle is generally not used in citations of the speaker's own previous speech.

```
(xx7) [fántà bày] kîyé-só-y,

[Fanta Dat] say-Perf2-1SgS,

á:mádù ńné-èrè-∅ (?#wà)

Amadou go-Perf1a-3SgS (say)

'I told Fanta that Amadou has gone.'
```

Dat]

e.g. to escape arrest in (xx9), /wa/ reappears.

'Tell her that I have gone.'

[3Sg

In (xx8.a), presence of /wa/ is possible (though not obligatory). If the subject of the quoted proposition is switched to first person (xx8.b), absence of /wa/ is preferred.

```
á:mádù
                                                ńnέ-ὲrὲ-∅
(xx8) a. [ńné
                   báy]
                         kîyâ,
                                                                 wà
                         say.Imprt, Amadou
                   Dat]
                                                go-Perf1a-3SgS
           [3Sg
                                                                 say
           'Tell her that Amadou has gone.'
                                                          (?#wà)
       b. [ńné
                     báy]
                                        ńné-èrè-ỳ
                            kîyâ,
```

say.Imprt,

The difference is undoubtedly due to the different **epistemic status** of the quoted proposition in the two cases, rather than an automatic effect of subject type. If the context in (xx8.b) is adjusted so that it involves telling a falsehood,

go-Perf1a-1SgS

(?#say)

```
(xx9) [gendarme bày] kîyâ, ńné-èrè-ỳ wà

[policeman Dat] say.Imprt, go-Perf1a-1SgS say

'Tell the policeman that I have gone away.'
```

Similar nuances have been found with second person (=current addressee) as subject of the quoted proposition. (xx10.a) was elicited in the following context: the addressee has been injured in an accident, and is being informed that help is on the way. Quotative wa was usually absent, though possible, in this context. Here the speaker and especially the addressee are quite well aware of the injury having occurred. When the context was changed, so that the injury report was in fact a lie, presence of wa was consistent (xx10.b).

```
(xx10) a. á:mádù [d\hat{g}\hat{g}\hat{f}\hat{f}\hat{g}\hat{f}] \hat{b}\hat{a}y \hat{k}\hat{i}y\hat{\epsilon} \hat{i}\hat{\epsilon}-\emptyset,
```

```
Amadou [doctor Dat] say RecPf-3SgS

[ú (wá) bár¹ím-y-èrè-ẁ (?#wà)]

[2SgS (say) be.hurt-MP-Perf1a2SgS (?#say)]

'Amadou has already told (= informed) the doctor that you-Sg are injured.'
```

```
b. á:mádù [dògòtórô bày] kîyé-só-∅,
Amadou [doctor Dat] say-Perf-3SgS,
[ú bár¹ímíy¹-èrè-ẁ wà]
[2SgS be.hurt-Perf-2SgS say]
'Amadou has told the doctor (falsely) that you-Sg are injured.'
```

17.1.4 Jussive complement (reported imperative or hortative)

17.1.4.1 Quoted imperative

rewrite (third person Hortative)

When the quoted material represents an original **positive imperative**, the verb is not in the original Imperative form (§10.6.1). Instead, the verb appears in the same form as that used for third-person subject hortative as in wishes of the 'may God VERB you!' type. The morpheme in question is expressed segmentally as a suffix -y with Cv:- verbs (with shortened vowel), a mutation of a final vowel to /i/, or no segmental change. There is also a characteristic tonal contour. Examples: gŏ-y '(told X) to go out' (gŏ:-), gùrî '(told X) to steal' (gùró-), pîgîrè '(told X) to screw in' (pîgîré-). For the forms, see (xx1) in §10.6.4.

In quoted imperatives, this suffix is used regardless of original addressee number. The quotative particle /wa/ is added clause-finally, without a prosodic break, and it constitutes part of the phonological environment for syncope of the stem-final /i/ in forms like gùri.

An overt pronominal referring to the original addressee is not obligatory, as seen in (xx1).

```
(xx1) [ńnś báy] kîyâ,
[3Sg Dat] say.Imprt,
[tè: gú] náŋ wá
[tea.L Def.InanSg] put.up.Hort.3rd say
'Tell him to put the tea (kettle) up (on the burner)!' (náŋî-)
```

Very often, however, the original addressee is indexed in the form of a preverbal **independent pronoun**. This is a somewhat pro forma quoted vocative ('hey you!'), converted to the appropriate pronoun in the current speech event's deictic structure (hence usually converted from second to third person).

```
bày]
                                                    kiyέ-só-y,
(xx2)
       a.
            [yì-tègè
                           bû:
                                                    say-Perf2-1SgS,
            [child-Pl.L
                           Def.AnPl
                                         Dat]
            [bû:
                       wàl
                                 [gó-ý
                                                      wá]
            [3P1
                       say]
                                 [go.out-Hort.3rd
                                                      say
            'I told the children to go out.'
            (lit.: "I told the children: hey them! Go out!"")
```

- b. [ńnś báy] kîyâ, [ńnś wá] yé-ý wá [3Sg Dat] say.Imprt, [3SgS say] come-Hort.3rd say 'Tell him/her to come.'
- c. ńné kiyésêwndè, [i:n wá] [yé-y wá] 3SgS say, [1Sg say] [come-Hort.3rd say] 'He/She told me to come.'
- d. [bû: bày] kîyέ-só-ý say-Perf2-1SgS [3P1 Dat] yû: [bû: wà] tór wá [3P] millet pound.Hort.3rd savl sav 'I told them them to pound the millet!.' (/tóri/ <tóró-)

In a **reported prohibitive**, the verb appears in singular-subject prohibitive ((i.e., negative imperative) form, regardless of embedded-subject number. An independent pronoun denoting the original addressee, ostensibly a quoted vocative but somewhat pro forma, is often present as in reported positive imperatives. (xx3.a) has a singular embedded addressee, (xx4.b) a plural one.

```
(xx3) a.
            ſńnέ
                       báy]
                                  kîyâ,
            [3Sg
                       Dat]
                                  say.Imprt,
            (ńnέ
                       wá) [tè:
                                      gúl
                                                      nánî-r<sup>n</sup>á
                                                                          wá
            (3Sg)
                       say) [tea.L Def.InanSg]
                                                      put.up-Imprt.Neg say
            'Tell him not to put the tea (kettle) up (on the burner)!'
```

```
b [bû: bày] kîyâ,

[3Pl Dat] say.Imprt,

(bû: wà) [tè: gú] náŋî-r<sup>n</sup>á wá
```

(3Pl say) [tea.L Def.InanSg] put.up-Imprt.Neg say 'Tell them not to put the tea (kettle) up (on the burner)!'

17.1.4.2 Embedded hortative

In **reported positive hortatives** attributed to a third person speaker, the originally imperative verb appears with suffix $-\hat{\eta}$, regardless of original first-person inclusive number. Like other $/\eta$ / suffixes and clitics, this suffix may be reduced to nasalization of the final vowel. Examples of this form, which I called Quotative Hortative (QHort), are in (xx1), along with the dual and 3+ first person inclusive hortatives for comparison. The vocalism and tone contours of the stem are identical in the three cases. It is therefore reasonable to think of $-\hat{\eta}$ as historically a reduction of one or both of the Hortative suffixes. Compare the m/η alternation in the Imperfective verb paradigm, or in the conjugated 'it is' clitics.

(xx1)	gloss	dual hortative	3+ hortative	QHort
	'run' 'go out' 'buy' 'go'	yògó-má gŏ:-má éwé-má ńné-má	yògó-mày ⁿ gŏ:-mày ⁿ éwé-mày ⁿ ńné-mày ⁿ	yògó-ń (wà) gŏ:-ń (wà) éwé-ń (wà) ńné-ń (wà)
	'give' 'go in' 'scrub'	ńdé-má núy ⁿ ó-má púgúsó-má	ńdé-mày ⁿ núy ⁿ ó-mày ⁿ púgúsó-mày ⁿ	ńdέ-ή (wà) núy ⁿ ó-ή (wà) púgúsó-ή (wà)

The Quotative Hortative suffix It is followed by the quotative particle /wa/, which has low tone in spite of the preceding high-toned suffix. (This detail suggests that $-\acute{\eta}$ is specifically a reduction of the 3+ Hortative suffix -màyⁿ, rather than Dual -má.) There is usually no preverbal independent pronoun.

```
kîyésêwndè,
(xx2) a.
           ńnέ
                                              [ńné-ń
                                                            wà]
            3SgS
                        say,
                                              [go-QHort
                                                            say]
            'He/She said, let's go!'
        b. nné
                      kiyésêwndè,
                                        [pèrgé
                                                     έwέ-ή
                                                                       wà]
                                        sheep
                                                     buy-QHort
                                                                       say
            'He/She said, let's buy a sheep!'
           á:mádù
                     bàr<sup>n</sup>i
                                      kîyésêwndè,
                               ńnέ
                                                       [gŏ:-ń
                                                                    wàl
```

```
Amadou 1Sg.Dat 3Sg say, [go-QHort say] 'Amadou said to me, let's (=him and me) go out!'
```

A second person attributed speaker gets the same treatment, as long as the original first inclusive did not include the current speaker (xx3).

```
(xx3) [á:mádù bày] ú kíyésêwndè [ńné-ń wà]

[Amadou Dat] 2Sg say, [go-QHort say]

'You-Sg said to Amadou, let's (= you and Amadou) go!'
```

With attributed first-person speaker (self-quotation), which entails that the speaker was also part of the original group of prospective agents, the -ŋ suffix is not used. Instead, we get the original Hortative verb form, with suffix -má for first inclusive dual and -màyⁿ for 3+ first inclusive plural. Quotative /wa/ is absent.

- (xx4) a. [á:mádù bày] ĭ:ⁿ kîyésêwndè ńné-má [Amadou Dat] 1Sg say go-Hort.Du 'I said to Amadou, let's (= the two of us) go!'
 - b. [yì-tègê bày] i:n kiyésêwndè ńné-màyn [child-Pl Dat] 1Sg say go-Hort.3+
 'I said to the children, let's (= all of us) go!'

When the attributed speaker is second (but not third) person, if the original first inclusive included the current speaker, we again revert to the original hortative form used. Therefore (xx4.a) has the (unembedded) hortative form. By contrast, (xx4.b) with third person attributed speaker has the -\(\delta\) suffix.

- (xx4) a. bàrⁿî ú kîyésêwndè ńné-mà 1Sg.Dat 2Sg say go-Hort.Du 'You-Sg said to me, let's (= you and me) go!.'
 - b. bàrⁿi ńné kiyésêwndè ńné-ń wà 1Sg.Dat 2Sg say go-QHort say 'He said to me, let's (= him and me) go!.'

For **reported hortative negative**, the original hortative negative -ndǎ:-má or (3+ plural) -ndǎ:-màyⁿ, with variants -rá-má\-rá-màyⁿ and -ndá-má\-nd-màyⁿ depending on stem shape, is simplified to invariant -ndà:, which I gloss as Quotative Hortative Negative (QHortNeg). It is followed by the quotative particle /wa/. This -ndǎ:- pattern is used under the same syntactic conditions as

is the reported positive hortative in $-\hat{\mathbf{y}}$ (xx5.a-b). With e.g. first person attributed speaker (self-quotation), the original full form (dual or 3+ plural) is used (xx5.c).

```
kiyésêwndè
(xx5) a.
           tà-tã:
                         ńnέ
            hyena
                         3SgS
                                    say
            [jòmó
                                   ńné-ndà:
                        wá]
                                                     wà
            [hare
                                   go-QHortNeg
                        say]
                                                     say
            'Hyena said: hey hare, let's not go!'
```

- b. bàrⁿi ńné kîyésêwndè ńné-ndà: wà 1Sg.Dat 3SgS say go-Hort.Neg say 'He/She said to me, let's (= the two of us) not go!'
- c. i:n kîyésêwndè ńné-ndă:-má
 1SgS say go-Hort.Neg-Hort.Du
 'I said (to one other person), let's go!'

17.2 Factive (indicative) complements

17.2.1 'Know that ...' complement clause

The factive complement of 'know' contains a verb with regular AN (aspect-negation) inflection, but without pronominal-subject marking (except for 3Pl). A pronominal subject is expressed by an independent pronoun, as in relative clauses. The complement clause as a whole functions as a noun, and is followed by a pronominal possessor agreeing with the subject of the complement clause, along with an inanimate singular possessive classifier, even though this subject is already expressed by a NP or pronoun at the beginning of the clause.

An example with **Imperfective Negative** verb is (xx1). A super-literal gloss would be "I know [your [you not coming]]."

If we replace the 2Sg subject in (xx1) with the various other pronominal categories, we get the factive complements in (xx2), in all cases subordinated to a 'know' matrix clause (not shown). The various first and second person categories, and 3Sg, have their usual postnominal possessor forms with Inanimate Singular possessive classifier $\frac{1}{3}$ (kô). In the 3Pl, we get a low-

toned, short-voweled form bù gò, distinct from the usual 3Pl possessor combination bû: gò seen e.g. in [ńdô bû: gò] 'their house'. Note also that the 3Pl subject form has subject agreement on the verb (-ŋ-ɛ:), whereas all other categories have invariant -ŋò:. The special treatment of 3Pl subject in negative forms is also found in relative-clause participles (§14.xxx).

```
(xx2) a. [î:n yé:-ŋð: kồ:] 'that I am not coming' 'that we are not coming' [î: yé:-ŋð: î: gð] 'that we are not coming' 'that you-Pl are not coming' b. [ńné yé:-ŋð: nð] 'that he/she is not coming' c. [bû: yé:-ŋ-è: bù gð] 'that they are not coming'
```

An example with the **Perfective Negative** (-ri) is (xx3).

```
(xx3) [ú yè:-rî ú gô] júgó-m-∅
[2SgS come-PerfNeg 2SgP Poss.InanSg] know-Impf-1SgS
'I know that you-Sg did not come.'
```

Replacing 2Sg by other pronominal-subject categories, we have the data in (xx4). Again, the 3Pl is the only category requiring its specific pronominal-subject suffix on the verb.

```
(xx4) a. [î:n yè:-rî kɔ̃:] 'that I did not come'
[î: yè:-rî î: gɔ̃] 'that we did not come'
[û: yè:-rî û: gɔ̃] 'that you-Pl did not come'

b. [ńnɛ́ yè:-rî nɔ̃] 'that he/she did not come'

c. [bû: yè:-ndú bù gɔ̃] 'that they did not come'
```

An example with **(positive) reduplicated Imperfective** verb is (xx5). The Imperfective ending is $-\dot{m}$ (not $-\dot{\eta}$).

```
(xx5) [ú yè-yé:-mì ú gô] júgó-mì-∅
[2SgS come-Impf 2SgP Poss.InanSg] know-Impf-1SgS
'I know that you-Sg will come.'
```

With the other pronominal categories, the outputs are in (xx6). The 3Pl form again has its inflection on the verb.

```
(xx6) a. [i:n yè-yé:-m kɔ̃:] 'that I will come'
```

```
[î: yè-yé:-m̂ î: gò] 'that we will come' 

[û: yè-yé:-m̂ û: gò] 'that you-Pl will come'

b. [ńné yè-yé:-m̀ nò] 'that he/she will come'

c. [bû: yè-yé:-m-è bù gò] 'that they will come'
```

A (positive) **Perfective-2** example is (xx7). With pronominal categories other than 2Sg, the forms are in (xx8). Again, the 3Pl form stands out.

```
(xx7)
       Γú
                yě:-só
                              ú
                                       gŝ]
                                                       júgó-m̀-∅
        [2SgS come-Perf2
                                       Poss.InanSg]
                                                       know-Impf-1SgS
                             2SgP
        'I know that you-Sg came.'
       a. [i:" yě:-só kɔ̃:]
                                        'that I came'
(xx8)
            [î: yě:-só î: gò]
                                        'that we came'
                                        'that you-Pl came'
            [û: yě:-só û: gò]
        b. [ńné yě:-só nò]
                                        'that he/she came'
           [bû: yě:-s-é bù gò]
                                        'that they came'
```

Other AN inflections beyond the four illustrated above can also be used in this construction. I have verified this for the Perfective-1a (-èrè), Perfective-1b (-tì), and the Progressive (-sò), in each without pronominal-subject suffixation except in the 3Pl.

In English, there is a basic distinction between 'X know that P' and 'X know whether P' for some proposition P. In the first case, but not the second, the current speaker presupposes the truth of P (cf. also 'X realize that P'). In Nanga, as generally in languages of the region, this distinction is not made. Therefore e.g. 'X doesn't know that [Y is sick]' (when Y is in fact sick) is often expressed by the same construction that translates 'X doesn't know whether [Y is sick]'. In this construction, the factive complement is followed by polar interrogative particle /ma/. The philosophical significance of this distinction between English and Nanga is evident; in Nanga, one simply reports the state of mind of X, while in English the speaker superimposes his/her own knowledge on X's state of mind.

```
(xx9) ńnś [ú sèllè-ri mà∴]
3SgS [2SgS be.healthy-Perf.Neg Q]
júgó-ŋò:-∅
know-Impf.Neg-3SgS
'He/She doesn't know whether (= that) you-Sg are sick.'
```

The same factive clause type is used under some conditions with 'see' (§17.2.2.2).

17.2.2 'See (find, hear) that ...'

17.2.2.1 Direct-perception type (relative-clause complement)

When the complement of 'see' denotes a durative activity (or an imminent event at the point where it was about to happen) that was viewed by the agent, we get the construction in (xx1) with Durative subordinator $/m\delta/$ (§15.2.1).

```
(xx1) a. [nàná yégé mò] yì:-só-ý [cow.L fall while] see-Perf2-1SgS 'I saw the cow falling (or: about to fall).'
```

b. [yì-tègê gìyé bû: gîyé mò] yì:-só-ý [child-Pl dance(noun) 3PlS dance while] see-Perf2-1SgS 'I saw the children dance (= dancing).'

17.2.2.2 Recognition (inference, hearsay) construction

When the agent sees that an event has already taken place, the **passive construction** with -y $\hat{\epsilon}$ after a high-toned verb stem, and with tone-dropped subject NP ($\S 9.3.2$), may be used. In (xx1), what the agent sees is a cow on the ground in a position other than the usual resting position.

```
(xx1) [nàŋà yégé-yé] yì:-só-ý

[cow.L fall-Pass] see-Perf2-1SgS

'I saw the fallen cow.' (or: 'I saw that a cow had fallen.')
```

In (xx2), the speaker (who had expected the addressee to take the motorcycle away), arrives and sees that the motorcycle is still in its normal place. Here we get a **factive** complement of the same type used with 'know' (§17.2.1, above).

'I saw (= see) that you-Sg didn't take the motorcycle away.'

In (xx3), the speaker reports what he/she has heard from others. Again, the complement is **factive** in form.

(xx3) [ú bàmàkô=∅ ńnî-m ú gô] nǔyⁿ-só-ý [2SgS B≡in go-Impf 2SgP Poss.InanSg] hear-Perf2-1SgS 'I heard (= hear) that you-Sg are going to Bamako.'

17.2.3 Factive complement with tîlây≡∅ 'it is certain'

tilây=∅ 'it is certain' may precede an ordinary indicative clause, denoting a future event that is (all but) certain, or a situation or a past event that one infers from strong evidence or reasoning. tilây is a regionally ubiquitous form.

- (xx1) a. tîlây=Ø á:ndê=Ø ńnî-m-Ø certainty=it.is Anda=in go-Impf-1SgS 'It's certain (definite) that I will go to Anda.'
 - b. tîlây=⊘ séwá:rì=yè gó-èrè-⊘ certainty=it.is S go.out-Perfla-3Sgs 'He/She has certainly left Sevare (by now).'

17.3 Verbal Noun (and other nominal) complements

17.3.1 Structure of Verbal Noun Phrase

Subjects, objects, and other preverbal phrases are unchanged from main clauses to verbal-noun clauses. There is no compounding (or incorporation) of object nouns. A direct object may optionally have Accusative marking, and this is obligatory for 1Sg pronoun objects (which have a different stem-shape in the accusative) (xx1.a-c). A subject NP may also be present, in its usual form (xx1.d).

- (xx1) a. [yî: súyó-ndé] èsì≡ndó-∅ [child hit-VblN] good≡not.be-3SgS 'Hitting a child is not good.'
 - b. [nji-nj súyó-ndé] èsi=ndó-∅
 [1Sg-Acc hit-VblN] good=not.be-3SgS
 'Hitting me isn't good.

- c. [nàmá kúwó-ndɛ́] jɔ́rɔ́-m̀-∅ [meat eat.meat-VblN] want-Impf-1SgS 'I want to eat some meat.'
- d. [árⁿà yǎ-ý súyó-ndɛ́] ɛ̀sì=ndó-Ø
 [man woman-Sg hit-VblN] good=not.be-3SgS
 'A man's hitting a woman isn't good.'

In construction involving a different subjects in the main clause and the subordinated clause, other than those of the type (xx1.a), the verbal-noun clause is additionally followed by a pronominal possessor, as with 'prevent' (see just below).

17.3.2 'Prevent' (késé-, gă:ndi-)

The verb késé- 'cut' can be used in the sense 'prevent (motion), block', and more generally 'prevent (an activity)'. The complement is expressed as a verbal-noun clause, with the agent appearing as a postnominal possessor. The possessed NP as a whole is optionally followed by Definite Inanimate Singular gù (low-toned form).

- (xx1) a. [wàgárî≡yè] ńné-ndé kɔ̃: gù]
 [[W≡in go-VblN 1SgP.InanSg Def.InanSg]
 bòndî kèsè-∅
 rain cut.Perf.L-3SgS
 'The rain prevented me from going (= blocked my going) to
 Wakara.'
 - b. [yù:-wórî kồ: gù] [millet.L-farming 1SgP.InanSg Def.InanSg]
 ùsú kú-ń kèsè-∅
 sun InanSg-Acc cut.Perf.-3SgS
 'My farming millet, the (hot) sun blocked me from doing it (= farming).'

Another 'prevent' verb is gă:ndî-, which also takes verbal-noun complements ending in a possessor (xx2).

'They prevented Amadou from sleeping.'

17.3.3 'Dare' (dă:rî-)

This verb takes verbal noun complements. The subjects of the two clauses are coindexed. There is no possessor following the verbal noun.

- (xx2) a. sigé-ndé dă:rá-ŋ-è: go.down-VblN dare-Impf.Neg-3PlS 'They don't dare go down.'
 - b. [pà:ŋgŏ: báy] bèré-ndíyé-ndé dă:rá-m̂-w mà [elephant Dat] approach-Inch-VblN dare-Impf-2SgS Q
 'Do you-Sg dare get close to the elephant(s)?'
 - c. [bărⁿ tîŋ-ndɛ̃] dă:r-só-Ø [1SgDat speak-VblN] dare-Perf2-3SgS 'He/She dared to speak to me.' (bàrⁿ1)

17.3.4 'Consent' (àwá)

The verb awá- 'accept, receive, take (sth given)' is also used in the sense 'consent, give one's agreement' with a clauseal complement. In (xx1), the complement's subject is coindexed with that of the higher clause, and there is no possessor of the verbal noun.

In (xx2), the two subjects are not coindexed. In one construction, the lower-clause is a verbal noun with following possessor (xx2.a). In another, the complement takes Imperfective -mi on the verb.

```
(xx2) a. [i:
                             bà:]
           [1PlP
                             father.L]
           [sîkásò
                         î:
                                   ńnέ-ndέ
                                              î:
                                                       gò]
           [Sikasso
                         1PlS
                                   go-Impf
                                              1PlP
                                                       Poss.InanSg]
           àwá
                      įὲ-Ø
                      RecPf-3SgS
           accept
```

'Our father has consented to our going to Sikasso.'

b. [yà:jî: î: kárⁿi-mi] àwà-rî-Ø
 [marriage 1PIS do-Impf] accept-PerfNeg-3SgS
 'He did not agree that we do the marriage.'

17.3.5 'Cease' (dògó-)

dògó- means 'cease, stop (an activity)'. In many contexts it implies an abrupt or definitive abandonment of the activity, as opposed to 'finish, complete'. It takes a verbal-noun complement, without a possessor (xx1.a). In the perfective positive, the preferred inflection is Perfective-1a -ti- rather than Perfective-2-só-. This construction gets some competition from a monoclausal transitive construction with a cognate nominal of a verb (xx1.b).

- (xx1) a. [nàmá kúwó-ndé] dògó-tĭ-∅ [meat eat-VblN] cease-Perf-3SgS 'He/She has ceased to eat meat.'
 - b. nùŋá dò-dógó-m-∅ song Rdp-cease-Impf-1SgS 'I will stop singing.'

17.3.6 'Want' (jòró)

jòró- 'want' also takes verbal noun complements. In (xx1), the lower-clause subject is coindexed with the higher-clause subject.

- (xx1) a. nné-ndè jóró-m-∅ go-VblN want-Impf-1SgS 'I want to go.'
 - b. [nné-ή yî:-ndé] jóró-ŋò-y∴
 [3Sg-Acc see-VblN] want-Impf.Neg-1PlS
 'We don't even want to see him/her.'

In (xx2), the two clauses have different subjects, so the lower-clause subject is expressed as a possessor of the verbal noun.

(xx2) [ú bâ:] [ŋgá yé:-ndé ú gô]

```
[2SgP father] [here come-VblN 2Sg Poss.InanSg] jɔrɔ́-ŋɔ̄:-Ø want-Impf.Neg-3SgS 'Your-Sg father doesn't want you-Sg to come here.'
```

17.3.7 'Forget' (irέ)

A partial paradigm of this verb is (xx1). It is transitive, taking an Accusative object, but it uses the Perfective-1a.

(xx1) 'Forget'

Perfective-1a řré-èrè-Perfective Negative řrè-ri-Imperfective Negative řrè-ŋò:-Imperative îrà Imperative Negative řrì-ndá

In the Perfective positive, this verb is commonly chained with a following unsuffixed Perfective besi- (xx2.a,c). The verb besi- elsewhere means 'bury' or 'set down (for storage)', implying that the item in question will be not be removed from its storage place any time soon.

The 'forget' verb takes **verbal noun complements** when the lower-clause subject is coindexed to the higher-clause subject and the complement denotes an action that was intended to be carried out.

- (xx2) a. [[bidò:n gú] témbiri-ndé]
 [[jug.L Def.AnSg] moisten-VblN]

 řré bèsì-ỳ
 forget bury.Perf.L-1SgS
 'I forgot to moisten the jug.'
 - b. [[sá:gù ú gô] jé:-ndé] irí-ndá [[sack 2Sg Poss.InanSg] bring-VblN] forget-Imprt.Neg 'Don't forget to bring your sack.'
 - c. yé:-ndé řré bèsì-Ø
 come-VblN forget bury.Perf.L-3SgS
 'He/She forgot to come.'

When the complement denotes a fact that the agent has forgotten, as opposed to an intended action that slipped his or her mind, 'forget' takes a different type of complement. In one construction, the complement is a headless relative clause ending in Inanimate Singular Definite gú, which forces tone-dropping on the verbal participle (xx2.a). In another, the complement-clause verb has an AN suffix but no pronominal-subject marking, as in relatives, but ends in a postnominal possessor expressing the subject (xx2.b-c). This pattern is possible even when the subject of the complement clause is coindexed with the possessor (xx2.d). A third construction is translatable 'forget whether P' rather than 'forget that P', and the complement contains polar interrogative /ma/ (xx2.e).

- (xx2) a. [ńné sèllè-ri gú]
 [3SgS be.healthy-Ppl.PerfNeg.L Def.InanSg]
 iré bèsi-ỳ
 forget bury.Perf.L1SgS
 'I forgot that he/she was sick.'
 - b. [sèllè-rî nò] iré bèsi-ỳ
 [be.healthy-PerfNeg 3SgPoss] forget bury.Perf.L1SgS
 [= (a)]
 - c. [bû: sɛllà-ndú bû: gɔ]
 [3PlS be.healthy-PerfNeg.3PlS 3PlP Poss.InanSg
 irɛ́ bɛ̃si-ŷ
 forget bury.Perf.L1SgS
 'I forgot that they were sick.'
 - d. [sèllè-rî kɔ̃:] řiré bèsì-ỳ
 [be.healthy-PerfNeg 1SgP.Poss.InanSg] forget bury.Perf.L-1SgS
 'I forgot that I am sick.'
 - e. [ú yě:-só-w mã:] ĭrɛ́ bɛ̀sĭ-ỳ
 [2SgS come-Perf2-2SgS Q] forget bury.Perf.L-1SgS
 'I forget whether (= that) you-Sg came.'

'Remember' is illî-rî- (an irregular reversive of 'forget'), or an unrelated verb stem nángírî-. The more general verb mă:ndî- 'think' can also be used in the sense 'remember' (cf. English *think of*). Yet another expression for 'remember' is iré bèllî-, where iré- 'forget' is chained to what appears to be an irregular reversive of bèsî-.

17.3.8 Obligational (wá:jîbì 'duty')

The noun wá:jîbì 'duty, obligation, ultimately from Arabic and familiar in languages of the zone, indicates that the agent in question has a duty to carry out the activity. The free translation is of the type 'X must VP'. In Nanga, the complement is expressed by a verbal noun clause, the agent appearing as a postnominal possessor. The NP headed by the verbal noun can be taken as the subject (or topic) of wá:jîbì=ŋ 'it is a duty' (xx1.a). Alternatively, wá:jîbì can take the Purposive postposition to form an adverbial phrase within a larger indicative clause. This construction is useful when the event in question has already transpired (xx1.b).

```
(xx1) a. [séwá:ri=yè ńné-ndé kɔ̃:] wá:jibi=j

[S=in go-VblN 1SgP.Poss.InanSg] duty=it.is

'I must go to Sevare.' ("My going to Sevare is a duty.")
```

```
b. [o: go] [wá:jîbi dèr<sup>n</sup>i] nnè-y<sup>n</sup>
[field.L in] [duty for] go.Perf.L-1SgS
'I had to go to the field.' ("I went to the field out of a duty.")
```

17.3.9 'Be afraid to' (ú:-yέ-)

When the complement denotes a hypothetical act whose subject is coindexed with the subject of 'fear' ('be afraid to ...'), we get a **verbal noun** complement.

```
(xxx) a. ŋgá yé:-ndé ú:-y-èrè-∅
here come-VblN be.afraid-MP-Perf1a-3SgS
'He/She was afraid to come here.'
```

When the complement denotes any other type of eventuality, we get an **Imperfective relative complement**, with a pronominal-subject (if any) expressed as an independent pronoun.

```
(xx1) a. [nji-nj bû: súyó-mi] ú-ùwà-y

[1Sg-Acc 3PlS hit-Impf] Rdp-fear.Stat-1SgS

'I'm afraid he/she will hit me.'
```

```
b. [\inf - \inf ] i: \sup - \min ] ú-ùwà-\emptyset [3Sg-Acc 1SgS hit-Impf] Rdp-fear.Stat-3SgS 'He<sub>x</sub>'s afraid I will hit \lim_{x/y}.'
```

```
17.3.10 'Begin' (tóró-)
```

The 'begin' verb is tóró-. It is paired with its antonym dimé- 'finish, end' in (xx1), which has no overt complement clause for either.

```
(xx1) wàgàti àrnáná tóró-m-w,
time.L where? begin-Impf-2SgS
wàgàti àrnáná dimé-m-w
time.L where? finish-Impf-2SgS
'What time do you-Sg begin, (and) what time do you end?'
```

The complement clause can end in a bare verb stem that is **chained** with the 'begin' verb (xx2).

```
(xx2) [nă: kố:] tốró-số-∅
[meal eat.meal] begin-Perf2-3Sgs
'He/She began to eat (the meal).'
```

However, the bulk of my examples involve a **verbal-noun or other nominal complement**. If there is a cognate nominal associated with a verb, the cognate nominal by itself is sufficient as complement of 'begin' (xx2.a). A compound including an incorporated object can also serve as the complement (xx2.b). The verbal noun in -ndɛ́ can be used with any verb (xx2.c-e).

- (xx2) a. kòyô tóró-ti-∅ weeping **begin**-Perf-3SgS 'He/She began to weep.' (cognate nominal, cf. kòyô kóyó-)
 - b. ńné pèrgè-sémê tóró-só-∅
 3SgS sheep.L-slaughter begin-Perf2-3SgS
 'He/She began to slaughter the sheep.'
 - c. [nă: kó:-ndέ] tóró-tì-∅
 [meal eat-VblN] begin-Perf1b-3SgS
 'He/She began to eat (the meal).'
 - d. [nùŋá núŋú-ndé] tɔ́rɔ́-ti-∅
 [song sing-VblN] begin-Perf1b-3SgS
 'He/She began to sing (a song).'
 - e. [ńné pèrgé sémé-ndé] tóró-só-∅

[3SgS sheep slaughter-VblN] **begin-**Perf2-3SgS 'He/She began to slaughter the sheep.'

17.3.11 'Finish' (dimέ-)

The complements are parallel to those of 'begin' (just above). Verbal-noun complements are exemplified in (xx1).

- (xx1) a. [nă: k5:-ndɛ́] dîmɛ́-nɔ̂:-∅ [meal eat-VblN] finish-Impf.Neg-3SgS 'He/She doesn't stop (= keeps on) eating.'
 - b. [bû: wárá-ndé] dǐmé-èr-à
 [3PlS do.farm.work-VblN] finish-Perfla-3PlS
 'They have finished farming (=weeding).'
 - c. [tòŋó tóŋ-ndɛ́] dǐmè-rʰí-ý
 [writing write-VblN] finish-Perf.Neg-1SgS
 'I havent' (yet) finished writing.'

It is also possible to express similar concepts using a possessed verbal noun or other nominal that functions as the subject of 'finish'.

```
(xx2) [wórî bû: gò] dimé-èrè-∅

[farming 3Pl Poss.InanSg] finish-Perf-3SgS

'Their farming work is finished.' (= 'They have finished farming.')
```

Recent Perfect $j\grave{\epsilon}$ - ($\S10.xxx$) is sometimes used in senses approaching 'finish VP-ing', as in $k\acute{\delta}$: $j\grave{\epsilon}$ -y., which can mean 'we have (already) eaten' or 'we have finished eating'.

17.4 Locative verbal noun or other nominal complement

17.4.1 'Help' (bă:rî-)

This verb normally takes nominal complements with final **Locative** postposition /gá/ or variant 'in' (§8.2.3), added to a verbal noun (which allows complements), or to some other nominal. Verbal-noun examples are in (xx1).

[sit-MP-VblN in] 1Sg-Acc help-Perf2-3SgS 'He/She helped me to sit down.'

- b. á:mádù [ú-ŋ jóŋ-ndɛ gá] njî-ŋ bă:rî-só-∅
 Amadou [2SgO treat-VblN in] 1SgO help-Perf2-3SgS
 'Amadou helped me to treat you-Sg (medically)'
- c. [nàŋá pági-ndé gá] nìji-ŋ bă:ri-só-∅
 [cow tie-VblN in] 1SgO help-Perf2-3SgS
 'He/She helped me to tie up the cow.'

In (xx2), the nominal takes the form of a compound with incorporated object ($\S5.1.3$).

- (xx2) a. [lɛ̃:tẽrẽ-tóŋô ŋð] ú-ŋ băːrá-m̀-∅ [letter.L-write.Nom.HL in] 2Sg-Acc help-Impf-1PlS 'I will help you-Sg to write the letter.'
 - b. [pèrgè-sémê ŋà] njî-ý bă:ră
 [sheep.L-slaughter in] 1Sg-Acc help.Imprt
 'Help-2Sg me to slaughter the sheep!'
 - c. [nàŋà-págâ gá] ǹjî-ŋ́ bǎ:rî-só-Ø [cow.L-tie in] 1Sg-Acc help-Perf2-3SgS 'He/She helped me to tie up the cow.'

A noun wórî, elsewhere used mainly as cognate nominal for wará- 'do farm work', is the complement in (xx3).

(xx3) á:mádù [wór gó] ú-ý bǎ:r-só-∅ Amadou [farming in] 2Sg-Acc help-Perf2-3SgS 'Amadou helped you-Sg do the farming.'

17.5 Chained-verb complement clause

These are simply special cases of direct verb chains (§15.1), but with a fixed final verb.

For the occasional chain construction with 'begin' and 'help', see §17.3.10 and §17.xxx, respectively. However, the most consistent use of this construction is with the 'can (be able to)' verb.

17.5.1 'Be able to, can' (beré-)

The verb beré- 'get, obtain' is also used with VP complement in the sense 'can, be able to'. The VP complement ends in a bare verb stem. The lower-clause subject must be coindexed to that of the higher-clause, and the single audible occurrence is attributable to the higher clause, although it appears to the left of the first VP.

- (xx1) a. [bîrá bǐré] béré-ŋɔ̂:-∅ [work(noun) work] can-Impf.Neg-3SgS 'He/She cannot work.'
 - b. ndé béré-m-w mà
 go.up can-Impf-2SgS Q
 'Can you-Sg go up?'
 - c. [\(\xi\):\(\eta\)i y\(\xi\): \\ \text{b\(\xi\)t\(\xi\)-y\(\ni\)} \\ \text{can-Impf.Neg-1SgS} \\ \(\text{'I can't come tomorrow.'}\)
 - d. [kúrⁿô îriyé-mî] béré-m-^w mà [stone rise-Caus] **can**-Impf.2SgS Q 'Can you-Sg lift the stone?'
 - e. [njî-rj bă:r] bɛrɛ́-m-w mà
 [1SgO help] can-Impf.2SgS Q
 'Can you-Sg help me?' (bă:ri)

A relative-clause example is (xx2), showing that a subject pronoun in a non-subject relative has its usual position just before the final verb in the chain (i.e. the participle).

(xxx) ùsù yě: ú béré-mì day.L come 2SgS can-Ppl.Impf 'the day you-Sg can come'

17.6 Purposive, causal, and locative clauses

17.6.1 Purposive clauses with postposition der"i 'for'

17.6.1.1 Positive purposive clause with Imperfective -m and der 17.6.1.1

In the basic purposive-clause type, the Purposive postposition derni 'for' is added to a clause whose verb is in pronominally-uninflected Imperfective form, e.g. ending in invariant -m. In the frequent case where the two clauses have the same subject, the subject is not overtly expressed in the purposive clause. Therefore the purposive clause [[ndô pó:-m] derni] 'in order to replaster the house' in (xx1) remains unchanged when the pronominal category of the subject (expressed on 'go up' at the end) is changed to 3Sg, 1Pl, or whatever.

(xx1) [[ńdô pó:-m] dèrⁿi] [[bàrkô: gò] ńdé-m-è] [house replaster-Impf] for] [[barrel in] go.up-Impf-2PIS] 'They will go up (and stand) on the barrel in order to replaster the house (= ceiling).'

Another example of this construction is (xx2).

```
(xx2) [[ámbéri bày] dàmá dámá-m] dèr<sup>n</sup>i]
[[chief Dat] speaking speak-Impf] for]
yè-ỳ∴
come.Perf.L-1PIS
'We have come in order to speak with the chief.'
```

When the main clause and the purposive clause have distinct subjects, the purposive clause expresses a pronominal subject as an independent pronoun, leaving the Imperfective verb unconjugated.

```
b. [[á:ndê≡Ø
                                                           dèr"i]
                                            ńnî-m]
                                î:
    [[Anda≡in
                   manner.L
                               1PIS
                                            go-Impf]
                                                           for]
    [lègèsô:
                                             î:-'n
                                                        ńdî-só-∅
                            gŝ]
                  ReflSgP Poss.InanSg]
                                             1PlO
                                                        give-Perf2-3SgS
    [bicycle
    'He<sub>x</sub> gave us his<sub>x</sub> bicycle, so that we (could) go to Anda.
```

A low-toned dàyⁿ (cf. noun dăyⁿ 'limit, boundary' or 'manner') occurs in (xx3.b). This form occurs optionally in any purposive clause with dèrⁿî, whether or not the subjects of the two clauses are identical. This suggests that the purposive construction can be analysed syntactically as a PP with a relative clause as complement, e.g. 'for [(a) limit that ...]', cf. English toward the end that ... For -mì ~ -m in Imperfective relative-clause participles, see §14.1.7.2. However, in Nanga, dàyⁿ occurs in only a minority of my elicited purposive clause examples.

17.6.1.2 Negative clause with Imperative Negative -rá and dèrⁿí

Purposive dèrⁿi may follow a prohibitive (Negative Imperative) clause to produce a negative purposive clause. The verb has -rá suffix, which (in this construction) is used for plural as well as singular subject. In other words, Negative Imperative Plural -rá-ndi is not used in such clauses.

In (xx1.a-b), the subjects of the two clauses are distinct. In (xx1.b), note suffix -rá even with (animate) plural subject.

```
(xx1) a. dèwî ké:ndé-m-ĭy
roof fix-Impf-1PIS

[[gã:n yègĭ-rá] dèrnî]

[[beam fall-ImprtNeg] for]

'We'll fix the roof, so the roof beam(s) won't fall.'
```

```
b. [ndo-no:
                                         kέsέ
                                                   dògò-y∴,
                         gú
                       Def.InanSg]
    [house.L-mouth.L
                                         cut
                                                   leave.Perf.L-1PIS,
    [[pèrgè
                 bû:]
                                [màngòrò
                                              gú]
                 Def.AnPl]
                                [mango.L
    [[sheep.L
                                              Def.InanSg]
    kó:-rá
                       der<sup>n</sup>i]
    eat-Imprt.Neg
                       for]
    'We (have) blocked the doorway, so that the sheep-Pl will not eat
    the mango(s).'
```

A same-subject example is (xx2).

```
(xx2) [nă: áyá-m-iy] [[[ósú gó] túy-rá] dèr<sup>n</sup>i] [meal take-Impf-1PIS] [[[road in] die-ImprNeg] for] 'We'll take food (along), so as not to die (= starve) on the way.'
```

17.6.1.3 Negative clause with Imperfective Negative -ŋɔ: and der i

Another negative purposive clause type has uninflected Imperfective Negative suffix $-\eta\delta$: on the verb. If the subject is pronominal, it is expressed as a preverbal pronoun. This construction is as an alternative to the type with Imperative Negative $-r\acute{a}$, discussed just above. In (xx1.a), the subjects of the two clauses are disjoint; in (xx1.b), the subjects are coindexed.

```
(xx1) a. [[gǐrè-nîyʰɛ̂ dàyʰ ú nîyɛ́-ŋð:] dèrʰi]
[[sleeping manner.L 2SgS sleep-Impf.Neg] for]
jĩnjâ jĩnjî-m̀-Ø
noise make.noise-Impf-1SgS
'I will make noise, so that you-Sg do not sleep.'
```

```
b. [chaise
                             ná:-m̀-∅,
                  gàl
                             spend.night-Impf-1SgS
    [chair
                  inl
                        d\grave{a} y^n
    [[girè-niy<sup>n</sup>ê
                                      ĭ:n
                                                niyέ-ηὸ:]
                                                                       dèr<sup>n</sup>i]
                                      1SgS
                                              sleep-Impf.Neg]
    [[sleeping
                        manner.L
                                                                       for]
    'I will spend the night in a chair, so as not to sleep.'
```

17.6.2 Purposive clauses with Imperfective participle, without derⁿi

A clause with Imperfective -mi may also be used in contexts allowing a purposive construal. However, in cases like (xx1) it is difficult to teast apart the purposive element ('sit in order to eat') from the mere temporal simultaneity of 'sit' and 'eat', in the absence of an explicit Purposive morpheme.

```
(xx1) [nă: kó:-mì] èw-yè-yè
[meal eat-Impf] sit-MP.Stat-3PlS
'They are sitting in order to eat.'
```

17.6.3 Purposive clause with with gó 'in' after motion verb

In the purposive clause-type favored with main-clause motion verbs (especially 'go' and 'come'), with positive purposive clause, and with subjects coindexed ('X went in order to VP'), the verb of the purposive clause occurs in its bare stem, but is tone-dropped, and is followed by go. The latter looks like one of the allomorphs of the Locative postposition, but in this construction its form is invariant (no nasalization of g to g after nasal syllable, no vocalic assimilation).

- - b. [gö: giyè gó] nnè-y
 [fire kill.L Purp] go.Perf.L-1SgS
 'I went in order to put out the fire.'
 - c. [nî: nò: gó] y-ò:
 [water drink.L Purp] come.Perf.L-3PlS
 'They came in order to drink the water.'

This construction has no negative counterpart.

17.6.4 Causal ('because') clause (ságù)

ságù 'because' follows the 'because' clause, which has the form of a normal main clause.

```
(xx1)
       ósî
                 ὲsì≡ndó-Ø
                                          ságù,
                 good≡StatNeg-3SgS
                                          because,
        road
        îsè
                     gó]
                               ńné
                                         béré-ŋò-y∴
                     in
                                         can-ImpfNeg-1PIS
        [village.L
                               go
        'We can't go to the village because the road isn't good.'
```

17.6.5 'Because of' (der"i)

The Purposive postposition $\frac{der^n}{1}$ can be glossed 'because of X', with NP complement X, in examples like (xx1).

```
(xx1) [ndo go] [bondi derni] nù-yn.:

[house in] [rain for] go.in.Perf.L-1PIS

'We went into the house because of the rain.'
```

17.7 Clause-final nà: 'though'

write

18 Anaphora

The most important anaphoric morphemes are Singular á and Plural â:. They can be third-person reflexive (generally coindexed to a clausemate subject), or logophoric (coindexed to the author of a quoted utterance or thought). Their functions are covered in this chapter, along with Reciprocals and certain emphatic pronoun.

18.1 Reflexive

18.1.1 Reflexive object (á-ή, Plural â:-ὴ)

If the subject is first or second person, the regular object form of the pronominal is used for a coindexed direct object, with no explicit reflexive marking (xx1). As always, the Accusative suffix - $\hat{\eta}$ may or may not be audible. For 1Sg, the accusative form has a special stem shape, so it is always recognizable even when the - $\hat{\eta}$ is not audible.

```
(xx1) a. njî-ń késé-só-ý
1Sg-Acc cut-Perf2-1SgS
'I cut myself.'
b. î:-n) késé-só-ý. 'We cut ourselves.'
c. ú-ń késé-só-w 'You-Sg cut yourself.'
d. û:-n) késé-só-w.: 'You-Pl cut yourselves.'
```

If the subject is anything other than first or second person (NP, third person pronominal), the coindexed direct object is the accusative form of Reflexive Singular á or Reflexive Plural â:, depending on grammatical number. In (xx2.d), the Reflexive pronoun could also be interpreted as Logophoric, since both coindexation relationships are present. I gloss á and â: as "ReflSg" and "ReflPl," respectively, in interlinears, but it should be kept in mind that they are specifically third person forms.

```
(xx2) a. á-ý késé-só-Ø

ReflSg-Acc cut-Perf2-3SgS

'He cut himself.' or 'She cut herself.'
```

- b. â:-ŋ késé-s-é ReflPl-Acc cut-Perf2-3PlS 'They cut themselves.'
- c. á:mádù á-ý késé-só-Ø A **ReflSg-Acc** cut-Perf2-3SgS 'Amadou cut himself.'
- d. á:mádù ńné kíyésêwndè [á-ń késé-só-∅] wà A 3SgS say [**ReflSg-Acc** cut-Perf2-3SgS] say 'Amadou said that he cut himself.'

18.1.2 Reflexive PP complement (á, â:)

The examples in (xx1) involve dative complements that are coindexed to the clausemate subject. With a first or second person pronominal, like 1Sg in (xx1.a), the regular dative form is used. When a third person pronominal is coindexed to the subject, the regular 3Sg ńn $\hat{\epsilon}$ is replaced by Reflexive Singular \hat{a} (xx1.b), and the regular 3Pl b \hat{u} : is replaced by Reflexive Plural \hat{a} : (xx1.c).

- (xx1) a. kě:rè bàrⁿi tîy-só-ý money 1Sg.Dat send-Perf2-1SgS 'I sent the money to myself.'
 - b. á:mádù kĕ:rè [á báy] tîy-só-∅
 Amadou money [ReflSg Dat] send-Perf2-3SgS
 'Amadou sent the money to himself.'
 - c. kě:rè [â: bày] tîy-s-é money [**ReflPl** Dat] send-Perf2-3PlS 'They sent the money to themselves.'

The same third person reflexive forms occur with other postpositions (adpositions), like 'under' in (xx2).

(xxx) á:mádù [dòsî á gô] sùŋgó bèrè-∅ Amadou [under **ReflSg** Poss.Inan] boubou get.Perf.L-3SgS 'Amadou found (a/the) boubou under himself.'

18.1.3 Reflexive possessor (á, â:)

When the possessor of a direct object or other non-subject NP is coindexed to the clausemate subject, if the possessor is pronominal we get the same patterns seen for accusatives. For first or second person, the regular possessor forms are used (xx1). As a reminder, some pronominal possessors precede the possessed noun if it is a kin term ($\S6.2.2$), otherwise the pronominal possessor combines with a possessive classifier and follows the noun ($\S6.2.1$).

```
(xx1) a. [nèr<sup>n</sup>i yè:] yǐ:-só-ý [dog 1SgP.Poss.AnSg] see-Perf2-1SgS 'I saw my dog.'
```

```
b. [ú bâ:] yǐ:-só-ẃ
[2SgP father.HL] see-Perf2-2SgS
'You-Sg saw your father.'
```

For third person possessor, if the possessor is coindexed to the clausemate subject, we get Reflexive pronouns (á, â:) as possessors. The positioning of the reflexive-possessor pronoun is the same as for those other pronominals (1Pl, 2Sg, 2Pl, and 3Pl) that precede kin terms but follow other nouns.

- (xx2) a. [nèr^mi á yê] yĭ:-só-∅ [dog ReflP Poss.AnSg] see-Perf2-3SgS 'She saw her (own) dog.'
 - c. [nèr"î â: yè] yǐ:-s-e [dog ReflP Poss.AnPl] see-Perf2-3PlS 'They saw their (own) dog.'
 - d. [ndô á gô] yĭ:-só-∅ [house ReflP Poss.InanSg] see-Perf2-3SgS 'He saw his (own) house.'
 - d. [ńdô â: yɛ] yǐ:-s-ɛ́ [house ReflP Poss.InanPl] see-Perf2-3PlS 'They saw their (own) houses.'
 - e. [á bâ:] yǐ:-só-Ø [ReflSgP father.HL] see-Perf2-3SgS 'She saw her (own) father.'

```
f. [â: bâ:] yǐ:-s-ɛ́
[ReflPIP father.HL] see-Perf2-3PIS
'They saw their (own) father(s).'
```

18.1.4 Emphatic pronouns

18.1.4.1 With màr 's elf'

Emphatics with adverbial màrⁿá following an independent pronoun are in (xx1.a-b). When the referent is spelled out by a name or other nonpronominal NP, this is topicalized and is followed by a pronoun with màrⁿá (xxx.c). The specific type of emphasis here is **exclusionary**. Where it might have been expected that X will perform the activity with the help of others, he/she does it without help (for this sense, see also the construction with tùmâ, discussed below). Or, where it might have been expected that X would have someone else perform the activity, he/she does it in person.

- (xx1) a. [yî: á yê] tiyè-rî-∅, [child ReflSg Poss.Qn] send-Perf.Neg-3SgS, [ńné màrⁿá] yè:-∅ [3SgS self] come.Perf.L-3SgS 'He didn't send his son, (rather) he came himself.
 - b. [i: màrⁿá] wárá-m-iy [1Pl self] farm-Impf-1PlS 'We will do the farming ourselves.'
 - c. hàmidû [ńné màrⁿá] ǹnè-∅ Hamidou [ReflSg self] go.Perf.L-3SgS 'Hamidou went himself (in person).'

Most examples involve subjects, but this is not a syntactic requirement. In (xx2), the relevant pronoun is an accusative functioning as direct object.

```
(xx2) [yî: yɛ̃:] nà:rna-rnī-Ø,

[child 1SgPoss.Anim] call-Perf.Neg-3SgS

[njî-ŋ marna] nà:rni-Ø

[1Sg-Acc self] call.Perf.L-3SgS

'She didn't call my son, she called me myself (i.e. directly).'
```

18.1.4.2 With tùmâ (tùmáyⁿ) 'one/alone'

The stem tùmâ 'one', which elsewhere patterns as a modifying adjective, figures in expressions of the type 'X alone', 'X by him/herself' (i.e. without accompaniment or assistance). In one pattern, tùmâ 'one' (hence 'single', 'alone') follows an independent pronoun. Here, tùmâ is optionally followed by sày 'only', though this is somewhat redundant.

In this construction, tuma has a specifically adverbial variant form $tumay^n$ that is not used as a numeral (xx1.b).

```
    b. [i:n tùmáyn] [jinjà gú]
    [1Sg alone] [water.jar.L Def.InanSg]
    äy iriyé-m béré-ŋò-yn
    take rise-Caus can-Impf.Neg-1SgS
    'I can't lift the water jar by myself.'
```

If the subject is nonpronominal, it is treated as a topicalized NP and resumed by a third person Reflexive pronoun (xx2).

```
(xx2) a. [yì-tègè
                         bû:1
                                               tùmâ]
                                      Γâ:
                                               alone]
                        Def.AnPl]
                                     [ReflPl
            [child-Pl.L
            [jinjà
                         gú]
                                                  béré-n-è:
                                       ăу
            [water.jar.L Def.InanSg]
                                       take
                                                  can-Impf.Neg-3PIS
            'The children can't pick up the water jar by themselves (=without
           help).'
```

```
    b. [bă: yɛ̃:] [á tùmáy¹] wàrá
    [father 1SgP] [ReflSg alone] do.farm.work
    béré-ŋɔ̃:-Ø
    can-Impf.Neg-3SgS
    'My father cannot do the farming by himself.'
```

18.1.4.3 With kû: 'head'

Possessed forms of 'head' are used as emphatic alternatives to ordinary pronominal possessors (including third person reflexive possessors) when

coindexed to the clausemate subject. In (xx1.a-b), the possessed noun is a cognate nominal associated with the verb.

- (xx1) a. [[kû: î: gò] bìrà] bîré-m-iy [[head 1PIP Poss.InanSg] work(noun).L] work-Impf-1PIS 'We work for ourselves.' (lit.: "we do the work of our head")
 - b. [[kû: â: gò] bìrà] biré-m-è [[head ReflPlP Poss.InanSg] work(noun).L] work-Impf-3PlS 'They work for themselves.'
 - c. [[kû: kɔ̃:] birà] biré-m̀- \varnothing [[head 1SgP.Poss.InanSg] work(noun).L] work-Impf-1SgS 'I work for myself.'

18.2 Logophoric and indexing pronouns

18.2.1 True third person logophoric function

In a quotation (of speech or thought) attributed to one or more third persons (i.e. not the current speaker or addressee), any occurrence of a pronoun coindexed to the attributed speaker(s) takes logophoric form. The forms are singular á and plural â:, which we have already seen in third-person reflexive function (§18.xxx). In effect, á represents an embedded 'I', and â: an embedded 'we'.

In non-subject functions, á and â: have the same linear positions as other pronouns. For example, they have regular accusative and dative forms (accusative á-ý and â:-ỳ, dative á báy and â: bày, respectively). They pattern with pronouns rather than with nouns as **possessors**, specifically by combining with possessive classifiers and following possessed nouns (except kin terms), whereas nonpronominal NPs precede the possessed noun (xx1).

(xx1)	a.		ńdô ńdô	ú û:	gô gò	'your-Sg house' 'your-Pl house'
	b.		ńdô ńdô	á â:	gô gò	'his-Logo house' 'their-Logo house'
	c.	á:mádù yă:	ndò ndò			'Amadou's house' '(a/the) house of women'

However, logophorics pattern like nouns in **subject** function, preceding the verb and requiring the verb (in a simple main clause) to agree with them, i.e. á requires a 3Sg suffix while â: requires a 3Pl suffix on the verb.

```
nèr<sup>n</sup>i
                                        súvó-só-w
                                                            'you-Sg hit (a/the) dog.'
(xx2)
         a.
                           nèrni
                                        súy5-só-ẃ∴
                                                            'you-Pl hit (a/the) dog.'
          b.
                           nèr<sup>n</sup>î
                                        súvó-só-Ø
                                                            'He-Logo hit (a/the) dog.'
               â:
                           nèr<sup>n</sup>î
                                        súy5-s-έ
                                                            'They-Logo hit (a/the) dog.'
                           nèr<sup>n</sup>î
                                                            'Amadou hit (a/the) dog.'
              á:mádù
                                        súyó-só-Ø
               yă:
                           nèr<sup>n</sup>î
                                        súy5-s-έ
                                                            '(The) women hit (a/the) dog.'
```

In subject function in non-subject relative clauses, logophorics do not behave like nonpronominal NPs. The latter precede all verbs in a direct chain, frequently with a resumptive third person subject pronoun immediately preceding the final verb (i.e. the participle); see §14.1.8. Rather, logophorics take the same position as do other subject pronouns in relatives, i.e. immediately preceding the final participle. (xx3.a-c) are non-relative clauses, given for comparison. (xx4.a-c) are versions of the same as in adverbial (hence non-subject) relatives. Observe that 3Sg ńné, either as simple pronominal subject (xx4.a) or as resumptive (xx4.b), and Logophoric pronouns, occur immediately before sigé- 'go down', following the chained verb tómbó 'jump'. Therefore Logophoric á has a different position relative to the nonfinal chained verb in (xx3.c) and (xx4.c).

- (xx3) a. tómbó sigé-só-∅ jump go.down-Perf2-3SgS 'He/She jumped down.'
 - b. $[ar^na]$ $n\epsilon]$ tómbó sigé-só- \varnothing [man.L Def.AnSg] jump go.down-Perf2-3Sgs 'The man jumped down.'
 - c. á tómbó sigé-só- \varnothing wà LogoSg jump go-down-Perf2-3Sgs say 'He_x said that he_x jumped down.'
- (xx4) a. ùsù tómbó ńné sìgè-sè gú day.L jump **3SgS** go.down.Ppl.Perf.L Def.InanSg 'the day he/she jumped down'
 - $\begin{array}{ccc} b. & usu & [ar^na & n\epsilon] \\ & day.L & [man.L & Def.AnSg] \end{array}$

```
tómbó (ńné) sigè-sè gú
jump (3SgS) go.down.Ppl.Perf.L Def.InanSg
'the day the man jumped down'
```

c. [ùsù tómbó á sǐgè-sè gú] wà [day.L jump **LogoSgS** go.down-Ppl.Perf.L Def.InanSg say 'He_x said, the day he_x jumped down, ...'

So, overall, Logophoric pronouns have some pronominal features and some nonpronominal features, in those syntactic positions where pronouns and nonpronominal NPs behave differently.

A logophoric is not used when the attributed speaker is the current speaker or addressee.

b. yě: béré-ŋò-wⁿ wà come can-Impf.Neg-2SgS say 'You-Sg said that you-Sg can't come.'

If the pronominal is plural, and its reference includes the attributed speaker (along with at least other person), the logophoric plural category is applied. In other words, an embedded 'we' in a quotation attributed to a single speaker appears as logophoric plural.

```
(xx6) a. á:mádù
                          ńnέ
                                       kîyésêwndè
            Amadou
                          3Sg
                                       say
                                    yè]
            Γâ:
                         [móti
                                                ńnî-m-ε]
                                                                  wà
            [LogoPlS [Mopti
                                                go-Impf-3PIS]
                                    to]
                                                                  sav
            'Amadou<sub>x</sub> said that they<sub>xy</sub> (e.g. Amadou and Seydou) are going to
            Mopti.'
```

18.2.2 Non-logophoric topic-indexing function

The subject of a relative clause can be expressed as a third person Reflexive pronoun to coindex it to the subject of the main clause. In (xx1.a), the subject of both clauses is 1Sg. As usual in non-subject relative clauses, a pronominal subject takes independent-pronoun form. When 1Sg is replaced by a singular

third person subject, the relative clause has Reflexive Singular \acute{a} as subject (xx1.b). A plural third person subject requires Reflexive Plural $\^{a}$: (xx1.c).

```
(xx1) a. [dày<sup>n</sup> ĭ:<sup>n</sup> gór<sup>n</sup>ó-mĭ] kár<sup>n</sup>î-m̀-∅

[limit.L 1SgS be.able-Ppl.Impf] do-Impf-1SgS

'I will do as much as I can.'
```

```
b. á:mádù [dày<sup>n</sup> á gór<sup>n</sup>ó-mì]

Amadou [limit.L ReflSgS be.able-Ppl.Impf]

kárî-<sup>n</sup>

do-Impf-3SgS

'Amadou will do as much as he can.'
```

c. [nù bû:] [dàyⁿ â: góró-mǐ]
[person Def.AnPl] [limit.L ReflSg be.able-Ppl.Impf]
kárⁿí-m-è
do-Impf-3PlS
'The people will do what they can.'

Examples showing that the targeted relative-clause subject must be coindexed to the subject, not some other NP, in the main clause are in (xx2). The requirement is met in (xx2.a), but not in (xx2.b), which therefore has an ordinary 3Sg subject pronoun in the relative clause.

```
(xx2) a. [\tilde{l}i:g\tilde{i} á \tilde{j}iy\hat{\epsilon}-s\tilde{\epsilon}] \tilde{n}j\hat{i}-\tilde{\eta} \tilde{k}\hat{\epsilon}:r\hat{i}-s\hat{o}-\varnothing [bird.L LogoSgS kill-Ppl.Perf] 1Sg-Acc show-Perf2-3SgS 'He<sub>x</sub> showed me a bird that he<sub>x</sub> (had) killed.'
```

```
b. [nàma] nné kàr^ni-sè gú] [damage.L 3SgS do-Ppl.Perf.L Def.InanSg] nné-n ké:ri-só-y 3Sg-Acc show-Perf2-1SgS 'I showed him_x the damage that he_x (or: she_y) had made.'
```

18.3 Reciprocal

18.3.1 Simple reciprocals (tũ:)

The Reciprocal is used when a plural direct object or other nonsubject NP is coindexed in a distributive fashion with a plural clausemate subject. The Reciprocal form is invariant til for any pronominal category of subject. This

form differs (slightly) in tone from the noun tǔ: 'agemate'. It behaves like a noun, and may take a postposition or Accusative -ŋ.

```
(xx1) a. tũ: yǐ:-só-y∴
Recip see-Perf2-1PIS
'We saw each other.'
```

- b. [tǔ: yàŋà] jòrîyé-s-£ [Recip with] fight-Perf2-3PlS 'They fought each other.'
- c. tũ:-ỳ súyó-só-ý∴ Recip-Acc hit-Perf2-1PIS 'We hit each other.'

The Reciprocal pronoun is optionally preceded by a plural pronominal possessor. A preceding pronominal possessor is grammatically correct for possession of a kin term or similar relationship term (including tǔ: 'agemate'); see §6.2.2.1. The normal forms of such possessor pronouns are 1Pl î:, 2Pl û:, and (third person) Reflexive Pl â:, with {HL} tone. However, in combination with the Reciprocal the pronominal is H-toned. The Reciprocal morpheme is L-toned tù:, as it would have been as a possessed noun after a {HL}-toned possessor, and in this combination it does not allow the Accusative morpheme. Thus first person î: tù:, second person ú: tù:, third person á: tù:. This tonal quirk permits audible distinctions between reciprocals like (xx2.a) and simple combinations of a pronominal possessor with 'agemate', as in (xx2.b). With a pronominal possessor other than 1Pl, 2Pl, or Reflexive Plural, only the latter reading would be possible anyway, as in (xx2.c).

- (xx2) a. [i: tù:] yĭ:-só-ý∴ [1PlP Recip] see-Perf2-1PlS 'We saw each other.'
 - b. [î: tù:] yĭ:-só-ý∴
 [1PIP Recip/agemate.HL] see-Perf2-1PIS
 'We saw our agemate(s).'
 - c. [bû: tù:] yǐ:-s-£
 [3PIP Recip/agemate.HL] see-Perf2-3PIS
 'They_x saw their_y (another group's) agemate(s).'

18.3.2 'Together' (bendey)

This is expressed with a morpheme bendey preceded by a pronominal denoting a nonsingular set. It is used in intransitive and transitive clauses where the subjects acted in concert. Since the subject is coindexed with the 'together' pronominal, the Reflexive Plural form of the pronominal is required for third persons. The forms are irregular and are given in (xx1). One expects î: bendey, û: bendey, and â: bendey, with the usual long vowel and falling tone of these pronouns. Instead, the vowel is short and high-toned. Nevertheless, the low tone on bendey is consistent with an original falling tone on a preceding possessor. In other contexts, the shortening of the plural pronoun's vowel might lead to confusion with the corresponding singular (especially 2Sg ú, Reflexive Singular á), but since bendey is used only in the context of collective action, no misparsing is possible.

```
(xx1) category form

1Pl 1 bèndèy
2Pl ú bèndèy
Reflexive Plural á bèndèy
```

Examples are in (xx2).

```
(xx2) a. [i bèndèy] yè-ỳ∴
[1Pl together] come.Perf.L-1PIS
'We came together.'
```

```
b. [nù bû:] [á bèndèy]
[person.L Def.AnPl] [ReflPl together]
yû: wàr-à
millet farm.Perf.L-3PlS
'The people farmed (= raised) millet together.'
```

The construction can also be used when the antecedent is the direct object (xx3).

```
(xx3)
       [[úwâ
                    yò]
                               [gâ:n
                                        yò]]
        [[leaf.L
                    and]
                               onion
                                        and]]
        Γá
                       bèndèy]
                                       dàŋi-ỳ∴
        [ReflPl
                       together]
                                       cook.Perf.L-1PIS
        'We cooked leaves and onions together.'
```

18.4 Restrictions on reflexives

18.4.1 No antecedent-reflexive relation between coordinands

Parallel to 'I and my father' (xxx.a), we have the third person pattern 'Amadou_x and his_x father' in (xxx.b), where (in the primary reading intended) the possessor of the right coordinand is coindexed with the left coordinand. Note that (xxx.b) uses the ordinary 3Sg possessor form $n\delta$, which does not specifically coindex the possessor to a particular antecedent. Therefore (xxx.b) also has a reading involving a possessor for 'father' other than Amadou.

```
b. [\acute{a}:m\acute{a}d\grave{u} \quad y\grave{o}] \quad [b\check{a}: \quad n\grave{o} \quad y\grave{o}] [Amadou and] [father 3SgPoss and] 'Amadou<sub>x</sub> and his<sub>x</sub> father.' (or: 'Amadou<sub>x</sub> and her<sub>y</sub> father')
```

19 Grammatical pragmatics

19.1 **Topic**

```
19.1.1 Topic (gây, gâ)
```

This particle is especially common with pronouns, but may occur after other NPs and adverbials: [[ú dê:] gày] 'as for your mother', [[àrnà né] kây]. The final /y/ is often omitted in allegro speech but is common in careful styles. The particle has F-tone after a final high (including rising) tone on the preceding word, L-tone after a final L-tone (including F-tone).

The topicalized constituent may be presentential, in which case it may correspond to a resumptive pronoun that would not otherwise be necessary. Or it may be clause-internal. It is difficult to make the distinction with NPs that function as subject in the clause, since the "resumptive" pronoun in this case is just the obligatory pronominal-subject suffix on the verb.

Combinations with independent pronouns are in (xx1).

(xx1)	category	independent (e.g. subject)			
	1Sg 1Pl 2Sg 2Pl	ĭ: ⁿ gây î: gày ú gây û: gày			
	3Sg 3Pl InanSg InanPl LogoSg LogoPl	ńné gây bû: gày kú gây kû: gày á gây â: gày			

The Topic particle can also follow e.g. **accusative** nouns and pronouns, showing clearly that the NP containing the Topic particle can function as a clause-internal argument. This is easiest to hear with 1Sg object, which has a distinctive form.

```
[njî-ý gây] sùyò-rî-∅
[1SgO Topic] hit-Perf.Neg-3SgS
'He/She hit my friend, but me he/she didn't hit.'
```

19.1.2 'Now' (náyⁿ)

The temporal adverb 'now' is nîŋèyⁿ. However, a shorter form náyⁿ (always high-toned) is used, in close conjunction with a preceding NP (often a pronoun), as an alternative to the Topic particle gây. Thus 1Sg ĭ:ⁿ náyⁿ 'me now, ...', 1Pl î: náyⁿ 'us now, ...', and so forth. This form is common in narratives.

When the 'now' particle functions as a topical element by itself, it appears in the form náyⁿ gây, with variants ná gây and né gà.

```
19.1.3 'Also, even' (yana)
```

This particle can follow any NP or adverbial constituent. It must be distinguished from Instrumental yana (§8.1.2).

When added directly to a noun or pronoun, the 'also, even' particle gets its tones by spreading from the left, and this high or low tone extends across both syllables: i:n yáná 'me too', i: yànà 'us too', é:ní yáná 'tomorrow too'. It is therefore analysed as **atonal**. However, it is also low-toned after the Accusative suffix -n, even when the latter is high-toned due to spreading from its left (xx1).

```
(xx1) ńnś-ń ńdi-tì-w-ndè [nji-ń yana] ńdi
3Sg-Acc give-Perf1b-2SgS-if [1SgO too] give.Imprt
'If you give (some) to him/her, give (some) to me too!'
```

The particle may follow a PP, including one with the (partially) homophonous Instrumental postposition yana.

```
wárá-m-Ø,
(xx2)
       [nàná
                  yàŋà]
                              farm(verb)-Impf-1SgS,
       Cow
                  with]
       [[ògònònó
                        yànà]
                                      yàŋà]
                                                     wárá-m-Ø
       [[camel
                         with]
                                      too
                                                     farm(verb)-Impf-1SgS,
       'I do farm work with an ox, (and) I do farm work with a camel also.'
```

My assistant did not allow /yaŋa/ after verbs, e.g. with clausal scope. Since every verb has a natural complement, often a referentially unspecific cognate nominal, it is probably always possible to find a nominal to serve as the

immediate scope of /yaŋa/. In (xx3), note 'meal' and 'place' in the interlinears (omitted from the free English translation).

(xx3) nă: dánî-n, [î: yànà] sémbî-n meal cook-Impf.3SgS, [place also] sweep-Impf.3SgS 'He/She cooks, and he/she sweeps too.'

/yaŋa/ can also be used in the emphatic sense 'even X', which is closely related logically to 'also, too'.

- (xx4) a. [yì-tègè èwrè bû: yàŋà] wórî wárá-m-è [child-Pl.L small.L Def.AnPl even] farming farm-Impf-3PlS 'Even the little kids will do farm work (= weeding).'
 - b. [pŏ: yáŋá] pò:-mè-rⁿ1-∅ [greeting even] greet-Caus-Perf.Neg-3SgS 'He/She didn't even say hello.'

19.2 Presentential discourse markers

19.2.1 'As much as ...' (hálì ~ hálè)

This particle, which occurs in various forms in all languages of the zone, may occur at the beginning of a constituent phrase (NP or adverbial). In this section the focus is on discourse extensions of the usual sense 'as far as (location)' or 'up until (time)'. The clause itself may be negated, resulting in the sense 'not as much as' (hence 'not even').

- (xxx) a. [háli njí-í pŏ:] kiyè-ri-∅ [as.much.as 1Sg-Acc greeting] say-Perf.Neg-3SgS 'He/She didn't say as much as hello to me.'
 - b. [hálì bú:dù wŏy] njî-ń ndè-rì-∅
 [as.much.as riyal two] 1Sg-Acc give-Perf.Neg-3SgS
 'He/she didn't give me as much as a red cent.'
 - c. [háli nà tùmâ yàŋà] yè:-rí-∅ [as.much.as time one even] come-Perf.Neg-3SgS 'He/She didn't come as much as (=even) once.'
 - d. [háli nùná] nùni-só-Ø

```
[as.much.as song] sing-Perf2-3SgS 'He/She even sang (a song).'
```

```
19.2.2 'Well, ...' (háyà)
```

This is the common 'well, ...' expression, giving the speaker time to formulate a clause. This is a regional form also common in e.g. Fulfulde and Jamsay.

```
19.2.3 'So, ...' (wálà:)
```

French voilà appears in the form wálà:, also in regional use.

19.2.4 Clause-initial emphatic particle (péy, fés)

péy can be used with positive and negative clauses. pés is used with negative clauses only ('not at all').

- (xx1) a. péy séllé-só-y Emph be.healthy-Perf2-1SgS 'I'm perfectly healthy.'
 - b. péy nă: kà:-rǐ-y
 Emph meal eat-Perf.Neg-1SgS
 'I haven't eaten at all.'
 - c. pés nă: kò:-rì-y

19.2.5 'But ...' (mě:)

Younger speakers use mě: (French mais). Unlike the French original, mě: is prononced at the end of the preceding clause.

(xx1) yě:-só-
$$\varnothing$$
 mě: bè:-rî- \varnothing

come-Perf2-3SgS but remain-PerfNeg-3SgS 'He/She came but did not stay.'

19.2.6 'Indeed'

This particle, which occurs in slightly different forms in most local languages, is used in narrative at the beginning of a clause introducing a dramatic or surprising new event.

19.3 Pragmatic adverbials or equivalents

19.3.1 '(Not) again', 'on the other hand'

'Again', i.e. 'one more time' or 'a second time', is expressed by the adverb bendî (also the adjective 'other') or by the chained verb bindé 'go back, return'. bendî is preferred in negative or other modally problematic contexts (xx1.a-b), while bindé is usual in positive contexts (xx1.c).

- (xx1) a. bèndi òmbòri≡yê ńni-ŋò-yⁿ other Hombori≡in go-ImpfNeg-1SgS 'I won't go to Hombori again.'
 - b. bèndî ŋgá yè-w-ndè, há:jè béré-m-w other here come.Perf.L-2SgS-if, problem get-Impf-2SgS 'If you-Sg come here again, you'll get trouble.'
 - c. [nă: [á báy] kó:-ŋ́]
 [meal [ReflSg Dat] eat-and.SS]
 [bìndé ŋgá kòè-Ø]
 [go.back here eat.Perf.L-3SgS]
 'Having eaten at home, he ate here again.'

19.4 'Only' particles

19.4.1 'Only' (săy)

There is a particle săy (as in Jamsay) that follows the constituent (X), either an NP or an adverbial, that it has scope over. The sense is 'only X'.

```
(xxx) a. [tɛ́:mdɛ́rɛ̀ saˇy] sŏ-y
[hundred only] have-1SgS
'I have only one hundred (riyals).'
```

săy can indirectly have scope over a VP by being added to a cognate nominal (or other conventionalized object associated with the verb).

(xxx) a. bîrὲ-ŋò:-∅, [gừrè-nîyⁿὲ săy] nîyⁿε-ὴ work-Impf.Neg-3SgS, [eye-sleeping only] sleep-Impf.3SgS 'He/She doesn't work, he/she just sleeps.'

As is true in all languages of the zone, 'only X' is also expressed by a negated clause combined with an 'if not' (i.e. 'unless') conditional.

(xxx) búrà [ò:ndó ndò:-∅ ndè] kò bèndî mùrá-∅
B [honey not.be-3SgS if] thing.L other not.want-3SgS 'Boura wants nothing other than honey.'

```
19.4.2 'Just (one)' (léη)
```

This is a colorful intensifier for tùmá 'one'.

(xxx) a. [nàŋà tùmá léŋ] sò-y
[cow.L single mere] have-1SgS
'I have a single (=only one) cow.'

The stylistic flavor is along the lines of colloquial English 'I have one lousy cow', where 'lousy' has logical focus on 'one' (expressing the meager number of cows) rather than on 'cow'.

19.5 Phrase-final emphatics

19.5.1 Phrase-final já:ťi 'exactly'

This emphatic, common in Fulfulde and (through borrowing) some other languages of the zone, is often used as a single-word confirmation of a proposition uttered by an interlocutor ('Exactly!'). It may also be added to a phrase or clause with similar sense.

- (xxx) a. kòkòsú-wⁿ já:tì viper-be.InanS exactly 'Yes, indeed it is a viper.'
 - b. ú já:tì
 2Sg exactly
 'precisely you-Sg'

19.5.2 Clause-final kòy

This clause-final emphatic particle is used to give strong assent to a proposition by an interlocutor, or to a polar interrogative.

```
(xxx) kě:rè nŏm kòy
money difficult Emph
'Money is indeed hard to come by.' (nòmî)
```

This is a regional word (Jamsay, Fulfulde, Humburi Senni, etc.).

19.5.3 Clause-final dè

This clause-final emphatic particle is more adversative or admonitive than kòy, suggesting that the addressee is unaware of something. (xxx) might be said to someone who had just announced plans to mount an expensive project.

- (xx1) a. kě:rê nŏm dè money difficult Emph '(But) money is hard to come by!'
 - b. [gùrî báy] ú-ý háybà dè [thief Dat] 2SgS-Acc watch.over.Imprt Emph 'Watch out for the thief (or: thieves)!'

19.6 Backchannel and uptake checks

19.7 Greetings

Time-of-day related greetings ('good morning!' etc.) and their responses (R) are in (xx1). In form, the unmarked pŏ: is used in the middle of the day, while the greetings used early in the day and at night have a retrospective time perspective: the 'good morning!' expressions are based on ná:- 'spend night', while the 'good night' expressions are based on dèrné- 'spend mid-day'. By contrast, the late-afternoon greeting uses the term for 'late afternoon' (dèndèsî). The Imperative Plural suffix -ndî on the plural-addressee versions of the greetings suggests that all the greetings are imperative in form.

```
(xx1)
                                       situation
             greeting
             náv<sup>n</sup>
                                       morning 6-9 AM (Sg addressee)
             ná:-nđi
                                       (Pl addressee)
         R: àwâ:
         R: ná:-kò:
                                       (archaic)
             pŏ:
                                       mid-day 9AM to 4PM (Sg addressee)
             pŏ:-ndi
                                       (Pl addressee)
         R: àwâ:
         R: pŏ: bèrè-mì
                                       (archaic)
                                       late afternoon 4PM to sunset (Sg addressee)
             dèndèsî pŏ:
             dèndèsî pŏ:-ndì
                                       (Pl addressee)
         R: àwâ:
         R: dèndèsî pŏ: bèrè-mì
                                       (archaic)
             dèr<sup>n</sup>éy<sup>n</sup>
                                       sunset to 4 AM (Sg addressee)
             dèr<sup>n</sup>éy<sup>n</sup>-ndì
                                       (Pl addressee)
         R: àwâ:
         R: dér<sup>n</sup>î bèrè-mì
                                       (archaic)
```

The '(have a) good night!' expressions are in (xx2). They may be addressee-directed (imperative), or hortative and inclusive ('let's ...').

```
    (xx2) jáŋà nâ: final 'good night!' (Sg addressee)
    R: àmî:<sup>n</sup>
    (Pl addressee)
    gaŋà ná:-má
    final 'good night!' Hortative ('we' dual)
```

```
jáŋà ná:-mày<sup>n</sup> Hortative ('we' 3+)
R: àmi:<sup>n</sup>
```

Also using the verb ná:- 'spend night' are the questions in (xx3), which can be added to a 'good morning!' greeting.

```
(xx3) jáŋà nà:-w<sup>n</sup> mà jáŋà nà:-w<sup>n</sup> (Sg addressee, respectful)
(û:) jáŋà nà:-w<sup>n</sup>... (Pl addressee)
R: ná:-só-ý 'I slept well' (from a friend)
R: ná:-só 'I slept well' (respectful)
R: ná:-s-é 'we slept well' (respectful)
```

Activity- and/or place-specific greetings and their responses are of two basic structures. One type is based on the locative form (postposition gó, gá, etc.) of a noun like 'field(s)', 'market', 'well', or 'work' (xx4). These greetings are primarily used when the addressee is at the relevant location or is performing the indicated activity, though $[\grave{o}: g\acute{o}]$ pǒ: can also be addressed to someone returning from the fields.

```
(xx4) [ô: gố] pố: (at, or returning from, fields)
R: àwâ:

[éwé gá] pố: (at the market)
R: àwâ:

[ě: gá] pố: (at the well)
R: àwâ:

[bìré gá] pŏ: (at work) (with bìré, not bìrá)
R: àwâ:
```

The greetings in (xx5) are of the form 'you and {fields, work, market, water}' and are generally addressed to someone returning from the place and activity in question.

```
(xx5) [ú yò] [ŏ: yò] (at, or returning from, fields)
R: àwâ:

[ú yò] [bîrá yò] (coming back from work)
R: àwâ:

[ú yò] [éwé yò] (returning from the market))
R: àwâ:
```

```
[ú yò] [nî: yò] (coming back from well)
R: àwâ:
```

A traveler arriving at a house is greeted with [[ńdó gó] dô:] 'arrive at (or: approach) the house!', an invitation to come in and deposit one's baggage. One who is leaving on a long trip is sent off with the phrase [jáŋà dô:], literally 'arrive (there) in health!'. The reply is à:mî:n 'amen!'.

One gives condolences to a survivor of a deceased person with the phrase [[pòr"5 ŋ5] pŏ:], literally 'greetings in high worth'. The visitor who has presented condolences and is about to leave is told: [děnjê [ú gô] gùró-ndîyé-mì 'may God lengthen your (life)', and/or [děnjê [î bèndéy kò] wàgá-ndîyé-mì] 'may God put distance between us (i.e. us and the dead person)'. On returning home from giving condolences in another village, one is greeted with [yógî gó] pŏ:], literally 'greetings in running'.

The usual **Islamic greetings** and similar formulaic phrases, from Arabic, are present. ásàlâ:màlé:kùm (Arabic 'peace to you-Pl') is the formal greeting, especially on entering the presence of a group of men. The reply is wă:lékùmàsàlâ:m. The Islamic formula for inviting someone to come in, to join in a meal, etc., is bǐsímílà (Arabic 'in God's name'). àlbárkà 'thank you!' (Arabic 'blessing') is used to thank someone for a meal or a gift, and in markets as a polite refusal to buy.

20 Text

Text recorded in 2007. Some direct quotations from hyena are in Jamsay, or in a peculiar Jamsay-Nanga mix. These foreign items are italicized, as is the odd French word (e.g. *donc*, *encore*).

```
(xxx) té:njè:-ni:,
     [formulaic story opening phrase, cf. té:njè 'tale'; audience should respond
     yáwò⇒]
(xxx) [jòmó yò tà-tã: yò] [sàmàr<sup>17</sup>i-bîrá
                                                     gá] nn-ò,
        [hare and hyena and] [day.labor.L-work in] go.Perf.L-3PIS,
                      [sàmàr<sup>n</sup>i-birá
     [â:
               wŏyl
                                           gál
                                                  ńné
                                                      ή,
     [ReflPl
              two
                       [day.labor.L-work in]
                                                       and.SS,
                                                  go
     [sàmàr<sup>n</sup>i-birá
                      gá]
                          [bû:
                                     ŋ]
                                             ày-à,
     [wage.L-work
                      in
                           [3P1
                                     Obi1
                                            receive.Perf-3PIS
     làgàrà-mé
                      kémé-m-ù,
     apiary
                      build-Impf-3PIS
     'Hare and hyena went to (get) day-labor work. The two of them went to
     (get) paid work, and they (= people) took them in paid work. They were
     going to build apiaries (man-made beehives).'
     [X yò Y yò 'X and Y' §xxx, topic-indexing Reflexive Plural â: §xxx; ń
     'and.SS' in same-subject VP chains §xxx]
                                              [nă:
(xxx) donc [làgàrà-mé
                            kémè-kèmè]
                                                      dégé-ŋ-è:
                                                                        wàl,
                            build.HL-build.L] [hand lick-FutNeg-3PIS say],
              [apiary
        so
     wó:ti-vô:
                  wà.
                           [lògòrò-mé
                                             kémè-kèmè]
                                             build.HL-build.L]
     all.right
                           [apiary
                  say,
     tà-tã:
                                                   dègè-Ø,
                  [nă:
                           á
                                    kŝ]
                           ReflP
                                    Poss.InanSg] lick.Perf-3SgS
     hyena
                  [hand
     làgàrà-mé
                    yègé
                              sìgè-∅,
                              go.down.Perf.L-3SgS,
     apiary
                    fall
     donc
              [[ńné
                                [bèndî
                                            dègi-rá
                        wál
                                                              wá]]
                                            lick-ImprtNeg
             [[3Sg
                        say]
                                other
                                                              say]]
     so
     [wó:ti-yô:
                    wàl,
                                [kú
                                            mày<sup>n</sup>]
     [all.right
                                           like]
                    say],
                                [InanSg
     kémè-kèmè-kèmè-kèmè
     build.HL-build.L-build.L
```

```
[ndò kú] kémé gò-nd-ò,
```

[house.L Def.InanSg] build go.out-Caus.Perf.L-3PIS

'So (the bosses) said, when building the apiary, they (= hare and hyena) will (= must) not lick their hands. They said, all right. When building the apiary, hyena licked his hand. The apiary fell down (= collapsed). (The bosses) said, hey you (= hyena), don't lick (your hand) again! In that way they kept building and building, and they finished building that house (= apiary).

[iterated verb stem with {HL} then all-low tones §xxx; 'fall' and 'go down' in verb chain §xxx; 3Sg pronoun for original 2Sg vocative 'hey you!']

```
(xxx) kémé
                gò-ndó
                             ń,
                                       donc
                                               [sàrâ
                                                       bû:
                                                               gò],
        build
                go.out-Caus and.SS,
                                                        3PlP
                                                              Poss.InanSg]
                                       so
                                               [pay
     ńnέ
              dš:
                        nà.
     3SgS
              arrive
                        then.DS,
                                               dègέ≡bέ-Ø
     ſńnέ
             wá]
                     [tà-tã:
                               wàl
                                       [nă:
                                                                  wá],
             say]
                     [hyena
                                       [hand
                                               lick≡Past-3SgS
     [3Sg
                               say]
                                                                  say],
                        bêr
                                sárá
     [[tà-tã:
                ŋ]
                                        ή]
     [[hyena
                Obi1
                                pay
                                        and.SS]
                        goat
                        nàŋá
     [[jòmó
                                nd-a],
                ήÌ
     [[hare
                Obi]
                        cow
                                give.Perf.L-3PIS
```

'Having finished building, their pay (= wages), when it (= payday) arrived, (they) said: hey you, hyena, you licked your hand previously. (So) having paid hyena (with) a goat, they gave (= paid to) hare a cow.'

[Hyena was paid less because of his having licked; nà 'then.DS' in different-subject clause §xxx; Past clitic §10.xxx]

```
(xxx)
      [â:
                wŏy] yê:-yè:
                                           [ńdô
                                                   òsil
                                                             wŏ:
                                                                     ή,
       [ReflPl two] come.HL-come.L
                                         [house
                                                   road.L]
                                                             catch
                                                                     and.SS,
     ſâ:
             wòy] yê:-yè:,
                                        yé:
                                              mòl
                                                    [vé:
                                                           mò] [vé:
     [ReflPl two] come.HL-come.L, [come Dur] [come Dur] [come Dur],
     bònđi-úsúr<sup>n</sup>ò ìrìy-Ø,
                                      bòndi-úsúr<sup>n</sup>ò
                                                    ńné îriy
                                                                  nà,
                   get.up.Perf-3SgS, rain.L-wind
     rain.L-wind
                                                    3SgS get.up then.DS,
     èmći
              [nàŋá
                                                   nòŋgiyè-∅,
                                   yê]
                                   Poss.AnSg]
     hare
              cow
                       ReflSgP
                                                   mount.Perf-3SgS,
     [tà-tã:
               [bêr
                                       yê]
                                       Poss.AnSg]
     [hyena
               [goat
                       ReflSgP
     bàsέ:
                ńnέ
                            ďέ:
                                         nà]
                                                    ńnî-ŋὸ:-∅,
                                                    go-ImpfNeg-3SgS
     pull.Dur
                3SgS
                            be.tired
                                         then.DS]
     cmći
               ſńnέ
                       wál
                               màrnándiyé
                                              yè
                                                         ńné
                                                                 wá]
     [hare
               [3Sg
                       say
                               have.courage
                                              come.L
                                                         go
                                                                 say
```

```
[wó:ti-yô: wà],
[all.right say]
```

'As the two of them were coming (back), they took the road of (= to) the houses (= village). The two of them were coming. They kept coming and coming, (then) a rain storm arose. When the storm arose, hare mounted his cow. Hyena tugged on his goat until he (= hyena) was exhausted (= for a long time), (but) it wouldn't go. Hare said: hey you, have courage, come and (let's) go! He (= hyena) said, all right.'

[mò in durative background clause §15.2.xxx; -é: §xxx; verb-verb chain with low-toned nonfinal verb 'come' §xxx]

```
(xxx) básâ-<sup>n</sup>
                       ńnέ
                                      bú
                                                mò]
        pull-Impf
                       3SgS
                                      be
                                                while]
                                biyé
     [bèr
                  nέ]
                                              dě:
                                                        jὲ-Ø,
                   Def.AnSg]
                                lie.down
                                                        finish.Perf.L-3SgS,
     [goat.L
                                              be.calm
     biyé
                                 dě:
                    ńnέ
                                                 jέ
                                                            nà,
     lie.down
                    3SgS
                                                 finish
                                                            then.DS,
                                 be.calm
     [[bèr
                nέ]
                             biyé
                                          ńnέ
                                                   dě:
                                                             įέ
                                                                     nà]
                Def.AnSg]
                            lie.down
                                          3SgS
                                                   be.calm
                                                            finish then.DS]
     [[goat.L
     Cmcil
                ſńnέ
                        wál
                                  Tvè
                                               ńnî-ŋò:
                                                                ndèl
                [3Sg
                                               go-ImpfNeg
     [hare
                                                                if]
                        say
                                  [come.L
               ſńnέ
                            dàgá
                                    ťi
                                           kày]
                                                  ńnî:-n
                                                                  wà],
                      ή]
                     Obj] leave
     [LogoSg [3Sg
                                    Perf1b Top] go-Impf.3SgS
                                                                  say
     [wó:ti-yô:
                        wà],
     [all.right
                        savl
```

'As he (= hyena) was pulling, the goat lay down motionless (refusing to budge). When it lay down motionless—, when the goat lay down motionless, hare said: hey you (= hyena), if you (= hyena) won't come and go (with hare), I (= hare) will leave you (here) and go. He (= hyena) said: all right.'

[Imperfective plus bú 'be' §10.xxx; dě: jé- is fixed idiomatic verb chain; Logophoric pronoun indexing author of quotation §xxx; Perfective-1b tí as linker in VP chain §xxx; low-toned ndè 'if' §xxx]

```
(xxx) [bàsá
                  ńnέ
                             kán
                                        nà]
       [pull
                 3SgS
                             do
                                        then.DS]
    [[b\ear
                           yé:-ηò:
                                                        kán
             nέ]
                                               ńnέ
                                                                  nà]
                           come-ImpfNeg
                                               3SgS
             Def.AnSg]
                                                        do
                                                                  then.DS]
    [goat
    [[b\ear
                             lásà
                                       tùmà]
                                                  [gùwà
                                                                 tùmâ]
               nέ
               Def.AnSg]
                                                  [front.leg.L
    [[goat.L
                             leg.HL
                                       one.L]
                                                                 one]
    álá
                kùwè-Ø,
                                       [álá
                                                    kúwó
                                                               ή]
               eat.meat.Perf.L-3SgS,
                                                               and.SS]
    yank.off
                                       [yank.off
                                                    eat.meat
```

```
go-ImpfNeg-3SgS
                               say,
     'When he (= hyena) had pulled and the goat wouldn't come, he (= hyena)
     vanked off and ate one of the goat's legs, one front leg. Having vanked it
     off and eaten it, (hare) said, come and (let's) go. Hare said, hey you, once
     again, come and (let's) go. He (= hyena) said: I am not going.'
     [bare verb stem plus kán- 'do' §xxx; 'one of goat's legs' possessed core
     NP with noun and adjective §xxx]
                    básâ-n
                                 básâ-n
(xxx) [básâ-n
                                               ńnέ
                                                         bú
                                                               mò]
        [pull-Impf
                    pull-Impf
                                 pull-Impf
                                               3SgS
                                                         be
                                                               Dur]
     [[b\ear
                  né]
                                  yé:-ndέ
                                                    àwà-rí-∅],
                  Def.AnSg]
                                  come-VblN
                                                    accept-PerfNeg-3SgS],
     [[goat.L
                    àwà-rî
                                                       kán
     [yé:-ndέ
                                             ńnέ
                                                               nà]
                    accept-PerfNeg-3SgS]
                                                       do
     [come-VblN
                                             3SgS
                                                               then.DS]
     [bèr-dè:
                          nέ
                                            ή]--
     [goat-mother
                          Def.AnSg
                                            Obj]—
              [gùwà
                          tùmâ]
                                     álá
                                                 kùwè-∅,
     îyê
                                                 eat.meat.Perf.L-3SgS,
              [front.leg.L one]
                                     yank.off
     again
     [néy
                gày]
                           [lòsô
                                     wŏy]
                                               wàsὲ-∅,
                                               remain.Perf.L-3SgS,
     [now
                 Top]
                          [foot
                                     two
     [[lòsô
             tà:ndĭ:] ndŏ:] [néy
                                    gày]
                                           [lòsô wǒy] wàsè-Ø,
                     not.be] [now
                                    Top] [foot two]
     [[foot
                                                         remain.Perf.L-3SgS,
            three
     [[lòsò
                wòy
                       gú]
                                       ńnέ
                                                wàsá
                                                           nà]
                                                remain
               two.L Def.InanSg]
                                                           then.DS]
     [[foot.L
                                       3SgS
                                 màr<sup>n</sup>á]
     [ber-de:
                                          sέmέ
                                                         ĵiyέ
                                                                 ή,
                                                         kill
     [goat-mother.L
                     Def.AnSg entire]
                                          cut.throat.of
                                                                 and.SS,
     [bèr-sùŋùr<sup>n</sup>i
                     tùmâ]
                               [jòmó
                                         ή]
                                                   ndi-Ø,
                                                   give.Perf.L-3SgS.
     [goat.L-ear.L
                      one
                               [hare
                                         Obil
     'As he (= hyena) was pulling and pulling, the goat refused to come. When
     it refused to come, he (= hyena) again yanked off and ate the whole goat—
     (or rather) one (= the other) front leg (of the goat). Now two hind legs
     remained. Not three legs, now (just) two remained. When the two legs
     remained, he (= hyena) cut its throat to kill it, and he gave one goat-ear to
     [awa 'accept' with verbal-noun complement §xxx; negative clause plus
     kán- 'do' §xxx; "goat-mother" = 'entire goat' §xxx]
(xxx) [italicized sequence is in Jamsay]
```

ńnê-ⁿ

go-Impf

wà],

say], hare

òmćį

wà,

ńnέ

3Sg

wá

say

îyê

again

encore.

still,

[yè

[néy

gày

[come.L

ńnî-ŋò:-∅

gŝ]

Γá

```
[ReflSgP
[now
            Top]
                                          Poss.InanSg]
[ber-de:
                                   kúwó-jè
                    nέ]
                                                         ŋ]
[goat.L-mother.L
                    Def.AnSg]
                                   eat.meat-RecPerf
                                                         and.SS]
             wòyl
                        [yé:
                                   mò]
                                            [yé:
                                                         mò]
[ReflPl
             two]
                        [come
                                   while] [come
                                                         while]
[dέmì⇒
               ńné
                            nà]
                            then.DS]
[a.little
               go
[jòmó
        wá
              [bèr-súŋúr<sup>n</sup>i
                               á
                                                          ńdî
                                                                 wá],
                                          gŝ]
[hare
        say
              [goat.L-ear
                               LogoSgP
                                          Poss.InanSg] give
                                                                 say]
[jòmó
           wá
                   kúwó-jè
                                            mă:ndî
                                                         wá]
                   eat.meat-RecPerf
[hare
           say
                                            think
                                                         say]
[jòmó
           gò-ndó
                            ſńnέ
                                              \hat{n}di-\emptyset],
                                      ή]
[hare
           go.out-Caus
                            [3Sg
                                      Obj]
                                              give.Perf.L-3SgS],
hà:
          [nîη
                  kè]
                           dôm
                                       gá:jè
                                                  kún-á:r<sup>n</sup>à-m
                                                                  wà,
hah!
          [now
                  Top]
                           up.to.now joking
                                                  do-Impf-1SgS
wó:ti-yô:
                   wà,
all.right
                   say,
```

'Now, when he (= hyena) had finished eating his thing (= meal), (eating) the goat, the two of them (= hyena and hare) were coming, (and) when they had gone a little way, he (hyena) said to hare: give me (= hyena) (back) my goat-ear! He (= hyena) thought (= assumed) that hare had already eaten (it), (but) hare took (it) out and gave it to him (= hyena). (Hyena) said (in Jamsay): *I am just kidding!* (Hare) said: all right.'

```
(xxx) [â:
                             mò] [yé:
               wòv] [vé:
                                         mò] [yé:
                                                      mò] [yé:
       [ReflPl two] [come Dur] [come Dur] [come Dur] [come Dur]
           tà-tã:
                 [jòmó wá]
                               [súŋúr<sup>n</sup>i á
                                                               ńdî
                                                                    wá,
                                                  gŝ]
    again hyena [hare
                         say]
                               ear
                                        LogoSgP Poss.InanSg] give say,
    èmćį
               îyê
                      ſńnέ
                               ήÌ
                                      gò-ndó
                                                       ndi-Ø,
    hare
               again [3Sg
                               Obj] go.out-Caus
                                                       give.Perf.L-3SgS,
    dôm
                  gá:jè
                               kún-á:r<sup>n</sup>à-m
                                                  wà.
                  kidding
                               do-Impf-1SgS
    up.to.now
                                                  say,
                                              [yé:
    [yé:
            mà]
                  [yé:
                          mò] [yé:
                                      mà]
                                                      mò] [yé:
                                              [come Dur] [come Dur]
    [come Dur] [come
                          Dur] [come Dur]
    'The two of them were coming and coming. Again hyena said: hey hare,
    give (me) my (goat-)ear! Hare again took (it) out and gave it to him.
    (Hyena) said (in Jamsay): I am just kidding! They were coming and
    coming.
```

```
(xxx) [[kú
                   mày<sup>n</sup>]
                             yě:
                                               [[ńdó
                                                        gó]
                                                               d-à:],
                                     ή]
                             come and.SS]
        [[InanSg like]
                                               [house in]
                                                               arrive.Perf.L-3PIS
     ſńdó
                 gó]
                             yě:
                                          dă:
                                                        ή--,
```

```
and.SS-.
[house
           in]
                      come
                                  arrive
ſńdó
                      yě:
           gól
                                  dă:
                                               ń,
[house
           in]
                      come
                                  arrive
                                               and.SS,
                                                        ńďi
[jàmá
               [ber-namá
                              á
                                         gŝ]
                                                               wá,
[hare
        say
               [goat.L-meat LogoSgP Poss.InanSg] give
                                                               say,
                             gò-ndó
[jòmó
           ſńnέ
                   ŋ!]
                                               ndi-Ø]
           [3Sg
                   Obj]
                             go.out-Caus
                                               give.Perf.L-3SgS]
[hare
                                          kúr<sup>n</sup>i-sò
Γá
             [nîŋèy
                     ỳŋà]
                               gá:jè
                                                         wà],
[LogoSg
             [now
                      with]
                               kidding
                                          do-Perf2
                                                         say],
'In that way they came and arrived at the houses (= village). Having come
and arrived at the houses—. Having come and arrived at the houses,
(hyena) said: hey hare, give (me) my goat meat (= ear). Hare took (it) out
and gave it to him, saying that now he (= hare) was kidding (= joking).'
[Hare speaks in Nanga]
```

```
(xxx) [[kú
                    mày<sup>n</sup>]
                                [ùsú
                                           wŏy]
                                                        dă:
                                                                   ήŢ
                               [day
        [[InanSg like]
                                           two]
                                                        arrive
                                                                   and.SS]
                             dê:]
     [jòmó
               [[[á
                                            sè:mbè-bì:<sup>n</sup>]
                                                                ŋò]
     [hare
               [[[ReflSgP
                             mother.HL] cotton.basket.L]
                                                                in]
     [ńné
                           kún
                                         dògè-Ø.
               ή]
              and.SS]
                                         leave.Perf.L-3SgS,
     [go
                           put.in
     [[sè:mbè-bì:
                         ŋò]
                                  kún
                                          dàgá
                                                   ή]
     [[cotton.basket
                         in]
                                  put.in
                                          leave
                                                   and.SS]
     [jòmó
                 gŏ:
                             ńnέ
                                          èrέ
                                                       nà],
     [hare
                                          Perf1a
                                                       then.DS1.
                 go.out
                             3SgS
                                                   yě:
     òmó
                 [['njó
                                       nò]
                                                                ήÌ
     hare
                 [[younger.sibling
                                       3SgP]
                                                                and.SS]
                                                    come
     [[sùŋùr<sup>n</sup>i
                   kú]
                                   gò-ndó
                                                   kùwè-Ø],
     [[ear.L
                   Def.InanSg]
                                                   eat.meat.Perf.L-3SgS],
                                  go.out-Caus
     'When two days had arrived (= passed) in that way, hare went and put and
     left (the goat-ear) in his mother's cotton-gear basket. When hare had put
     and left (it) in the cotton-gear basket, and had gone out (completely),
     hare's younger brother came and took out the (goat) ear and ate (it).
     [ετέ as linking equivalent of Perfective-1 -ετε- in chains, §xxx; 'hare<sub>x</sub> [his<sub>x</sub>
     brother]' with topicalized possessor, cf. the simpler phrasing jòmó ńjò
     'hare's brother']
```

```
kúwó
(xxx) gò-ndó
                        ńnέ
                                                 nà,
                        3SgS
        go.out-Caus
                                   eat.meat
                                                 then.DS,
     tà-tã:
                              [ber-súnúr"i á
             [yě:
                                                   gŝ]
                                                                 ńdî
                                                                         wá,
                     ή]
     hyena [come and.SS] [goat.L-ear
                                            ReflP Poss.InanSg] give
     [jàmá
              [yě:
                       ήl
                                ŋìr<sup>n</sup>é
                                        ńnέ
                                                kán
                                                         nà]
```

```
[hare
         come
                  and.SS] look
                                   3SgS
                                            do
                                                    then.DS]
kùr<sup>n</sup>∂-ndó-Ø,
be.put.in-StatNeg-3SgS,
                                                            tásî
      [ńné wá] [bèr-súnúr<sup>n</sup>i
                                             gŝ]
                                                                  wá,
       [3Sg say] [goat.L-ear
                                  LogoSgP Poss.InanSg] pay
so
                                                                  say,
                               wà,
iòmó
             wó:tì-yô:
hare
             all.right
                               say,
                                              bàsá
jòmó
         [[nàŋá
                             yê]
                                                         ή]
hare
         [[cow
                    ReflP
                             Poss.AnSg]
                                              pull
                                                         and.SS]
                           gól
                                   ńné
[[â:
           wŏy]
                   [ò:
                                            ή]--,
[[ReflPl
           two]
                   [fields in]
                                   go
                                            and.SS]—,
[[â:
           wŏy]
                   ſò:
                           gól
                                   ńné
                                            ή],
[[ReflPl
           two]
                   [fields in]
                                   go
                                            and.SS],
jòmó
        [tà-tã:
                 wà]
                        [ùsi
                                túmbó
                                          yè:-mì
                                                           gú]
                        [sun.L sun.rise come-Ppl.Impf Def.InanSg]
hare
        [hyena
                 say
vî:-sò
             mâ:
                       wà,
             O
see-Prog
                       say,
```

'After he took (it) out and ate (it), hyena came and said: give (me) my goat-ear! Hare came and looked, (but) it (= ear) wasn't in (the basket). (Hyena) said: so, hey you, pay for (= replace) my goat-ear! Hare said, all right. Hare pulled his cow, the two of them (= hare and hyena) went to the field—. The two of them went to the field, and hare said: hey hyena, do you see that sun which is rising (and coming)?'

[kùrno-ndó-, negation of stative yá kùrno- 'be (put) in'; same-subject ý in 'went to the field' with sloppy coindexation of subjects, singular versus plural §xxx; imperfective relative clause 'sun which is rising and coming' §xxx]

```
(xxx)
                 tà-tã:
                             Γá
                                          yî:-sò]
                                                       wà,
                 hyena
                             [LogoSgS
                                          see-Impf]
        yes
                                                       say,
     donc
            ſńnέ
                   wál
                          [ùsi
                                   túmbó
                                             vè:-mì
                                                               gú]
                                                               Def.InanSg]
     so
            [3Sg say]
                                   sun.rise come-Ppl.Impf
                          [sun.L
     [[ùsi
                gú]
                               gõ:≡Ø
                                           wà]
     [[sun.L
               Def.InanSg]
                               fire≡it.is
                                           say]
     [ńné wá] [[kú
                              ńné jὲ:
                         ή]
                                               gáy
                                                        yě-y
                                                                       wá,
     [3Sg say] [[InanSg Obj] go get.coals.L then.SS] come-Hort.3Sg say,
     'Hyena said, yes, he saw (it). (Hare) said, that sun which was rising, the
     sun was fire; he (= hyena) should go get that (= fire, i.e. hot coals) and
     come.'
```

[gáy 'and then' after tone-dropped verb §15.2.6; third-person Hortative in jussive, i.e. in quoted command, §10..6.4 and §17.1.4.1]

```
(xxx)
       tà-tã:
                [gò:
                        gúl
                                     [iὲ:
                                                  gól
                                                         ńnέ
                                                               ńné
                [fire.L Def.InanSg] [get.coals.L Purp] 3SgS go
       hyena
                                                                     then.DS
    jòmó,
            jòmó [[nàŋá
                                                   sémé
                            á
                                      yê]
                                                                 ή],
                            ReflSgP Poss.AnSg] cut.throat.of and.SS]
            hare [[cow
    [[nàmà
                 ý1
                               úrś
                                               įε
                                                         ŋ]
    [[meat.L
                Def.InanPl]
                               skin&butcher
                                               RecPf
                                                         and.SS]
              èsì
    [[nàmà
                       ý٦
                                    gò-ndó
                                                           [sîmbî
                                                  ή]
                                                                    ή],
    [[meat.L good.L Def.InanPl] go.out-Caus
                                                  and.SS] [roast
                                                                    and.SS]
    [[tùmà-pómbó
                      gó] ndé
                                  ή]
                                           [k<sub>ó</sub>r
                                                       dàgá
                                                                 ή],
    [[tree.L-hole
                           go.up and.SS] [hang.up
                                                       leave
                                                                 and.SS]
                      in]
    [[bindi
                ý⇒Ì
                              [kù:
                                      ý]
                                                    biyni
                                                             ή],
    [[entrails Def.InanPl]
                             [head
                                      Def.InanPl] bury
                                                             and.SS]
    kîrâ
              kě:y
                                 dògè-∅,
                                 leave.Perf.L-3SgS
    horn
              sticking.out
    'When hyena had gone in order to get fire (= hot coals), hare—, hare
    slaughtered his cow, he finished skinning and butchering the meat cuts
    (= sections), he took out and roasted the choice meat cuts, he went up into
    a tree hollow, (and) he hung up and left (the choice meat cuts). He buried
    the entrails (and) the heads, and left the horns sticking out (of the ground).'
    [Purposive clause with Locative gó §17.5.2]
```

```
(xxx)
        [kě:y
                               dògó
                                             ή],
                                             and.SS]
         [sticking.out
                               leave
     tà-tã:
               [[ńné
                          ή]
                                        gòŋir<sup>n</sup>i
                                                       dè:-∅]
                          and.SS]
                                                       be.tired.Perf.L-3SgS]
     hyena
               [[go
                                        go.around
     [gồ:
                  bèrè-ri-Ø]
                  get.PerfNeg-3SgS]
     fire
     [ùsî
               nà:]
                          [gõ:
                                     ndŏ:
                                                     yè],
                          [fire
     sun
               rather]
                                     not.be
                                                     rather.than]
     [kú
                   mày<sup>n</sup>]
                                tà-tã:
                                             [yě:
                                                           ή]
     [InanSg
                   like]
                                hvena
                                             [come
                                                           and.SS1
                               gõ:
                                           bèrè-ri
     ſά
                    kòy]
                                                               wá,
     [LogoSg
                    Top]
                               fire
                                           get-PerfNeg
                                                               say,
                    wó:tiyô:
     jàmá
                                   wà,
     hare
                    all.right
                                   say,
     bèndi
                   déy<sup>n</sup>⇒
                                kò-kámâ
                                                 ndŏ:
                                                              wá.
     other
                   apart
                                thing-any
                                                not.be
                                                              say,
```

'He (= hare) left (the bones) sticking out. Hyena went walking around until he was exhausted (i.e. trying in vain to get fire from the sun), (but) he didn't get any fire. It was the sun, not fire. In that way hyena came and said: as for me, I haven't gotten any fire. Hare said, all right, it's nothing special (= it doesn't matter).'

[ý 'and.SS' at end of same-subject clause sequence; [[X nà:] [Y ndŏ: (yè)]] 'it is X, rather than Y' §xxx]

```
(xxx) donc [[ńné
                                    [kirà
                                               bù-mù
                          wá]
                                                              gú]
              [[3Sg
                          say]
                                    [horn.L
                                               be-Ppl.Impf
                                                              Def.InanSg]
        so
                                           Γé
     kἔy⇒
                    yî:-sò
                                  mà]
                                                    yî:-sò
                                                                wàl,
     sticking.out
                    see.Prog
                                  Q1
                                           yes
                                                    see-Prog
                                                                say],
     [donc
                  [[kú
                                námà
                                             gù]
                                                              kéréw]
                                             Def.InanSg.L]
     [so
                  [[InanSg
                                meat.HL
                                                              all]
     ká
                               kùr<sup>n</sup>ò
                                                  wàl
                  yá
     there.Def
                  Exist
                               be.put.in
                                                  say
     ſńnέ
                wá]
                            [gò-ndê
                                                   wàl
                                                            wó:tiyô:,
     [3Sg
                say]
                            [go.out-Caus-Hort.3Sg say]
                                                            all.right,
     '(Hare) asked him: so, do you see those horns that are (there) sticking out?
     (Hyena) said: yes, I see it. (Hare) said: so, all that meat is in there
     (= underground), you (= hyena) should take it out. (Hyena) said, all right.'
     [combination of preposed possessor, noun, and low-toned Definite
     morpheme §xxx; Existential particle yá §xxx]
```

- (xxx)[[kú màyⁿ] ńnέ kán nàl [[InanSg 3Sg do then.DS] like [[bòndi dùgi] yè:- \emptyset], [[bòndi dùgi] ńné yě: come.Perf.L-3SgS,, [rain.L big] [[rain.L big] 3SgS come then.DS] [jòmó [á gày] [témbè gà] ńdê-n wà], [LogoSg Top] [above in] go.up-Impf.3SgS [hare savl. gày] [[dósú gó] bê:-n [wò:tiyô: wà] tà-tã: [á wà], [all.right say] hyena [LogoSg Top] [[belowin] stay-Impf.3SgS] say 'After it happened like that, a big rain came. Hare said: as for me, I will go up above (= into the tree). (Hyena) said, all right. Hyena said: as for me, I will stay below.'
- cmci] (xxx) [témbè 'ndέ gàl ŋ] [above in] and.SS] [hare go.up [nàmá á kùwò-ndé, gŝ] [meat ReflSgP Poss.InanSg] eat.meat-if, [kirnà [[tà-tã: gà] gisê-n, gúl kù:] [horn.L Def.InanSg] [[hyena head.HL] in throw-Impf.3SgS, wò⇒-wôy ké] [àrⁿà démbéré] mirⁿá:- r^n à- \varnothing kòv, [îyé ouch! [today Top] [rain.L big] rain.fall-Impf-3SgS Emph, [dé jùmò] *ε*:-rà-w mà, [elder.sibling hare] see-Impf-2SgS Q, àⁿháⁿ òmó Γá yî:-sò] wà,

```
uhhuh
             hare
                          [LogoSgS see-Prog]
                                                      say,
òmó
          [nàmá
                                                                    ndé,
                                  gŝ]
                                                     kùwò
hare
          [meat
                     ReflSgP
                                  Poss.InanSg]
                                                     eat.meat.L
                                                                    if,
                                                              gîsê-n,
[kir<sup>n</sup>à
           gúl
                          [[kû:
                                    nò]
                                              gàl
                                                      tác!
[bone.L
           Def.InanSg] [[head
                                    3SgP] in]
                                                      thud! throw-Impf.3SgS
            [àr<sup>n</sup>à
                                        mìr<sup>n</sup>á:-r<sup>n</sup>à-Ø
éy⇒
                      démbéré]
                                                          wà,
                               rain.fall-Impf-3SgS
hey!
            [rain.L big]
                                                           say,
\eth^n h \acute{5}^n
            jòmó
                                        yî:-sò
                                                        wà,
uhhuh!
            hare
                      LogoSgS
                                        see-Prog
                                                        say,
```

'Hare went up above, and when he had eaten his meat, he was throwing the bones (down) on hyena's head. (Hyena thought:) woo-woo, today a big rainstorm sure is falling! (Hyena mistakes the bones for hailstones.) (Hyena) said: elder brother hare, do you see? Uh-huh, hare said, I see (it). When hare had eaten his meat, he was throwing it down with a thud on his (= hyena's) head. (Hyena) said, hey, a big rainstorm is falling! Uh-huh, hare said, I see (it).'

[high-toned ndé 'if' §xxx]

```
mày<sup>n</sup>]
                                  kúwó
(xxx) [[kú
                                                   ή]
        [[InanSg
                     like
                                  eat.meat
                                                   and.SS1
     [kirnà
                    dèmbirè
                                                    [tùndú
                                   gú]
                                                                  gó],
     [bone.L
                    big.L
                                   Def.InanSg]
                                                    [down
                                                                  in],
     [[tà-tã:
                kù:]
                                    tây
                                               ńnέ
                                                         gisé
                                                                   nà,
                             gà
     [[hyena
                head..L]
                             in]
                                    thud!
                                               3SgS
                                                         throw
                                                                   then.DS,
                    ón⇒
     tà-tã:
                                  Γdé
                                                        [cmci
     hyena
                    oh!
                                  [elder.sibling
                                                        hare.L]
     ſú≡ỳ
                   тî
                            kám-á:rnà-wn
                                                    mà⇒]
                                                                wà,
     2Sg≡Foc
                   1SgO
                            throw-Impf-2SgS
                                                    Q1
                                                                say,
     é⇒
              wá,
                       d'accord
                                    [yèré
                                              sîgé
                                                         wá]
     yes
                       okay
                                    [come
                                              go.down
              say,
                                                         say]
                          yì-yî:-<sup>n</sup>
     ſńnέ
              wál
                                              wà.
              say]
                          Rdp-see-Impf
     [3Sg
                                              say,
```

'In that way he ate, and at the end (of the meal) he threw the big bone down hard on hyena's head. Hyena said: oh, elder brother hare, is it you [focus] who is throwing (bones) on me? (Hare) said: yes. (Hyena) said: all right, come down (here)! You (= hare) will see!'

[yèré sigé is a mix of Jamsay and Nanga; yi-yi:- n reduplicated Imperfective $\S10.xxx$]

```
ńnέ
(xxx)
        [kú
                   mày<sup>n</sup>]
                             jàmá
                                     sîgé
                                                          yě:
                                                                   nà,
        [InanSg
                   like]
                             hare
                                     go.down
                                                 3SgS
                                                          come
                                                                   then.DS,
     á
                   ſńnέ
                                          kúwô-n,
                              ή]
```

```
LosoSgS
                      Obi]
            [3Sg
                                 eat.meat-Impf.3SgS,
ſńnέ
                  ùmăy
                              Γá
                                                kùwò
                                                             ndé,
         wá]
                                         ή]
[3Sg
         say]
                  like.this
                              [LogoSg Obj] eat.meat.L
                                                             if,
Γá
           [dòsù
                        ká]
                                             yègè
                                                       ndé]
[LogoSg
           [down.L
                         on]
                                LogoSgS
                                             fall.L
                                                       if]
ſńnέ
              Γá
                              mònè-r<sup>n</sup>i
       wál
                         ήÌ
                                                 wá
                                                        quoi,
              [LogoSg Obj] get.even-PerfNeg say
[3Sg
                                                        Emph,
donc
            ſńnέ
                         wá],
so
            [3Sg
                         say],
           [[pà:ndi
                      siridi] gò] [á
[ńné wá]
                                             ήÌ
                                                   pági
                                                          wá
                                                                quoi,
                              in] [LogoSg Obj] tie
[3Sg say]
           [[thread.L thin]
                                                          say
                                                                Emph,
                                                       ndé,
[[pà:ndi
           siridi]
                    gò]
                           Γá
                                              pàgi
                                        ήÌ
[[thread.L
           thin]
                    in
                           [LogoSg
                                        Obj]
                                             tie.L
                                                       if,
                                                     ndé]
[[tir"i
                 èsi]
                            ńnέ
                                        jè:
[[firewood.L
                            3SgS
                                        bring.L
                                                     if
                 good
                       dà:
                                     ndé]
[[á
             ήŢ
             Obj]
                       roast.L
                                    if]
[[LogoSg
kúwî
             wá
                          quoi,
eat.Imprt
                          Emph,
             say
```

'In that way, when hare came down, (hyena said:) I will devour you (= hare). (Hare) said: if you devour me like that, if I fall down there, you won't have gotten (back at) me. So, you there, tie me up with thin thread. When you have tied me up with thin thread, have brought some good firewood, and have roasted me, (then) eat me!'

[the final string of 'if/when' clauses is completed by an imperative; pragmatically the entire sequence is imperative]

```
< [[tir"i èsi]
                                          gó]-- >
(xxx) tà-tã:
                                 jàrà
        hyena < ... >
     [[ńnέ
               ή]
                           pági
                                       dògó
                                                  ήŢ
                                                  and.SS1
     [[3Sg
               Obj]
                           tie
                                       leave
     [[[tir"i
                      èsi]
                                                   ńnέ
                                                           ńné
                             jàrà
                                          gól
                                                                  nà]
     [[[firewood.L good] look.for.L Purp]
                                                                  then.DS]
                                                  3SgS
                                                           go
              párá-gí-tì-∅],
                                      párá-gî
                                                  yògó
                                                           màrè-∅,
     [jàmá
     [hare
              snap-Caus-Perf1b-Ø, snap-Caus run
                                                           be.lost.Perf.L-3SgS,
     tà-tà: encore
                     îyê
                                                                yi:-Ø,
                             [\acute{n}n\acute{\epsilon} \acute{\eta}]
                                            [jòrá
                                                       ή]
     hyena again
                      again [3Sg Obj]
                                           [look.for and.SS] see.Perf.L-3SgS
     [[ńnέ
                      jòró
                                                yĭ:
              ή]
                                                         ń,
                                 ή]
     [[3Sg
              Obj]
                     look.for
                                 and.SS]
                                                see
                                                         and.SS,
                            jórð-jòrð-jòrð-jòrð]
     [[ńné
                  ή]
                            look.for.HL-look.for.L-look.for.L-look.for-L
     [[3Sg
                  Obj]
```

'Hyena tied up and left him, then he (= hyena) went in search of good firewood. Hare snapped (the rope). He snapped (it) and ran (away) and disappeared. Hyena looked for him (= hare) again. He looked for him, he was looking and looking and looking for him.'

[jɔ́rɔ-jɔ̀rɔ-... iteration of verb with HL then all-low tones (on the tape the speaker began with the lexical LH tone but corrected this during transcription]

```
gól
(xxx) [yá
                ńné
                       ήÌ
                                 [[ndò-piré
                                                          ńné
                                                                  ήÌ
                       and.SS] [[house.L-inside
                                                                  and.SS]
        [there
                go
                                                    in]
                                                          go
     [jòmó
                [séwrè
                                gàl
                                        ńnέ
                                                έnià
                                                                   mà]
                                                be.slipped.in.Stat while]
                [ceiling.pole
     [hare
                                in]
                                        3SgS
     ſńnέ
               ή]
                          tèmbi-Ø
                                                quoi,
               Obj]
                          find.Perf.L-3SgS
                                                Emph,
     [3Sg
     [séwrè
                       gàl
                               ńnέ
                                        έnjà
                                                              mò]
     [ceiling.pole
                       in
                               3SgS
                                        be.slipped.in.Stat
                                                              while]
                              wá]
     [dé
                     èmćį
                                      [yè
                                                  sîgî
                                                                wá],
     [elder.sibling
                     hare
                              say]
                                      [come.L
                                                  go.down
                                                                say
     òmó
                á
                              sîgé-ŋò:
                                                     wà,
     hare
                LogoSgS
                              go.down-ImpfNeg
                                                     say,
```

'He (= hyena) went there, he went inside a house, and found him (= hare) slipped in among the thin ceiling poles. While he (= hare) was in among the thin ceiling poles, (hyena) said: elder brother hare, come down! Hare said: I won't go down.'

[Stative form of verb §10.3]

```
(xxx)
        ſńnέ
                                              ùmăy
                                                        yègé sigè
                                                                            ndé]
                 wá] [niŋèy
        [3Sg
                 say] [now
                                 LogoSgS like.this fall
                                                               go.down.L if]
     [[nàmá
                 á
                                 gŝ]
                                                   pàmá-èrè-Ø
                                                   be.ruined-Perf1a-3SgS]
     [[meat
                 LogoSgP
                                 Poss.InanSg]
     ſńnέ
               wál
                      dùvá
                                 ńné
                                         bàrá
                                                    jè:
                                                                 ndé,
               say]
                      ashes
                                         gather
                                                    bring.L
                                                                 if,
     [3Sg
                                 go
     [dòsù
                   gá]
                             ńnέ
                                         tiy
                                                       ndé,
     [below
                   in
                             3SgS
                                         dump.L
                                                       if,
                     [dùyà
                                               yégê-<sup>n</sup>l
     Γá
                                     gá]
                     [ashes.L
                                     in]
                                               fall-Impf]
     [LogoSgS
                                   bàr<sup>n</sup>ùmè-r<sup>n</sup>i]
     [[á
                   ή]
                                   wound-PerfNeg-Ø
     [[LogoSg
                   Obj]
     [ńné wá]
                   [nàmá
                                                       wŏ: kúwô-n
                                        gŝ]
     [3Sg say] [meat
                             LogoSgP Poss.InanSg] catch eat.meat-Impf
                                                                               say,
                                   t \grave{\partial} v^n \acute{\partial} = \grave{v}^n
     wó:tiyô:
                 [kò
                          ké]
                                                    wà
     all.right
                 [Nonh Top]
                                   truth≡it.is
                                                    say
```

'(Hare) said: if I fall down like that now, my flesh will be ruined. When you (= hyena) have gone and gathered some ashes and brought them, and when you have dumped them down below (on the ground), (then) I will fall (= land) on the ashes and it will not have injured me, (then) you will catch me and eat my meat. (Hyena) said (in Jamsay): all right, that's true.'

```
(xxx)
        donc
                 tà-tã:
                          [ńné
                                   ή]
                                             dùyá
                                                        [ńné
                                                                 ή]
                                                                 and.SS]
        so
                 hyena
                          [go
                                   and.SS]
                                             ashes
                                                        [go
     jòró
                      jě:
                                    ή]
     look.for
                      bring
                                    and.SS]
                 [ndò-pirè
     [dùyá
                                    gá]
                                            mâ:
                                                           tiy
                                                                    ή],
                                                                    and.SS],
     [ashes
                 [house.inside
                                    in
                                            pouring.lots
                                                           pour
     [jòmó
                gày]
                         [séwrè
                                          gà]
                                                   yá
                                                           nàηà-\emptyset,
                         [ceiling.poles
                                                           be.up.on.Stat-3SgS,
     [hare
                Top]
                                          in]
                                                   Exist
                                       yè
     ſńnέ
                wá]
                         kûngà
                                                 sîgî
                                                                       wá,
     [3Sg
                         all.clear
                                       come
                                                 go.down.Hort.3Sg say,
                say]
     jàmá
               [dùyà
                         gá]
                                  pûy
                                           ńnέ
                                                    kán
                                                             nà,
                                           3SgS
     hare
               [ashes
                         in]
                                  thud!
                                                   do
                                                             then.DS,
                                                           kàr<sup>™</sup>i-Ø]
     [[[tà-tã:
                        gò]
                              dùyá
                                        yέgî⇒
                girè]
     [[[hyena
               eye.L]
                        in
                              ashes
                                        powder.get.in
                                                           do.Perf.L-3SgS]
     [jòmó
                   ká
                                              màrè-∅],
                                   yàgá
     [hare
                   there.Def
                                   run
                                              be.lost.Perf.L-3SgS]
```

'So, hyena went and looked for ashes, and brought (them), and dumped (them) all around inside the house. Hare for his part was (still) up among the ceiling poles. (Hyena) said: the coast is clear, (now) to come (= drop) down! When hare made a thud (falling) on the ashes, the dust (kicked up from the ashes) got into hyena's eyes (half-blinding him). Hare ran (from) there and disappeared.'

[kûŋgà, used in contexts like 'the coast is clear', obscurely related to kú=ŋ 'it's that']

(xxx)[té:njè mélèm][dùmá mélèm]quoi[storysubmerged][finish(noun)submerged]Emph[story-closing formula]